# SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549

# FORM 10-K

FOR ANNUAL AND TRANSITION REPORTS PURSUANT TO SECTIONS 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

# (Mark One)

Securities register

Ame

# [x] Annual Report pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 for the fiscal year ended December 31, 2002 or

### [] Transition report pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934 for the transition period from \_\_\_\_ to \_\_\_\_

Commission <u>File Number</u> 333-47647	Registrant, State of Incorporation Address and Telephone Number American States Water Company (Incorporated in California) 630 East Foothill Boulevard, San Dimas 91773 (909)394-3600	<b>IRS Employer</b> <u>Identification No.</u> 95-4676679
000-01121	Southern California Water Company (Incorporated in California) 630 East Foothill Boulevard, San Dimas 91773 (909)394-3600	95-1243678
red pursuant to Section 12(b) of the Act:		
Title of Each Class erican States Water Company Common Shares,		Name of Each Exchange on Which Registered
\$2.50 Stated Value		New York Stock Exchange

\$2.50 Stated Value Rights to Purchase Junior Participating Preferred Stock

## Securities registered pursuant to Section 12(g) of the Act: None

Indicate by check mark whether Registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that Registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days.

American States Water Company	Yes [x] No [ ]
Southern California Water Company	Yes [x] No [ ]

Indicate by check mark if disclosure of delinquent filers pursuant to Item 405 of Regulation S-K is not contained herein, and will not be contained, to the best of Registrant's knowledge, in definitive proxy or information statements incorporated by reference in Part III of this Form 10-K or any amendment to this Form 10-K.

Indicate by check mark whether the Registrant is an accelerated filer (as defined in Rule 12b-2 of the Act.).

American States Water Company	Yes [x] No [ ]
Southern California Water Company	Yes [ ] No [x]

The aggregate market value of the total voting common stock held by non-affiliates of American States Water Company was approximately \$401,258,000 and \$370,146,800 on June 28, 2002 and April 4, 2003, respectively. The closing price per Common Share on April 4, 2003, as quoted in the Western Edition of *The Wall Street Journal*, was \$24.35. As of April 4, 2003, the number of Common Shares of American States Water Company, \$2.50 Stated Value, outstanding was 15,201,102. As of that same date, American States Water Company owned all 110 outstanding Common Shares of Southern California Water Company. The aggregate market value of the total voting stock held by non-affiliates of Southern California Water Company was zero on June 28, 2002 and April 4, 2003.

## **Documents Incorporated by Reference:**

Portions of the Proxy Statement of American States Water Company will be subsequently filed with the Securities and Exchange Commission as to Part III, Item Nos. 10, 11, 12 and 13, in each case as specifically referenced herein.

AMERICAN STATES WATER COMPANY
and
SOUTHERN CALIFORNIA WATER COMPANY

FORM 10-K

INDEX

Page No.

Item 3:	Legal Proceedings	3 - 5
Item 4:	Submission of Matters to a Vote of Security Holders	6
Part II		
Item 5:	Market for Registrant's Common Equity and Related Stockholder Matters	6 - 7
Item 6:	Selected Financial Data	7 - 8
Item 7:	Management's Discussion and Analysis of Financial Conditions and Results of Operation	8-34
Item 7A:	Quantitative and Qualitative Disclosures about Market Risk	34 - 35
Item 8:	Financial Statements and Supplementary Data	36 - 68
Item 9:	Changes in and Disagreements with Accountants on Accounting and Financial Disclosure	69
Part III		
Item 10:	Directors and Executive Officers of Registrant	69
Item 11:	Executive Compensation	69
Item 12:	Security Ownership of Certain Beneficial Owners and Management	69
Item 13:	Certain Relationships and Related Transactions	70
Item 14:	Controls and Procedures	70
Item 15:	Exhibits, Financial Statement Schedules and Reports on Form 8-K	71-72
	<u>Signature(s)</u>	73
	Certifications	74–75
	Report of Independent Accounts on Financial Statement Schedules	76
	Schedule I – Condensed Financial Information of Parent	77–79

## PART I

## Item 1. Business

This annual report on Form 10-K is a combined report being filed by two separate Registrants: American States Water Company (hereinafter "AWR") and Southern California Water Company (hereinafter "SCW"). References in this report to "Registrant" are to AWR and SCW, collectively, unless otherwise specified. SCW makes no representations as to the information contained in this report relating to AWR and its subsidiaries, other than SCW.

AWR makes its periodic reports, Form 10-Q and Form 10-K, and current reports, Form 8-K, available free of charge through its website, www.aswater.com, as soon as material is electronically filed with or furnished to the Securities and Exchange Commission (SEC).

#### General

AWR, incorporated in 1998, is engaged in the business of holding, for investment, the stock primarily of utility companies. AWR's principal investment is the stock of SCW. SCW is a California public utility company engaged principally in the purchase, production, distribution and sale of water (SIC No. 4941). SCW also distributes electricity in one customer service area (SIC No. 4911). SCW is regulated by the Public Utilities Commission of the State of California (CPUC) and was incorporated on December 31, 1929. SCW is organized into one electric customer service area and three water service regions operating within 75 communities in 10 counties in the State of California and provides water service in 21 customer service areas. Region I incorporates 7 customer service areas in northern and central California; Region II has 4 customer service areas located in Los Angeles County; Region III incorporates 10 water customer service areas in eastern Los Angeles County, and in Orange, San Bernardino and Imperial counties. SCW also provides electric service to the City of Big Bear Lake and surrounding areas in San Bernardino County through its Bear Valley electric service division.

AWR also owns two other subsidiaries. American States Utility Services, Inc. (ASUS) performs non-regulated, water related services and contract operations on a contract basis ranging from services, such as billing and meter reading, to full operation of water utility related systems. Chaparral City Water Company (CCWC) is an Arizona public utility company serving the town of Fountain Hills, Arizona and a portion of the City of Scottsdale, Arizona (SIC No. 4941). The Arizona Corporation Commission (ACC) regulates CCWC. Neither AWR nor ASUS is directly regulated by either the CPUC or the ACC.

SCW served 248,511 water customers and 21,987 electric customers at December 31, 2002, or a total of 270,498 customers, compared with 268,546 total customers at December 31, 2001. CCWC served 11,834 water customers as of December 31, 2002, compared with 11,353 customers at December 31, 2001. ASUS has approximately 90,000 accounts under contract.

#### **Competition**

The businesses of SCW and CCWC are substantially free from direct and indirect competition with other public utilities, municipalities and other public agencies. AWR's other subsidiary, ASUS, actively competes for business with other investor-owned utilities, other third party providers of water and wastewater services, and governmental entities on the basis of price and quality of service.

# **Employee Relations**

SCW had 488 employees as of December 31, 2002 as compared to 481 at December 31, 2001. Eighteen positions in SCW's Bear Valley Electric customer service area are covered by a collective bargaining agreement, which would have expired on December 31, 2002, with the International Brotherhood of Electrical Workers (IBEW). SCW and IBEW are currently renegotiating this contract and have mutually agreed to extend the existing contract until a new one is ratified. Sixty-one positions in SCW's Region II ratemaking district are covered by a collective bargaining agreement, which expires in 2004, with the Utility Workers of America. SCW has no other unionized employees.

CCWC had fifteen employees as of December 31, 2002, all of whom are non-unionized

# Item 2. Properties

# **Franchises**

SCW holds certificates of public convenience and necessity granted by the CPUC in each of the ratemaking districts it serves. CCWC holds certificates of public convenience and necessity granted by the ACC for the areas in which it serves. Both SCW and CCWC hold franchises, easements and rights of way pursuant to the terms of agreements that must periodically be renewed. These agreements are subject to suspension or termination in certain circumstances if SCW or CCWC, as applicable, violate the terms of these agreements.

### **Electric Properties**

SCW's electric properties are all located in the Big Bear area of San Bernardino County in California. As of December 31, 2002, SCW operated 29 miles of overhead 34.5 kv transmission lines, 1 mile of underground 34.5 kv transmission lines, 174 miles of 4.16 kv or 2.4 kv distribution lines, 42 miles of underground cable and 14 sub-stations. Neither AWR nor any of its subsidiaries

# Office Buildings

Registrant's general headquarters are housed in a single-story office building located in San Dimas, California. The land and the building are owned by SCW. SCW also owns and/or leases certain facilities housing regional, district and customer service offices. CCWC owns its primary office space.

## Water Properties

As of December 31, 2002, SCW's physical properties consisted of water transmission and distribution systems which included 2,651 miles of pipeline together with services, meters and fire hydrants and approximately 430 parcels of land, generally less than 1 acre each, on which are located wells, pumping plants, reservoirs and other water utility facilities, including five surface water treatment plants.

As of December 31, 2002, SCW owned 277 wells. Certain wells have been removed from service due to water quality problems. For further information, see the section entitled *"Environmental Matters"* included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation. All wells are equipped with pumps with an aggregate capacity of approximately 206 million gallons per day. SCW has 58 connections to the water distribution facilities of the Metropolitan Water District of Southern California (MWD) and other municipal water agencies. SCW's storage reservoirs and tanks have an aggregate capacity of approximately 107 million gallons. SCW owns no dams in its customer service areas. The following table provides, in greater detail, selected water utility plant of SCW for each of its water ratemaking districts:

\_\_\_\_\_

	Pu	ımps	Di	stribution Fac	ilities	Res	ervoirs
District	Well	Booster	Mains	Services	Hydrants	Tanks	Capacity
Arden Cordova	25	17	93	14,838	1,206	5	9,500
Barstow	23	36	162	8,553	1,013	13	7,525
Bay Point	3	12	31	4,900	343	7	4,046
Calipatria	0	8	25	1,168	128	8	13,241
Claremont	25	32	134	10,573	1,266	12	6,689
Clearlake	0	13	37	2,087	75	4	883
Desert	17	20	92	3,350	576	11	1,477
Los Osos	11	11	38	3,204	173	8	1,422
Metro	69	75	945	99,205	8,196	32	23,940
Ojai	5	12	45	2,789	350	5	1,494
Orange	31	38	416	41,505	4,602	15	11,900
San Dimas	11	38	222	15,915	1,600	15	10,147
San Gabriel	18	8	94	11,973	805	3	1,520
Santa Maria	30	21	183	12,780	783	8	3,076
Simi	2	24	97	13,064	891	8	8,250
Wrightwood	7	5	38	2,623	249	7	1,546
Total	277	370	2,652	248,527	22,256	161	106,656

Capacity is measured in thousands of gallons. Mains are in miles.

As of December 31, 2002, CCWC's physical properties consisted of water transmission and distribution systems, which included 180 miles of pipeline, together with services, meters, fire hydrants, wells, reservoirs with a combined storage capacity of 7.05 million gallons and other water utility facilities including a surface water treatment plant, which treats water from the Central Arizona Project.

#### Mortgage and Other Liens

As of December 31, 2002, SCW had no mortgage debt outstanding, and its properties were substantially free of any encumbrances or liens securing indebtedness. For further information, see the section entitled "Contractual, Obligations and Other Commitments" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

As of December 31, 2002, substantially all of the utility plant of CCWC was pledged to secure its Industrial Development Authority Bonds. The Bond Agreement, among other things, restricts CCWC's ability to incur debt and make liens, sell, lease or dispose of assets, or merge with another corporation, and pay dividends.

As of December 31, 2002, neither AWR nor ASUS had any mortgage debt or liens securing indebtedness outstanding.

#### **Condemnation of Properties**

The laws of the State of California and the State of Arizona provide for the acquisition of public utility property by governmental agencies through the power of eminent domain, also known as condemnation. Registrant has not been, within the last three years, involved in activities related to the condemnation of any of its water customer service areas or in its Bear Valley Electric customer service area.

### Item 3. Legal Proceedings

## Water Quality-Related Litigation

SCW is a defendant in twenty-two lawsuits involving claims pertaining to water quality. Nineteen of the lawsuits involve customer service areas located in Los Angeles County in the southern portion of the State of California that have been filed in Los Angeles Superior Court: Robert Arenas, et al. v. Suburban Water Systems, Inc., et al., Case No. KC037559, Anthony John Bell, et al. v. City of Pomona, et al., Case No. KC038796, Adler, et al. v. Southern California Water Company, et al., Case No. BC169892, Santamaria, et al. v. Suburban Water Systems, et al., Case No. KC028524, Abarca, et al. v. City of Pomona, et al., Case No. K027795, Celi, et al. v. Southern California Water Company, et al., Case No. GC021657, Anderson, et al. v. Suburban Water Systems, et al., Case No. KC027318, Demciuc et al. v. Suburban Water Systems, et al., Case No. KC027312, Antoinette Adejare, et al. v. City of Pomona, et al., Case No. KC034096, Almelia Brooks, et al. v. Suburban Water Systems, et al., Case No. KC028732, Antoinette Adejare, et al. v. City of Pomona, et al., Case No. KC034096, Almelia Brooks, et al. v. Suburban Water Systems, et al., Case No. KC032915, Lori Alexander, et al. v. Suburban Water Systems, et al., Case No. KC035296, Georgianna Company, et al., Case No. KC031130, David Arnold, et al. v. City of Pomona, et al., Case No. KC034963, Gilda Ambrose-Dubre, et al. v. City of Pomona, et al., Case No. KC032906, Melissa Garrity Alvarado, et al. v. Suburban Water Systems et al., Case No. KC034953, Charles Alexander, et al. v. City of Pomona, et al., Case No. KC035987. The lawsuits filed in Los Angeles County Superior Court are based on the allegations that SCW and the other defendants have provided and continue to provide plaintiffs with contaminated water from its wells, several of which are located in an area of the San Gabriel Valley that has been designated a federal superfund site, that the maintenance of this contaminated well wear thas resulted in contamination of the soil, subsurface soil and surrounding air with

On September 1, 1999, the Court of Appeals in San Francisco held that the CPUC had preemptive jurisdiction over regulated public utilities with respect to water quality matters and ordered dismissal of a series of these lawsuits. On October 11, 1999, one group of plaintiffs appealed this decision to the California Supreme Court. On February 4, 2002, the California Supreme Court concluded that (i) the CPUC had preemptive jurisdiction over claims seeking injunctive relief and claims based on the theory that a public utility regulated by the CPUC provided unsafe drinking water even though it had complied with federal and state drinking water standards, but (ii) the CPUC did not have preemptive jurisdiction over damage claims based on allegations of violations of federal and state drinking water standards may proceed while the other claims must be dismissed.

In light of the breadth of plaintiff's claims, the lack of factual information regarding plaintiffs' claims and injuries, if any, the impact of the California Supreme Court decision on plaintiffs' claims and the fact that investigation is presently underway, SCW is unable at this time to determine what, if any, potential liability it may have with respect to these claims. Based upon the information currently available to it, Registrant believes that these claims are without merit and intends to vigorously defend against these claims.

SCW is subject to self-insured retention provisions in its applicable insurance policies and has either expensed the self-insured amounts or has reserved against payment of these amounts as appropriate. SCW's various insurance carriers have, to date, provided reimbursement for costs incurred above the self-insured amounts for defense against these lawsuits, subject to a reservation of rights.

# Order Instituting Investigation (OII)

In March 1998, the CPUC issued an OII to regulated water utilities in the state of California, including SCW. The purpose of the OII was to determine whether existing standards and policies regarding drinking water quality adequately protect the public health and whether those standards and policies were being uniformly complied with by those water utilities. On November 2, 2000, a final decision from the CPUC concluded that the CPUC has the jurisdiction to regulate the service of water utilities with respect to the health and safety of that service; that the California Department of Health Services requirements governing drinking water quality adequately protect the public health and safety; and that, based on an extensive review of 25 years of water quality records, regulated water utilities, including SCW, have satisfactorily complied with past and present drinking water quality requirements, except for one small unrelated water utility in Northern California.

The CPUC had previously authorized establishment of memorandum accounts to capture expenses related to the OII. Under the memorandum account procedure, SCW may recover litigation costs from ratepayers to the extent authorized by the CPUC. As of December 31, 2002, SCW had incurred a net cost of \$890,000 related to the OII, for which a reserve of the same amount was established. SCW filed for recovery of these amounts, through a special condition surcharge over a 12 month period, with the CPUC in its application filing for its Region III and general office functions. See the section entitled "*Regulatory Matters*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Conditions and Results of Operation.

## Other Water Quality Litigation

On October 25, 1999, SCW filed a lawsuit against the State of California and its State Water Resources Control Board, Central Valley Regional Water Quality Control Board, and Department of Toxic Substances Control (collectively, the State) alleging that the State had substantially participated in a project to inject chemical pollution into portions of the Sacramento County groundwater basin and that pollution is progressively destroying the groundwater supply in SCW's Rancho Cordova water system. SCW and the State have entered into a comprehensive \$2,475,000 settlement of all of SCW's claims against the State, contingent upon the court's approval of the good faith of the State's settlement offer.

In a separate case, also filed on October 25, 1999, SCW sued Aerojet-General Corporation (Aerojet) for causing the contamination of eastern portions of the Sacramento County groundwater basin. A cross complaint

4		
4		

filed by Aerojet against SCW for negligence and constituting a public nuisance was dismissed by the court in October 2002.

The CPUC has authorized memorandum accounts to allow for recovery, from customers, of costs incurred by SCW in prosecuting the suits filed against the State and Aerojet, less any recovery from the defendants or others. As of December 31, 2002, approximately \$13.7 million in legal and other related costs has been recorded in the deferred charges.

The CPUC has authorized SCW to increase rates, effective April 28, 2001, for recovery over a six-year period of approximately \$1.8 million, in expenses that were incurred on or before August 31, 2000, in the Aerojet matter. The remaining costs were included in SCW's January 31, 2003 filing with the CPUC of its Notice of Intent to increase rates in Region I. Management believes these costs are recoverable based on past practices of the CPUC but cannot give assurance that the CPUC will ultimately allow recovery of all or any of the remaining costs through rates.

The compound MTBE has been detected in a well serving SCW's Los Osos water system. For some time SCW has been working with the Regional Water Quality Control Board as well as the owner of a service station. Although the owner of the service station has attempted remediation with funds provided through the California Underground Storage Tank Fund ("CUSTF"), it appears there will be insufficient funds from the CUSTF to complete remediation sufficient to return the well to service. In order to prevent the running of the statute of limitations, on August 12, 2002 SCW filed suit in the Superior Court of the State of California for the County of San Luis Obispo against the operators and owners of the service station facility, and Chevron USA Products, Inc., the supplier.

Volatile Organic Compounds (VOC) and perchlorate have been detected in 2 wells servicing SCW's San Gabriel System. SCW filed suit, along with two other affected water purveyors and the San Gabriel Basin Water Quality Authority (WQA), in the federal court against some of those responsible for the contamination. Some of the other potential defendants settled with SCW, other water purveyors and the WQA on VOC related issues prior to the filing of the lawsuit. In response to the filing of the Federal lawsuit, the Potentially Responsible Party (PRP) defendants filed motions to dismiss the suit or strike certain portions of the suit. Following a hearing on these motions on March 31, 2003, the judge issued a ruling on April 1, 2003 granting in part and denying in part the defendant's motions. A key ruling of the court was that the water purveyors, including the Registrant, by virtue of their ownership of wells contaminated with hazardous chemicals are themselves PRPs under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA). Registrant is currently evaluating the potential impact of this decision, its opportunity to appeal the decision and the opportunity to amend its suit to claim certain affirmative defenses as an "innocent" party under CERCLA. Registrant is presently unable to predict the outcome of this ruling on its ability to fully recover from the PRPs future costs associated with the treatment of these wells.

#### Electric Service Litigation

SCW has been, in conjunction with the Southern California Edison (Edison) unit of Edison International, planning to upgrade transmission facilities to 115kv (the 115kv Project) in order to meet increased energy and demand requirements for SCW's Bear Valley Electric Service area. On December 27, 2000, SCW filed a lawsuit against Edison for declaratory relief and seeking damages for breach of contract as a result of delays in the 115kv Project, violations of good faith and fair dealing, negligent misrepresentation, intentional misrepresentation and unjust enrichment. Subsequently Edison filed a cross-complaint against SCW for breach of contract, anticipatory breach, and quantum meruit. SCW has also sought declaratory relief from Edison's claims. To date, SCW has spent approximately \$1.7 million in this matter, all of which has been expensed. This matter is expected to go to trial during the second quarter of 2003.

#### Other Litigation

Registrant is also subject to ordinary routine litigation incidental to its business. Other than as disclosed above, no other legal proceedings are pending, which are believed to be material.

5

## Item 4. Submission of Matters to a Vote of Security Holders

No matter was submitted during the fourth quarter of the fiscal year covered by this report to a vote of security holders through the solicitation of proxies or otherwise.

PART II

# Item 5. Market for Registrant's Common Equity and Related Shareholder Matters

### Market Information Relating to Common Shares -

Common Shares of American States Water Company are traded on the New York Stock Exchange (NYSE) under the symbol AWR. The intra-day high and low NYSE prices on the Common Shares for each quarter during the past two years were:

	High	Low
2002		
First Quarter	\$25.06	\$21.94
Second Quarter	28.00	23.38
Third Quarter	27.55	20.25
Fourth Quarter	\$27.20	\$23.15
2001		
First Quarter	\$24.92	\$19.17
Second Quarter	22.67	19.00
Third Quarter	26.40	21.60
Fourth Quarter	\$25.33	\$21.47

## Approximate Number of Holders of Common Shares

As of March 26, 2003, there were 3,438 holders of record of Common Shares of American States Water Company. AWR owns all of the authorized and outstanding Common Shares of SCW, CCWC and ASUS.

### Frequency and Amount of Any Dividends Declared and Dividend Restrictions

For the last three years, AWR has paid dividends on its Common Shares on March 1, June 1, September 1 and December 1. The following table lists the amount of dividends paid on Common Shares of American States Water Company, adjusted by the 3-for-2 stock split completed on June 7, 2002, for the last two years:

	2002	2001
First Quarter	\$ 0.217	\$ 0.217
Second Quarter	0.217	0.217
Third Quarter	0.217	0.217
Fourth Quarter	0.221	0.217
Total	\$ 0.872	\$ 0.867

Neither AWR nor ASUS is subject to any contractual restriction on its ability to pay dividends. SCW's maximum ability to pay dividends is restricted by certain Note Agreements to the sum of \$21 million plus 100% of consolidated net income plus the aggregate net cash proceeds received from capital stock offerings or other instruments convertible into capital stock from various dates. Under the most restrictive of the Note Agreements, \$154.7 million was available to pay dividends to AWR. For further information, see (3) of the section entitled "*Contractual Obligations and Other Commitments*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Conditions and Results of Operation.

6

The ability of AWR, ASUS and SCW to pay dividends is also restricted by California law. Under restrictions of the California tests, approximately \$86.3 million of retained earnings for AWR was available to pay dividends to Common Shareholders at December 31, 2002. Approximately \$84.2 million was available from the retained earnings of SCW to pay dividends to AWR. At December 31, 2002, ASUS was not allowed to pay dividends to AWR under the California tests.

CCWC is subject to contractual restrictions on its ability to pay dividends. CCWC's maximum ability to distribute dividends is limited to maintenance of no more than 55% debt in the capital structure for the quarter immediately preceding the distribution. The ability of CCWC to pay dividends is also restricted by Arizona law. Under restrictions of the Arizona tests, approximately \$4 million was available to pay dividends to AWR at December 31, 2002. See (8) of the section entitled "*Contractual Obligations and Other Commitments*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Conditions and Results of Operation.

AWR paid \$13.2 million and \$28,700 in common and preferred dividends, respectively, to shareholders for the year ended December 31, 2002, as compared to \$13.1 million and \$84,700 for the year ended December 31, 2001. All outstanding preferred shares were redeemed during 2002. SCW paid dividends of \$14.6 and \$13.5 million to AWR in 2002 and 2001, respectively.

Information concerning securities authorized for issuance by AWR under its equity compensation plans is included in AWR's proxy statement for its 2003 annual meeting and is incorporated by reference herein.

#### **Other Information**

The certifications of Registrant's Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, were filed as correspondence with the Securities and Exchange Commission concurrent with this Form 10-K.

## Item 6. Selected Financial Data

( in thousands, except per share amounts)	2002	2001 Restated (1)	2000 Restated (1)	1999 Restated (1)	1998 Restated (1)
Income Statement Information					
Total Operating Revenues	\$ 209,20	\$ 197,514	\$ 183,960	\$173,421	\$ 148,060
Total Operating Expenses	171,55	159,813	149,708	144,960	124,178
Operating Income	37,64	37,701	34,252	28,461	23,882
Other Income (Loss)	39	) (510)	(99)	532	769
Interest Charges	17,69	15,735	14,122	12,945	11,207
Net Income	20,33	21,456	20,031	16,048	13,444
Preferred Dividends	(29	9) (84)	(86)	(88)	(90)
Earnings Available for Common Shareholders	\$ 20,31		\$ 19,945	\$ 15,960	\$ 13,354
Basic Earnings per Common Share	\$ 1.34		\$ 1.42	\$ 1.19	\$ 0.99
Dividends Declared per Common Share	\$ 0.872	\$ 0.867	\$ 0.857	\$ 0.853	\$ 0.84
Average Shares Outstanding	15,14	15,120	14,070	13,437	13,437
Average Number of Diluted Shares Outstanding	15,15		14,070	N/A	N/A
Fully Diluted Earnings per Common Share	\$ 1.34	\$ 1.41	\$ 1.42	N/A	N/A
Balance Sheet Information		<b>* * * * * *</b>		¢ 500 404	<b>.</b>
Total Assets	\$ 701,65	. ,	\$ 616,646	\$ 533,181	\$484,671
Common Shareholders' Equity	213,27	,	196,386	160,564	156,070
Long-Term Debt	231,08		176,452	167,363	120,809
Preferred Shares-Not Subject to Mandatory Redemption	-	- 1,600	1,600	1,600	1,600
Preferred Shares-Mandatory Redemption	_	- 280	320	360	400

7

(1) Restated due to overstatements in net deferred tax liabilities and current taxes payable for prior years. See the section entitled "Introduction" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation and Note 2 of "Notes to Financial Statements" included in Part II, Item 8 in Financial Statements and Supplementary Data.

# Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operation

Unless specifically noted, the following discussion and analysis provides information on AWR's consolidated operations and assets. For the twelve months ended December 31, 2002, there is generally no material difference between the consolidated operations and assets of AWR and the operations and assets of SCW. However, where necessary, the following discussion and analysis includes references specifically to AWR's other subsidiaries – CCWC and ASUS.

## **Forward-Looking Information**

Certain matters discussed in this report (including the documents incorporated herein by reference) are forward-looking statements intended to qualify for the "safe harbor" from liability established by the Private Securities Litigation Reform Act of 1995. These forward-looking statements can generally be identified as such because the context of the statement will include words such as Registrant "believes," "anticipates," "expects" or words of similar import. Similarly, statements that describe Registrant's future plans, objectives, estimates or goals are also forward-looking statements. Such statements address future events and conditions concerning capital expenditures, earnings, litigation, rates, water quality and other regulatory matters, adequacy of water supplies, SCW's ability to recover electric and water supply costs from ratepayers, liquidity and capital resources, and accounting matters. Actual results in each case could differ materially from those currently anticipated in such statements, by reason of factors such as changes in utility regulation, including ongoing local, state and federal activities; future economic conditions, including changes in customer demand and changes in water and energy supply cost; future climatic conditions; and legislative, regulatory and other circumstances affecting anticipated revenues and costs.

#### **Introduction**

As described in Note 2 to the accompanying consolidated financial statements included in Item 8 of this Annual Report on Form 10-K, Registrant has restated its financial statements for the years ended December 31, 2001 and 2000. In connection with a deferred tax analysis of Registrant's tax accounts performed in 2002, Registrant determined that it had miscalculated its net accumulated deferred income taxes and current tax payable account over several years, resulting in taxes on income not being recognized in the proper period, and overstating Registrant's net deferred tax liabilities and current taxes payable. Registrant further determined that the amount of the overstatements, which totaled approximately \$4.7 million and \$4.9 million for AWR and SCW, respectively, arose in prior years and that prior years' financial statements should be restated to reduce net accumulated deferred income taxes, taxes payable and the related tax expense to reflect amounts currently due and deferred. The financial information for all periods included in these consolidated financial statements gives effect to the restatement.

The following table summarizes the effects of the restatement on Registrant's consolidated financial statements for 2001 and 2000:

		December 31, 2001				
		AWI	R	SC	w	
ollars in thousands)	Previou Report		Restated	Previously Reported	Restated	
onsolidated Balance Sheets:						
Regulatory tax-related assets	\$ 15,8	43	\$ 13,908	\$ 15,843	\$ 13,908	
Taxes payable	(5,3	39)	(404)	(5,599)	(1,194)	
Accumulated deferred income taxes - net	(53,4	44)	(51,541)	(52,075)	(49,399)	
Regulatory tax-related liability	(1,7	73)	(2,054)	(1,773)	(2,054)	
Earnings reinvested in the business	(74,5	44)	(79,216)	(72,716)	(77,581)	
Lannings remvested in the business	(74,-	(17)	(7),210)	(12,110)	(77,-	

8

		Year Ended December 31, 2001			
	A	AWR		W	
(dollars in thousands except EPS)	Previously Reported	Restated	Previously Reported	Restated	
Statements of Income:					
Taxes on income	\$ 15,379	\$ 14,370	\$ 15,066	\$ 14,057	
Operating income	36,692	37,701	35,030	36,039	
Net income	20,447	21,456	19,829	20,838	
Basic earnings per share	\$ 1.35	\$ 1.41	\$ 185,318	\$ 194,748	
Diluted earnings per share	\$ 1.33	\$ 1.41	N/A	N/A	

		Year Ended December 31, 2001			
		AWR		SCW	
ars in thousands except EPS)	Previous Reported		stated	Previously Reported	Restated
tements of Income:					
axes on income	\$ 15,12	7 \$ 1	3,182	\$ 14,881	\$ 12,743
Derating income	32,30	7 3	34,252	32,175	34,313
let income	18,08	6 2	20,031	17,685	19,823
asic earnings per share	\$ 1.2	8 \$	1.42	\$ 176,850	\$ 198,230
Diluted earnings per share	\$ 1.2	8 \$	1.42	N/A	N/A

# **Results of Operations**

Basic earnings per common share in 2002 decreased by 5% to \$1.34 as compared to \$1.41 per common share for the comparable period of 2001. The decreases in the recorded results primarily reflect the lack of timely approval of revenue increases by the CPUC, and the impact of the removal of balancing account treatment for supply costs enacted by the CPUC in November 2001. Partially offsetting these items was the reversal of a reserve established in 2001 for potentially unrecoverable purchased power costs and various other reasons as discussed below. For the year ended December 31, 2002, fully diluted earnings were \$1.34 per common share as compared to \$1.41 per common share for the comparable period of 2001. Earnings per common share reflect the 3-to-2 stock split of common shares completed on June 7, 2002 for both periods.

Water operating revenues increased by 3.1% in 2002 to \$187.1 million from the \$181.5 million reported in 2001 due to an increase of 3.3% in consumption. There were no significant water rate increases effective during 2002. Differences in temperature and rainfall in Registrant's service areas will impact sales of water to customers and may cause fluctuations in Registrant's revenues and earnings between comparative periods. For more information, see the section entitled "*Water Supply*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Electric revenues increased by 39.2% to \$21.3 million in 2002 as compared to \$15.3 million in 2001. The increases reflected (i) a rate increase of 12.5% effective May 24, 2001 and an additional 14.8% increase effective August 23, 2001 authorized by the CPUC to recover previously under-collected energy costs, (ii) rate increases to cover purchased power costs under various power supply agreements, and (iii) an increase of 3.9% in kilowatt-hour consumption, primarily due to heavier use of snow making machines at ski resorts in the area in the fourth quarter of 2002. For further information, see the section entitled "*Electric Energy Situation in California*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Registrant relies upon rate increases for water and electric services approved by regulatory bodies in California and Arizona in order to recover operating expenses and provide for a return on invested and borrowed capital used to fund utility plant. Without such adequate rate relief granted in a timely manner, revenues and

9

earnings can be negatively impacted. For more information, see the section entitled "Regulatory Matters" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Purchased water costs in 2002 increased by 14.1% to \$42.9 million as compared to \$37.6 million in 2001, reflecting an increase in purchased water volume resulting from (i) higher water consumption, (ii) additional purchased water necessary in 2002 to replace pumped water supply lost due to wells being removed from service as a result of water quality issues and mechanical problems, particularly in SCW's Orange County and San Dimas customer service areas, (iii) refunds received from SCW's wholesale water suppliers during 2001 of approximately \$777,000 for which there was no counterpart in 2002, and (iv) costs of approximately \$1.3 million, incurred most significantly in connection with the trucking of water, in SCW's Wrightwood customer service area due to a sudden and unexplained drop in groundwater level in the area during the early part of the third quarter of 2002.

Registrant has had to take a number of wells out of service from time-to-time due to various causes, including contamination by third parties. Changes in the water resource mix between water supplied from purchased sources and that supplied from Registrant's own wells can increase actual supply-related costs relative to the supply-related costs approved for recovery through rates, thereby impacting earnings either negatively or positively. Registrant has the opportunity to change the supply-related costs recovered through rates by application to the appropriate regulatory body. Registrant believes that its applications for recovery of supply-related costs accurately reflect the water supply situation as it is known at the time. However, it is impossible to adequately protect earnings from adverse changes in supply costs related to unforeseen contamination. For more information, see the sections entitled "*Regulatory Matters*," "*Environmental Matters*" and "*Water Supply*" in Part II, Item 7 of Management's Discussion and Analysis of Financial Condition and Results of Operation.

Costs of power purchased for pumping increased by 10.4% to \$10.6 million in 2002 as compared to \$9.6 million recorded in 2001, due to the rate increases implemented in 2001 by SCW's energy suppliers, in particular Southern California Edison Company and Pacific Gas and Electric Company. In 2001, the CPUC approved a portion of SCW's advice letters to increase revenues by approximately \$1.4 million annually to recover the costs of purchased power for certain of its water ratemaking districts. A credit of \$440,000 recorded in the fourth quarter of 2001 for the sale of groundwater in the Chino Basin also impacted the comparison. There was no such sale in 2002. For further information, see the sections entitled "*Regulatory Matters*" and "*Electric Energy Situation in California*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Costs of power purchased for resale to customers in SCW's Bear Valley Electric division in 2002 decreased by 7.1% to \$18.3 million from the \$19.7 million recorded in 2001 due primarily to lower power costs in the first quarter of 2002 under SCW's power supply agreements to cover peak winter demand as compared to the costs for energy purchased on the spot market in the same period of 2001. Offsetting the decrease were (i) an unrealized loss of \$2.5 million recorded for SCW's power purchase agreements effective November 2002 with Pinnacle West Capital Corporation (PWCC), which qualify as derivative instruments under the Statement of Financial Accounting Standards No. 133, "Accounting for Derivative Instruments and Hedging Activities", and (ii) a one-time sale of energy on the spot market that generated a \$644,000 gain in April 2001. The gain is the subject of a complaint filed at the Federal Energy Regulatory Commission by Mirant Americas Energy Marketing LP (Mirant) where Mirant, the purchaser of the energy, is seeking to be refunded all or a portion of the gain. The sale of excess energy on the spot market in 2001 resulted from a one-month overlap of energy purchase agreements. For further information, see the sections entitled "*Electric Energy Situation in California*" included in Part II, Item 7 and "*Quantitative and Qualitative Disclosures About Market Risk*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation, as well as Note 1 of "*Notes to Financial Statements*" included in Part II, Item 8 in Financial Statements and Supplementary Data.

Groundwater production assessments increased by 8.8% to \$7.4 million in 2002 from \$6.8 million in 2001. The increase reflects (i) an increase in well production and higher assessments levied against production for the water year 2002/2003 as compared with the previous year in SCW's Metropolitan customer service area, and (ii) increased pumped costs in SCW's Orange County customer service area due to an in-lieu program implemented by the Metropolitan Water District of Southern California (MWD) to help preserve groundwater levels in the area, which allows SCW to purchase treated water from MWD essentially at SCW's groundwater

10

pumping costs. The increases were partially offset by revenues from leases of unused water rights to third parties in SCW's Barstow customer service area in 2002.

A positive entry for the provision for supply cost balancing accounts reflects recovery of previously under-collected supply costs. Conversely, a negative entry for the provision for supply costs balancing accounts reflects an under-collection of previously incurred supply costs. The negative entries for 2001 and 2002 primarily reflect untimely-recovery of electric power costs discussed previously. At December 31, 2002, SCW had a combined net under-collected position of \$27.5 million in both its water and electric balancing accounts, primarily due to the increases in energy costs. For further information, see the sections entitled "*Accounting for Supply Costs*", "*Regulatory Matters*" and "*Electric Energy Situation in California*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Other operating expenses decreased slightly in 2002 to \$17.0 million as compared with \$17.2 million in 2001 due primarily to (i) a refund for a sewer service overpayment, (ii) reimbursement received during 2002 from the settling parties for expenses incurred in the San Gabriel basin to meet water quality standards for which there was no counterpart in 2001, and (iii) a decrease in the accrual for bad debt, net of increased water treatment costs, chemical costs and higher labor costs.

Administrative and general expenses decreased by 14.5% in 2002 to \$30.0 million from \$35.1 million recorded in 2001 reflecting reversal of the remaining reserve of \$6.5 million for potential non-recovery of electric power costs incurred to serve customers at SCW's Bear Valley Electric customer service area. The reserves were established in 2001 to offset future impacts to earnings for the difference between authorized rates and SCW's actual electric power costs under its agreement with Mirant Americas Energy Marketing, LP. SCW charged off \$1.7 million against this reserve in 2002. The agreements SCW entered into with PWCC in September 2002 enable it to purchase power at a cost lower than that authorized by the CPUC, thereby removing the need for the reserve. The decrease was partially offset by increased reserves for self-insured worker's compensation liabilities, higher labor and benefits related costs, and increased outside service expenses. Registrant anticipates that administrative and general administrative and staff personnel, and costs associated with pursuit of additional business lines not subject to regulation. Registrant believes that prudent administrative expenses incurred in the operation and management of its regulated subsidiaries will be recovered through water and electric rates. For further information, see the sections entitled "*Regulatory Matters*" and "*Electric Energy Situation in California*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation and Note 11 of "*Notes to Financial Statements*" included in Part II, Item 8 in Financial Statements and Supplementary Data.

Depreciation expense in 2002 increased by 2.0% to \$18.3 million reflecting, among other things, the effects of recording approximately \$38.0 million in utility plant additions during 2001, depreciation on which began in January 2002. Registrant anticipates that depreciation expense will continue to increase due to Registrant's on-going construction program at its regulated subsidiaries. Registrant believes that depreciation expense related to property additions approved by the appropriate regulatory bodies will be recovered through water and electric rates. For more information, see the section entitled "*Construction Program*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Conditions and Results of Operation. The comparison was also impacted by the elimination, effective January 1, 2002, of amortization of the goodwill recorded in AWR's acquisition of CCWC. During 2001, amortization of this goodwill was \$331,073. Pursuant to Financial and Accounting Standards Board No. 142, AWR has determined that this goodwill was not impaired at January 1, 2002 and January 1, 2003. For further information, see Note 1 of "*Notes to Financial Statements*" included in Part II, Item 8 in Financial Statements and Supplementary Data.

As compared to 2001, maintenance expense increased by 13.9% to \$9.8 million due principally to the termination of SCW's Cash Preservation Program ("CPP") in August 2002 following the CPUC's approval of rate increases permitting SCW to begin recovery of power costs incurred during the energy crisis of 2000-2001 to serve customers of its Bear Valley Electric division. The CPP

was initially implemented in 2001 to control costs and temporarily limit capital and maintenance expenditures principally to those projects that were believed necessary to meet public safety and health requirements or otherwise provide for continued service. The CPP impacted both the electric and water businesses of SCW. For further information, see the sections entitled "*Electric Energy Situation in California*" and "*Regulatory Matters*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation. In addition to maintenance routinely

11

performed on Registrant's facilities at its regulated subsidiaries, maintenance expense can be impacted by water main breaks or other incidents related to older infrastructure. Registrant's construction program is in part designed to minimize the impact the aging infrastructure has on maintenance costs. For more information, see the section entitled "Construction Program" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Taxes on income decreased by 9.9% to \$12.9 million in 2002 as compared to \$14.4 million in 2001 due to an approximate 9.5% decrease in pre-tax operating income. Registrant estimates its taxes payable and exposure based on a variety of factors and judgments applied by management and reviewed by Registrant's tax advisor. During 2002, Registrant determined that current taxes payable and deferred tax payable had been overstated cumulatively by \$4.7 million. This determination was based on a reconciliation of the balance in the account to amounts payable to Registrant's taxing jurisdictions, made in conjunction with the results of an analysis of Registrant's deferred tax for a period of 20 years. Registrant further determined that the amount of the overstatement arose in prior years and that prior years' financial statements should be restated to reduce taxes payable and the related current tax expense to reflect amounts currently due. For more information, see the section entitled "*Introduction*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation and Note 2 of "*Notes to Financial Statements*" included in Part II, Item 8 in Financial Statements and Supplementary Data.

Property and other tax expense increased slightly by 1.7% in 2002 to \$7.7 million reflecting principally increased property taxes due to higher property valuation assessments, and higher payroll taxes based on increased labor costs.

As compared to 2001, the change in other income reflects (i) the sale of a parcel of non-operating property in SCW's Metropolitan customer service area in the first quarter of 2002 for which there was no equivalent sale in 2001, (ii) recording certain non-regulated expenses to administrative and general expenses of ASUS, a non-regulated subsidiary, and (iii) the write-off of expenses associated with the termination of the acquisition of Peerless Water Co. in the fourth quarter of 2001.

Interest expense increased by 12.7% in 2002 to \$17.7 million as compared to \$15.7 million recorded in 2001 due to the issuance of \$50 million in long-term debt by SCW in December 2001, partially offset by a reduction in short-term borrowing costs. Registrant anticipates that interest costs will increase in future periods due to the need for additional external capital to fund its construction program and to the likelihood that interest rates will increase at some future point. Registrant believes that costs associated with capital used to fund construction at its regulated subsidiaries will be recovered in water and electric rates charged to customers. For more information, see the section entitled *"Liquidity and Capital Resources"* in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

## Years Ended December 31, 2001 and 2000

Basic earnings per common share in 2001 decreased slightly by 0.7% to \$1.41 as compared to \$1.42 per common share for the comparable period of 2000. The decrease in the recorded results primarily reflects the booking of a reserve for unrecoverable purchased power costs incurred to serve customers at SCW's Bear Valley Electric customer service area. The decrease was offset by various rate increases authorized by the CPUC for SCW, additional revenues generated by CCWC since the acquisition in October 2000, improvement in operating margins and, to some extent, SCW's cash preservation plan which was terminated in August 2002. For the year ended December 31, 2001, fully diluted earnings were \$1.41 per common share as compared to \$1.42 per common share for the comparable period of 2000. For further information, see the section entitled "*Liquidity and Capital Resources*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation. Earnings per common share have been adjusted by the 3-to-2 stock split of common shares completed on June 7, 2002.

Water operating revenues increased by 7.5% in 2001 to \$181.5 million from the \$168.8 million reported in 2000 due to increases in water rates authorized by the CPUC, and an additional \$5 million in revenues generated by CCWC. New rates, representing an annualized increase of \$3.1 million, in the customer service areas that comprise SCW's Region I were implemented during 2001. Rate increases, representing an annualized increase of \$2.9 million, for SCW's Region II and rate increases for SCW's Region III, representing an annualized increase of \$3.9 million, were also implemented at various times during 2001. The additional revenues generated by rate

12

increases were partially offset by a 3.4% reduction in water sales throughout most of SCW's customer service areas in 2001 due to relatively mild weather. For further information, see the section entitled "Regulatory Matters" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Electric revenues increased by 6.3% to \$15.3 million in 2001 as compared to \$14.4 million in 2000. The increases reflected a rate increase of 12.5% effective May 24, 2001 and an additional 14.8% increase effective August 23, 2001 authorized by the CPUC to recover previously under-collected energy costs. The increases were partially offset by a decrease of 5.3% in kilowatt-hour consumption, primarily due to heavier winter snows experienced in SCW's service area in 2001, which decreased the use of snow making machines at ski resorts in the area. For further information, see the sections entitled "*Regulatory Matters*" and "*Electric Energy Situation in California*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Purchased water costs in 2001 decreased by 9.6% to \$37.6 million as compared to \$41.6 million in 2000 reflecting a decrease in purchased water volume resulting from both lower sales and less purchased water in SCW's supply mix due to additional well production capability coming on-line during the year as well as refunds received from Registrant's wholesale water suppliers during 2001 of approximately \$770,000. There was no similar refund in 2000. Purchased water expense at CCWC was approximately \$497,000 in 2001.

Costs of power purchased for pumping increased by 27.7% to \$9.6 million in 2001 as compared to \$7.5 million recorded in 2000, due to the rate increases implemented by SCW's energy suppliers pursuant to CPUC decisions, and increased pumped water in SCW's supply mix. In 2001, the CPUC approved SCW's Advice Letters to increase revenues by approximately \$1.4 million annually to recover the costs of purchased power for its water ratemaking districts. For further information, see the sections entitled "*Regulatory Matters*" and "*Electric Energy Situation in California*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation. During 2001, CCWC incurred approximately \$423,000 in power costs used for pumping.

Costs of power purchased for resale to customers in SCW's Bear Valley Electric division in 2001 increased by 84.1% to \$19.7 million from the \$10.7 million recorded in 2000 due primarily to significant increases in wholesale market prices for energy in the State of California. The increase was partially offset by a one-time sale of energy on the spot market that resulted in a \$644,000 pre-tax gain in April 2001. The sale of excess energy on the spot market resulted from a one-month overlap of energy purchase agreements. For further information, see the sections entitled "*Liquidity and Capital Resources*", "*Regulatory Matters*" and "*Electric Energy Situation in California*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Groundwater production assessments decreased by 9.3% to \$6.8 million in 2001 from \$7.5 million in 2000. The decrease occurred principally in SCW's San Gabriel and San Dimas customer service areas due to reduced pumped water volume in their supply mix and lower administrative assessments levied against production for the water year 2001 as compared with 2000.

A positive entry for the provision for supply cost balancing accounts reflects recovery of previously under-collected supply costs. Conversely, a negative entry for the provision for supply cost balancing accounts reflects an under-collection of previously incurred supply costs. The negative entries for 2001 primarily reflect untimely recovery of electric power costs discussed previously. At December 31, 2001, SCW had a net under-collected position of \$25.8 million in both its water and electric balancing accounts primarily due to the increases in energy costs. For further information, see the sections entitled "Accounting for Supply Costs", "Liquidity and Capital Resources", "Regulatory Matters" and "Electric Energy Situation in California" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Other operating expenses increased by 3.0% in 2001 to \$17.2 million as compared with \$16.7 million in 2000 due primarily to operating expenses at CCWC, offset by the effects of the cash preservation plan at SCW that reduced or deferred a number of operating projects.

Administrative and general expenses increased by 34.5% to \$35.1 million in 2001 from \$26.1 million recorded in 2000 reflecting (i) reserves of \$7.9 million established for potential non-recovery of electric power costs incurred to serve customers at SCW's Bear Valley Electric customer service area, (ii) increased reserves for

self-insured worker's compensation liabilities, and (iii) additional costs from CCWC. The reserves were established to offset future impacts to earnings in the event that SCW was unable to fully recover all of its purchased power costs through rates. For further information, see the sections entitled "Regulatory Matters" and "Electric Energy Situation in California" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation and Note 11 of "Notes to Financial Statements" included in Part II, Item 8 in Financial Statements and Supplementary Data.

Depreciation expense in 2001 increased by 17.6% to \$18.0 million reflecting, among other things, the effects of recording approximately \$40.1 million in net plant additions at SCW during 2000, depreciation on which began in January 2001, and additional depreciation associated with CCWC's plant. In addition, amortization of goodwill, which represents the difference between the purchase price of the common equity of CCWC and CCWC's book equity at the time of closing, began in October 2000. Amortization of this goodwill was \$331,073 in 2001.

As compared to 2000, maintenance expense decreased by 16.5% to \$8.6 million due primarily to the implementation of SCW's cash preservation plan in April 2001. The cash preservation plan impacted both the electric and water businesses of SCW. Management estimates that the cash preservation plan, through deferral of capital projects alone, reduced cash expenditures in 2001 by approximately \$20 million.

Taxes on income increased slightly by 9% to \$14.4 million in 2001 as compared to \$13.2 million in 2000 due primarily to an increase of 9.1% in pre-tax operating income. For more information, see the section entitled "*Introduction*" in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation and Note 2 of "*Notes to Financial Statements*" included in Part II, Item 8 in Financial Statements and Supplementary Data.

Property and other tax expense increased by 7.0% in 2001 to \$7.6 million reflecting principally increased property taxes due to higher property valuation assessments, and additional property and payroll taxes at CCWC.

The loss recorded in other income for 2001 was due principally to the write-off of expenses associated with the termination of the acquisition of Peerless Water Co. The loss also reflects the effects of recording amortization and interest expenses, starting January 2000, on SCW's 500 acre-foot entitlement in the State Water Project (SWP). During the fourth quarter of 2001, SCW signed an agreement with a property developer requiring an assured water supply to complete construction of a development. Under the terms of this Agreement, the developer will reimburse SCW for the costs related to 350 acre-feet of SCW's 500 acre-foot SWP entitlement.

Interest expense increased by 11.3% in 2001 to \$15.7 million as compared to \$14.1 million recorded in 2000 due to (i) short-term borrowing to fund capital expenditures, (ii) the issuance of \$20 million in long-term debt by SCW in January 2001, (iii) the issuance of \$50 million in long-term debt by SCW in December 2001, and (iv) the inclusion of long-term debt at CCWC in AWR's financial results.

#### Accounting for Supply Costs

As permitted by the CPUC, SCW has historically maintained water and electric supply balancing accounts to account for under-collections and over-collections of revenues designed to recover such costs. Costs have been recorded in income and charged to balancing accounts when such costs were incurred. The balancing accounts were reversed when such costs were recovered through rate adjustments or through refunds of previously incurred costs. SCW accrued interest on its supply cost balancing accounts at the rate prevailing for 90-day commercial paper. CCWC does not maintain a supply cost balancing account.

On November 29, 2001, the CPUC ordered water utilities with existing water supply balancing accounts to cease booking amounts to such accounts. In its place, water utilities are now required to establish a memorandum account. The over- or under-collection of water supply costs are recorded in this memorandum account in a manner similar to the balancing account. However, the CPUC has not issued a final decision regarding the mechanism for recovery of costs recorded in the newly established memorandum account. SCW's recovery of deferred water supply costs for providing water service may be reduced if SCW is earning more than its authorized rate of return on a weather normalized basis. The income statements of SCW will no longer include entries reflecting differences between authorized water supply unit costs included in rates and actual water supply costs. As a result, any changes in water supply costs as well as any future authorized revenue increases for supply expenses may directly impact earnings. SCW's basic earnings per share for the twelve

months ended December 31, 2002 were \$0.16 less than they would have been if the November 2001 CPUC order had not been issued. SCW anticipates recovering the under-collection tracked in the water supply cost memorandum account as part of Region III rate case application filing submitted in October 2002. SCW may not, however, be able to fully recover the under-collection of supply costs if it is earning a rate of return in excess of that allowed.

SCW had a net under-collection position of \$2.6 million recorded in its water supply balancing account at December 31, 2002 related to pre-November 29, 2001 activities. Of these amounts, recovery of approximately \$0.3 million is currently included in rates. On December 17, 2002, the CPUC issued an order authorizing water utilities, including SCW, to file advice letters, within 90 days, requesting recovery of the balancing accounts prior to November 29, 2001. SCW filed such advice letters for the unrecovered balance of \$2.3 million on March 17, 2003.

Electric power costs incurred by SCW's Bear Valley Electric division continue to be charged to its electric supply cost balancing account. The under-collection in the electric balancing account is \$24.9 million at December 31, 2002. The CPUC has authorized SCW to collect a surcharge from its customers of 2.2¢ per kilowatt hour for a period of up to ten years to enable SCW to recover this under-collection. SCW sold 131,826,000 and 126,911,000 kilowatt hours of electricity to its Bear Valley Electric division customers in 2002 and 2001, respectively, and anticipates that electricity sales during the ten year period will be sufficient to enable SCW to recover the amount of this under-collection during this ten year period. For further information, see the sections entitled *"Regulatory Matters"* and *"Electric Energy Situation in California"* included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

#### **Critical Accounting Policies**

Critical accounting policies are those that are important to the portrayal of AWR's financial condition, results of operations and cash flows, and require the most difficult, subjective or complex judgments of AWR's management. The need to make estimates about the effect of items that are uncertain is what makes these judgments difficult, subjective and/or complex. Management makes subjective judgments about the accounting and regulatory treatment of many items. The following are examples of accounting policies that are critical to the financial statements of AWR. For more information regarding the significant accounting policies of Registrant, see Note 1 of "*Notes to Financial Statements*" included in Part II, Item 8 in Financial Statements and Supplementary Data.

### Accounting for Rate Regulation

- Management believes that regulation and the effects of regulatory accounting have the most significant impact on the financial statements. When either SCW or CCWC file for adjustments to
  rates, capital assets, operating costs and other matters are subject to review, and disallowances could occur. Regulatory disallowances in the past have not been frequent but have on occasion
  significantly impacted AWR's results of operations.
- The utility subsidiaries, SCW and CCWC, have incurred various costs and received various credits reflected as regulatory assets and liabilities. Accounting for such costs and credits as regulatory assets and liabilities is in accordance with Statement of Financial Accounting Standards No. 71 "Accounting for the Effects of Certain Types of Regulation" (SFAS 71). This statement sets forth the application of generally accepted accounting principles for those companies whose rates are established by or are subject to approval by an independent third-party regulator. Under SFAS 71, utility companies defer costs and credits on the balance sheet as regulatory assets and liabilities when it is probable that those costs and credits will be recognized in the rate making process in a period different from the period in which they would have been reflected in income by an unregulated company. These deferred regulatory assets and liabilities are then reflected in the income statement in the period in which the same amounts are reflected in the rates charged for service. If rate recovery is no longer probable, Registrant is required to write off the related regulatory asset.
- Registrant's income tax calculations require estimates due principally to the regulated nature of the operations of SCW and CCWC, the multiple states in which Registrant operates, and
  potential future tax rate changes. Registrant uses the asset and liability method of accounting for income taxes under which deferred tax assets and liabilities are recognized for the future tax
  consequences attributable to differences

between the financial statement carrying amounts of existing assets and liabilities and their respective tax bases. Deferred tax assets and liabilities are measured using enacted tax rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. A change in the regulatory treatment, or significant changes in tax-related estimates or assumptions or changes in law, could have a material impact on the financial position and results of operations of Registrant.

Depreciation is computed at composite rates, based on depreciable plant at the beginning of each year, considered sufficient to amortize costs over the estimated remaining lives of assets. Depreciation studies are performed periodically and prospective changes in rates are estimated to make up for past differences. These studies are reviewed and approved by either the CPUC or the ACC. Changes in estimates of depreciable lives or changes in depreciation rates mandated by regulations could impact results of operations of Registrant in periods subsequent to the change. Depreciation is computed on the straight-line, remaining-life basis, As prescribed by the CPUC under its Uniform System of Accounts for Water Utilities, SCW is allowed to capitalize a portion of general costs such as engineering, supervision, general office salaries and expenses, legal expenses, insurance, injuries and damages, pensions and benefits, and taxes, as overhead construction costs included in SCW's Utility Plant. All overhead construction costs are charged to jobs on the basis of the amounts of such overhead expenses reasonably applicable thereto, so that each job bears its equitable proportion of such costs and its direct costs.

#### Accounting for Derivative Instruments

In June 1998, the FASB issued SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities". A derivative financial instrument or other contract derives its value from another investment or designated benchmark. SFAS No. 133 requires companies to record derivatives on the balance sheet as assets and liabilities, and to measure those instruments at their fair value. SCW is a party to various block-forward purchase power contracts. Certain of these contracts qualify as an exception provided under the SFAS No. 133 for activities that are considered normal purchases and normal sales. These contracts are reflected in the statements of income at the time of contract settlement. Contracts that do not qualify for the normal purchases and normal sales exception have been recognized at fair market value on the balance sheet as an asset or liability and an unrealized gains or losses will be recorded and the unrealized loss will be reversed. The net pre-tax unrealized loss recorded as a component of Purchased Power for Resale for the year ended December 31, 2002 was approximately \$2.5 million.

## Accounting for Asset Retirement Obligations

SFAS No. 143, "Accounting for Asset Retirement Obligations," requires businesses to record the fair value of a liability for a legal obligation to retire an asset in the period in which the liability is incurred. A legal obligation is a liability that a party is required to settle as a result of an existing or enacted law, statute, ordinance or contract. When the liability is initially recorded, the entity capitalizes the cost by increasing the carrying amount of the related long-lived asset. Over time, the liability is accreted to its present value each period, and the capitalized cost is depreciated over the useful life of the related asset. Upon settlement of the liability, an entity either settles the obligation for its recorded amount or incurs a gain or loss upon settlement. Registrant adopted SFAS No. 143 on January 1, 2003, as required. Registrant's legal obligations for retirement reflect principally the retirement of wells, which by law need to be disposed of at removal. As such, the regulated subsidiaries of Registrant incur asset retirement obligations. Retirement cost has historically been recovered through rates at the time of retirement. As a result, the cumulative effect upon adoption on January 1, 2003, Registrant recorded an asset retirement obligation of \$13.2 million at its net present value of \$2.7 million, increased depreciable assets by \$0.4 million for asset retirement costs, increased accumulated depreciation by \$0.2 million. Amounts recorded under SFAS 143 are subject to various assumptions and determinations, such as determining whether a legal obligation exists to remove assets, and estimating the fair value of the costs of removal, when final removal will occur and the credit-adjusted risk-free interest rates to be utilized on

16

discounting future liabilities. Changes that may arise over time with regard to these assumptions will change amounts recorded in the future.

### Accounting for Pension and Other Postretirement Benefits

Registrant records plan assets, obligations and expenses related to its pension and other postretirement benefit plans based on actuarial valuations. Key assumptions in these valuations include discount rates, expected returns on plan assets, compensation increases and health care cost trend rates. Registrant believes that the assumptions utilized in recording obligations under the plans are reasonable based on prior experience, market conditions and from the advice of plan actuary. Registrant discounted its future pension and other postretirement plan obligations using a rate of 6.75% at December 31, 2002 as compared to 7.25% at December 31, 2001. Both pension liability and future pension expense increase as the discount rate is reduced. At December 31, 2002, Registrant assumed that its plans' assets would generate a long-term rate of return of 7%, except for the non-union postretirement medical plan which used 4.2%, as compared to 8% assumed at December 31, 2001. Pension expenses increases in health care costs, Registrant increased the medical trend rate to 12% for pre-65 aged retirees at December 31, 2002, as compared to 8.5% used at December 31, 2001. An increase in assumed health care cost trend rates would increase the postretirement benefit obligation as well.

#### **Unbilled Revenues**

SCW and CCWC record water and electric utility operating revenues when the service is provided to customers. Operating revenues include unbilled revenues which are earned (service has been provided) but not billed by the end of each accounting period. By using the billed revenues based on the last meter reading and billed customer numbers, an average amount billed per customer is used to estimate the unbilled revenues for estimated average remaining days to the end of the reporting period. Unbilled revenues are recorded for both monthly and bi-monthly customers.

## Liquidity and Capital Resources

### AWR

AWR funds its operating expenses and pays dividends on its outstanding Common Shares primarily through dividends from its subsidiaries, principally SCW. AWR has a Registration Statement on file with the Securities and Exchange Commission (SEC) for issuance, from time to time, of up to \$60 million in Common Shares, Preferred Shares and/or debt securities. As of December 31, 2002, approximately \$31.1 million remained for issuance under this Registration Statement.

During 2001, AWR maintained a \$25 million credit facility, \$20 million of which was outstanding at December 31, 2001. This credit facility expired on January 2, 2002. In June 2002, AWR established a \$75 million revolving credit facility which matures in June 2005. Up to \$15 million of this facility may be used for letters of credit. As of December 31, 2002, an aggregate of \$35 million in cash borrowing was outstanding. Approximately \$8 million of letters of credit were outstanding under this facility.

On April 19, 2002, AWR completed the redemption of all of its outstanding 4%, 41/4% and 5% series of preferred shares.

# SCW

SCW funds the majority of its operating expenses, payments on its debt, and dividends on its outstanding Common Shares through internal sources. Internal sources of cash flow are provided primarily by retention of a portion of earnings from operating activities. Internal cash generation is influenced by factors such as weather patterns, environmental regulation, litigation, changes in supply costs and regulatory decisions affecting SCW's ability to recover these supply costs, and timing of rate relief. For further information, see the sections entitled "*Risk Factors*" and "*Electric Energy Situation in California*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

SCW also relies on external sources, including equity investments and short-term borrowings from AWR, long-term debt, contributions-in-aid-of-construction, advances for construction and install-and-convey advances,

17

to fund the majority of its construction expenditures. In January 2001, SCW issued \$20 million of long-term debt in a public offering with the proceeds used to reduce bank borrowings. On March 30, 2001, AWR made an additional \$25 million equity investment in SCW. On November 14, 2001, SCW filed a Registration Statement with the SEC for issuance, from time to time, of up to \$100 million in debt securities. In December 2001, SCW issued \$50 million of long-term debt under this Registration Statement that initially reduced bank borrowings incurred to fund capital expenditures and purchased power costs.

# CCWC

CCWC funds the majority of its operating expenses, payments on its debt and dividends, if any, through internal sources. CCWC also relies on external sources, including long-term debt, contributions-in-aid-of-construction, advances for construction and install-and-convey advances, to fund the majority of its construction expenditures.

#### ASUS

ASUS funds its operating expenses primarily through contractual management fees and investments by or loans from AWR.

In addition to contractual maturities, Registrant has certain debt instruments that contain annual sinking fund or other principal payments. Registrant believes that it will be able to refinance debt instruments at their maturity through public issuance, or private placement, of debt or equity. Annual principal payments are generally made from cash flow from operations. In the case of the \$12,500,000 in notes due October 2003, SCW intends to refinance this obligation through issuance of notes in the public or private markets.

The following table reflects Registrant's contractual obligations and commitments to make future payments pursuant to contracts as of December 31, 2002. All obligations and commitments are obligations and commitments of SCW unless otherwise noted.

(\$ in thousands)	Payments/Commitments Due by Period (1)					
	Total	Less th	an 1 Year	1 - 3 Years	4 - 5 Years	After 5 Years
Notes/Debentures(2)	\$185,600	\$	12,500	_	_	\$ 173,100
Private Placement Notes(3)	28,000		_	_	_	28,000
Tax-Exempt Obligations(4)	19,453		91	194	218	18,950
Other Debt Instruments(5)	2,452		188	404	453	1,407
Purchased Power Contracts(6)	71,893		11,973	23,973	23,947	12,000
Operating leases (7)	6,831		1,794	3,257	1,501	279
Other Commitments (8)	77,262		_	_	_	—
Chaparral City Water Company (9)	8,889		526	1,158	620	6,585
TOTAL	\$400,380	\$	27,072	\$ 28,986	\$ 26,739	\$ 240,321

(1) Excludes interest, dividends, commitment, and facility fees.

(2) The Notes and Debentures are issued under an Indenture dated as of September 1, 1993. The Notes and Debentures do not contain any financial covenants that Registrant believes to be material, or cross default provisions.

(3) The private placement notes are issued pursuant to the terms of Note Agreements with substantially similar terms. The Note Agreements contain restrictions on the payment of dividends, minimum interest coverage requirements, maximum debt to capitalization ratio and a negative pledge. Pursuant to the Note Agreements, SCW must maintain a minimum interest coverage ratio of two times interest expense. SCW does not currently have any outstanding mortgages or other encumbrances on its properties.

(4) Consists of obligations under a loan agreement supporting \$8 million in debt issued by the California Pollution Control Financing Authority, \$6 million in obligations supporting \$6 million in certificates of participation issued by the Three Valleys Municipal Water District and \$5.5 million of obligations

18

incurred by SCW with respect to its 500 acre foot entitlement to water from the State Water Project (SWP). Except as described in paragraph (6) below, these obligations do not contain any financial covenants believed to be material to Registrant or any cross default provisions. SCW's obligations with respect to the certificates of participation issued by the Three Valleys Municipal Water District are supported by a letter of credit issued by Wells Fargo Bank. In regards to its SWP entitlement, SCW has entered into agreements with various developers for 422 acre-feet, in aggregate, of its 500 acre-foot entitlement to water from the SWP.

- (5) Consists of \$1.5 million outstanding under a fixed rate obligation incurred to fund construction of water storage and delivery facilities with the Three Valleys Municipal Water District, \$0.6 million outstanding under a variable rate obligation incurred to fund construction of water delivery facilities with the Three Valleys Municipal Water District and an aggregate of \$0.4 million outstanding under capital lease obligations. These obligations do not contain any financial covenants believed to be material to Registrant or any cross default provisions.
- (6) Consists of \$71.9 million purchased power contracts with Pinnacle West Capital Corporation from November 2002 to December 2008.
- (7) Reflects Registrant's future minimum payments under non-cancelable operating leases.
- (8) Other commitments of Registrant consist of (i) \$75 million syndicated revolving credit facility, expiring in June 2005, (ii) a \$966,534 irrevocable letter of credit, which is reviewed at the end of each year for adjustment, for its self-insured workers compensation plan, (iii) an amount of \$296,000 with respect to a \$6,296,000 irrevocable letter of credit issued by Wells Fargo Bank to support the certificates of participation of Three Valleys Municipal Water District (the other \$6,000,000 is reflected under tax-exempt obligations), (iv) an irrevocable letter of credit in the amount of \$400,000 that expires on October 2003 for the deductible in Registrant's business automobile insurance policy (v) an irrevocable letter of credit that expires March 31, 2005 for its energy scheduling agreement with Automated Power Exchange as security for the purchase of power; the amount of the credit is \$585,000 for the months from November to March, and \$270,000 to cover the months from April to October, and (vi) outstanding performance bonds of \$14,290 to secure performance under franchise agreements with governmental agencies. All of the letters of credit are issued pursuant to the syndicated revolving credit facility. The syndicated revolving credit facility contains restrictions on prepayments, deposition of property, mergers, liens and negative pledges, indebtedness and guaranty obligations, transactions with affiliates, minimum interest coverage requirements, a maximum total funded debt ratio, of 0.65 to 1.00 and a minimum debt rating of Baa1 or BBB+.
- (9) Consists of \$7.9 million of obligations under a loan agreement supporting Industrial Development Revenue Bonds due in 2006 and a \$1.0 million repayment obligation to the United States Bureau of Reclamation. The loan agreement contains provisions that establishes a maximum of 65% debt in the capital structure, limits cash distributions when the percentage of debt in the capital structure exceeds 55% and requires a debt service coverage ratio of two times. The Bureau of Reclamation obligation does not contain any financial covenants believed to be material to Registrant or any cross default provisions.

Under the terms of its power purchase contracts with Mirant Americas Energy Marketing, LP and Pinnacle West Capital Corporation, SCW is required to post security, at the request of the seller, if SCW is in default under the terms of the contract and the future value of the contract is greater than the future value of contracts of a similar term on the date of default. SCW will be in default under the terms of these contracts if its debt is rated less than BBB- by Standard & Poor's Ratings Service ("S&P") or Fitch, Inc. ("Fitch") or less than Baa3 by Moody's Investor Services, Inc ("Moody's"). SCW currently has a rating of A+ by S & P and A2 by Moody's. Fitch does not rate SCW.

S&P debt ratings range from AAA (highest rating possible) to D (obligation is in default). Moody's debt ratings range from Aaa (best quality) to C (lowest quality). Securities ratings are not recommendations to buy, sell or hold a security and are subject to change or withdrawal at any time by the rating agency.

### **Electric Energy Situation in California**

### **Background Information**

On January 17, 2001, the Governor of the State of California proclaimed a state of emergency in California due to shortages of electricity available to certain of California's utilities (resulting in blackouts), the unanticipated and dramatic increases in electricity prices and the insufficiency of electricity available from certain of California's utilities to prevent disruption of electric service in California. The Federal Energy Regulatory Commission ("FERC") also implemented a number of changes to the tariff for the California Independent Operator System ("Cal ISO") beginning in December 15, 2000 in an attempt to stabilize the market. The reasons for the high cost of energy are under investigation but are reported to include, among other things, limited supply caused by a lack of investment in new power plants to meet growth in demand, planned and unplanned outages of power plants, decreased availability of hydroelectric power from the Pacific Northwest due to lower than usual precipitation and higher demand for electricity in the region, transmission line constraints, increased prices for natural gas, the fuel used in many of the power plants serving the region, a dysfunctional power market and market manipulation through the withholding of generation and a variety of "gaming" strategies.

As of December 31, 2002, SCW had accrued \$24.9 million in under-collected power costs that SCW mostly incurred during the energy crisis in connection with providing service to its Bear Valley Electric customers. In addition, SCW established \$7.9 million in reserves in 2001 for potential non-recovery of these power costs. In 2002, SCW charged \$1.7 million against the reserves due to its inability to recover all of its costs associated with entering into a long-term power contract with Mirant Americas Energy Marketing, LP (Mirant) in March 2001, and reversed the balance due to various new contracts entered into with Pinnacle West Capital Corporation (PWCC) in September 2002 as discussed below.

On July 17, 2002, the FERC extended and modified the mitigation measures that were set to expire on September 30, 2002, citing delays in construction of new generation resources in California and throughout the West, delays in adopting a new market design and market rules by the Cal ISO, transmission line constraints, constraints on natural gas pipeline capacity and continuing

dysfunctions in the California power market. It remains unclear how long the FERC will leave its mitigation measures in place. The premature termination of such mitigation measures could result in a substantial increase in spot market prices and the prices of long-term contracts for power and capacity. In addition, FERC is considering a number of market reforms, some of which could materially increase SCW's costs for power.

#### Power Supply Arrangements at SCW's Bear Valley Electric Service Area

All electric energy sold by SCW to customers in its Bear Valley Electric customer service area is purchased from others. In May 2000, SCW entered into a one-year, block forward purchase contract with Dynegy Power Marketing, Inc. (DYPM) for 12 MWs of electric energy at a price of \$35.50 per MWh. This contract expired April 30, 2001. Under the terms of contracts with DYPM that expired on April 30, 2002, DYPM provided electric energy to SCW, acted as scheduling coordinator and provided other ancillary services. Various disputes arose between the parties regarding the services provided by DYPM and the amounts charged by DYPM. As a result of these disputes, SCW withheld payment from DYPM of certain amounts invoiced by DYPM. On November 1, 2002, DYPM accepted a payment of \$3.5 million from SCW as payment in full for the disputed amounts.

SCW entered into a five-year nine-month, block forward purchase contract with Mirant for 15 MWs of electric energy at a price of \$95 per MWh beginning April 1, 2001 through December 31, 2006. On December 20, 2001, SCW filed a complaint with FERC seeking to reduce the amount charged by Mirant under the terms of this contract to a just and reasonable price. A hearing was conducted on this matter in October 2002. On December 19, 2002, the administrative law judge issued an initial decision denying SCW's request to reduce the charges under the contract. SCW has appealed the initial decision which results in the complaint being reviewed by the full Commission. SCW anticipates that it is unlikely that FERC will grant its request to reduce these charges.

In June 2001, SCW executed an agreement with PWCC for an additional 8 MWs of electric energy to meet peak winter demands. The contract provided for pricing of \$75 per MWh from November 1, 2001 to March

20

#### 31, 2002, \$48 per MWh from November 1, 2002 to March 31, 2003, and \$36 per MWh from November 1, 2003 to March 31, 2004.

In September 2002, SCW entered into a series of purchased power contracts with PWCC. Under the agreements, SCW will sell 15 MWs to PWCC of electric energy at a price of \$95 per MWh beginning November 1, 2002 through December 31, 2006, and the 8 MWs of electric energy covered under the energy purchase agreement with PWCC discussed previously. In return, PWCC will supply SCW's BVE customer service area with 15 MWs of electric energy at a price of \$74.65 per MWh beginning November 1, 2002 through December 31, 2008, and an additional 8 MWs at \$74.65 per MWh beginning on November 1, 2002 through March 31, 2003 and each succeeding November 1 through March 31 period through March 31, 2008, and for the period November 1, 2008 through December 31, 2008. These contracts do not qualify for the normal purchases and normal sales exception, under the Statement of Financial Accounting Standards No. 133, "Accounting for Derivative Instruments and Hedging Activities", and have been recognized at fair market value on the balance sheet as a liability at December 31, 2002 and an unrealized loss of \$2.5 million against earnings.

The average minimum load at SCW's Bear Valley Electric customer service area has been approximately 14 MWs. The average winter load has been 18 MWs with a winter peak of 39 MWs when the snowmaking machines at the ski resorts are operating. In addition to the power purchase contracts, SCW buys additional energy from the spot market to meet peak demand and sells surplus power to the spot market as well. The average cost of power purchased including the transactions in the spot market was approximately \$95.38 per MWh for 2002. SCW's average energy cost can be impacted by pricing fluctuation at the spot market.

SCW is a party to the FERC proceedings in which refunds are being sought on purchases of power in the spot market in California from October 2, 2000 through June 2001. It is anticipated that a decision will be issued by FERC by the end of the second quarter of 2003. SCW was not a direct participant in the spot market. As a result, SCW has not yet been able to determine what, if any effect this decision will have on SCW.

#### Transmission Constraints

Demand for energy in SCW's Bear Valley Electric customer service area generally has been increasing. However, the ability of SCW to deliver purchased power to these customers is limited by the ability of the transmission facilities owned by Southern California Edison Company to transmit this power. On December 27, 2000, SCW filed a lawsuit against Edison for breach of contract as a result of delays in upgrading these transmission facilities as well as for violations of good faith and fair dealing, negligent misrepresentation, intentional misrepresentation and unjust enrichment. As a result of the delays and the pending litigation, it is highly unlikely that the transmission upgrade will occur. Therefore, SCW is considering the construction of a natural gas-fueled 8.4 MW generation facility to be owned by SCW as a means of meeting these increasing demands for energy. A Certificate of Public Convenience and Necessity was filed on March 28, 2002 seeking the CPUC's authorization for construction of the generation facility. If approved, it will result in the need to file for further increases in electric energy prices for customers of SCW's Bear Valley Electric customer service area. For more information, see the section entitled "*Regulatory Matters-Pending Rate Changes*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

#### Construction Program

SCW maintains an ongoing water distribution main replacement program throughout its customer service areas based on the priority of leaks detected, fire protection enhancement and a reflection of the underlying replacement schedule. In addition, SCW upgrades its electric and water supply facilities in accordance with industry standards, local requirements and CPUC requirements. SCW's Board of Directors has approved anticipated net capital expenditures of approximately \$81.6 million for 2003 principally reflecting the 2003 infrastructure replacement program in SCW's Metropolitan customer service area, a water treatment plant upgrade, water supply related projects, and security related costs. However, approved capital expenditures may be limited pending CPUC approval of SCW's general rate case filings. For further information, see the section entitled *"Regulatory Matters-Pending Rate Changes"* included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

CCWC's Board of Directors has approved a net capital budget of \$3.0 million for 2003 primarily reflecting improvement and upgrade of a water treatment plant. AWR and ASUS have no material capital

21

commitments. However, ASUS actively seeks opportunities to own, lease or operate water and wastewater systems for governmental entities, which may involve significant capital commitments.

Neither SCW nor CCWC has material capital expenditures for specific environmental control facilities or measures.

#### **Regulatory Matters**

## Rate Regulation

SCW is subject to regulation by the CPUC, which has broad powers with respect to service and facilities, rates, classifications of accounts, valuation of properties, the purchase, disposition and mortgaging of properties necessary or useful in rendering public utility service, the issuance of securities, the granting of certificates of public convenience and necessity as to the extension of services and facilities and various other matters. CCWC is subject to regulation by the ACC.

Rates that SCW and CCWC are authorized to charge are determined by the CPUC and the ACC, respectively, in general rate cases and are derived using rate base, cost of service and cost of capital, as projected for a future test year in California and using an historical test year, as adjusted in Arizona. Rates charged to customers vary according to customer class and rate jurisdiction and are generally set at levels allowing for recovery of prudently incurred costs, including a return on rate base. Rate base generally consists of the original cost of utility plant in service, plus certain other assets, such as working capital and inventory, less accumulated depreciation on utility plant in service, deferred income tax liabilities and certain other deductions.

Neither AWR nor ASUS are directly regulated by the CPUC. The CPUC does, however, regulate certain transactions between SCW and its affiliates. The ACC also regulates certain transactions between CCWC and its affiliates.

For rate-making purposes, the 22 customer service areas of SCW are grouped into 9 water districts and 1 electric district. Water rates vary among the 9 water ratemaking districts due to differences in operating conditions and costs. SCW monitors operations on a regional basis in each of these districts so that applications for rate changes may be filed, when warranted. Under the CPUC's practices, rates may be increased by three methods: (i) general rate case increases (GRC's), (ii) offsets for certain expense increases including but not limited to supply cost offset and balancing account amortization, and (iii) advice letter filings related to certain plant additions and other operating cost increases. GRC's are typically for three-year periods, which include step increases for the second and third year. Rates are based on a forecast of expenses and capital costs. GRC's have a typical regulatory lag of one year. Offset rate increases and advice letter filings typically have a two to four month regulatory lag. In addition, recovery of water supply costs incurred after November 29, 2001 will be considered as part of GRC's.

### Changes in Rates

The following table lists information on estimated annual rate changes for SCW as approved by the CPUC during 2002, 2001 and 2000.

(S in 000's) Year	Supply Cost Offset	Balancing Account Amortization	General and Step Increases	Advice Letters	Total
2002	_	\$ (1,926)	\$ 322	\$ 6,029	4,425
2001	1,364	4,422	6,943	595	13,324
2000	_	(1,474)	6,973	1,040	6,539

New water rates with an annual increase of approximately \$2.5 million for seven ratemaking districts in SCW's Region I and step increases of approximately \$1.7 million for SCW's Region III was in effect from February 2001. An attrition increase of approximately \$2.8 million for Region II was in effect from February 2001. Step increases of approximately \$321,600 for four of seven ratemaking districts in SCW's Region I were implemented in January 2002.

22

On January 16, 2003, the CPUC approved rate increases of approximately \$2.7 million annually, effective January 22, 2003, in SCW's Metropolitan district to recover costs associated with an increase in Region II rate base due to SCW's 2002 infrastructure replacement program and, additionally, to recover general increases in operating expenses. On January 31, 2003, the CPUC also approved SCW's Advice Letter filed for the 2003 infrastructure replacement program with rate increases of \$3.5 million annually effective February 4, 2003.

In March 2001, the CPUC approved SCW's Advice Letters to recover increased costs of purchased power by \$761,351 incurred to pump water for its water customers included in base water rates for each of its ratemaking districts. In April 2001, SCW filed additional Advice Letters by ratemaking areas to increase water rates by approximately \$2.3 million company-wide to recover additional electric base rate increases, authorized by the CPUC for the Southern California Edison Company and the Pacific Gas and Electric Company. The CPUC approved in the fourth quarter of 2001 increases of approximately \$672,900 in base water rates. For further information, see the section "*Electric Energy Situation in California*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

As of December 31, 2002, SCW had accrued approximately \$24.9 million in under-collected purchased power costs included in the electric balancing account. In 2001, the CPUC approved two Advice Letters which allow SCW to collect \$11.1 million in aggregate over five years, equivalent to 2.2¢ per kilowatt hour, for recovery of its under-collection in the electric balancing account.

On August 17, 2001, SCW filed an application with the CPUC seeking recovery of an average cost of \$87 per megawatt hour (MWh) for electric energy purchased pursuant to power purchase contracts with Mirant Americas Energy Marketing, LP and Pinnacle West Capital Corporation (PWCC). On July 17, 2002, CPUC approved a settlement agreement reached among SCW, all intervening parties and the Office of Ratepayer Advocates (ORA). The authorization permits SCW to recover \$77 per MWh of purchased power costs through rates. SCW is allowed to include its actual purchased power costs up to an average annual weighted cost of \$77 per MWh each year, through August 2011, for 10 years in its balancing account. To the extent SCW's actual average annual weighted costs for purchased power is less than \$77 per MWh, the differential would offset amounts included in the electric supply balancing account. Conversely, to the extent that actual average annual weighted costs for power purchased exceed the \$77 per MWh amount, SCW would not be able to include these amounts in its balancing account and such amounts would be expensed against income.

The CPUC settlement also extended the previously approved surcharges of 2.2¢ per kilowatt hour for an additional five years through August of 2011 to allow SCW an opportunity to collect amounts remaining in its electric cost balancing account. SCW anticipates that electricity sales will be sufficient to enable SCW to fully recover the under-collection in its balancing account during this ten year period.

SCW previously established approximately \$8.2 million in reserves against potential non-recovery of electric power costs. In 2002, \$1.7 million was written off against the reserve based on the settlement with the CPUC as previously discussed. In September 2002, SCW entered into various agreements with PWCC, which enable SCW to purchase energy at a lower cost. As a result, the remaining reserve of \$6.5 million was reversed into income in 2002.

The settlement requires SCW to pursue its complaint filed with FERC in which SCW has requested FERC to reduce the prices in its power purchase contract with Mirant to a just and reasonable price. If FERC rules a price reduction, SCW is required, based on the settlement, to file an Advice Letter to notify the CPUC within 30 days of the ruling. On December 19, 2002, the Administrative Law Judge issued an initial decision denying SCW's request to reduce the charges under the contract. SCW has appealed the initial decision which results in the complaint being reviewed by all member commissions at the FERC. SCW anticipates that it is unlikely that FERC will grant its request to reduce these charges. See the section entitled *"Electric Energy Situation in California"* included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

# 23

# Pending Rate Changes

On March 28, 2002, SCW filed for a Certificate of Public Convenience and Necessity (CPCN) seeking the CPUC's authorization to construct an 8.4 MW natural gas-fueled generation facility on a portion of its property in the City of Big Bear Lake. The capital cost of the generating facility is estimated to be approximately \$13 million, which, if approved by the CPUC, will authorize construction and enable SCW to file a rate application subsequently, to generate an annual revenue increase of about \$2.4 million. The rate application will be subject to additional review by the CPUC.

In October 2002, SCW filed an Application to increase water rates in the customer service areas that comprise Region III. SCW also filed a concurrent application requesting a rate increase applicable to SCW's entire customer base to recover costs associated with the general office functions of SCW. With a proposed return on equity of 12.45%, the new water rates in these filings, if fully approved by the CPUC, would generate an initial annual increase in revenues of approximately \$19.8 million for the Region III customer service areas, and \$6.0 million for all the other customer service areas of SCW. A final decision on these applications is not anticipated until the third quarter of 2003.

On January 31, 2003, SCW also filed Notices of Intent to the CPUC to increase water rates in the customer service areas that comprise its Region I and Region II. New rates, if approved by the CPUC, are not anticipated until the second quarter of 2004. As part of the Region I filing, SCW requested amortization of the balance of costs included in the Aerojet litigation memorandum account. For more information, see the section entitled "*Other Water Quality Litigations*" included in Part I, Item 3 in Legal Proceedings.

In 1993, the CPUC disallowed \$1.6 million of costs incurred in construction of a water treatment facility in SCW's Clearlake customer service area and Registrant wrote off the disallowed amount at that time. Based on new water quality standards, in 2000, SCW re-applied to the CPUC for inclusion of the disallowed amount in rate base. A draft decision issued on March 30, 2001 by the CPUC would have allowed SCW to include \$500,000 of the \$1.6 million in the regulated rate base. An alternate draft decision issued by one of the CPUC Commissioners proposed to deny the relief sought by SCW in its application. An Administrative Law Judge subsequently reopened the proceeding in August 2001 requiring additional information. A final order is anticipated during 2003.

#### Other Regulatory Matters

On November 29, 2001, the CPUC adopted an Order Instituting Rulemaking (OIR) to (i) evaluate existing practices and policies, (ii) determine whether new procedures or policies for processing offset rate increases and balancing accounts should be made and (iii) determine whether the new memorandum account procedures adopted on November 29, 2001 should be made permanent. On December 17, 2002, the CPUC issued an order authorizing water utilities, including SCW, to file advice letters within 90 days requesting recovery of any under-collections remaining in the balancing accounts for water supply costs incurred prior to November 29, 2001. On March 17, 2003, SCW filed advice letters seeking recovery of \$2.3 million. For more information, see the section entitled "*Accounting for Supply Costs*" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

There are no active regulatory proceedings affecting CCWC or its operations.

#### **Environmental Matters**

## 1996 Amendments to Federal Safe Drinking Water Act

On August 6, 1996, amendments (the 1996 SDWA amendments) to the Safe Drinking Water Act (the SDWA) were signed into law. The 1996 SDWA amendments revised the 1986 amendments to the SDWA with a new process for selecting and regulating contaminants. The U.S. Environmental Protection Agency (EPA) can only regulate contaminants that may have adverse health effects, are known or likely to occur at levels of public health concern, and the regulation of which will provide a meaningful opportunity for health risk reduction. The EPA has published a list of contaminants for possible regulation and must update that list every five years. In addition, every five years, the EPA must select at least five contaminants on that list and determine whether to regulate them. This law allows the EPA to bypass the selection process and adopt interim regulations for

24

contaminants in order to address urgent health threats. The California Department of Health Services (DOHS), acting on behalf of the EPA, administers the EPA's program in California.

The 1996 SDWA amendments allow the EPA to base primary drinking water regulations on risk assessment and cost/benefit considerations and on minimizing overall risk. The EPA must base regulations on best available, peer-reviewed science and data from best available methods. For proposed regulations that involve the setting of maximum contaminant levels (MCL's), the EPA must use, and seek public comment on, an analysis of quantifiable and non-quantifiable risk-reduction benefits and costs for each such MCL.

SCW and CCWC currently test their wells and water systems according to requirements listed in the SDWA. Water from wells found to contain levels of contaminants above the established MCL's is treated to reduce contaminants to acceptable levels before it is delivered to customers. If treatment is not possible, the wells are shut down. Since the SDWA became effective, SCW has experienced increased operating costs for testing to determine the levels, if any, of the constituents in SCW's sources of supply and additional expense to lower the level of any contaminants in order to meet the MCL standards. Such costs and the costs of controlling any other contaminants may cause SCW to experience additional capital costs as well as increased operating costs. The CPUC and ACC ratemaking processes provide SCW and CCWC with the opportunity to recover prudently incurred capital and operating costs associated with water quality. Management believes that such incurred and expected future costs should be authorized for recovery by the CPUC and ACC, as appropriate.

#### **Enhanced Surface Water Treatment Rules**

The EPA has adopted Enhanced Surface Water Treatment Rules (ESWTR), which require increased surface-water treatment to decrease the risk of microbial contamination. These rules apply to each of SCW's four surface water treatment plants and CCWC's surface water treatment plant. Registrant anticipates that all plants will achieve compliance within the three-year to five-year time frames identified by EPA. Registrant is required to be in compliance by June 1, 2006. SCW is planning to build a new treatment plant in the Calipatria-Niland customer service area (see further discussion below under *Regulation of Disinfectant/Disinfection By-Products*) to bring that facility into compliance. Once this project is completed, all the surface water plants in SCW and CCWC should be in compliance with these rules.

#### **Regulation of Disinfection/Disinfection By-Products**

SCW and CCWC are also subject to regulations concerning disinfection/disinfection by-products (DBP's). Stage I of the regulations was effective in November 1998 with full compliance required for systems serving 10,000 or more persons by 2002 and for systems serving fewer than 10,000 persons by 2004. Stage I requires reduction of trihalomethane contaminants from 100 micrograms per liter to 80 micrograms per liter. SCW has already implemented modifications to the treatment process in its Bay Point and Cordova systems to achieve compliance and the Calipatria plant is undergoing treatment modifications, which will also address the requirements under the ESWTR, in order to comply with the DBP's by 2004.

The EPA is not allowed to use the new cost/benefit analysis provided for in the 1996 SDWA amendments for establishing the Stage II rules applicable to DBP's but may utilize the regulatory negotiating process provided for in the 1996 SDWA amendments to develop the Stage II rule. The final rule is not expected until 2004.

#### Ground Water Rule

On May 10, 2000, the EPA published the proposed Ground Water Rule (GWR), which establishes multiple barriers to protect against bacteria and viruses in drinking water systems that use ground water. The proposed rule will apply to all U.S. public water systems that use ground water as a source. The proposed GWR includes system sanitary surveys conducted by the state to identify significant deficiencies; hydrogeologic sensitivity assessments for undisinfected systems; source water microbial monitoring by systems that do not disinfect and draw from hydrogeologically sensitive aquifers or have detected fecal indicators within a distribution system; corrective action; and compliance monitoring for systems which disinfect to ensure that they reliably achieve 4-log (99.99%) inactivation or removal of viruses. The GWR is scheduled to be issued as a final regulation during 2003. While no assurance can be given as to the nature and cost of any additional compliance

7	э.

measures, if any, SCW and CCWC do not believe that such regulations will impose significant compliance costs, since they already currently engage in disinfection of the majority of their groundwater systems.

### Regulation of Radon and Arsenic

On October 31, 2001, EPA announced that the arsenic standard in drinking water would be 10 parts per billion (ppb). Compliance with an MCL of 10 ppb will require implementation of wellhead treatment remedies for eight affected wells in SCW's system and two wells in CCWC's system. The effective date for utilities to comply with the standard will be January 2006. In California, the Office of Environmental Health Hazard Assessment (OEHHA) issued a Public Health Goal for arsenic of 4 parts per trillion in March 2003 that may result in California adopting a lower MCL for arsenic.

The EPA has proposed new radon regulations following a National Academy of Sciences risk assessment and study of risk-reduction benefits associated with various mitigation measures. The National Academy of Sciences study is in agreement with much of EPA's original findings but has slightly reduced the ingestion risk initially assumed by EPA. EPA established an MCL of 300 Pico Curies per liter based on the findings and has also established an alternative MCL of 4000 Pico Curies per liter, based upon potential mitigation measures for overall radon reduction. Registrant is currently waiting for the EPA to establish a MCL to determine the impact.

#### Voluntary Efforts to Provide Treated Surface Water Below Minimum Surface Water Treatment Requirements

SCW is a voluntary member of the EPA's Partnership for Safe Water, a national program designed to further protect the public from diseases caused by cryptosporidium and other microscopic organisms. As a volunteer in the program, SCW commits to treat surface water to levels much lower than the minimum operating requirements governing surface water treatment, optimize surface water treatment plant operations and seeks to have its surface water treatment facilities perform as efficiently as possible.

#### **Unregulated Contaminants Monitoring Rule**

EPA has revised the Unregulated Contaminant Monitoring Rule (UCMR), as required by the 1996 SDWA amendments. The data generated by the UCMR will be used to evaluate and prioritize contaminants on the Drinking Water Contaminant Candidate List, a list of contaminants EPA is considering for possible new drinking water standards. This data will help to ensure that future decisions on drinking water standards are based on sound science.

A tiered approach will be utilized with the three monitoring lists to provide the maximum capability to monitor up to the statutory limit of no more than 30 contaminants in any 5-year monitoring cycle. Therefore, as List 3 contaminants are found to occur in public water systems, they may move up to List 2, and likewise, List 2 contaminants may move up to List 1, in 2004, when the UCMR is revised again. The law requires that EPA publish a new contaminant-monitoring list every 5 years. When the EPA adds contaminants to their list, they will also include a compliance date. Registrant will evaluate the impact and necessary actions once the EPA makes its new announcement.

### Fluoridation of Water Supplies

SCW is required to fluoridate water supplies for public water systems serving more than 10,000 service connections if funds have been made available to cover capital and operating costs. The CPUC is required to authorize cost recovery through rates should funds for the operation of fluoridation facilities, once installed, become unavailable in future years. Currently, funds have been made available to public agencies only. SCW has not been offered any funding and has not installed any fluoridation facilities.

## Perchlorate Action Level Activities

In January 2002, DOHS reduced the action level from 18 ppb to a level of 4 ppb, based upon new reference dose for health risk information from EPA. The Governor in California recently signed into law a bill requiring DOHS to establish a MCL for perchlorate by January 1, 2004. SCW has removed eight wells from service in its Rancho Cordova system and six additional wells in various other systems since they contained perchlorate in amounts in excess of this reduced action level of 4 ppb. On December 6, 2002, the OEHHA

26

published a revised draft perchlorate Public Health Goal of 2 to 6 ppb. This is the first step in the establishment of an MCL in California. SCW is continuing to periodically monitor all of its wells to determine that levels of perchlorate are below the action level currently in effect. Perchlorate is known to be used in oxidizing rocket fuels.

### Nitrosodimethylamine (NDMA) Action Level

In February 2002, DOHS increased the action level from 2 parts per trillion (ppt) to 10 ppt. NDMA is an additional by-product from the production and use of rocket fuels.

#### Matters Relating to SCW's Arden-Cordova Water System

In SCW's Rancho Cordova system, four wells have been removed from service and destroyed due to contamination from perchlorate. The supply has been replaced for three of these wells. An additional four wells are currently out of service due to perchlorate levels above the reduced action level of 4 ppb, and four wells are out of service due to detectable levels of NDMA. SCW continues to monitor all of its active groundwater wells in the Rancho Cordova system for perchlorate and NDMA level. Aerojet-General Corp. (Aerojet) has, in the past, used ammonium perchlorate in oxidizing rocket fuels. NDMA is an additional by-product from the production of rocket fuels and it is believed that such contamination is also related to the activities of Aerojet. SCW has filed suit against Aerojet for contamination of SCW's ground water supply in its Rancho Cordova system.

To date, Aerojet has reimbursed SCW for constructing a pipeline to interconnect with the City of Folsom water system to provide an alternative source of water supply in SCW's Rancho-Cordova customer service area and has reimbursed SCW for costs associated with the drilling and equipping of new wells. As of December 31, 2002, Aerojet had previously reimbursed SCW \$4.5 million of the approximately \$21.2 million in costs SCW has incurred. The remainder of the costs is subject to further reimbursement pending outcome of the litigation previously discussed. Reimbursements received from Aerojet have been applied directly to reduce SCW's costs of utility plant and purchased water. For further information regarding litigation related to contamination of ground water in Sacramento County, see the section entitled "*Other Water Quality Litigation*" included in Part II, Item 3, Legal Proceedings.

The EPA issued an order to Aerojet in August 2002 that requires Aerojet to ultimately restore the contaminated aquifer and minimize further loss of water supply wells, and requires Aerojet to provide an alternative water supply if more water supply wells are lost.

### Matters Relating to SCW's Culver City Water System

The compound, methyl tertiary butyl ether (MTBE), an oxygenate used in reformulated fuels, has been detected in the Charnock Basin, located in the vicinity of the City of Santa Monica and within SCW's Culver City customer service area. At the request of the Regional Water Quality Control Board, the City of Santa Monica and the California Environmental Protection Agency, SCW removed two of its wells in the Culver City water system from service in October 1996 to help in efforts to avoid further spread of the MTBE contamination plume. Neither of these wells has been found to be contaminated with MTBE. SCW is purchasing water from the Metropolitan Water District of Southern California (MWD) at an increased cost to replace the water supply formerly pumped from the two wells removed from service.

On September 22, 1999, the U.S. EPA and the Los Angeles Regional Water Quality Control Board ordered Shell Oil Company, Shell Oil Products Company, Equilon Enterprises LLC and others to provide replacement drinking water to both SCW and the City of Santa Monica due to MTBE contamination in the Charnock Basin. The EPA has ordered Shell Oil and others to reimburse SCW for water replacement costs. In March 2002, SCW reached a settlement agreement with the City of Santa Monica, in which SCW assigns its rights against all the potentially responsible parties, who stored, transported and dispensed gasoline containing methyl tertiary butyl ether (MTBE) in underground storage tanks, pipelines or other related infrastructure, and its water rights in the Charnock Basin to the City of Santa Monica and Santa Monica takes over the prosecution against the potentially responsible parties in exchange for an assignment payment. The settlement agreement is pending the CPUC's approval. In a proposed decision issued on March 19, 2003, the Administrative Law Judge of the CPUC finds that the settlement is reasonable, and directs SCW to file a new proceeding at a later date that

27

sets forth specific details as to SCW's plans to reinvest the net proceeds from the settlement agreement with the City of Santa Monica. The filing will be subject to the CPUC's review.

#### Matters Relating to SCW's Yorba Linda Water System

The compound MTBE has been detected in three wells serving SCW's Yorba Linda water system. Two of the wells are standby wells and to date the third well has not shown MTBE above the DOHS secondary standard of 5.0 ppb. SCW has constructed an interconnection with the Metropolitan Water District of Southern California to provide for additional supply in the event the third well experiences levels of detection in excess of the DOHS standard.

SCW has met with the Regional Water Quality Control Board, the Orange County Water District, the City of Anaheim, the DOHS and three potentially responsible parties (PRP's) to define the extent of the MTBE contamination plume and assess the contribution from the PRP's. The PRP's have voluntarily initiated a work plan for regional investigation. While there have not been significant disruptions to the water supply in Yorba Linda at this point in time, no assurances can be given that MTBE contamination will not increase in the future.

#### Matters Relating to SCW's Los Osos Water System

The compound MTBE has been detected in a well serving SCW's Los Osos water system. For some time SCW has been working with the Regional Water Quality Control Board as well as the owner of a service station. Although the owner of the service station has attempted remediation with funds provided through the California Underground Storage Tank Fund ("CUSTF"), it appears there will be insufficient funds from the CUSTF to complete remediation sufficient to return the well to service. In order to prevent the running of the statute of limitations, on August 12, 2002 SCW filed suit in the Superior Court of the State of California for the County of San Luis Obispo against the operators and owners of the service station facility, and Chevron USA Products, Inc., the supplier.

#### Matters Relating to SCW's San Gabriel Water Systems

Volatile Organic Compounds (VOC) and perchlorate have been detected in 2 wells servicing SCW's San Gabriel System. SCW filed suit, along with two other affected water purveyors and the San Gabriel Basin Water Quality Authority (WQA), in federal court against some of those responsible for the contamination. Some of the other potential defendants settled with SCW, other water purveyors and the WQA on VOC related issues prior to the filing of the lawsuit resulting in reimbursement to SCW of the \$1 million in capital cost of VOC treatment facilities and contribution of approximately \$450,000 towards future operating and maintenance (O&M) costs of operating the facilities.

In response to the filing of the Federal lawsuit, the Potentially Responsible Party (PRP) defendants filed motions to dismiss the suit or strike certain portions of the suit. Following a hearing on these motions on March 31, 2003, the judge issued a ruling on April 1, 2003 granting in part and denying in part the defendant's motions. A key ruling of the court was that the water purveyors, including the Registrant, by virtue of their ownership of wells contaminated with hazardous chemicals are themselves PRPs under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA). Registrant is currently evaluating the potential impact of this decision, its opportunity to appeal the decision and the opportunity to amend its suit to claim certain affirmative defenses as an "innocent" party under CERCLA. Registrant is presently unable to predict the outcome of this ruling on its ability to fully recover from the PRPs future costs associated with the treatment of these wells.

Three other wells serving customers in SCW's San Gabriel customer service area are also impacted by VOC contamination. A settlement with several potentially responsible parties together with federal funds administered by WQA resulted in reimbursement of 100% of the \$1.1 million in capital costs, \$205,000 of past O&M costs and 100% of future O&M costs for a period of up to 30 years for VOC treatment facilities at these wells.

# Security Issues

Since the tragic events of September 11, 2001, water utilities, including Registrant, have been advised to increase security at key facilities in order to avoid contamination of water supplies and other disruptions of

service. In compliance with "*The Public Health Security and Bioterrorism Preparedness Act of 2002*" (HR 3448), Registrant has continued to implement measures to increase security, which includes conducting vulnerability assessments for its large systems by March 31, 2003. Large systems are defined as systems serving a population of 100,000 or more. These formal assessments will be submitted in accordance with the Federal deadline. In addition to large system assessments, all systems operated by Registrant were assessed to identify potential areas requiring enhancements. These assessments resulted in a prioritized listing of recommended facility upgrades to enhance the safety of water system operations. Costs associated with capital improvements identified as a result of the assessment process, approximately \$15 million, has been included in an application filed with the CPUC in October 2002. Registrant is unable to predict if the CPUC will authorize recovery of any or all of these costs.

## Water Supply

#### SCW's Water Supply

During 2002, SCW supplied a total of 87,754,000 CCF of water, or approximately 180 million gallons per day, on average. Of this amount, approximately 55.8% came from pumped sources and 41.3% was purchased from others, principally the Metropolitan Water District of Southern California (MWD). The Bureau of Reclamation (the Bureau) supplied the remaining amount under a no-cost contract. During 2001, SCW supplied 84,103,000 CCF of water, 58.5% of which came from pumped sources, 39.6% was purchased, and the Bureau supplied the remainder.

The MWD is a water district organized under the laws of the State of California for the purpose of delivering imported water to areas within its jurisdiction. Registrant has 58 connections to the water distribution facilities of MWD and other municipal water agencies. MWD imports water from two principal sources: the Colorado River and the State Water Project (SWP). Available water supplies from the Colorado River and the SWP have historically been sufficient to meet most of MWD's requirements. However, California missed a federal deadline of December 31, 2002 to demonstrate how it will reduce the Colorado River water that it takes beyond its legal entitlement after negotiations on how to divide the water in Southern California broke down. The federal government began to cut Southern California's Colorado River supplies in January 2003. MWD has publicly stated that Southern California should have ample water supplies for 20 years despite this cut by stepping up a number of efforts including desalination, conservation, recycling, transfer and storage. In addition, efforts are under way to resolve the impasse that led to the reduction in Colorado River supplies to Southern California.

SCW's water supply and revenues are significantly affected, both in the short-run and the long run, by changes in meteorological conditions. The current water year, October 2002 to September 2003 is projected to be significantly better than the past water year. Most of California now has average to above average precipitation and all but the southwest corner of Arizona has moved away from levels below 50% of normal. The California snow pack is at 85% of normal levels in February 2003. Lingering effects of the past drought year are reflected in reservoir levels that are still low with California at 85% of average and Arizona at only 50% of average.

The general outlook for January through May has been for wetter and warmer than normal weather throughout the Southwestern United States. Forecasts are for spring runoff to be generally near normal with some areas in Northern California experiencing above normal stream flow.

Although overall groundwater conditions remain at adequate levels, certain of SCW's groundwater supplies have been affected to varying degrees by various forms of contamination which, in some cases, have caused increased reliance on purchased water in its supply mix. For further information, see Part I, Item 3 in Legal Proceedings.

A sudden and unexplained drop in groundwater water supply severely impacted SCW's Wrightwood customer service area in the third quarter of 2002. In response to this emergency situation, SCW undertook a number of steps to provide water service, including trucking water into the area from nearby sources, bringing a formerly shut down well into service, taking steps to increase capacity at existing wells and expediting the drilling and equipping of a new well. SCW experienced increased operating costs associated with the trucking of water of approximately \$1.3 million. Due to the actions of SCW and conservation efforts of consumers, the

29

situation has stabilized. Management is unable to predict the extent to which additional costs may be incurred or the extent to which additional problems may be encountered. The cause of the unexplained and sudden drop in groundwater supply has not been identified, but it may be drought-related.

#### CCWC's Water Supply

For the 2002 water year, Arizona experienced an extreme drought with precipitation almost nonexistent in most areas, and less than 50% of normal level in others, reservoirs at less than 25% of normal capacity, and snow pack well below normal. The Colorado River flow into Lake Powell for the 2002 water year was only 4,049,700 AF or 23% of normal.

The Arizona outlook is more optimistic with drought conditions expected to end or at least ameliorate as El Nino weather patterns develop. For the 2003 water year precipitation levels have improved, although are still below normal levels, and the Colorado River flow into Lake Powell has been 1,580,000 AF, through February 2003, or 67% of normal. More importantly the National Weather Service expects Spring and Summer runoff to be between 90%-109% of normal.

The Arizona Water Banking Authority (AWBA) was created to store Arizona's unused Colorado River water entitlement in western, central and southern Arizona to develop long-term storage credits to: (i) firm existing water supplies for municipal and industrial users during Colorado River shortages or Central Arizona Project (CAP) service interruptions; (ii) help meet the water management objectives of the Arizona Groundwater Code; and (iii) assist in the settlement of American Indian water rights claims. The ABWA successfully banked up to Arizona's full allocation of Colorado River entitlement in 2002, and expects the same result in 2003 as a hedge against drought conditions. Further, the first curtailment of CAP deliveries in the event of shortage would occur to non-Indian agricultural users. Such users accounted for 36% of CAP deliveries in 2002, creating a buffer for users such as CCWC. Though it is difficult to predict drought conditions with certainty, the activities of AWBA, and the priority for users of CAP, such as CCWC, provides an improved outlook for CCWC supplies.

CCWC obtains its water supply from three operating wells and from Colorado River water delivered by the CAP. The majority of CCWC's water supply is obtained from its CAP allocation and well water is used for peaking capacity in excess of treatment plant capability, during treatment plant shutdown, and to keep the well system in optimal operating condition.

CCWC has an Assured Water Supply designation, by decision and order of the Arizona Department of Water Resources, providing in part that, subject to its requirements, CCWC currently has a sufficient supply of ground water and CAP water which is physically, continuously and legally available to satisfy current and committed demands of its customers, plus at least two years of predicted demands, for 100 years. Notwithstanding such a designation, CCWC's water supply may be subject to interruption or reduction of CAP water. In the event of interruption or reduction of CAP water, CCWC can currently rely on its well water supplies for short-term periods. However, in any event, the quantity of water CCWC supplies to some or all of its customers may be interrupted or curtailed, pursuant to the provisions of its tariffs.

#### **Risk Factor Summary**

You should carefully read the risks described below and other information in this Form 10-K in order to understand certain of the risks of our business.

#### Our liquidity and earnings could be adversely affected by increases in water supply costs

Prior to November 29, 2001, we recovered certain water supply costs through a balancing account mechanism. Water supply costs include the cost of purchased water and power and groundwater production assessments. Changes in the unit costs did not directly affect earnings. The balancing account was not, however, designed to insulate SCW's earnings against changes in supply mix. As a result, SCW was not permitted to recover increased costs due to increased use of purchased water, which is generally more expensive than groundwater, through the balancing account mechanism. Increased cost of this nature are only recoverable, on a prospective basis, through the filing of a general rate case.

30

On November 29, 2001, the CPUC ordered us to suspend the use of the current water balancing account, and instead started a memorandum account for each offsettable expense of purchased water, purchased power and pump tax for its water service areas. However, the CPUC has not issued a final decision regarding the mechanism for recovery of costs recorded in the newly established memorandum account. Our recovery of deferred water supply costs for providing water service could be reduced if we are earning more than our authorized rate of return on a weather normalized basis. Additionally, changes in water supply costs compared to the authorized amount, as well as any future authorized offset increases may directly affect our earnings. SCW's earnings in 2002 were reduced by \$0.16 per share as a result of this change in the methodology for recovering water supply costs. While we cannot determine the outcome, SCW has filed various rate cases which include requests to recover the above mentioned memorandum account dollars.

Under California law, we are permitted to file for a rate increase to recover electric power costs not being recovered in current rates. Increases in electric power costs generally have no direct impact on profit margins, unless recovery of these costs is disallowed, but do affect cash flows and can therefore impact the amount of our capital resources. Electric power costs increased substantially in California during the fall of 2000 until the summer of 2001. On July 17, 2002, the FERC extended and modified the mitigation measures that were set to expire on September 30, 2002, citing delays in construction of new generation resources in California and throughout the West, delays in adopting a new market design and market rules by the Cal ISO, transmission line constraints, constraints on natural gas pipeline capacity and continuing dysfunctions in the power market. It remains unclear how long the FERC will leave its mitigation measures in place. The premature terforms are under consideration at FERC. Certain of these reforms, if adopted, could result in significantly increased electric supply reserve requirements that have the potential for further cost increases.

On July 17, 2002, the CPUC approved a settlement authorizing us to include \$0.077 per kilowatt-hour (KWh) in rates to recover electric power costs at our Bear Valley Electric operation. If our actual annual costs exceed this amount, we cannot recover the excess and the amount will be expensed against income. If our actual annual energy costs are less than \$0.077 per KWh, we can use this difference to collect amounts previously included in the balancing account. The new rates have been in effect since July 17, 2002. As of December 31, 2002, we have accrued \$24.9 million in under-collected power costs in our electric supply balancing account. The CPUC has authorized SCW to collect a surcharge from its customers of 2.2¢ per Kwh for a period of up to ten years to enable SCW to recover this under-collection. SCW anticipates that the 2.2¢ surcharge on its electricity sales during the ten year period will be sufficient to enable SCW to recover the amount of this under-collection, with interest, during this ten year period.

The purchased power agreements with Pinnacle West Capital Corporation (PWCC) enable SCW to purchase a contractual volume of power at \$74.65 per megawatt hour (MWh). The agreements with PWCC have not been reviewed by the CPUC and we are unable to predict if the regulatory body will authorize recovery of any or all of these costs.

SCW still needs to buy additional energy from the spot market to meet peak demand and sell surplus power to the spot market from time to time. The average cost of power purchased including the transactions in the spot market was approximately \$95.38 per MWh for 2002. SCW's energy cost can be impacted by pricing fluctuation, and amount of energy purchased and sold at any given time in the spot market.

#### Significant claims have been asserted against us in water quality litigation

SCW and others have been sued in twenty-two water quality related lawsuits alleging personal injury and property damage as a result of the delivery of water that was allegedly contaminated. Nineteen of the lawsuits involve plaintiffs who received water from two groundwater basins in Los Angeles County. The other lawsuits involve plaintiffs in Sacramento County.

In March 1998, the CPUC issued an Order Instituting Investigation as a result of water quality lawsuits being filed against water utilities in California. On November 2, 2000, the CPUC issued a final order concluding that (i) the CPUC has jurisdiction to regulate the service of water utilities with respect to the health and safety of

31

that service; (ii) DOHS requirements governing drinking water quality adequately protect the public health and safety; and (iii) regulated water utilities, including SCW, have satisfactorily complied with past and present drinking water quality requirements.

On February 5, 2002, the California Supreme Court ruled that water utilities regulated by the CPUC may only be sued for damages based on allegations and ultimate proof that the utility failed to comply with federal and state safe drinking water requirements. As a result, plaintiffs may proceed on their claims against SCW to the extent that these claims are based on violations of federal and state law. The extent which SCW and other CPUC regulated water utilities may be subject to liability under the ruling is currently being addressed by the trial court in the area that gave rise to the Supreme Court ruling.

SCW is unable to predict the outcome of any of this litigation or the extent to which it will be able to recover its litigation costs from ratepayers or other third parties.

# Our operating costs have increased and are expected to continue to increase as a result of groundwater contamination

SCW's operations have been impacted by groundwater contamination in certain of its service territories. We have taken a number of steps to address this contamination, including the removal of wells from service, the construction of water treatment facilities and securing alternative sources of supply from other areas not affected by the contamination.

In some cases, potentially responsible parties have reimbursed us for our costs. In other cases, we have taken legal action against parties that we believe to be potentially responsible for the contamination.

The Potentially Responsible Parties, i.e. the polluters, have raised the possibility that water producers, such as SCW and CCWC, may have liability under certain environmental statutes if their pumping operations affect the movement of the contamination. SCW has been required to remove certain wells from service because its pumping activities might affect the movement of contamination. Currently, neither the Environmental Protection Agency nor any other governmental agency has identified SCW or, to our knowledge, any other water producer, as a potentially responsible party. We cannot assure you, however, that SCW or CCWC will not be identified as a potentially responsible party in the future. Our future results of operations could be adversely affected if either SCW or CCWC is required to pay clean-up costs and is not allowed to recover such costs in rates. To date, the CPUC has permitted SCW to establish memorandum accounts for recovery of these types of costs, but actual recovery of these costs from ratepayers has been limited to filing only with a General Rate Case, generally every 3 years.

## Environmental regulation has increased, and is expected to continue to increase, our operating costs

SCW and CCWC are subject to increasingly stringent environmental regulations that will result in increasing capital and operating costs. These regulations include:

- · The 1996 amendments to the Safe Drinking Water Act that require increased testing and treatment of water to reduce specified contaminants to maximum contaminant levels
- Approved regulations requiring increased surface-water treatment to decrease the risk of microbial contamination; these regulations affect SCW's five surface water treatment
  plants and one CCWC plant
- Additional regulation of disinfection/disinfection byproducts
- Additional regulations expected to be adopted requiring disinfection of certain groundwater systems
- Regulation of arsenic and radon
- Changes in the action level and the proposed adoption of maximum contamination levels for perchlorate

SCW and CCWC may be able to recover costs incurred to comply with these regulations through the ratemaking process. We may also be able to recover certain of these costs under our contractual arrangements

32

with municipalities. In certain circumstances, we may be able to recover costs from parties responsible or potentially responsible for contamination, either voluntarily or through specific court action.

# The adequacy of our water supplies depends upon a variety of factors beyond our control

The adequacy of our water supplies varies from year to year depending upon a variety of factors, including:

- Rainfall
- Availability of Colorado River water
- The amount of water stored in reservoirs and groundwater basins
- The amount of water used by our customers and others
- Water quality
- Legal limitations on use

Population growth and increases in the amount of water used have increased limitations on use to prevent over-drafting of groundwater basins. The import of water from the Colorado River, one of SCW's important sources of supply, is expected to decrease in future years due to the requirements of the Central Arizona Project ("CAP") and disputes affecting the amount of water that the Metropolitan Water District of Southern California is entitled to take from the Colorado River. We have also taken wells out of service due to groundwater contamination.

CCWC obtains its water supply from operating wells and from the Colorado River through the CAP. CCWC's water supply may be subject to interruption or reduction if there is an interruption or reduction in CAP water.

Water shortages may affect us in a variety of ways:

- They adversely affect supply mix by causing us to rely on more expensive purchased water.
- They adversely affect operating costs.
- They may result in an increase in capital expenditures for building pipelines to connect to alternative sources of supplies and reservoirs and other facilities to conserve or reclaim water.

We may be able to recover increased operating and construction costs for our regulated systems through the ratemaking process. We may also be able to recover certain of these costs from third parties that may be responsible, or potentially responsible, for groundwater contamination.

## Our earnings are greatly affected by weather during different seasons

The demand for water and electricity varies by season. Therefore, the results of operations for one period may not indicate results to be expected in another period. For instance, most water consumption occurs during the third quarter of each year when weather tends to be hot and dry. On warm days, use of water by residential and commercial customers may be significantly greater than on cold days because of the increased use of water for outdoor landscaping. Likewise the demand for electricity in our Big Bear service area is greatly affected by winter snows. An increase in winter snows reduces the use of snow making machines at ski resorts in the Big Bear area and, as a result reduces electric revenues. Likewise, unseasonably warm weather during a skiing season may result in temperatures too high for snowmaking conditions also reducing electric revenues.

Variability of weather from normal temperatures or changes in snow or rainfall can materially impact results of operations. As a result, weather has been and will continue to be one of the dominant factors in our financial performance.

## Our business is heavily regulated and, as a result, decisions by regulatory agencies and changes in laws and regulations can significantly affect our business

33

Our revenues depend substantially on the rates that we are permitted to charge our customers and our ability to recover our costs in these rates, including the ability to recover the costs of purchased water, groundwater assessments and electric power costs in rates. We have filed for increased water rates to recover operating costs from customers in Region III as well as from all customers for costs associated with general office activities. We also filed Notices of Intent to increase rates in Region I and Region II. In addition, we are seeking CPUC authorization to construct an 8.4 MW natural gas-fueled generator facility to meet increasing demand in our Bear Valley customer service area. The CPUC's resolution ordering SCW to suspend the use of the current water balancing account, and instead to start a memorandum account for our supply costs has and will directly impact our earnings. The recovery of deferred water supply costs may be reduced if we are earning an amount in excess of our authorized rate of return.

We have been adversely affected by electric restructuring in California and the escalation of energy costs that we have only recently begun recovering in rates. SCW has also filed a complaint with FERC seeking a reduction of the rates in its power purchase contract with Mirant Americas Energy Marketing, L.P. (Mirant) to a just and reasonable price, which SCW expects to be denied. There is a complaint by Mirant against us pending at the FERC. There are also proceedings pending before FERC to restructure the California energy markets.

## Our business requires significant capital expenditures

The utility business is capital intensive. On an annual basis, we spend significant sums for additions to or replacement of property, plant and equipment. During calendar years 2002, 2001 and 2000, we spent \$40,655,000, \$47,570,000, and \$46,091,000, respectively, for these purposes. Our budgeted capital expenditures for calendar year 2003 for these purposes are approximately \$84,637,000. There is no capital expenditure for environmental control facilities budgeted for 2003.

We obtain funds for these capital projects from operations, contributions by developers and others and advances from developers (which must be repaid). We also periodically borrow money or issue equity for these purposes. In addition, we secured a syndicated bank facility that we can use for these purposes. We cannot assure you that these sources will continue to be adequate or that the cost of funds will remain at levels permitting us to earn a reasonable rate of return.

#### New Accounting Pronouncements

We are subject to newly issued as well as changes in existing requirements issued by the Financial Accounting Standard Board. Differences in financial reporting between periods could occur unless and until the CPUC and the ACC approve such changes for conformity through regulatory proceedings. See *Note 1* of Notes to Consolidated Financial Statements.

### Item 7A. Quantitative and Qualitative Disclosures About Market Risk

Registrant has certain block-forward purchase power contracts that qualify as derivative instruments under Statement of Financial Account Standards No. 133 (SFAS No. 133), "Accounting for Derivative Instruments and Hedging Activities." A derivative financial instrument or other contract derives its value from another investment or designated benchmark. SCW is a party to various block-forward purchase power contracts. Power purchase contracts with Mirant America Energy Marketing, L.P. (Mirant) and Pinnacle West Capital Corporation (PWCC) executed in 2001 qualify for the exception provided under SFAS No. 133 for activities that are considered normal purchases and normal sales. These contracts are reflected in the statements of income at the time of contract settlement. Contracts with PWCC executed in September 2002, however, do not qualify for the normal purchases and normal sales exception and, as a result, has been recognized at fair market value on the balance sheet as of December 31, 2002. This resulted in a pre-tax unrealized loss of \$2.5 million in 2002 recorded as a component of Power Purchased for Resale. On a quarterly basis, the related asset or liability will be adjusted to reflect the fair market value at the end of the quarter. As this contract is settled, the realized gains or losses will be recorded and the unrealized gain or loss

Under the terms of its power purchase contracts with Mirant and PWCC, SCW is required to post security, at the request of the seller, if SCW is in default under the terms of the applicable contract. In addition, SCW's liquidity, and in certain circumstances, earnings could be adversely affected by increases in electricity prices in California. For further information, see the section entitled "Critical Accounting Policies" and "Electric Energy Situation in

34

California" included in Part II, Item 7 in Management's Discussion and Analysis of Financial Condition and Results of Operation.

Except as discussed above, Registrant has no other derivative financial instruments, financial instruments with significant off-balance sheet risks or financial instruments with concentrations of credit risk.

# Item 8. Financial Statements and Supplementary Data

American States Water Company

Consolidated Balance Sheets - December 31, 2002 and 2001

Consolidated Statements of Income - for the years ended December 31, 2002, 2001 and 2000

Consolidated Statements of Changes in Common Shareholders' Equity - for the years ended December 31, 2002, 2001 and 2000

Consolidated Statements of Cash Flows - for the years ended December 31, 2002, 2001 and 2000

# Southern California Water Company

Balance Sheets - December 31, 2002 and 2001

Statements of Capitalization - December 31, 2002 and 2001

Statements of Income - for the years ended December 31, 2002, 2001 and 2000

Statements of Changes in Common Shareholder's Equity - for the years ended December 31, 2002, 2001 and 2000

Statements of Cash Flows - for the years ended December 31, 2002, 2001 and 2000

Notes to Consolidated Financial Statements

Report of Management

Report of Independent Accountants

36

# AMERICAN STATES WATER COMPANY CONSOLIDATED BALANCE SHEETS

	Decen	ıber 31,
(in thousands)	2002	2001 Restated Note 2
Assets		
Utility Plant, at cost Water	\$ 693,949	\$ 645,185
Electric		,
Electric	41,017	38,525
	734,966	683,710
Less – Accumulated depreciation	(206,873)	(190,656)
	528,093	493,054
Construction work in progress	35,218	46,788
Net utility plant	563,311	539,842
Other Property and Investments	22,670	24,104
Current Assets		
Cash and cash equivalents	18,397	30,496
Accounts receivable-customers (less allowance for doubtful accounts of \$769 in 2002 and \$972 in 2001)	10,833	10,557
Unbilled revenue	12,277	12,141
Other accounts receivable	2,411	1,997
Taxes receivable	557	_
Materials and supplies, at average cost	936	970
Supply cost balancing accounts – current	3,215	5,505
Prepayments	3,220	2,493
Total current assets	51,846	64,159
Deferred Charges and Other Assets		
Regulatory tax-related assets	12,828	13,908
Supply cost balancing accounts	25,248	20,321
Other accounts receivable	3,370	3,309
Other	22,377	16,186
Total deferred charges	63,823	53,724
Total Assets	\$ 701,650	\$ 681,829

The accompanying notes are an integral part of these consolidated financial statements.

# AMERICAN STATES WATER COMPANY CONSOLIDATED BALANCE SHEETS

		December 31,	
(in thousands)	2	2002	2001 Restated Note 2
Capitalization and Liabilities			

Capitalization		
Common shareholders' equity	\$ 213,279	\$ 204,654
Preferred Shares	_	1,600
Preferred Shares – mandatory redemption	_	280
Long-term debt	231,089	245,692
Total capitalization	444,368	452,226

# **Current Liabilities**

Notes payable to banks	35,000	20,000
Long-term debt and Preferred Shares – current	13,305	800
Accounts payable	11,600	13,086
Taxes payable	_	404
Accrued employee expenses	5,105	4,290
Accrued interest	1,941	1,945
Reserves for potential disallowance of power costs	_	7,948
Other	12,569	10,178
Total current liabilities	79,520	58,651

# Other Credits

Advances for construction	70,208	69,436
Contributions in aid of construction – net	47,751	43,723
Accumulated deferred income taxes – net	53,817	51,541
Unamortized investment tax credits	2,791	2,882
Regulatory tax-related liability	2,011	2,054
Other	1,184	1,316
Total other credits	177,762	170,952

# Commitments and Contingencies (Notes 11 and 12)

Total Capitalization and Liabilities	\$ 701,650	\$ 681,829

The accompanying notes are an integral part of these consolidated financial statements.

38

# AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF CAPITALIZATION

	Decen	nber 31,
(in thousands)	2002	2001 Restated Note 2
Common Shareholders' Equity:		
Common Shares, no par value, \$2.50 stated value:		
Authorized: 30,000,000 shares		
Outstanding: 15,180,835 shares in 2002 and 15,119,444 shares in 2001	\$ 25,341	\$ 25,199
Additional paid-in capital	101,606	100,239
Earnings reinvested in the business	86,332	79,216
	213,279	204,654
Preferred Shares: \$25 par value		
Authorized: 0 shares in 2002 and 64,000 shares in 2001		
Series 4% - 0 and 32,000 shares outstanding in 2002 and 2001, respectively	_	800
Series 4 1/4 % - 0 and 32,000 shares outstanding in 2002 and 2001, respectively		800
	_	1,600

Less: Preferred Shares to be redeemed within one year  Long-Term Debt (All are of SCW unless otherwise noted) Notes/Debentures: 5.82% notes due 2003 6.64% notes due 2013 6.80% notes due 2013 6.80% notes due 2013 6.80% notes due 2023 7.00% notes due 2023 7.00% notes due 2025 7.65% notes due 2025 6.81% notes due 2025 6.81% notes due 2029 7.875% notes due 2030 7.23% notes due 2031 Private Placement Notes: 9.56% notes due 2031 Tax-Exempt Obligations: 5.50% notes due 2026		(40)
Notes/Debentures:           5.82% notes due 2003           6.64% notes due 2013           6.80% notes due 2013           6.80% notes due 2023           7.00% notes due 2023           7.00% notes due 2023           7.55% notes due 2025           6.81% notes due 2028           6.59% notes due 2029           7.875% notes due 2030           7.23% notes due 2031           Private Placement Notes:           9.56% notes due 2031           Tax-Exempt Obligations:		280
Notes/Debentures:           5.82% notes due 2003           6.64% notes due 2013           6.80% notes due 2013           6.80% notes due 2023           7.00% notes due 2023           7.55% notes due 2025           7.65% notes due 2025           6.81% notes due 2028           6.59% notes due 2029           7.875% notes due 2030           7.23% notes due 2031           Private Placement Notes:           9.56% notes due 2031           Tax-Exempt Obligations:		280
Notes/Debentures:           5.82% notes due 2003           6.64% notes due 2013           6.80% notes due 2013           6.80% notes due 2023           7.00% notes due 2023           7.00% notes due 2023           7.55% notes due 2025           7.65% notes due 2025           6.81% notes due 2028           6.59% notes due 2029           7.875% notes due 2030           7.23% notes due 2031           Private Placement Notes:           9.56% notes due 2031           Tax-Exempt Obligations:	12 500	
5.82% notes due 2003         6.64% notes due 2013         6.80% notes due 2013         6.87% notes due 2023         7.00% notes due 2023         7.55% notes due 2025         7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	12 500	
6.64% notes due 2013         6.80% notes due 2013         6.87% notes due 2023         7.00% notes due 2023         7.55% notes due 2025         7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	12 500	
6.80% notes due 2013         6.87% notes due 2023         7.00% notes due 2023         7.55% notes due 2025         7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:		12,500
6.87% notes due 2023         7.00% notes due 2023         7.55% notes due 2025         7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	1,100	1,100
7.00% notes due 2023         7.55% notes due 2025         7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	2,000	2,000
7.55% notes due 2025         7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	5,000	5,000
7.65% notes due 2025         6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	10,000	10,000
6.81% notes due 2028         6.59% notes due 2029         7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	8,000	8,000
6.59% notes due 20297.875% notes due 20307.23% notes due 2031Private Placement Notes:9.56% notes due 2031Tax-Exempt Obligations:	22,000	22,000
7.875% notes due 2030         7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	15,000	15,000
7.23% notes due 2031         Private Placement Notes:         9.56% notes due 2031         Tax-Exempt Obligations:	40,000	40,000
Private Placement Notes: 9.56% notes due 2031 Tax-Exempt Obligations:	20,000	20,000
9.56% notes due 2031 Tax-Exempt Obligations:	50,000	50,000
Tax-Exempt Obligations:		
	28,000	28,000
5.50% notes due 2026		
	7,945	7,950
Variable rate obligation due 2014	6,000	6,000
State Water Project due 2035	5,508	6,870
Other Debt Instruments:	,	,
8.50% fixed rate obligation due 2013	1,530	1,630
Variable rate obligation due 2018	554	589
Capital lease obligations	368	425
Chaparral City Water Company:		
4% to 4.85% serial bonds due 2007	1,100	1,295
5.20% term bonds due 2011	1,000	1,000
5.40% term bonds due 2022	4,610	4,610
4.65% term bonds due 2006	150	185
5.30% term bonds due 2022	1,015	1,015
3.34% repayment contract due 2006	1,014	1,283
sis no tepaj inter contrat da 2000		1,205
	244,394	246,452
Less: Current maturities	(13,305)	(760)
	231,089	245,692
Total Capitalization	\$ 444,368	\$ 452,226

The accompanying notes are an integral part of these consolidated financial statements.

39

# AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF INCOME

	For the y	For the years ended December 31,	
(in thousands, except per share amounts)	2002	2001 Restated Note 2	2000 Restated Note 2
Operating Revenues			
Water	\$ 187,061	\$ 181,474	\$ 168,795
Electric	21,298	15,251	14,366
Other	846	789	799
Total operating revenues	209,205	197,514	183,960
Operating Expenses			
Water purchased	42,859	37,609	41,592
Power purchased for resale	18,348	19,662	10,664
Power purchased for pumping	10,576	9,592	7,509
Groundwater production assessment	7,416	6,847	7,489
Supply cost balancing accounts	(3,406)	(14,681)	(6,371)
Other operating expenses	16,983	17,162	16,748
Administrative and general expenses	30,010	35,108	26,135
Depreciation and amortization	18,302	17,951	15,339
Maintenance	9,839	8,640	10,280
Taxes on income	12,949	14,370	13,182
Property and other taxes	7,681	7,553	7,141
Total operating expenses	171,557	159,813	149,708

Operating Income	37,648	37,701	34,252
Other Income			
Total other income (loss) - net	390	(510)	(99)
Total other meonic (1655) - net		(510)	())
Income before interest charges	38,038	37,191	34,153
Interest Charges		12.105	
Interest on long-term debt	16,972	13,497	11,623
Other interest and amortization of debt expense	727	2,238	2,499
T-41	17 (00	15 725	14 122
Total interest charges	17,699	15,735	14,122
Net Income	20,339	21,456	20,031
Dividends on Preferred Shares	(29)	(84)	(86)
Femilian Andlehle Fen Common Chambeldon	¢ 20.210	¢ 01 270	¢ 10.045
Earnings Available For Common Shareholders	\$ 20,310	\$ 21,372	\$ 19,945
Weishand American Neuroban of Commune Shares Outstanding	15 144	15 120	14.070
Weighted Average Number of Common Shares Outstanding	15,144 \$ 1.34	15,120 \$ 1.41	14,070
Basic Earnings Per Common Share	\$ 1.34	\$ 1.41	\$ 1.42
	16.167	15 100	14.070
Weighted Average Number of Diluted Shares Outstanding	15,157	15,122	14,070
Fully Diluted Earnings Per Common Share	\$ 1.34	\$ 1.41	\$ 1.42
Dividends Declared Per Common Share	\$ 0.872	\$ 0.867	\$ 0.857
	\$ 0.072	- 0.007	- 0.007

The accompanying notes are an integral part of these consolidated financial statements.

40

# AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF CHANGES IN COMMON SHAREHOLDERS' EQUITY

	Commo	Common Shares		Common Shares		nmon Shares			
(in thousands, except per share data)	Number of Shares	Amount	Additional Paid-in Capital	Earnings Reinvested in the Business	Total				
Balances at December 31, 1999 – Previously Reported	13,437	\$ 22,394	\$ 74,937	\$ 61,515	\$ 158,846				
Cumulative adjustment for taxes – Note 2				1,718	1,718				
Balances at December 31, 1999 (Restated–Note 2)	13,437	22,394	74,937	63,233	160,564				
Add:									
Net income (Restated – Note 2)				20,031	20,031				
Issuance of Common Shares for public offering	1,660	2,768	24,924		27,692				
Issuance of Common Shares, others	22	37	378		415				
Deduct:									
Dividends on Preferred Shares				86	86				
Dividends on Common Shares-\$0.857 per share		. <u></u>		12,230	12,230				
Balances at December 31, 2000 (Restated–Note 2)	15,119	25,199	100,239	70,948	196,386				
Add:									
Net income (Restated – Note 2)				21,456	21,456				
Deduct:									
Dividends on Preferred Shares				84	84				
Dividends on Common Shares-\$0.867 per share				13,104	13,104				
Balances at December 31, 2001 (Restated–Note 2)	15,119	25,199	100,239	79,216	204,654				
Add:									
Net income				20,339	20,339				
Issuance of Common Shares, others	62	142	1,367		1,509				
Deduct:									
Dividends on Preferred Shares				29	29				
Dividends on Common Shares-\$0.872 per share				13,194	13,194				
Balances at December 31, 2002	15,181	\$ 25,341	\$ 101,606	\$ 86,332	\$ 213,279				

41

# AMERICAN STATES WATER COMPANY CONSOLIDATED STATEMENTS OF CASH FLOWS

2000 Restated Note 2 5 \$ 20,03 5 \$ 20,035 5
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
$\begin{array}{cccccccccccccccccccccccccccccccccccc$
2 5,72 7 1,77 6) 27 8) (11 8) (88 8) (88 8) 1,02 9) (6,37 8) (2,37 5 (2,74
2 5,72 7 1,77 6) 27 8) (11 8) (88 8) (88 8) 1,02 9) (6,37 8) (2,37 5 (2,74
7       1,774         5)       276         8)       (11)         8)       (88)         3)       (88)         3)       (637)         5)       (2,37)         5)       (2,74)
b)       27         c)       (1)         c)       (88)         c)       (637)         c)       (6,37)         c)       (2,37)         c)       (2,74)
$\begin{array}{c} (1) \\ (3) \\ (8) \\ (8) \\ (8) \\ (6,37) \\ (2,37) \\ (2,74) \\ (2$
$\begin{array}{c} (1) \\ (8) \\ (8) \\ (8) \\ (8) \\ (6,37) \\ (2,37) \\ (2,74) \\ (2,74) \\ (2,74) \\ (1) \\ (1) \\ (1) \\ (2$
3 1,02 ) (6,37 3) (2,37 5 (2,74)
(6,37 (2,37 (2,74)
3) (2,37 5 (2,74
(2,74
) (1,66
1,12
31,22
) (46,09
- (18,48
- (1,65
)) (66,22
- 28,10
) —
2,25
(2,80)
5) (61 
) 24,00
3) (12,31
38,61
3,61
5 \$ 5,80
\$ 9,43
\$ 14,37
\$ 2,62
- \$ 48,042
- (25,35
- 22,68
- 22,68
- 14/11
- \$ 18,484
,089

The accompanying notes are an integral part of these consolidated financial statements.

2002	2001
	Restated
	Note 2

# Assets

Utility Plant, at cost		
Water	\$ 656,331	\$ 607,988
Electric	41,017	38,525
		<u> </u>
	697,348	646,513
Less – Accumulated depreciation	(196,660)	(181,371)
		165.1.10
	500,688	465,142
Construction work in progress	33,705	46,042
Net utility plant	534,393	511,184
Other Property and Investments	8,018	9,446

# **Current Assets**

Cash and cash equivalents	11,677	26,079
Accounts receivable-customers (less allowance for doubtful accounts of \$729 in 2002 and \$951 in 2001)	10,609	10,228
Unbilled revenue	12,060	11,940
Inter-company receivable	1,044	250
Other accounts receivable	2,234	1,893
Materials and supplies, at average cost	905	883
Supply cost balancing accounts – current	3,215	5,505
Prepayments	3,005	2,310
Total current assets	44,749	59,088

# Deferred Charges and Other Assets

Regulatory tax-related assets	12,828	13,908
Supply cost balancing accounts	25,248	20,321
Other accounts receivable	3,370	3,309
Other	21,405	15,433
Total deferred charges	62,851	52,971
Total Assets	\$ 650,011	\$ 632,689

The accompanying notes are an integral part of these financial statements

# 43

# SOUTHERN CALIFORNIA WATER COMPANY BALANCE SHEETS

	Decem	nber 31,
(in thousands)	2002	2001 Restated Note 2
Capitalization and Liabilities		
Capitalization		
Common shareholder's equity	\$ 207,562	\$ 200,972
Long-term debt	222,725	236,804
Total capitalization	430,287	437,776
Current Liabilities		
Long-term debt – current	12,780	260
Accounts payable	10,576	12,729
Inter-company payable	13,000	276
Taxes payable	1,099	1,194
Accrued employee expenses	5,033	4,254
Accrued interest	1,843	1,877
Reserves for potential disallowance of power costs	-	7,948
Other	12,276	9,977
Total current liabilities	56,607	38,515

Other Credits		
Advances for construction	59,640	58,570
Contributions in aid of construction – net	47,480	43,493
Accumulated deferred income taxes – net	51,195	49,399
Unamortized investment tax credits	2,791	2,882
Regulatory tax-related liability	2,011	2,054
Total other credits	163,117	156,398

Commitments and Contingencies (Notes 11 and 12)

Total Capitalization and Liabilities	\$ 650,011 \$ 632,689

The accompanying notes are an integral part of these financial statements.

44

# SOUTHERN CALIFORNIA WATER COMPANY STATEMENTS OF CAPITALIZATION

	Decem	nber 31,
(in thousands)	2002	2001 Restated Note 2
Common Shareholder's Equity:		
Common shares, no par value		
Outstanding: 110 shares in 2002 and 2001	\$ 123,391	\$ 123,391
Earnings reinvested in the business	84,171	77,581
	207,562	200,972
Long-Term Debt		
Notes/Debentures:		
5.82% notes due 2003	12,500	12,500
6.64% notes due 2013	1,100	1,100
6.80% notes due 2013	2,000	2,000
6.87% notes due 2023	5,000	5,000
7.00% notes due 2023	10,000	10,000
7.55% notes due 2025	8,000	8,000
7.65% notes due 2025	22,000	22,000
6.81% notes due 2028	15,000	15,000
6.59% notes due 2029	40,000	40,000
7.875% notes due 2030	20,000	20,000
7.23% notes due 2031	50,000	50,000
Private Placement Notes:		
9.56% notes due 2031	28,000	28,000
Tax-Exempt Obligations:		
5.50% notes due 2026	7,945	7,950
Variable rate obligation due 2014	6,000	6,000
State Water Project due 2035	5,508	6,870
Other Debt Instruments:		
8.50% fixed rate obligation due 2013	1,530	1,630
Variable rate obligation due 2018	554	589
Capital lease obligations	368	425
	235,505	237,064
Less: Current maturities	(12,780)	
	222,725	236,804
Total Capitalization	\$ 430,287	\$ 437,776

The accompanying notes are an integral part of these financial statements.

# SOUTHERN CALIFORNIA WATER COMPANY STATEMENTS OF INCOME

	For the	For the years ended December 31,				
(in thousands, except share and per share amounts)	2002	2001 Restated Note 2	2000 Restated Note 2			
Operating Revenues						
Water	\$ 180,904	\$ 175,204	\$ 167,529			
Electric	21,298	15,251	14,366			
Total operating revenues	202,202	190,455	181,895			
Operating Expenses						
Water purchased	42,182	37,112	41,450			
Power purchased for resale	18,348	19,662	10,664			
Power purchased for pumping	10,167	9,169	7,442			
Groundwater production assessment	7,416	6,847	7,489			
Supply cost balancing accounts	(3,406)	(14,681)	(6,371)			
Other operating expenses	15,823	16,111	16,306			
Administrative and general expenses	26,174	33,929	25,545			
Depreciation and amortization	17,384	16,710	15,086			
Maintenance	9,603	8,411	10,191			
Taxes on income	13,605	14,057	12,743			
Property and other taxes	7,368	7,089	7,037			
Total operating expenses	164,664	154,416	147,582			
Operating Income	37,538	36,039	34,313			
Other Income						
Total other income (loss) - net	318	(624)	(140)			
Income before interest charges	37,856	35,415	34,173			
Interest Charges						
Interest on long-term debt	16,520	13,015	11,512			
Other interest and amortization of debt expense	116	1,562	2,838			
Total interest charges	16,636	14,577	14,350			
Net Income	\$ 21,220	\$ 20,838	\$ 19,823			
Weighted Average Number of Common Shares Outstanding	110	107	100			
Basic and Fully Diluted Earnings Per Common Share	\$ 192,909	\$ 194,748	\$ 198,230			
Dividends Declared Per Common Share	\$ 133,000	\$ 123,000	\$ 129,000			

The accompanying notes are an integral part of these financial statements.

46

# SOUTHERN CALIFORNIA WATER COMPANY STATEMENTS OF CHANGES IN COMMON SHAREHOLDER'S EQUITY

		on Shares		
(in thousands, except share and per share data))	Number of Shares	Amount	Earnings Reinvested in the Business	Total
Balances at December 31, 1999 – Previously Reported	100	\$ 98,391	\$ 61,632	\$ 160,023
Cumulative adjustment for taxes – Note 2			1,718	1,718
Balances at December 31, 1999 (Restated–Note 2)	100	98,391	63,350	161,741
Add:				
Net income (Restated – Note 2)			19,823	19,823
Deduct:				
Dividends on common shares-\$129,000 per share			12,900	12,900
		·		
Balances at December 31, 2000 (Restated – Note 2)	100	98,391	70,273	168,664

Add:				
Net income (Restated – Note 2)			20,838	20,838
Issuance of common shares	10	25,000		25,000
Deduct:				
Dividends on common shares-\$123,000 per share			13,530	13,530
Balances at December 31, 2001 (Restated – Note 2) Add:	110	123,391	77,581	200,972
Net income			21,220	21,220
Deduct:				
Dividends on common shares-\$133,000 per share			14,630	14,630
Balances at December 31, 2002	110	\$ 123,391	\$ 84,171	\$ 207,562

The accompanying notes are an integral part of these financial statements.

47

# SOUTHERN CALIFORNIA WATER COMPANY STATEMENTS OF CASH FLOWS

	For the year	For the years ended December 31,		
(in thousands)	2002	2001 Restated Note 2	2000 Restated Note 2	
Cash Flows From Operating Activities:				
Net income	\$ 21,220	\$ 20,838	\$ 19,823	
Adjustments for non-cash items:	· · · ·	• • • • • •	,	
Depreciation and amortization	17,384	16,710	15,086	
Deferred income taxes and investment tax credits	2,742	7,171	5,642	
Other – net	2,702	(308)	801	
Changes in assets and liabilities:	_,,	(200)		
Accounts receivable – customers	(381)	(157)	64	
Unbilled revenue	(120)	(577)	(18	
Other accounts receivable	(402)	(105)	(822	
Prepayments and other current assets	(717)	1,602	1,209	
Supply cost balancing accounts	(3,406)	(14,681)	(6,371	
Deferred charges and other assets	(6,249)	(3,660)	(3,001	
Accounts payable				
Inter-company	(2,153) 11,930	2,726 (4,344)	(2,556 4,366	
Taxes payable	(95)	(767)	(2,120)	
Other current liabilities	(4,904)	7,582	976	
Net cash provided	37,551	32,030	33,079	
Net cash used	(39,488)	(47,029)	(44,077)	
Cash Flows From Financing Activities:				
Proceeds from issuance of common and preferred shares	_	25,000	_	
Proceeds from issuance of long-term debt	_	70,000	-	
Receipt of advances for and contributions in aid of construction	6,912	6,315	2,596	
Refunds on advances for construction	(3,188)	(3,019)	(2,807	
Repayments of long-term debt	(1,559)	(233)	(366	
Net change in notes payable to banks	_	(45,000)	24,000	
Common dividends paid	(14,630)	(13,530)	(12,900	
Net cash provided (used)	(12,465)	39,533	10,523	
Not increases (decreases) in cash and cash againstants	(14.402)	24 524	(175	
Net increase (decrease) in cash and cash equivalents	(14,402) 26,079	24,534 1,545	(475	
Cash and cash equivalents, beginning of year	28,079	1,545	2,020	
Cash and cash equivalents, end of year	\$ 11,677	\$ 26,079	\$ 1,545	
Faxes and Interest Paid:				
Income taxes paid	\$ 11,624	\$ 6,602	\$ 9,152	
Interest paid	\$ 17,101			
Non-Cash Transactions:				
Property installed by developers and conveyed to Company	\$ 2,247	\$ 1,919	\$ 2.627	
·····	÷ 2,2 17	,		

# AMERICAN STATES WATER COMPANY AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

### Note 1 - Summary of Significant Accounting Policies

<u>Nature of Operations</u>: American States Water Company (AWR) is the parent company of Southern California Water Company (SCW), American States Utility Services, Inc. (ASUS) and Chaparral City Water Company (CCWC). SCW is a public utility engaged principally in the purchase, production, distribution and sale of water in California. SCW also distributes electricity in several California mountain communities. The California Public Utilities Commission (CPUC) regulates SCW's water and electric business, including properties, rates, services, facilities and other matters. CCWC is a public utility regulated by the Arizona Corporation Commission (ACC) serving approximately 11,800 customers in the town of Fountain Hills, Arizona and a portion of the City of Scottsdale, Arizona. AWR completed the acquisition of the common stock of CCWC on October 10, 2000 for an aggregate value of \$31.2 million, including the assumption of approximately \$12 million in debt. ASUS performs non-regulated, water related services and operations on a contract basis. There is no direct regulatory oversight by either the CPUC or the ACC of ASUS or AWR. The consolidated financial statements include the accounts of AWR, SCW, ASUS and CCWC. AWR's assets and revenues are primarily those of SCW.

Basis of Presentation: The consolidated financial statements of AWR include the accounts of AWR and its wholly-owned subsidiaries SCW, ASUS and CCWC (collectively referred to as Registrant or the Company) and are prepared in conformity with accounting principles generally accepted in the United States of America. Inter-company transactions and balances have been eliminated. The preparation of these financial statements required the use of certain estimates by management in determining Registrant's assets, liabilities, revenues and expenses. Actual results could differ from those estimates. Certain prior year amounts have been reclassified to conform to current year presentation. None of these reclassifications had an impact on Registrant's Shareholders' Equity or Net Income. See Note 2 for discussion on restatement of previously issued financial statements.

Registrant's accounting policies conform to accounting principles generally accepted in the United States of America, including the accounting principles for rate-regulated enterprises, which reflect the rate-making policies of the CPUC, ACC and the Federal Energy Regulatory Commission (FERC). The utility subsidiaries, SCW and CCWC, have incurred various costs and received various credits reflected as regulatory assets and liabilities. Accounting for such costs and credits as regulatory assets and liabilities is in accordance with Statement of Financial Accounting Standards (SFAS) No. 71 "Accounting for the Effects of Certain Types of Regulation". This Statement sets forth the application of generally accepted accounting principles for those companies whose rates are established by or are subject to approval by an independent third-party regulator. Under SFAS 71, rate regulated entities defer costs and credits on the balance sheet as regulatory assets and liabilities when it is probable that those costs and credits will be recognized in the rate making process in a period different from the period in which the yould have been reflected in income by an unregulated company. These deferred regulatory assets and liabilities are then reflected in the income statement in the period in which the same amounts are reflected in the rates for the service.

<u>Property and Depreciation</u>: SCW and CCWC capitalize, as utility plant, the cost of additions and replacements of retirement units. Such cost includes labor, material and certain indirect charges. Depreciation is computed on the straight-line, remaining-life basis, based on depreciable plant as of the beginning of each year in accordance with Registrant's rate making process. For the years 2002, 2001 and 2000 the aggregate provisions for depreciation for SCW approximated 2.6% for its water distribution unit and 3.8% for its electric unit. The aggregate provision for depreciation for CCWC was 2.5% for each of the same three years.

Impairment of Long-Lived Assets: Long-lived assets are reviewed for impairment whenever events or changes in circumstances indicate that the carrying amount of an asset may not be fully recoverable. In August 2001, the Financial Accounting Standards Board (FASB) issued SFAS No. 144, "Accounting for the Impairment or Disposal of Long-Lived Assets," which supercedes SFAS No. 121, "Accounting for the Impairment of Long-Lived Assets and for Long-Lived Assets to be Disposed Of." This Statement retains the fundamental provisions of SFAS No. 121 for the measurement and recognition of the impairment of long-lived assets to be held and used, as well as the measurement of long-lived assets to be disposed of by sale. SFAS No. 144 resolves implementation issues related to SFAS No. 121, and retains the amendments in SFAS No. 121 pertaining to

49

regulatory assets under SFAS No. 71 and SFAS No. 90, "Regulated Enterprises – Accounting for Abandonments and Disallowances of Plant Costs." Effective January 1, 2002, Registrant adopted SFAS No. 144. The adoption of SFAS No. 144 did not have an impact on Registrant's consolidated financial statements.

Goodwill: At December 31, 2002 and 2001, Registrant had approximately \$12.3 million of goodwill included in "Other Property and Investments". The goodwill represents the difference between the purchase price of the common equity of CCWC and CCWC's book equity at the time of closing and was being amortized over a period of 40 years. Effective January 1, 2002, Registrant adopted SFAS No. 142, "Goodwill and Other Intangible Assets" and the provisions of SFAS No. 141, "Business Combinations," which were required to be adopted concurrent with the adoption of SFAS No. 142. Pursuant to SFAS No. 142, AWR completed its transitional goodwill impairment evaluation and determined that none of the recorded goodwill was impaired. As a result of the adoption of SFAS No. 142, the amortization of this goodwill ceased effective January 1, 2002. In accordance with SFAS No. 142, goodwill is tested for impairment at least annually and more frequently if circumstances indicate that it may be impaired. In the years 2001 and 2000, the recorded goodwill was amortized on a straight-line basis over a period of 40 years.

The following table reflects Registrant's earnings available for Common Shareholders and earnings per share adjusted to reflect the discontinuance of periodic goodwill amortization for the years ended December 31, 2002, 2001 and 2000, respectively:

(dollars in thousands except EPS)	2002 2001	2000
	Restated Note 2	Restated Note 2
Earnings available for Common Shareholders:		
As reported	\$ 20,310 \$ 21,37	2 \$ 19,945
Add back goodwill amortization, net of tax	— 19	1 35
	·	
As adjusted	\$20,310 \$21,56	3 \$ 19,980
Basic earnings per share:		
As reported	\$ 1.34 \$ 1.4	1 \$ 1.42
Add back goodwill amortization, net of tax	— 0.0	1 —
As adjusted	\$ 1.34 \$ 1.4	2 \$ 1.42
Diluted earnings per share:		
As reported	\$ 1.34 \$ 1.4	1 \$ 1.42
Add back goodwill amortization, net of tax	— 0.0	1 —
As adjusted	\$ 1.34 \$ 1.4	2 \$ 1.42

Interest: Interest is generally not capitalized for financial reporting purposes as such policy is not followed in Registrant's ratemaking process

<u>Revenues</u>: Revenues include amounts billed to customers for services provided and unbilled revenues representing estimated amounts to be billed for usage from the last meter reading date to the end of the accounting period.

Basic Earnings Per Common Share: Basic earnings per Common Share are based upon the weighted average number of Common Shares outstanding and net income after deducting preferred dividend requirements.

<u>Fully Diluted Earnings Per Common Share</u>: Diluted earnings per Common Share are based upon the weighted average number of Common Shares including both outstanding and shares potentially issued in connection with stock options and restricted common shares granted under Registrant's 2000 Stock Incentive Plan, and net income after deducting preferred dividend requirements. At December 31, 2002, 2001 and 2000, there were 237,610, 135,970 and 67,985 options outstanding, respectively. There were also 1,050 restricted shares issued in 2002.

50

Supply Cost Balancing Accounts: As permitted by the CPUC prior to November 29, 2001, Registrant maintained water and electric supply cost balancing accounts for SCW to account for under-collections and over-collections of revenues designed to recover such costs. Costs were recorded in operating expenses and charged to balancing accounts when such costs were incurred. The balancing accounts were reversed when such costs were recovered through rate adjustments. Registrant accrued interest on its supply cost balancing accounts at the rate prevailing for 90-day commercial paper. On November 29, 2001, the CPUC issued a resolution ordering that water utilities with existing balancing accounts cease booking amounts into these accounts. In its place, the CPUC required water utilities to start a new memorandum account that would work in a manner similar to the balancing account to track the under-collection or over-collection of supply costs for its water utility service areas. In accordance with this resolution, all supply costs will be accounted for and included in the income statement, just as is the case currently. However, the income statements will not include entries reflecting the differences between actual unit supply costs included in rates and those accounts. Recovery of deferred supply costs recorded in the memorandum accounts could be impacted if SCW is earning more than its authorized rate of return on a weather normalized means test basis. On December 17, 2002, the CPUC issued an order authorizing that utilities including SCW should request recovery of the pre-November 29, 2001 balancing accounts within 90 days. SCW filed advice letters on March 17, 2003. Registrant does not maintain a supply cost balancing account for CCWC.

Electric power costs up to \$77 per megawatt hour incurred by SCW's Bear Valley Electric (BVE) division will continue to be charged to an electric supply cost balancing account for recovery at a later date in accordance with the pre-November 29, 2001 procedure applicable to both its water and electric supply balancing accounts.

The amounts included in the supply cost balancing accounts that will be collected over a period exceeding one year have been reclassified as a long-term asset as of December 31, 2002 and 2001.

Debt Issue Expense and Redemption Premiums: Original debt issue expenses are capitalized and amortized over the lives of the respective issues. Premiums paid on the early redemption of debt, which is reacquired through refunding, are deferred and amortized over the life of the debt issued to finance the refunding.

Other Credits: For SCW, advances for construction represent amounts advanced by developers, which are generally refundable at rates ranging from 10% to 22% of the revenue received from the installations for which funds were advanced or in equal annual installments over periods of time ranging from 10 to 40-year periods. Contributions-in-aid of construction are similar to advances, but require no refunding and are amortized over the useful lives of the related property.

For CCWC, advances for construction represent amounts advanced by developers which are refundable over 10 to 20 years. Refund amounts under the contracts are based on annual revenues from the extensions.

Cash and Cash Equivalents: For purposes of the Statements of Cash Flows, cash and cash equivalents include short-term cash investments with an original maturity of three months or less. At times, cash and cash equivalent balances may be in excess of federally insured limits. The Company's cash and cash equivalents are held with financial institutions with high credit standings.

Derivative Instrument: Registrant has certain block-forward purchase power contracts that qualify as derivative instruments under SFAS No. 133, "Accounting for Derivative Instruments and Hedging Activities", as amended by SFAS No. 138. A derivative financial instrument or other contract derives its value from another investment or designated benchmark. SFAS No. 133 requires companies to record derivatives on the balance sheet as assets and liabilities, and to measure those instruments at their fair value. This Statement, as amended, became effective for Registrant's financial statements at the beginning of fiscal year 2001. This Statement did not have any impact on Registrant's consolidated financial statements at the time of adoption.

During 2002, SCW became a party to block-forward purchase power contracts that qualify as derivative instruments under SFAS No. 133 when it buys electricity that is scheduled for delivery in future periods. Certain of these contracts qualify as an exception provided under the Statement for activities that are considered normal purchases and normal sales. These contracts are reflected in the statements of income at the time of contract settlement. Contracts with Pinnacle West Capital Corporation (PWCC) which became effective in November 2002 do not qualify for the normal purchases and normal sales exception and, as a result, have been recognized

at fair market value on the balance sheet as of December 31, 2002. This resulted in a pre-tax unrealized loss of \$2.5 million in 2002 recorded as power purchased for resale. On a quarterly basis, the related asset or liability will be adjusted to reflect the fair market value at the end of the quarter. As this contract is settled, the realized gains or losses will be recorded and the unrealized gains or losses will be reversed. The market prices used to determine the fair value for this derivative instrument was estimated based on independent sources such as broker quotes and publications.

Under the terms of its power purchase contracts with Mirant Americas Energy Marketing, LP (Mirant Marketing) and PWCC, SCW is required to post security, at the request of the seller, if SCW is in default under the terms of the contract and the future value of the contract is greater than the future value of contracts of a similar term on the date of default. SCW will be in default under the terms of these contracts if its debt is rated less than BBB- by Standard & Poor's Ratings Service ("S&P") or Fitch, Inc. ("Fitch") or less than Baa3 by Moody's Investor Services, Inc ("Moody's"). SCW currently has a rating of A+ by S & P and A2 by Moody's. Fitch does not rate SCW.

Except as discussed above, Registrant has no other derivative financial instruments, financial instruments with off-balance sheet risk or financial instruments with concentrations of credit risk.

Fair Value of Financial Instruments: For cash and cash equivalents, accounts receivable, accounts payable and short-term debt, the carrying amount is used as approximates fair value due to the short-term nature of the amounts. The table below estimates the fair value of long-term debt held by Registrant. Rates available to Registrant at December 31, 2002 and 2001 for debt with similar terms and remaining maturities were used to estimate fair value for long-term debt. Changes in the assumptions will produce differing results.

		2002			2001	
(dollars in thousands)	Carry	ing amount	Fair value	Carry	ing amount	Fair value
Financial liabilities:						
Long-term debt	\$	244,394	\$ 282,507	\$	246,452	\$ 260,274
	_			_		

Stock Options: Registrant has a Stock Incentive Plan, which is described more fully in Note 10. Registrant applies Accounting Principles Board Opinion (APB) No. 25, "Accounting for Stock Issued to Employees", in accounting for its stock options. Accordingly, no compensation cost for the Plan has been recognized for options granted at fair value at the date of grant. Registrant has also adopted the disclosure only requirements of SFAS No. 123, "Accounting for Stock-Based Compensation". If Registrant had elected to adopt the optional recognition provisions of SFAS No. 123 for its stock options, net income and earnings per share applicable to common shareholders would have been changed to the proforma amounts indicted below.

dollars in thousands except EPS)	2002	2001	2000
		Restated Note 2	Restated Note 2
arnings available for common:			
As reported	\$ 20,310	\$ 21,372	\$ 19,945
Add: Stock-based compensation expense included in reported net income, net of tax	_	_	
Less: Stock-based compensation expense determined under the fair-value accounting method, net of tax	(258)	(139)	(134)

51

	\$ 2	0,052 3	\$ 21,233	\$ 19	Э,811
					—
asic earnings per share:					
As reported	\$	1.34 \$	5 1.41	\$	1.42
Pro forma	\$	1.32 \$	5 1.40	\$	1.41
Diluted earnings per share:					
As reported	\$	1.34 \$	5 1.41	\$	1.42
Pro forma	\$	1.32 \$	5 1.40	\$	1.41

52

New Accounting Pronouncements: The following are recent accounting pronouncements that will be effective for Registrant's years ending after December 31, 2002:

- SFAS No. 143, "Accounting for Asset Retirement Obligations": This Statement requires businesses to record the fair value of a liability for a legal obligation to retire an asset in the period in which the liability is incurred. A legal obligation is a liability that a party is required to settle as a result of an existing or enacted law, statute, ordinance or contract. When the liability is initially recorded, the entity capitalizes the cost by increasing the carrying amount of the related long-lived asset. Over time, the liability is accreted to its present value each period, and the capitalized cost is depreciated over the useful life of the related asset. Upon settlement of the liability, an entity either settles the obligation for its recorded amount or incurs a gain or loss upon settlement.

Registrant's legal obligations for retirement reflect principally the retirement of wells, which by law need to be disposed of at removal. As such, the regulated subsidiaries of Registrant incur asset retirement obligations. Retirement cost has historically been recovered through rates at the time of retirement. At implementation of SFAS No. 143, the cumulative effect will be reflected as a regulatory asset or liability. Registrant will also reflect the gain or loss at settlement as a regulatory asset or liability on the balance sheet.

Registrant adopted SFAS No. 143 on January 1, 2003, as required. Upon adoption of SFAS No. 143 on January 1, 2003, Registrant recorded an asset retirement obligation of \$13.2 million at its net present value of \$2.7 million, increased depreciable assets by \$0.4 million for asset retirement costs, increased regulatory assets by \$2.5 million and increased accumulated depreciation by \$0.2 million. Amounts recorded under SFAS 143 are subject to various assumptions and determinations, such as determining whether a legal obligation exists to remove assets, and estimating the fair value of the costs of removal, when final removal will occur and the credit-adjusted risk-free interest rates to be utilized on discounting future liabilities. Changes that may arise over time with regard to these assumptions will change amounts recorded in the future.

- SFAS No. 145, "Rescission of SFAS Nos. 4, 44, and 64, Amendment of SFAS No. 13, and Technical Corrections": Among other things, SFAS No. 145 rescinds various pronouncements regarding early extinguishment of debt and allows extraordinary accounting treatment for early extinguishment only when the provisions of APB No. 30 are met. This Statement also amends SFAS No. 13 to require sale-leaseback accounting for certain lease modifications that have economic effects that are similar to sale-leaseback transactions. SFAS No. 145 provisions are generally effective for fiscal years beginning after May 15, 2002. At this time, Registrant does not expect the adoption of SFAS No. 145 to have a material impact on the consolidated financial statements.

- SFAS No. 146, "Accounting for Costs Associated with Exit or Disposal Activities": This Statement addresses significant issues regarding the recognition, measurement, and reporting of costs associated with exit or disposal activities, and nullifies Emerging Issues Task Force Issue No. 94-3, "Liability Recognition for Certain Employee Termination Benefits and Other Costs to Exit an Activity (including Certain Costs Incurred in a Restructuring)." Costs addressed by SFAS No. 146 include costs to terminate a contract that is not a capital lease, costs of involuntary employee termination benefits pursuant to a one-time benefit arrangement, costs to consolidate facilities, and costs to relocate employees. This Statement will be effective for exit or disposal activities that are initiated after December 31, 2002. At this time, Registrant does not expect the adoption of SFAS No. 146 to have a material impact on the consolidated financial statements. However, unknown at this time, the timing of expense recognition in Registrant's financial statements for future restructuring, exit or disposal activities could differ significantly.

- SFAS No. 148, "Accounting for Stock-Based Compensation-Transition and Disclosure" - This Statement amends SFAS No. 123, "Accounting for Stock-Based Compensation," to provide alternative methods of transition for a voluntary change to the fair value based method of accounting for stock-based employee compensation. In addition, SFAS No. 148 amends the disclosure requirements of SFAS No. 123 to require prominent disclosures in both annual and interim financial statements about the method of accounting for stock-based employee compensation and the effect of the method used on reported results. Although Registrant is required to comply with the requirements of SFAS 148 beginning January 1, 2003, the Company has elected to continue to apply the recognition and measurement provisions of APB 25. Therefore, SFAS No. 148 is not expected to have a significant impact on the Registrant's consolidated financial statements. The amendments to SFAS No. 123 regarding disclosure are effective for financial statements for fiscal years ending after December

53

# 15, 2002. Registrant adopted the disclosure only requirements of SFAS No. 148, which are included in Note 1 above.

- FASB Interpretation No. 45: "Guarantor's Accounting and Disclosure Requirements for Guarantees, Including Indirect Guarantees of Indebtedness of Others". This interpretation elaborates on the disclosures to be made by a guarantor in its interim and annual financial statements about its obligations under certain guarantees that it has issued and requires that they be recorded at fair value. The initial recognition and measurement provisions of this interpretation are to be applied only on a prospective basis to guarantees issued or modified after December 31, 2002. The disclosure requirements of this interpretation are effective for financial statements of interim or annual periods ending after December 15, 2002. See Business Risks and Commitments in Note 11. Registrant does not expect the adoption to have a material impact on its consolidated financial statements.

- FASB Interpretation No. (FIN) 46, "Consolidation of Variable Interest Entities" – This interpretation expands upon existing guidance that addresses when a company should include in its financial statements the assets and liabilities of another entity. The primary objectives of FIN 46 are to provide guidance on the identification of entities for which control is achieved through means other than through voting rights ("variable interest entities") and to determine when and which business enterprise should consolidate the variable interest entity (the "primary beneficiary"). FIN 46 requires that both the primary beneficiary and all other enterprises with a significant variable interest make additional disclosures. The transitional disclosure requirements of FIN 46 are effective immediately. The effective date of the consolidation requirements of FIN 46 depends on the date the variable interest entity was created. FIN 46 is effective for all variable interest entities created after January 31, 2003. The Company is currently in the process of evaluating the impact of FIN 46 on Registrant's consolidated financial statements, but does not expect it to have a material impact.

# Note 2 - Restatement of Previously Issued Financial Statements

In connection with a deferred tax analysis of Registrant's tax accounts performed in 2002, Registrant determined that it had miscalculated its net accumulated deferred income taxes over several years, resulting in taxes on income not being recognized in the proper period, and overstating Registrant's net deferred tax liabilities and current taxes payable. Registrant further determined that the amount of the overstatements, which totaled approximately \$4.7 million and \$4.9 million for AWR and SCW, respectively, arose in prior years and that prior years' financial statements should be restated to reduce net accumulated deferred income taxes, taxes payable and the related tax expense to reflect amounts currently due and deferred. The financial information for all periods included in these consolidated financial statements gives effect to the restatement.

The following tables summarize the effects of the restatement on Registrant's financial statements for 2001 and 2000:

		December 31, 2001				
		AW	R	SC	w	
usands)	Previo Repor		Restated	Previously Reported	Restated	
s:						
tax-related assets	\$ 15	,843	\$ 13,908	\$ 15,843	\$ 13,908	
	(5	,389)	(404)	(5,599)	(1,194)	
ferred income taxes - net	(53	,444)	(51,541)	(52,075)	(49,399)	

Regulatory tax-related liability	(1,773)	(2,054)	(1,773)	(2,054)
Earnings reinvested in the business	(74,544)	(79,216)	(72,716)	(77,581)

54

		Year Ended December 31, 2001			
	AV	'R	sc	W	
rs in thousands except EPS)	Previously Reported	Restated	Previously Reported	Restated	
ements of Income:					
es on income	\$ 15,379	\$14,370	\$ 15,066	\$ 14,057	
perating income	36,692	37,701	35,030	36,039	
income	20,447	21,456	19,829	20,838	
ic earnings per share	\$ 1.35	\$ 1.41	\$ 185,318	\$ 194,748	
uted earnings per share	\$ 1.33	\$ 1.41	N/A	N/A	

		Year Ended D	ecember 31, 2000	)
	AV	AWR		w
dollars in thousands except EPS)	Previously Reported	Restated	Previously Reported	Restated
tatements of Income:				
Taxes on income	\$ 15,127	\$ 13,182	\$ 14,881	\$ 12,743
Operating income	32,307	34,252	32,175	34,313
Net income	18,086	20,031	17,685	19,823
Basic earnings per share	\$ 1.28	\$ 1.42	\$ 176,850	\$ 198,230
Diluted earnings per share	\$ 1.28	\$ 1.42	N/A	N/A

The effect of the restatement did not have an impact on cash flows from operating, investing or financing activities for the years ended December 31, 2001 and 2000.

# <u>Note 3 – Utility Plant</u>

The following table shows Registrant's consolidated utility plant by major class:

	As of Dec	ember 31,
(in thousands)	2002	2001
Water		
Land	\$ 8,181	\$ 7,689
Source of water supply	36,746	31,471
Pumping	83,299	75,739
Water treatment	32,079	27,316
Transmission and distribution	460,512	431,858
General	73,132	71,112
	693,949	645,185
Electric		
Transmission and distribution	36,846	34,566
General	4,171	3,959
	41,017	38,525
Less accumulated depreciation	(206,873)	(190,656)
Construction work in progress	35,218	46,788
Net utility plant	\$ 563,311	\$ 539,842

# Note 4 - Capital Stock

All Preferred Shares, including the 4%, 4¼% and 5% Series, were fully redeemed at the option of AWR in April 2002. At December 31, 2002, there are no Preferred Shares outstanding. The redemption price per share for each series of \$25 Preferred Shares was \$27.00, \$26.50 and \$25.25 for the 4%, 4¼% and 5% Series, respectively, plus accrued and unpaid dividends to the redemption date. The table below summarizes the activity of the preferred shares during the three years ended December 31, 2002:

(in thousands)		Preferred Shares Series 4%		Preferred Shares Series 4¼ %		Mandatorily Redeemable Preferred Shares Series 5%	
	Shares	Amount	Shares	Amount	Shares	Amount	

Balance at December 31, 1999	32,000	\$ 800	32,000	\$ 800	16,000	\$ 400
Redeemed at \$25 per share			_		(1,600)	(40)
Balance at December 31, 2000	32,000	\$ 800	32,000	\$ 800	14,400	\$ 360
Redeemed at \$25 per share	_				(1,600)	(40)
Balance at December 31, 2001	32,000	\$ 800	32,000	\$ 800	12,800	\$ 320
Fully redeemed	(32,000)	(800)	(32,000)	(800)	(12,800)	(320)
Balance at December 31, 2002		\$ —		\$ _	_	\$ —

On June 7, 2002, Registrant completed a 3-for-2 stock split of Common Shares for the holders of record on May 15, 2002. Earnings per share and number of shares have been retroactively restated for all periods presented to reflect the stock split.

AWR has a Registration Statement on file with the Securities and Exchange Commission for issuance, from time to time, of up to \$60 million in aggregate in Common Shares, Preferred Shares and/or debt securities. On August 16, 2000, AWR issued 1,107,000 shares under this Registration Statement. Net proceeds from this sale were used to fund a portion of the purchase price of CCWC and were invested in SCW. As of December 31, 2002, approximately \$31,080,000 remained for issuance under this Registration Statement.

During the year ended December 31, 2002, Registrant issued 30,180 and 31,211 Common Shares under the Registrant's Common Share Purchase and Dividend Reinvestment Plan (DRP) and 401(k) Plan, respectively. There are 709,379 and 63,906 Common Shares authorized but unissued under the DRP and the 401(k) Plan, respectively, at December 31, 2002. Shares reserved for the 401(k) Plan are in relation to Company matching contributions and for investment purposes by participants. During the year ended December 31, 2001, all shares issued under Registrant's DRP and the 401(k) Plan were purchased on the open market. During the year ended December 31, 2000, Registrant issued 10,442 and 11,996 Common Shares under the DRP and 401(k) Plan, respectively.

In 1998, the Board of Directors adopted a Shareholder Rights Plan (Rights Plan) and authorized a dividend distribution of one right (a Right) to purchase 1/1000<sup>th</sup> of Junior Participating Preferred Share for each outstanding Common Share. The Rights Plan became effective in September 1998 and will expire in September 2008. The Rights Plan is designed to provide shareholders' protection and to maximize shareholder value by encouraging a prospective acquirer to negotiate with the Board of Directors.

Each Right represents a right to purchase 1/1000<sup>th</sup> of Junior Participating Preferred Share at the price of \$120, subject to adjustment (the Purchase Price). Each Junior Participating Preferred Share is entitled to receive a dividend equal to 1000 times any dividend paid on each Common Share and 100 votes per share in any shareholder election. The Rights become exercisable upon occurrence of a Distribution Date event. A Distribution Date event occurs if (i) any person accumulates 15% of the then outstanding Common Shares, (ii) any person presents a tender offer which caused the person's ownership level to exceed 15% and the board determines the tender offer not to be fair to AWR's shareholders, or (iii) the board determines that a shareholder

55

maintaining a 15% interest in the Common Shares could have an adverse impact on AWR or could attempt to pressure AWR to repurchase the holder's shares at a premium.

Until the occurrence of a Distribution Date, each Right trades with the Common Share and is not separately transferable. When a Distribution Date occurs, AWR would distribute separately Rights Certificates to Common Shareholders. The Rights would subsequently trade separately from the Common Shares and each holder of a Right, other than the acquiring person whose Rights will thereafter be void, will have the right to receive upon exercise at its then current Purchase Price that number of Common Shares having a market value of two times the Purchase Price of the Right. If AWR merges into the acquiring person or enters into any transaction that unfairly favors the acquiring person or disfavors AWR's other shareholders, the Right becomes a right to purchase Common Shares of the acquiring person having a market value of two times the Purchase Price.

The Board of Directors may determine that, in certain circumstances, a proposal, which would cause a distribution of the Rights, is in the best interest of AWR's shareholders. Therefore, the board of directors may, at its option, redeem the Rights at a redemption price of \$0.01 per Right.

### Note 5 - Dividend Limitations

On January 28, 2003, AWR declared a regular quarterly dividend of \$0.221 per common share. The dividend, totaling approximately \$3.4 million, was paid on March 1, 2003 to common shareholders of record at the close of business on February 7, 2003. In 2002, 2001 and 2000, AWR paid quarterly dividends to the shareholders, totaling approximately \$13.2 million or \$0.872 per share, \$13.1 million or \$0.867 per share, and \$12.2 million or \$0.857 per share, respectively. AWR's ability to pay cash dividends on common stock outstanding depends primarily upon cash flows from its SCW subsidiary.

SCW is subject to contractual restrictions on its ability to pay dividends. SCW's maximum ability to pay dividends is restricted by certain Note Agreements to the sum of \$21 million plus 100% of consolidated net income plus the aggregate net cash proceeds received from capital stock offerings or other instruments convertible into capital stock from various dates. Under the most restrictive of the Note Agreements, \$154.7 million was available to pay dividends to AWR. Dividends in the amount of \$14,630,000, \$13,530,000, and \$12,900,000, were paid to AWR by SCW in 2002, 2001 and 2000, respectively.

The ability of AWR, ASUS and SCW to pay dividends is also restricted by California law. Under restrictions of the California tests, approximately \$86.3 million of retained earnings for AWR was available to pay dividends to Common Shareholders at December 31, 2002. Approximately \$84.2 million was available from the retained earnings of SCW to pay dividends to AWR. At December 31, 2002, ASUS was not allowed to pay dividends to AWR under the California tests.

CCWC is subject to contractual restrictions on its ability to pay dividends. CCWC's maximum ability to distribute dividends is limited to maintenance of no more than 55% debt in the capital structure for the quarter immediately preceding the distribution. The ability of CCWC to pay dividends is also restricted by Arizona law. Under restrictions of the Arizona tests, approximately \$4 million was available to pay dividends to AWR at December 31, 2002. Dividends in an amount of \$1,003,000 were paid to AWR in 2001 by CCWC. There were no dividends distributed from CCWC to AWR in 2002 or 2000.

## Note 6 – Bank Debt

AWR paid off a \$25 million non-revolving credit facility on January 2, 2002. In June 2002, AWR established a new \$75 million syndicated credit facility. Under the terms of this facility, either AWR or SCW may obtain letters of credit for up to \$15 million. At December 31, 2002, \$35 million in cash borrowing was outstanding under this facility, \$21 million of which was used to fund AWR's activities, \$13 million was used to fund SCW's operations and \$1.0 was used to fund ASUS activities. Additionally, SCW has obtained letters of credit, which expires May 31, 2003 in the amount of \$966,534 with an annual fee of 0.625% as security for its self-insured workers' compensation plan, (2) a letter of credit with a fee of 0.625%, which expires for the amount of \$6,296,000 to a trustee with respect to the variable rate obligation issued by the Three Valleys Municipal Water District, (3) a letter of credit with a nanual fee of 0.625%, which expires in October 1, 2003, in the amount of \$400,000 as security for the deductible in the Company's business automobile insurance policy, and (4) a letter of credit with a fee of 0.625%, which expires March 31, 2005 in an amount, which varies from \$270,000 to \$585,000 during

the year as security for the purchase of power. There were no compensating balances required. Loans can be obtained at the option of Registrant and bear interest at rates based on floating prime borrowing rates or at money market rates.

Registrant's short-term borrowing activities (excluding Letters of Credit) for the last three years were as follows:

Balance Outstanding at December 31,	\$ 35,000 \$ 20,000 \$ 45,000
Interest Rate at December 31,	2.16% 2.82% 7.19%
Average Amount Outstanding	\$ 6,811 \$ 34,748 \$ 38,531
Weighted Average Annual Interest Rate	2.49% 4.65% 7.11%
Maximum Amount Outstanding	\$ 35,000 \$ 50,000 \$ 50,000

There were no short-term borrowing activities at ASUS or CCWC.

# Note 7 - Long-Term Debt

Registrant's long-term debt consists primarily of Notes and Debentures. The Company summarizes its long-term debt in the Statements of Capitalization. No long-term debt was issued in 2002. In January 2001, \$20 million of Series C Medium Term Notes were sold with net proceeds from the issuance initially used to repay short-term bank borrowings and, after that, to fund construction expenditures. In 2001 SCW filed a Registration Statement for issuance from time to time of up to \$100 million in debt securities. SCW issued \$50 million in long-term debt in December 2001. The net proceeds were used to repay short-term bank borrowings, with the remaining proceeds used to refund certain existing long-term debt obligations, and for payments for construction, completion, extension or improvement of facilities. SCW has no mortgage debt, and leases and other similar financial arrangements are not material. Redemption of certain of the long-term debt issues outstanding as of December 31, 2002 and 2001 can be made in whole or in part at the option of SCW subject to redemption schedules imbedded in the agreements particular to each issue. With the exception of the 9.56% Notes, the redemption premiums in effect for 2003 range up to 7% of par value. The 9.56% Notes are subject to the make-whole premium based on 55 basis points above the applicable Treasury Yield if redeemed prior to 2021. After 2021, the maximum redemption premium is 3% of par value.

CCWC has long-term Industrial Development Authority Bonds (IDA Bonds) and a repayment contract due 2006. Substantially all of the utility plant of CCWC is pledged as collateral for its IDA Bonds. The Bond Agreement, among other things, (i) requires CCWC to maintain certain financial ratios, (ii) restricts CCWC's ability to incur additional debt, make liens, sell, lease or dispose of assets, merge with another corporation, and pay dividends, and (iii) requires CCWC to establish a debt service reserve fund held in trust for future payments, which totaled \$722,850 and \$720,463 as of December 31, 2002 and 2001, respectively.

Annual maturities of all long-term debt, including capitalized leases, amount to approximately \$13,305,000, \$851,000, \$905,000, \$664,000, \$627,000 and \$228,042,000 for the five years ending December 31, 2003 through 2006 and thereafter, respectively.

# Note 8 - Taxes on Income

Registrant provides deferred income taxes for temporary differences under SFAS No. 109, "Accounting for Income Taxes", for certain transactions which are recognized for income tax purposes in a period different from that in which they are reported in the financial statements. The most significant items are the tax effects of accelerated depreciation, the supply cost balancing accounts and advances for and contributions-in-aid-of-construction. SFAS No. 109 also requires that rate-regulated enterprises record deferred income taxes for temporary differences accorded flowthrough treatment at the direction of a regulatory commission. The resulting deferred tax assets and liabilities are recorded at the expected cash flow to be reflected in future rates. Since the CPUC has consistently permitted the recovery of previously flowed-through tax effects, SCW has established regulatory liabilities and assets offsetting such deferred tax assets and liabilities.

57

Deferred investment tax credits are being amortized to other income ratably over the lives of the property giving rise to the credits.

As discussed in Note 2, during 2002 it was determined that the consolidated accumulated deferred income taxes and current taxes payable had been overstated cumulatively by approximately \$4.7 million as of December 31, 2001. The Company further determined that the amount of the overstatement arose in prior years and that prior years' financial statements should be restated to reduce accumulated deferred income taxes, taxes payable and the related tax expense to reflect amounts currently due and deferred. The disclosures below reflect the restated amounts for 2001 and 2000.

The significant components of deferred tax assets and deferred tax liabilities, as reflected in the balance sheets, and the accumulated net deferred income tax liabilities at December 31, 2002 and 2001 were:

		ember 3	mber 31,	
dollars in thousands)	2002		2001 ited Note 2	
Deferred tax assets:				
ITC & Excess Deferred Taxes	\$ 2,054	\$	2,054	
Other	8,674		9,104	
	\$ 10,728	\$	11,158	
eferred tax liabilities				
Fixed assets	\$ (40,516)	\$	(35,782	
Regulatory tax liability – flow through	(12,828)		(13,899	
Other property related	3,342		(1,423	
Other non-property related	(2,864)		(1,072	
Balancing accounts	(11,679)		(10,523)	
	(64,545)		(62,699	
Accumulated deferred income taxes – net	\$ (53,817)	\$	(51,541	

The current and deferred components of income tax expense are as follows:

	Year	Ended Decemb	oer 31,
(dollars in thousands)	2002	2001 Restated Note 2	2000 Restated Note 2
Current			
Federal	\$ 6,709	\$ 4,569	\$ 6,720
State	1,769	1,387	1,886
Total current tax expense	\$ 8,478	\$ 5,956	\$ 8,606

Deferred – Federal and State:			
Accelerated depreciation	4,509	3,335	3,238
Balancing accounts	1,156	6,436	2,863
California privilege year franchise tax	(1,128)	(1,015)	(1,216)
Other	207	(657)	(392)
			<u> </u>
Total deferred tax expense	\$ 4,744	\$ 8,099	\$ 4,493
Total income tax expense	\$ 13,222	\$ 14,055	\$ 13,099
Income taxes included in operating expenses	12,949	14,370	13,182
Income taxes included in other income and expenses – net	273	(315)	(83)
			<u> </u>
Total income tax expense	\$ 13,222	\$ 14,055	\$ 13,099

The federal statutory income tax rate is reconciled to the effective tax rate in the following table:

Year E			er 31,
(dollars in thousands, except percent)	2002	2001 Restated Note 2	2000 Restated Note 2
Federal taxes on pre-tax income at statutory rates	\$ 11,746	\$ 12,426	\$ 11,595
Increase (decrease) in taxes resulting from:			
State income tax, net of federal benefit	1,859	1,885	1,769
Flow through on fixed assets	169	166	310
Other – net	(552)	(422)	(575)
Total income tax expense	\$ 13,222	\$ 14,055	\$ 13,099
Pre-tax income	\$ 33,561	\$ 35,511	\$ 33,130
Effective income tax rate	39.4%	6 39.6%	39.5%

# Note 9 - Employee Benefit Plans

Registrant maintains a pension plan (the "Plan") that provides eligible employees (those aged 21 and older, with one year of service) monthly benefits upon retirement based on average salaries and length of service. The normal retirement benefit is equal to 2% of the five highest consecutive years average earnings multiplied by the number of years of credited service, up to a maximum of 40 years, reduced by a percentage of primary social security benefits. There is also an early retirement option. Annual contributions are made to the Plan, which comply with the funding requirements of the Employee Retirement Income Security Act (ERISA).

Registrant also provides all active employees medical, dental and vision care benefits through a medical insurance plan. Eligible employees, who retired prior to age 65, and/or their spouses, were able to retain the benefits under the active plan until reaching age 65. Eligible employees upon reaching age 65, and those employees retiring at or after age 65, and/or their spouses, receive coverage through a Medicare supplement insurance policy paid for by Registrant subject to an annual cap limit.

The CPUC has issued a decision, which provides for the recovery in rates of tax-deductible contributions made to a separate trust fund. In accordance with that decision, Registrant established two separate trusts in 1995, one for those retirees who were subject to a collective bargaining agreement and another for all other retirees. Registrant's funding policy is to contribute annually an amount at least equal to the revenues authorized to be collected through rates for post-retirement benefit costs. Post-retirement benefit costs for 1993, 1994 and 1995 were estimated at a total of \$1.6 million and have been recorded as a regulatory asset for recovery over a 20-year period. The unamortized balance at December 31, 2002 was \$386,676.

At December 31, 2002, Registrant had 765 participants in the Plan, 67 of these are employees covered by collective bargaining agreements. One collective bargaining agreement expires in 2004 and one should have expired in 2002, but has been extended by mutual consent of the parties involved until a new contract is ratified. The following table sets forth the Plan's funded status and amounts recognized in Registrant's balance sheets and the components of net pension cost and accrued post-retirement liability at December 31, 2002 and 2001:

	Pensio	Pension Benefits		Other Postretire Benefits	
(dollars in thousands)	2002	2001	2002	_	2001
Change in Benefit Obligation:					
Benefit Obligation at beginning of year	\$ 43,830	\$ 40,521	\$ 4,939	9 \$	4,585
Service Cost	2,091	1,868	294	1	115
Interest Cost	3,061	2,877	507	7	335
Actuarial Loss/(Gain)	1,797	(81)	2,703	3	159
Plan Amendment		105	_	-	_
Benefits Paid	(1,706	) (1,460)	(283	3)	(255)
Benefit Obligation at end of year	\$ 49,073	\$ 43,830	\$ 8,160	) \$	4,939
Changes in Plan Assets:					
Fair Value of Plan Assets at beginning of year	\$ 42,004	\$ 45,105	\$ 2,192	2 \$	2,058
Actual Return of Plan Assets	(2,058	) (1,641)	32	2	(78)

Employer Contributions	1,518	_	741	468
Benefits Paid	(1,706)	(1,460)	(283)	(256)
Fair Value of Plan Assets at end of year	\$ 39,758	\$ 42,004	\$ 2,682	\$ 2,192
Reconciliation of Funded Status:				
Funded Status	(\$9,315)	(\$1,826)	(\$5,478)	(\$2,747)
Unrecognized Transition Obligation	_	_	5,030	5,449
Unrecognized Net Loss/(Gain)	9,358	2,137	1,627	(1,218)
Unrecognized Prior Service Cost	320	372	(2,629)	(2,829)
Prepaid/(Accrued) Pension Cost	\$ 363	\$ 683	(\$1,450)	(\$1,345)
Weighted-Average Assumptions as of December 31:				
Discount Rate	6.75%	7.25%	6.75%	7.25%
Long-term Rate of Return	7.00%	8.00%	*	8.00%
Salary Assumption	4.00%	4.00%	NA	NA

\* 7.0% for union plan, 4.2% for non-union, net of income taxes

A sliding scale for assumed health care cost increases was used for both periods. 2001 started at 8.5% then graded down ½ percentage point each year to the ultimate rate of 5% after 6 years. In 2002, health care cost increases started at 12.0% grading down to 5.0% in 10 years for those under age 65, and at 14.0% grading down to 4.5% in 10 years for post 65.

The components of net periodic post-retirement benefits cost for 2002, 2001, 2000 are as follows:

			Postretirement Benefits			
(dollars in thousands)	2002	2001	2000	2002	2001	2000
Components of Net Periodic Benefits Cost:						
Service Cost	\$ 2,091	\$ 1,868	\$ 1,530	\$ 294	\$ 115	\$103
Interest Cost	3,061	2,877	2,649	507	335	313
Expected Return on Plan Assets	(3,365)	(3,548)	1,336	(32)	78	(70)
Amortization of Prior Service Cost	52	44	(5,448)	78	(107)	23
Net Periodic Pension Cost	\$ 1,839	\$ 1,241	\$ 67	\$ 847	\$ 421	\$369

60

Assumed health care cost trend rates have a significant effect on the amounts reported for the health care plans. A one-percentage-point change in assumed health care cost trend rates would have the following effects:

(dollars in thousands)	centage- Increase	1-Percentage- Point Decrease	
Effect on Total of Service and Interest Cost Components	\$ 50	(\$60)	
Effect on Postretirement Benefit Obligation	\$ 531	(\$606)	

Registrant has a 401(k) Investment Incentive Program under which employees may invest a percentage of their pay, up to a maximum investment prescribed by law, in an investment program managed by an outside investment manager. Company contributions to the 401(k) are based upon a percentage of individual employee contributions and, for 2002, 2001 and 2000, totaled \$975,687, \$953,938, and \$968,019, respectively.

# Note 10 - Stock Compensation Plan

Registrant established a 2000 Stock Incentive Plan (the "2000 Plan") adopted at the Shareholders' Annual Meeting in 2000 to provide stock-based incentives as a means of promoting the success of the Company by attracting, retaining and aligning the interests of employees with those of shareholders generally. There are 375,000 Common Shares reserved for issuance under the 2000 Plan. An additional 375,000 Common Shares have been reserved for issuance, subject to shareholder approval.

The fair value of stock options used to compute pro forma net income and earnings per share disclosures discussed in Note 1, is the estimated fair value at grant date using the Black-Scholes option-pricing model with the following assumptions:

December 31,	2002	2001	2000
Weighted-average fair value of option granted	\$ 3.83	\$ 3.49	\$ 3.56
Risk-free rate of return	4.30%	4.86%	6.69%
Dividend yield	6.01%	6.94%	6.74%
Expected volatility	28.14%	27.84%	26.54%
Expected life	5	5	5

Stock option transactions relating to the 2000 Stock Incentive Plan are summarized below:

	2002 2001		1	2000		
	Option Shares	Weighted Price	Option Shares	Weighted Price	Option Shares	Weighted Price
Options outstanding at beginning of year	135,970	\$ 22.03	67,985	\$ 20.83	—	\$ —
Granted	116,775	23.43	68,985	23.21	68,985	20.83
Exercised	—	_	_	—	—	—
Cancelled	(15,135)	23.03	(1,000)	22.02	(1,000)	20.83
Options outstanding at end of year	237,610	\$ 22.66	135,970	\$ 22.03	67,985	\$ 20.83

One-third of the stock options granted become exercisable on each of the first three anniversaries of the grant date, but may be exercised earlier if there is a change in control of the Company. Weighted average remaining contractual life at December 31, 2002 was 8.31 years. Exercise prices of options outstanding at December 31, 2002 ranged from \$20.83 to \$23.43 per share.

## Note 11 - Business Risks and Commitments

Registrant's utility operations are engaged in supplying water and electric service to the public. Registrant is required to provide service and grant credit to customers within its defined service areas. Although Registrant has a diversified base of residential, industrial and other customers, revenues derived from commercial and

61

residential water customers accounted for approximately 91% of total water revenues in 2002, which is about the same percentage as in 2001. Registrant faces additional risks associated with weather conditions, adequacy and quality of water supplies, regulatory decisions, pronouncements and laws, water-related litigation, and general business conditions.

Approximately 41.3% of SCW's water supply is purchased from wholesalers of imported water, with the remainder produced from Company wells. The long-term availability of imported water supplies is dependent upon, among other things, drought conditions throughout the state of California, increases in population, water quality standards, legislation that may potentially reduce water supplies and issues related to California's allocation of Colorado River water. Reservoir storage in California statewide stood at 84% of normal in February of 2003. California missed a federal deadline of December 31, 2002 to show how it will reduce the Colorado River water that it takes beyond its entitlement after negotiations broke down on how to divide the water in Southern California's Colorado River supplies in January 2003.

CCWC also obtains its water from the Colorado River through a long-term water supply contract with the Central Arizona Water Conservation District. CCWC is entitled to take 6,978 acre feet of water per year from the Central Arizona Project (CAP) pursuant to the terms of this contract through September 2043. CCWC's water supply may be subject to interruption or reduction, in particular owing to interruption or reduction of CAP water. In the event of interruption or reduction of CAP water, CCWC can rely on its well water supplies for short-term periods. However, in any event, the quantity of water CCWC supplies to some or all of its customers may be interrupted or curtailed, pursuant to the provisions of its tariffs. For the water year of 2002, 4,069,700 AF of water flowed into Lake Powell from the Colorado River, or 23% of normal. For the 2003 water year precipitation levels have improved, although still below normal levels, and the Colorado River flow into Lake Powell has been 1,580,000 AF, through February 2003, or 67% of normal.

All electric energy sold by SCW to customers in its BVE customer service area is purchased from others. In March 2001, SCW entered into a five-year nine-month, block forward purchase contract with Mirant Marketing to supply its BVE customer service area with 15 MWs of electric energy at a price of \$95 per MWh beginning April 1, 2001 through December 31, 2006. SCW has filed a complaint with the FERC seeking to reduce the rates in the Mirant Marketing contract to a just and reasonable price. On December 19, 2002 the FERC issued an initial decision denying SCW's complaint. SCW has appealed the initial decision but anticipates that it is unlikely that FERC will grant its request to reduce these charges.

In June 2001, SCW executed a three-year, block forward purchase agreement with PWCC for an additional 8 MWs of electric energy to meet BVE's peak winter demands at a price of \$75 per MWh for the first year, \$48 per MWh for the second year and \$36 per MWh for the third year. Effective November 2002, SCW entered into a series of purchase power contracts with PWCC. Under the agreements, SCW will sell 15 MWs of electric energy to PWCC at a price of \$95 per MWh beginning November 1, 2002 through December 31, 2006, and 8 MWs of electric energy to PWCC at the prices set forth in the June 2001 energy purchase agreement with PWCC. In return, PWCC will supply SCW's BVE customer service area with 15 MWs of electric energy at a price of \$74.65 per MWh beginning November 1, 2002 through December 31, 2003 and each succeeding November 1, 2002 through March 31, 2008 and for the period November 1, 2008 through December 31, 2008. Total commitments under these agreements amount to \$71.9 million with \$12 million due each year through December 31, 2008.

The average minimum load at SCW's Bear Valley Electric customer service area has been approximately 14 MWs. The average winter load has been 18 MWs with a winter peak of 39 MWs when the snowmaking machines at the ski resorts are operating.

On July 17, 2002, the CPUC approved a settlement agreement reached among SCW, all intervening parties and the Office of Ratepayer Advocates ("ORA"), which permits SCW to recover \$77 per MWh of purchased power costs through rates, effective immediately thereafter. SCW will only be allowed to include up to a weighted annual energy purchase cost of \$77 per MWh each year for 10 years in its balancing account. To the extent SCW's actual average annual weighted costs for purchased power is less than \$77 per MWh, the differential will recover amounts included in the electric supply balancing account. Conversely, to the extent that actual average annual weighted costs for power purchased exceed the \$77 per MWh amount, SCW will not be able to include these

amounts in its balancing account and such amounts will be expensed against income. In 2002, approximately \$0.7 million was written-off against income. As a result of the settlement, SCW is also permitted to collect a surcharge of 2.2¢ per kilowatt hour from its customers for up to ten years commencing August 2001 to allow SCW an opportunity to collect amounts remaining in its electric cost balancing account, with interest, incurred by SCW during the energy crisis in 2000-2001.

Demand for energy in SCW's Bear Valley Electric customer service area generally has been increasing. However, the ability of SCW to deliver purchased power to these customers is limited by the ability of the transmission facilities owned by Southern California Edison Company to transmit this power. In order to meet these increasing energy demands, SCW is considering the construction of a natural gas-fueled generation facility owned by SCW. A Certificate of Public Convenience and Necessity was filed to seek the CPUC's authorization. If approved, it will authorize construction of the facility and enable SCW to file for further increases in electric energy prices for customers of SCW's Bear Valley Electric customer service area.

Operating Leases – Registrant leases equipment and facilities primarily for its Regional and District offices under non-cancelable operating leases with varying terms, provisions, and expiration dates. During 2002, 2001 and 2000, Registrant's consolidated rent expense was \$1,992,889, \$1,893,337 and \$1,794,113, respectively. Registrant's future minimum payments under long-term non-cancelable operating leases at December 31, 2002 are as follows:

2003	\$ 1,794,370
2004	1,739,878
2005	1,516,694
2006	991,583
2007	509,496
Thereafter	278,731
Total:	\$ 6,830,752

# Note 12 - Contingencies

SCW has been named as a defendant in twenty two lawsuits that allege that SCW and other water utilities, delivered unsafe water to their customers. Plaintiffs in these actions seek damages, including general, special, and punitive damages, according to proof at trial, as well as attorney's fees on certain causes of action, costs of suit, and other unspecified relief. Nineteen of the lawsuits involve customer service areas located in Los Angeles County in the southern portion of California; three of the lawsuits involve a customer service area located in Sacramento County in northern California. On September 1, 1999, the Court of Appeal in San Francisco held that the CPUC had preemptive jurisdiction over regulated public utilities and ordered dismissal of a series of these lawsuits. On October 11, 1999, one group of plaintiffs appealed the decision to the California Supreme Court.

On February 4, 2002, the California Supreme Court concluded that (i) the CPUC had preemptive jurisdiction over claims seeking injunctive relief and claims based on the theory that a public utility regulated by the CPUC provided unsafe drinking water even though it had complied with federal and state drinking water standards, but (ii) the CPUC did not have preemptive jurisdiction over damage claims alleging violations of federal and state drinking water standards by public utilities regulated by the CPUC. As a result, damage claims based on allegations of violations of federal and state drinking water standards may proceed while the other claims must be dismissed. In light of the breadth of plaintiffs' claims, the lack of factual information regarding plaintiffs' claims and

injuries, if any, the impact of the California Supreme Court decision on plaintiffs' claims and the fact that no discovery has yet been completed, SCW is unable at this time to determine what, if any, potential liability it may have with respect to claims based on allegations of violation of federal and state drinking water standards.

On October 25, 1999, SCW sued Aerojet-General Corporation (Aerojet) for causing the contamination of the Sacramento County Groundwater Basin. A cross complaint in 2000 filed by Aerojet against SCW for negligence and constituting a public nuisance was dismissed by the court in October 2002.

63

In a separate case, also filed on October 25, 1999, SCW filed a lawsuit against the State of California and its State Water Resources Control Board, Central Valley Regional Water Quality Control Board, and Department of Toxic Substances Control (collectively, the State) alleging that the State had substantially participated in a project to inject chemical pollution into portions of the Sacramento County groundwater basin and that pollution is progressively destroying the groundwater supply in SCW's Rancho Cordova water system. SCW and the State have entered into a comprehensive \$2,475,000 settlement of all of SCW's claims against the State contingent upon the court's approval of the good faith of the settlement.

The CPUC has authorized memorandum accounts to allow for recovery of costs incurred by SCW in prosecuting the cases against the State and Aerojet from customers, less any recovery from the defendants or others. As of December 31, 2002, approximately \$13.7 million has been recorded in the memorandum accounts as Other Deferred Charges. The CPUC has authorized SCW to increase rates, effective April 28, 2001, for recovery over a six-year period of approximately \$1.8 million, in expenses that were incurred on or before August 31, 2000. SCW will file to replace the current six year amortization to reflect the current memorandum account balance and maintain a long-tern amortization period in the next general rate case application for its Region I. The Notice of Intent for the application was filed with the CPUC on January 31, 2003. Management believes these costs are recoverable but cannot give assurance that the CPUC will ultimately allow recovery of all or any of the remaining costs through rates.

The compound MTBE has been detected in a well serving SCW's Los Osos water system. For some time SCW has been working with the Regional Water Quality Control Board as well as the owner of a gas service station. Although the owner of the service station has attempted remediation with funds provided through the California Underground Storage Tank Fund ("CUSTF"), it appears there will be insufficient funds from the CUSTF to complete remediation sufficient to return the well to service. In order to prevent the running of the statute of limitations, on August 12, 2002 SCW filed suit in the Superior Court of the State of California for the County of San Luis Obispo against the operators and owners of the service station facility, and Chevron USA Products, Inc., the supplier.

Volatile Organic Compounds (VOC) and perchlorate have been detected in 2 wells servicing SCW's San Gabriel System. SCW filed suit, along with two other affected water purveyors and the San Gabriel Basin Water Quality Authority (WQA), in the federal court against some of those responsible for the contamination. Some of the other potential defendants settled with SCW, other water purveyors and the WQA on VOC related issues, prior to the filing of the lawsuit. In response to the filing of the Federal lawsuit, the Potentially Responsible Party (PRP) defendants filed motions to dismiss the suit or strike certain portions of the suit. Following a hearing on these motions on March 31, 2003, the judge issued a ruling on April 1, 2003 granting in part and denying in part the defendant's motions. A key ruling of the court was that the water purveyors, including the Registrant, by virtue of their ownership of wells contaminated with hazardous chemicals are themselves PRPs under the Comprehensive Environmental Response, Compensation, and Liability Act (CERCLA). Registrant is currently evaluating the potential impact of this decision, its opportunity to appeal the decision and the opportunity to amend its suit to claim certain affirmative defenses as an "innocent" party under CERCLA. Registrant is presently unable to predict the outcome of this ruling on its ability to fully recover from PRPs future costs associated with the treatment of these wells.

SCW has been, in conjunction with the Southern California Edison (Edison) unit of Edison International, planning to upgrade transmission facilities to 115kv (the 115kv Project) in order to meet increased energy and demand requirements for SCW's Bear Valley Electric Service area. On December 27, 2000, SCW filed a lawsuit against Edison for declaratory relief and seeking damages for breach of contract as a result of delays in the 115kv Project as well as for violations of good faith and fair dealing, negligent misrepresentation, intentional misrepresentation and unjust enrichment. Subsequently Edison filed a cross-complaint against SCW for breach of contract, anticipatory breach, and quantum meruit. To date, SCW has spent approximately \$ 1.7 million in this matter, all of which was expensed. A trial date of May 5, 2003 has been set. SCW intends to fully prosecute this matter.

Management believes that proper insurance coverage and reserves are in place to insure against property, general liability and workers' compensation claims incurred in the ordinary course of business.

#### Note 13 - Construction Program

SCW's 2003 construction budget provides for gross expenditures of approximately \$88.3 million; \$6.7 million of this amount is anticipated to be obtained from developers and others. The increase in the capital budget for 2003 principally reflects the 2003 infrastructure replacement program in SCW's Metropolitan customer service area, a treatment plant upgrade, various water supply related projects, and security related costs. However, approved capital expenditures may be limited pending CPUC approval of SCW's general rate case filings. CCWC has a net capital budget of \$3.0 million for 2003 primarily reflecting improvement and upgrade of a water treatment plant. AWR and ASUS have no material capital commitments. However, ASUS actively seeks opportunities to own, lease or operate municipal water and wastewater systems, which may ultimately involve significant capital commitments.

6	Λ
U	7

## Note 14 - Business Segments

AWR has three principal business units: water and electric distribution units, through its SCW subsidiary, a water-service utility operation conducted through its CCWC unit, and a nonregulated activity unit through the ASUS subsidiary. All activities of SCW currently are geographically located within California. All activities of CCWC are located in the state of Arizona. All activities of ASUS are conducted in California and Arizona. Both SCW and CCWC are regulated utilities. On a stand-alone basis, AWR has no material assets other than its investments in its subsidiaries. The tables below set forth information relating to SCW's operating segments, CCWC and non-regulated businesses. Included in the amounts set forth, certain assets, revenues and expenses have been allocated. The identifiable assets are net of respective accumulated provisions for depreciation. Capital additions reflect capital expenditures paid in cash and exclude property installed by developers and conveyed to the Company.

(dollars in thousands)		Year Ended December 31, 2002						
	SC	SCW		SCW				
	Water	Electric	CCWC Water	Non- Regulated*	Consolidated AWR			
Operating revenues	\$ 180,904	\$ 21,298	\$ 6,157	\$ 846	\$ 209,205			
Operating income before income taxes	46,974	4,169	1,749	(2,295)	50,597			
Interest expense, net	15,275	1,361	482	581	17,699			
Identifiable assets	507,802	26,591	28,851	67	563,311			
Depreciation expense	15,914	1,470	918		18,302			
Capital additions	\$ 38,581	\$ 907	\$ 1,100	\$ 67	\$ 40,655			
(dollars in thousands)		Year I	Ended Decembe	er 31, 2001				
	SC	w						
	Water	Electric	CCWC Water	Non- Regulated*	Consolidated AWR			
Operating revenues	\$ 175,204	\$ 15,251	\$ 6,270	\$ 789	\$ 197,514			
Operating income before income taxes	54,037	(3,941)	1,915	60	52,071			
Interest expense, net	13,385	1,192	527	631	15,735			
Identifiable assets	484,002	27,182	28,658		539,842			
Depreciation expense	15,264	1,446	1,241		17,951			
Capital additions	\$ 44,773	\$ 2,256	\$ 541	_	\$ 47,570			

SCW CCWC Non- Consolidated Water Regulated\* AWR

	Water	Electric			
Operating revenues	\$ 167,529	\$ 14,366	\$ 1,266	\$ 799	\$ 183,960
Operating income before income taxes	42.542	4,520	\$ 1,200 292	80	47,434
Interest expense, net	13,176	1,174	125	(353)	14,122
Identifiable assets	453,538	26,531	29,027	_	509,096
Depreciation expense	13,685	1,401	253	—	15,339
Capital additions	\$ 41,774	\$ 2,303	\$ 2,014	_	\$ 46,091

\* Includes amounts from ASUS and AWR.

### Note 15 - Allowance for Doubtful Accounts

The table below presents Registrant's provision for doubtful accounts charged to expense and accounts written off, net of recoveries. Provisions included in 2002 and 2001 represent both SCW and CCWC. Provisions in 2000 represent SCW only.

	December 31,		
(dollars in thousands)	2002	2001	2000
Balance at beginning of year	\$ 972	\$ 510	\$ 487
Provision charged to expense	324	1,033	630
Accounts written off, net of recoveries	(527)	(571)	(607)
Balance at end of year	\$ 769	\$ 972	\$ 510

Neither AWR nor ASUS have established any provision for doubtful accounts.

## Note 16 - Selected Quarterly Financial Data (Unaudited)

The quarterly financial information presented below is unaudited. The business of Registrant is of a seasonal nature and it is management's opinion that comparisons of basic earnings for the quarterly periods do not reflect overall trends and changes in Registrant's operations.

	-	For The Year Ended December 31, 2002								
(in thousands, except per share amounts)	Operating Revenues			Basic Earnings Per Share						
First Quarter	\$ 44,485	\$ 7,918	\$ 3,807	\$ 0.25						
Second Quarter	52,802	9,886	5,435	0.36						
Third Quarter	61,612	11,864	7,639	0.50						
Fourth Quarter	50,306	7,980	3,458	0.23						
Year	\$ 209,205	\$ 37,648	\$ 20,339	\$ 1.34						

For The Veer Ended December 21, 2001

For The Year Ended December 31, 2001						
Operating Revenues	Operating	Operating Income Net Income			Basic Ea Per Sl	
	Previously Reported	Restated Note 2	Previously Reported	Restated Note 2	Previously Reported	Restated Note 2
\$ 40,291	\$ 7,223	\$ 7,580	\$ 3,117	\$ 3,474	\$ 0.21	\$ 0.23
49,870	9,013	9,456	5,053	5,496	0.33	0.36
59,410	13,372	13,651	9,454	9,733	0.63	0.64
47,943	7,084	7,014	2,823	2,753	0.18	0.18
\$ 197,514	\$ 36,692	\$ 37,701	\$ 20,447	\$ 21,456	\$ 1.35	\$ 1.41
	Revenues           \$ 40,291           49,870           59,410           47,943	Operating Revenues         Operating           Previously Reported         Previously Reported           \$ 40,291         \$ 7,223           49,870         9,013           59,410         13,372           47,943         7,084	Operating Revenues         Operating Income           Previously Reported         Restated Note 2           \$ 40,291         7,223         7,580           49,870         9,013         9,456           59,410         13,372         13,651           47,943         7,084         7,014	Operating Revenues         Operating Income         Net In           Previously Reported         Restated Note 2         Previously Reported           \$ 40,291         7,223         7,580         \$ 3,117           49,870         9,013         9,456         5,053           59,410         13,372         13,651         9,454           47,943         7,084         7,014         2,823	Operating Revenues         Operating Income         Net Income           Previously Reported         Restated Note 2         Previously Reported         Restated Note 2           \$ 40,291         7,223         7,580         3,117         \$ 3,474           49,870         9,013         9,456         5,053         5,496           59,410         13,372         13,651         9,454         9,733           47,943         7,084         7,014         2,823         2,753	Operating Revenues         Operating Income         Net Income         Basic E Per S           Previously Reported         Restated Note 2         Previously Reported         Restated Note 2

## Report of Management

The consolidated financial statements contained in the annual report were prepared by the management of American States Water Company, which is responsible for their integrity and objectivity. The consolidated financial statements were prepared in accordance with generally accepted accounting principles and include, where necessary, amounts based upon management's best estimates and judgments. All other financial information in the annual report is consistent with the consolidated financial statements and is also the responsibility of management.

Registrant maintains systems of internal control, which are designed to help safeguard, the assets of Registrant and provide reasonable assurance that accounting and financial records can be relied upon to generate accurate financial statements. These systems include the hiring and training of qualified personnel, appropriate segregation of duties, delegation of authority and an internal audit function, which has reporting responsibility to the Audit Committee of the board of directors.

The Audit Committee, composed of three outside directors, exercises oversight of management's discharge of its responsibilities regarding the systems of internal control and financial reporting. The committee periodically meets with management, the internal auditor and the independent accountants to review the work and findings of each. The committee also reviews the qualifications of, and recommends to the board of directors, a firm of independent accountants.

The independent accountants, PricewaterhouseCoopers LLP, have performed an audit of the consolidated financial statements in accordance with generally accepted auditing standards. Their audit gave consideration to Registrant's system of internal accounting control as a basis for establishing the nature, timing and scope of their work. The result of their work is expressed in their Report of Independent Accountants.

s/ FLOYD E. WICKS

Floyd E. Wicks President, Chief Executive Officer

April 7, 2003

### s/ McCLELLAN HARRIS III

McClellan Harris III Chief Financial Officer, Sr. Vice President - Finance, Treasurer and Corporate Secretary

67

### **Report of Independent Accountants**

To the Board of Directors and Shareholders

of American States Water Company

In our opinion, the accompanying consolidated balance sheets and consolidated statements of capitalization and the related consolidated statements of income, of changes in common shareholders' equity and of cash flows, present fairly, in all material respects, the financial position of American States Water Company and its subsidiaries (the Company) and Southern California Water Company (SCW) at December 31, 2002 and 2001, and the results of their operations and their cash flows for the each of the three years in the period ended December 31, 2002 in conformity with accounting principles generally accepted in the United States of America. These financial statements are the responsibility of the Company's and SCW's management; our responsibility is to express an opinion on these financial statements based on our audits. We conducted our audits of these statements in accordance with auditing standards generally accepted in the United States of America, which require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. An audit includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, and evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

As discussed in Note 1 to the accompanying consolidated financial statements, on January 1, 2002 the Company adopted Statement of Financial Accounting Standards No. 142 and changed its method of accounting for goodwill.

As discussed in Note 2 to the accompanying consolidated financial statements, the Company has revised its consolidated financial statements for the years ended December 31, 2001 and 2000 to decrease the provision for taxes on income and the related accumulated deferred income taxes – net and taxes payable balances.

PricewaterhouseCoopers LLP Los Angeles, California April 7, 2003

68

#### Item 9. Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

On May 13, 2003, the Registrants dismissed Arthur Andersen LLP as its independent auditor. The decision to dismiss Arthur Andersen was recommended by the Registrants' Audit Committee and approved by the Board of Directors.

Arthur Andersen's report on the financial statements of the Registrants for each of the years ended December 31, 2000 and December 31, 2001 did not contain an adverse opinion or a disclaimer of opinion and was not qualified or modified as to uncertainty, audit scope, or accounting principles.

During the years ended December 31, 2000 and December 31, 2001, and the interim period between December 31, 2001 and May 13, 2002, there were no disagreements between the Registrants and Arthur Andersen on any matter of accounting principles or practices, financial statement disclosure or auditing scope or procedure, which disagreements, if not resolved to the satisfaction of Arthur Andersen, would have caused it to make reference to the subject matter of the disagreements in connection with its report. During the years ended December 31, 2000 and December 31, 2001, and the interim period between December 31, 2001 and May 13, 2002, there were no reportable events (as defined in Item 304(a)(1)(v) of Regulation S-K promulgated by the SEC). In May 2002, Arthur Andersen furnished the Registrants with a letter addressed to the SEC stating that it agrees with the statements above. A copy of the letter was included as an exhibit to the Form 8-K filed by the Company in May 2002.

The Registrants engaged PricewaterhouseCoopers LLP as its independent auditor, effective May 13, 2002. During the years ended December 31, 2000 and December 31, 2001, and the interim period between December 31, 2001 and May 13, 2002, neither the Registrants nor anyone on their behalf consulted with PricewaterhouseCoopers LLP (i) regarding the application of accounting principles to a specified transaction, either completed or proposed, (ii) the type of audit opinion that might be rendered on the Registrants' financial statements, or (iii) any matter that was either the subject of a disagreement (as described above) or a reportable event.

## PART III

### Item 10. Directors and Executive Officers of the Registrant

Information responsive to Part III, Item 10 is included in the Proxy Statement, to be filed by Registrant with the Commission pursuant to Regulation 14A, under the captions therein entitled "Election of Directors" and "Executive Officers - Experience, Security Ownership and Compensation" and is incorporated herein by reference pursuant to General Instruction G(3).

#### Item 11. Executive Compensation

Information responsive to Part III, Item 11 is included in the Proxy Statement, to be filed by Registrant with the Commission pursuant to Regulation 14A, under the captions therein entitled "Election of Directors" and "Executive Officers - Experience, Security Ownership and Compensation" and "Performance Graph" and is incorporated herein by reference pursuant to General Instruction G(3).

#### Item 12. Security Ownership of Certain Beneficial Owners and Management and Related Stockholder Matters

Information responsive to Part III, Item 12 is included in the Proxy Statement, to be filed by Registrant with the Commission pursuant to Regulation 14A, under the captions therein entitled "Election of Directors" and "Executive Officers - Experience, Security Ownership and Compensation" and is incorporated herein by reference pursuant to General Instruction G(3).

## Item 13. Certain Relationships and Related Transactions

Information responsive to Part III, Item 13 is included in the Proxy Statement, to be filed by Registrant with the Commission pursuant to Regulation 14A, under the captions therein entitled "Election of Directors" and is incorporated herein by reference pursuant to General Instruction G(3).

## Item 14. Controls and Procedures

a) Within ninety days prior to the filing of this report, we carried out an evaluation, under the supervision and with the participation of our Chief Executive Officer and Chief Financial Officer, of the effectiveness of our disclosure controls and procedures. Disclosure controls and procedures are designed to ensure that information required to be disclosed in our periodic reports filed or submitted under the Securities Exchange Act of 1934 is recorded, processed, summarized and reported within the time periods specified in the Securities and Exchange Commission's rules and forms. Based upon that evaluation, our Chief Executive Officer and Chief Financial Officer concluded that our disclosure controls and procedures are effective and there are no significant deficiencies or material weaknesses.

During the 90 days preceding the filing of this report, Registrant noted one matter involving internal controls that is considered to be a weakness under standards established by American Institute of certificated Public Accountants. In connection with a deferred tax analysis of Registrant's tax accounts performed in 2002 for a period of 20 years, Registrant determined that it had miscalculated its net accumulated deferred income taxes over several years, resulting in taxes on income not being recognized in the proper period and overstating the net deferred tax liabilities and current taxes payable of AWR and SCW by \$4.7 million and by \$4.9 million, respectively, for prior years. Registrant further determined that the amounts of the overstatements arose in prior years and that prior years' financial statements should be restated to reduce net accumulated deferred income taxes, taxes payable and the related tax expense to reflect amounts currently due and deferred. Registrant is in the process of acquiring appropriate tax software and will add qualified staff as needed to ensure the accuracy in this regard.

There have been no other changes in our internal controls or in other factors that could significantly affect internal controls subsequent to the date we carried out this evaluation.

#### Item 15. Exhibits, Financial Statement Schedules and Reports on Form 8-K

- (a) 1. Reference is made to the Financial Statements incorporated herein by reference to Part II, Item 8 hereof.
  - 2. All required schedules may be found in the Financial Statements and Notes to Financial Statements incorporated herein by reference to Part II, Item 8 hereof or at the conclusion of this Item. Schedules II, III, IV, and V are omitted as they are not applicable.
- (b) No Reports on Form 8-K were filed during the fourth quarter of 2002.
- (c) Exhibits -
  - 3.1 By-Laws of American States Water Company incorporated herein by reference to Registrant's Form 8-K, dated November 2, 1998.
  - 3.2 Amended and Restated By-laws of Southern California Water Company.
  - 3.3 Amended and Restated Articles of Incorporation of American States Water Company.<sup>(1)</sup>
  - 3.3.1 Restated Articles of Incorporation of Southern California Water Company incorporated herein by reference to Registrant's Form 8-K, dated January 20, 1999<sup>(1)</sup>
  - 4.1 Amended and Restated Rights Agreement, dated January 25, 1999, by and between American States Water Company and ChaseMellon Shareholder Services, L.L.C., as Rights Agent incorporated by reference to Registrant's Form 10-K for the year ended December 31, 1998.

70

- 4.2 Indenture, dated September 1, 1993 between Southern California Water Company and Chemical Trust Company of California incorporated herein by reference to Registrant's Form 8-K.
- 10.1 Agreement of Merger dated as of June 25, 1998 by and among Southern California Water Company, SCW Acquisition Corp. and American States Water Company incorporated herein by reference to Registrant's Form 8-K, dated July 1, 1998.
- 10.2 Deferred Compensation Plan for Directors and Executives incorporated herein by reference to Registrant's Registration Statement on Form S-2, Registration No. 33-5151.<sup>(2)</sup>
- 10.3 Reimbursement Agreement, dated September 1, 2000, between Southern California Water Company and Bank of America, N.A. incorporated by reference to Registrant's Form 10-K with respect to the year ended December 31, 2000.
- 10.4 Second Sublease dated October 5, 1984 between Southern California Water Company and Three Valleys Municipal Water District incorporated herein by reference to Registrant's Registrant's Registration Statement on Form S-2, Registration No. 33-5151.
- 10.5 Note Agreement dated as of May 15, 1991 between Southern California Water Company and Transamerica Occidental Life Insurance Company incorporated herein by reference to Registrant's Form 10-Q with respect to the quarter ended June 30, 1991.
- 10.6 Schedule of omitted Note Agreements, dated May 15, 1991, between Southern California Water Company and Transamerica Annuity Life Insurance Company, and Southern California Water Company and First Colony Life Insurance Company incorporated herein by reference to Registrant's Form 10-Q with respect to the quarter ended June 30, 1991.
- 10.7 Loan Agreement between California Pollution Control Financing Authority and Southern California Water Company, dated as of December 1, 1996 incorporated by reference to Registrant's Form 10-K for the year ended December 31, 1998.
- 10.8 Agreement for Financing Capital Improvement dated as of June 2, 1992 between Southern California Water Company and Three Valleys Municipal Water District incorporated herein by reference to Registrant's Form 10-K with respect to the year ended December 31, 1992.
- 10.9 Water Supply Agreement dated as of June 1, 1994 between Southern California Water Company and Central Coast Water Authority incorporated herein by reference to Registrant's Form 10-K with respect to the year ended December 31, 1994.
- 10.10 Amended and Restated Retirement Plan for Non-Employee Directors of American States Water Company, dated as of October 25, 1999, incorporated herein by reference to Registrant's Form 10-K with respect to the year ended December 31, 1999. <sup>(2)</sup>
- 10.11 Dividend Reinvestment and Common Share Purchase Plan incorporated herein by reference to American States Water Company Rule 424 (b) (3) filing dated October 27, 1999.
- 10.12 Key Executive Long-Term Incentive Plan incorporated herein by reference to Registrant's 1995 Proxy Statement. <sup>(2)</sup>
- 10.14 Amended and Restated Change in Control Agreements, dated as of October 25, 1999, between American States Water Company, Southern California Water Company and certain executives incorporated herein by reference to Registrant's Form 10-K with respect to the year ended December 31, 1999. <sup>(2)</sup>

- 10.15 Amended and Restated Change in Control Agreements, dated as of October 25, 1999, between Southern California Water Company and certain executives incorporated by reference to Registrant's Form 10-K with respect to the year ended December 31, 1999. <sup>(2)</sup>
- 10.16 Southern California Water Company Pension Restoration Plan incorporated herein by reference to Registrant's Form 10-K with respect to the year ended December 31, 1999.<sup>(2)</sup>
- 10.17 American States Water Company Annual Incentive Plan as amended April 29, 2002 incorporated by reference to Registrant's Form 10-Q for the quarter ended March 31, 2002.<sup>(2)</sup>
- 10.18 American States Water Company amended 2000 Stock Incentive Plan incorporated herein by reference to Registrant's Form 10-Q for the quarter ended March 31, 2000.
- 10.19 Loan and Trust Agreement between The Industrial Development Authority of The County of Maricopa, Chaparral City Water Company and Bank One, Arizona, NA,, dated as of December 1, 1997 incorporated by reference to Registrant's Form 10-K with respect to the year ended December 31, 2000.

- 10.20 Delivery Agreement between Central Arizona Water Conservation District and Chaparral City Water Company, dated as of December 6, 1984 incorporated by reference to Registrant's Form 10-K with respect to the year ended December 31, 2000.
- 10.21 Repayment Contract between the United States Bureau of Reclamation and Chaparral City Water Company, dated as of December 6, 1984 for construction of a delivery and storage system to transport CAP water incorporated by reference to Registrant's Form 10-K with respect to the year ended December 31, 2000.
- 10.22 Energy Transaction Confirmation with Mirant Americas Energy Marketing, LP incorporated by reference to Registrant's Form 10-Q with respect to the quarter ended March 31, 2001.
- 10.23 Power Purchase Agreement in June 2001 between Southern California Water Company and Pinnacle West Capital Corporation incorporated by reference to Registrant's Form 10-Q with respect to the quarter ended June 30, 2001.
- 10.24 Western Systems Power Pool Agreement incorporated by reference to Registrant's Form 10-Q with respect to the quarter ended June 30, 2001.
- 10.25 Automated Power Exchange Master Service and Participation Agreement incorporated by reference to Registrant's Form 10-K with respect to the year ended December 31, 2001.
- 10.26 American States Water Company Three-Year Dividend Equivalent Right Certificate incorporated by reference to Registrant's Form 10-Q for the quarter ended March 31, 2002. <sup>(2)</sup>
- 10.27 Power Purchase Agreement dated September 3, 2002 between Southern California Water Company and Pinnacle West Capital Corporation. <sup>(1)</sup>
- 10.28 Credit Agreement between American States Water Company dated June 6, 2002 with Wells Fargo Bank, N.A., as Administrative Agent. <sup>(1)</sup>
- 21. Subsidiaries of Registrant incorporated herein by reference to Registrant's Form 10-K with respect to the year ended December 31, 1998.
- 23. Consent of Independent Public Accountants. <sup>(1)</sup>

(d) None.

(1) Filed concurrently herewith

(2) Management contract or compensatory arrangement

72

## SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, Registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

AMERICAN STATES WATER COMPANY and its subsidiary SOUTHERN CALIFORNIA WATER COMPANY

By: s/ McCLELLAN HARRIS III.

McClellan Harris III Sr. Vice President - Finance, Treasurer, Chief Financial Officer and Secretary Date: April 7, 2003

Pursuant to the requirements of the Securities Exchange Act of 1934, this report has been signed below by the following persons on behalf of Registrant and in the capacities and on the dates indicated.

s/ LLOYD E. ROSS.	Date:	April 7, 2003
Lloyd E. Ross Chairman of the Board and Director s/ FLOYD E. WICKS.	-	April 7, 2003
Floyd E. Wicks Principal Executive Officer; President, CEO and Director s/ McCLELLAN HARRIS III.	-	April 7, 2003
McClellan Harris III Principal Financial and Accounting Officer; CFO, Sr. VP - Finance, Treasurer and Secretary	-	•
s/ JAMES L. ANDERSON.	_	April 7, 2003
James L. Anderson, <i>Director</i> s/ JEAN E. AUER.		April 7, 2003
Jean E. Auer, <i>Director</i> s/ N. P. DODGE, JR.	-	April 7, 2003
N. P. Dodge, Jr., <i>Director</i> s/ ANNE M. HOLLOWAY.	-	April 7, 2003
Anne M. Holloway, <i>Director</i> s/ ROBERT F. KATHOL.	-	April 7, 2003
Robert F Kathol Director		

Robert F. Kathol, Director

## 73

## CERTIFICATIONS

I, Floyd E. Wicks, Chief Executive Officer of the Registrant, certify that:

- 1) I have reviewed this annual report on Form 10-K for the period ended December 31, 2002 (this "annual report");
- Based on my knowledge, this annual report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this annual report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this annual report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this annual report;
- 4) The Registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Section 240.13a-14of the General Rules and Regulations promulgated under the Securities Exchange Act of 1934) for the Registrant and we have:
  - a) designed such disclosure controls and procedures to ensure that material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
  - b) evaluated the effectiveness of the Registrant's disclosure controls and procedures as of a date within 90 days prior to the filing date of this annual report (the "Evaluation Date"); and
  - c) presented in this annual report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5) The Registrant's other certifying officers and I have disclosed, based on our most recent evaluation, to the Registrant's auditors and the audit committee of Registrant's board of directors (or persons performing the equivalent function):
  - a) all significant deficiencies in the design or operation of internal controls which could adversely affect the Registrant's ability to record, process, summarize and report financial data and have identified for the Registrant's auditors any material weaknesses in internal controls; and
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal controls; and
- 6) The Registrant's other certifying officers and I have indicated in this annual report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

Dated: April 7, 2003

By: /s/ FLOYD E. WICKS

Floyd E. Wicks Chief Executive Officer

74

## CERTIFICATIONS

I, McClellan Harris III, Chief Financial Officer of the Registrant, certify that:

- 1) I have reviewed this annual report on Form 10-K for the period ended December 31, 2002 (this "annual report");
- Based on my knowledge, this annual report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this annual report;
- 3) Based on my knowledge, the financial statements, and other financial information included in this annual report, fairly present in all material respects the financial condition, results of operations and cash flows of the Registrant as of, and for, the periods presented in this annual report;
- 4) The Registrant's other certifying officers and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Section 240.13a-14of the General Rules and Regulations promulgated under the Securities Exchange Act of 1934) for the Registrant and we have:
  - a) designed such disclosure controls and procedures to ensure that material information relating to the Registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this annual report is being prepared;
  - b) evaluated the effectiveness of the Registrant's disclosure controls and procedures as of a date within 90 days prior to the filing date of this annual report (the "Evaluation Date"); and
  - c) presented in this annual report our conclusions about the effectiveness of the disclosure controls and procedures based on our evaluation as of the Evaluation Date;
- 5) The Registrant's other certifying officers and I have disclosed, based on our most recent evaluation, to the Registrant's auditors and the audit committee of Registrant's board of directors (or persons performing the equivalent function):
  - a) all significant deficiencies in the design or operation of internal controls which could adversely affect the Registrant's ability to record, process, summarize and report financial data and have identified for the Registrant's auditors any material weaknesses in internal controls; and
  - b) any fraud, whether or not material, that involves management or other employees who have a significant role in the Registrant's internal controls; and
- 6) The Registrant's other certifying officers and I have indicated in this annual report whether or not there were significant changes in internal controls or in other factors that could significantly affect internal controls subsequent to the date of our most recent evaluation, including any corrective actions with regard to significant deficiencies and material weaknesses.

Dated: April 7, 2003

By: /s/ McCLELLAN HARRIS III

McClellan Harris III Chief Financial Officer, Senior Vice President-Finance, Treasurer and Corporate Secretary Our audits of the consolidated financial statements referred to in our report dated April 7, 2003 appearing in the 2002 Annual Report to Shareholders of American States Water Company (which report and consolidated financial statements are incorporated by reference in this Annual Report on Form 10-K) also included an audit of the financial statement schedules listed in Item 15(a)(2) of this Form 10-K. In our opinion, these financial statement schedules present fairly, in all material respects, the information set forth therein when read in conjunction with the related consolidated financial statements.

As discussed in Note 2 to the consolidated financial statements referred to above, the Company has revised its consolidated financial statements for the years ended December 31, 2001 and 2000 to decrease the provision for taxes on income and the related accumulated deferred income taxes – net and taxes payable balances.

PricewaterhouseCoopers LLP Los Angeles, California April 7, 2003

## AMERICAN STATES WATER COMPANY SCHEDULE I – CONDENSED FINANCIAL INFORMATION OF PARENT

### CONDENSED BALANCE SHEET

	Dec	ember 31,
(in thousands)	2002	2001 Restated (1)
Assets		
Cash and equivalents	\$ 2,29	7 \$ 523
Other current assets	13,93	2 525
	_	
Total current assets	16,22	9 1,048
	- ,	,
Investments in subsidiaries	231,73	7 225,378
Other deferred debits	40	1 115
Total assets	\$ 248,36	7 \$ 226,541
	_	

## Liabilities and Capitalization

Notes payable to banks	35,000	\$ 20,000
Accounts payable	59	7
Other current liabilities	29	
Total current liabilities	35,088	20,007
Common shareholders' equity	213,279	204,654
Preferred shares		1,880
Total capitalization	213,279	206,534
Total liabilities and capitalization	\$ 248,367	\$ 226,541

(1) Restated due to overstatements in net deferred tax liabilities and current taxes payable that occurred in prior years. See Note 2 of "Notes to Consolidated Financial Statements" included in Part II, Item 8 in Financial Statements and Supplementary Data.

## 77

## CONDENSED STATEMENTS OF INCOME

		December 31,		
(in thousands except per share amount)	2002	2001 Restated (1)	2000 Restated (1)	
Operating Revenues And Other Income	\$ —	\$ _	\$ —	
Operating Expenses	651	403	206	
Loss Before Equity in Earnings of Subsidiaries	(651)	(403)	(206)	
Equity in Earnings of Subsidiaries	20,990	21,859	20,237	
Net Income	20,339	21,456	20,031	
Dividends on Preferred Shares	(29)	(84)	(86)	
Earnings Available For Common Shareholders	\$ 20,310	\$ 21,372	\$ 19,945	
Weighted Average Number of Common Shares Outstanding	15,144	15,120	14,070	

Basic Earnings Per Common Share	\$ 1	34	\$ 1.41	\$	1.42
				-	
Weighted Average Number of Diluted Common Shares Outstanding	15,1	57	15,122	1	4,070
Fully Diluted Earnings per Common Share	\$ 1	34	\$ 1.41	\$	1.42

(1) Restated due to overstatements in net deferred tax liabilities and current taxes payable that occurred in prior years. See Note 2 of "Notes to Consolidated Financial Statements" included in Part II, Item 8 in Financial Statements and Supplementary Data.

## CONDENSED STATEMENTS OF CASH FLOWS

	_		
(in thousands)	2002	2001 Restated (1)	2000 Restated (1)
Cash Flows From Operating Activities	\$ 368	\$ 18,457	\$ 13,075
Cash Flows Used in Investing Activities		(25,000)	(24,340)
Cash Flows From Financing Activities	1,406	6,772	11,390
Increase in Cash and Equivalents Cash and Equivalents at Beginning of Period	1,774 523	229 294	125 169
Cash and Equivalents at the End of Period	\$ 2,297	\$ 523	\$ 294
Cash dividends received from Southern California Water Company and Chaparral City Water Company	\$ 14,630	\$ 14,533	\$ 12,900

(1) Restated due to overstatements in net deferred tax liabilities and current taxes payable that occurred in prior years. See Note 2 of "Notes to Consolidated Financial Statements" included in Part II, Item 8 in Financial Statements and Supplementary Data.

### CERTIFICATE OF AMENDMENT

#### TO

#### AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

### AMERICAN STATES WATER COMPANY

## Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly elected President and Chief Executive Officer, and the duly elected Chief Financial Officer, Vice President--Finance, Treasurer and Corporate Secretary, respectively, of American States Water Company, a California corporation (the "Company").

2. Article IV of the Company's Amended and Restated Articles of Incorporation is amended to add a new subsection (5) thereto to read in its entirety as follows:

(5) On May 15, 2002, every two shares of Common Shares of this Corporation shall be split into three Common Shares of this Corporation (the "Stock Split"). No certificates evidencing fractional shares will be issued in connection with this Stock Split, but instead, a certificate or certificates evidencing the aggregate of all fractional shares that would be issued (rounded, if necessary, to the next higher whole share) shall be issued to ChaseMellon Shareholder Services, Inc., or its nominee, as transfer agent, for the accounts of all holders of Common Shares of this Corporation otherwise entitled to have a fraction of a share delivered to them in connection with the Stock Split; and the transfer agent shall sell those shares as soon as practicable after receiving them at the then market price and remit the net proceeds to the shareholders entitled thereto.

4. The Company has only shares of Common Shares outstanding.

5. The foregoing amendments of the Amended and Restated Articles of Incorporation have been duly approved by the Board of Directors of the Company alone in accordance with Section 902(c) of the California General Corporation Law.

We further declare under penalty of perjury under the laws of the State of California that the matters set forth in this certificate are true and correct of our own knowledge.

DATED: May 3, 2002

/s/\_ Floyd E. Wicks

/s/ McClellan Harris III

#### CERTIFICATE OF AMENDMENT

#### TO

#### AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

### AMERICAN STATES WATER COMPANY

### Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly elected President and Chief Executive Officer, and the duly elected Chief Financial Officer, Vice President--Finance, Treasurer and Corporate Secretary, respectively, of American States Water Company, a California corporation (the "Company").

2. Article IV of the Company's Amended and Restated Articles of Incorporation is amended in its entirety to read as follows:

This Corporation is authorized to issue two classes of stock to be designated, respectively, "New Preferred Shares" and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,150,000; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000, and 30,000,000 shares are to be Common Shares with no par value with a stated value of \$2.50 per share and an aggregate stated value of \$75,000,000.

A statement of the preferences, privileges and restrictions granted to or imposed upon the respective classes or series of shares and/or upon the holders thereof is as follows:

(1) Subject to the provisions of this Article IV, New Preferred Shares of any particular series shall be entitled to such voting rights, if any, as may be specified for shares of such series in the certificate of determination of preferences of such series filed as provided below. All Common Shares shall be entitled to voting rights on the basis of one-tenth of one vote per share.

(2) New Preferred Shares may be issued from time to time in one or more series. Each such series shall be so designated as to distinguish it from other series of New Preferred Shares. Such designation may include an appropriate reference to the dividend rate and/or any other characteristics of such series. The Board of Directors is hereby authorized, within the limits of, but to the extent authorized by applicable law and within the limitations and restrictions, if any, stated in this Article IV, to fix or alter, from time to time, the dividend rights, dividend rate, conversion rights, voting rights, right and terms of redemption (including sinking fund provisions), redemption price or prices, and liquidation preferences, or any

of them, of any wholly unissued series of New Preferred Shares, and to fix the number of shares constituting any such unissued series, by a resolution or resolutions adopted by the Board of Directors in exercise of the authority hereby granted.

(3) Subject to the dividend preferences provided for all shares of each other class at the time outstanding and to the restrictions set forth in this Paragraph (3), the Common Shares shall be entitled to receive dividends when and as declared by the Board of Directors out of any funds of this Corporation legally available therefor. After payment of the full preferential amounts hereinabove provided for all shares of each other class outstanding at the time of any liquidation, dissolution or winding up of this Corporation, whether voluntary or involuntary, all then remaining assets of this Corporation available for distribution to its shareholders shall be distributed ratably upon the Common Shares. No dividend shall be declared on the Common Shares which, after giving effect to such declaration, would reduce the Common Stock Equity of this Corporation as of the end of the calendar month last preceding that in which such dividend was declared to an amount less than 25% of the Total Capitalization of this Corporation as of said last preceding month, except that any such dividend may be declared (a) which would reduce such Common Stock Equity to less than 25% but not less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during the 12 months period terminating at the end of such last preceding calendar month shall not exceed 75% of the net income of this Corporation applicable to its Common Shares for such period, or (b) which would reduce such Common Stock Equity to less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during said 12 months period shall not exceed 50% of the net income of this Corporation applicable to its Common Shares for such period; provided, however, that the foregoing restrictions of this sentence shall not apply to, nor in any way restrict, (a) the payment of any dividend on the Common Shares which is payable in shares of stock of this Corporation, or (b) any reclassification, subdivision, split-up or combination of the Common Shares, or (c) any transfer between the capital and surplus accounts of this Corporation in connection with any such reclassification, subdivisions, split-up or combination or payment of dividend in shares of stock of this Corporation. Common Stock Equity as herein used shall mean the aggregate of (i) par

value or stated capital of all outstanding Common Shares, and (ii) the surplus (including capital surplus, paid-in surplus and earned surplus) as shown by the books of this Corporation after giving effect to the declaration of the proposed dividend, and (iii) premium on Common Shares; less the remaining balance of the amount of organization expenses, as shown on said books. Total Capitalization as herein used shall mean the aggregate of (i) Common Stock Equity, (ii) premium on and the par value or stated capital of all outstanding shares of this Corporation of any and all classes having preferences over the Common Shares as to dividends or assets, and (iii) the principal amount of all outstanding debt maturing more than 12 months after the close of said 12 months period, all as shown by the books of this Corporation; less the remaining balance of organization expenses, as shown on said books. Net Income as herein used shall be determined in accordance with accepted accounting practice (but after provision for all taxes based upon or measured by income, and after annual interest charges adjusted by provision for amortization of bond discount and expense or of premium on indebtedness, and also after deduction of depreciation for said 12 months period as reported in the accounts of this Corporation as filed with the Public Utilities Commission of the State of California or other public authority of said state having jurisdiction to establish or approve the system of accounts of this Corporation). Net Income applicable to Common Shares as herein used shall mean net income after deduction therefrom of all dividends payable for the period involved on all outstanding shares of any and all classes of this Corporation having preference over the Common Shares as to dividends or assets.

(4) Unless such action has been approved by the affirmative vote of at least a majority of the Continuing Directors (as defined below), without the approval of Common Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares representing in the aggregate at least 66 2/3% of the combined voting power of this Corporation's outstanding Common Shares and the New Preferred Shares, voting together as a single class, this Corporation shall not

(i) subject to subparagraph (iii) below, sell, convey, lease or otherwise dispose of all or substantially all of its assets, property or business;

(ii) approve the sale, conveyance, lease or other disposition by any subsidiary of this Corporation of all or substantially all of such subsidiary's assets, property or business;

(iii) sell, transfer, convey or otherwise dispose of more than a majority of the outstanding capital stock of any subsidiary of the Corporation, if such subsidiary holds assets accounting for 50% or more of the Corporation's consolidated assets, other than to an entity the majority of the voting power of the capital stock or other equity interest of which is owned and controlled by this Corporation;

(iv) consolidate or merge with or into any other corporation or other business entity, except if, immediately after such consolidation or merger, the shareholders of this Corporation immediately prior to such consolidation or merger will own more than 60% of the voting power of the outstanding capital stock or other equity interest of or in the surviving entity; or

(v) approve the consolidation or merger of any subsidiary of this Corporation, if such subsidiary holds assets accounting for 50% or more of

the Corporation's consolidated assets, with or into any other corporation or other business entity.

For purposes of this paragraph (4) of Article IV, the term "Continuing Directors" shall mean any member of the Board of Directors of the Corporation (while such person is a member of the Board) who (i) is not an Acquiring Person, or an Affiliate or Associate of an Acquiring Person, or a representative of an Acquiring Person or of any such Affiliate or Associate, and (ii) either (A) was a member of the Board of Directors prior to the time any person became an Acquiring Person, or (B) became a member of the Board of Directors subsequent to the time any person became an Acquiring Person, if such person's nomination for election, or re-election, to the Board was recommended, or approved, by a majority of the Continuing Directors then in office. For purposes of the foregoing definition, (i) "Affiliate" and "Associate" shall have the respective meanings ascribed to such terms in Rule 12b-2 of the General Rules and Regulations under the Exchange Act, as in effect as of the date hereof; (ii) "Acquiring Person" shall mean any person or entity which, alone or together with all Affiliates and Associates of such person or entity, shall be the beneficial owner of 20% or more of the Corporation's voting stock, but shall not include (1) an Exempt Person or (2) any person or entity who or which acquires 20% or more of the Corporation's voting stock in connection with a transaction or series of transactions approved prior to such transaction or transactions by the Board of Directors of the Corporation; provided that no person or entity shall become an Acquiring Person solely as a result of a reduction in the number of shares of the Corporation's voting stock outstanding, unless and until such person or entity shall thereafter become the beneficial owner of additional shares constituting 1% or more of the general voting power of the Corporation. "Exempt Person" shall mean the Corporation, any majority-owned subsidiary of the Corporation, and any employee benefit plan or employee stock plan of the Corporation, or any trust or other entity organized, established or holding Common Shares by, for or pursuant to, the terms of any such plan.

3. Article VI of the Company's Amended and Restated Articles of Incorporation is amended in its entirety to read as follows:

Notwithstanding any contrary provision of these Articles of Incorporation, any amendment or repeal of paragraph (4) of Article IV, this Article VI or any amendment to these Articles of Incorporation providing for a classified board shall require the affirmative vote of shares representing not less than 66 2/3% of the combined voting power of the outstanding Common Shares and, unless otherwise provided in the certificate of determination, the New Preferred Shares, voting together as a single class.

Notwithstanding any contrary provision of these Articles of Incorporation, and except as otherwise expressly provided in the California Corporations Code, none of the following provisions of the Bylaws of the Company may be amended or repealed, except by a majority of the Board or by the shareholders upon the affirmative vote of shares representing at least 66 2/3% of the combined voting power of the outstanding Common Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares, voting together as a single class: (a) Section 2 of Article II, (b) Section 15 of Article II, and (c) Section 2 of Article III.

4. On the effective date of the filing of this Amendment to the Amended and Restated Articles of Incorporation, the Company's Preferred Stock will be eliminated in accordance with Section 510(b)(2)(A) of the California General Corporation Law. The Company has only shares of Common Shares outstanding, and there will be no effect on the outstanding Common Shares.

5. The foregoing amendments of the Amended and Restated Articles of Incorporation have been duly approved by the Board of Directors of the Company alone in accordance with Section 510(b)(2)(A) of the California General Corporation Law.

We further declare under penalty of perjury under the laws of the State of California that the matters set forth in this certificate are true and correct of our own knowledge.

DATED: April 29, 2002

/s/\_\_\_\_\_ Floyd E. Wicks

/s/\_\_\_\_\_\_ McClellan Harris III PWWT Contract No. 85502/65504/65505/65506/6507

[LOGO PINNACLE WEST CAPITAL CORPORATION] Ken Wolf [RE]
Marketing & Trading
P.O. Box 53999, M/S 9842
Phoenix, Arizona 85072-399
Telephone: (602) 250-4313
Facsimile: (602) 250-3199

September 4, 2002

### CONFIDENTIAL

To: Southern California Water Company

The following terms and conditions shall govern this agreement on September 3, 2002, between David Kolk, on behalf of Southern California Water Company ("SCWC"), and Ken Wolf, on behalf of Pinnacle West Marketing & Trading ("PWMT"), as follows:

Purchaser: Pinnacle West Capital Corporation	Seller: Southern California Water Company
400 N. 5th Street, M/S 9842	630 E. Foothills Blvd.
Phoenix, Arizona 55004	San Dimas, CA 91773-9016
Confirm	Confirm
Administrator: Tricie Henry	Administrator: Raymond P. Juels
(602) 250-2911 (phone)	(909) 394-3800 x 579 (phone)
(602) 371-5256 (fax)	(909) 394-3758 (Fax)
Preschedule: (602) 250-4371	Preschedule: (909) 280-9411
Real Time: (602) 250-4470	Real Time: (909) 280-9411
Price (S/MWh): \$95.00/MWh	Quantity (MWh): 547.920 MWh
Quantity (MWh): 15 Magawatts	Type of energy: CAISO Firm
Start date: November 1, 2002	End date: December 31, 2006
Day(s) of week: Monday through Sunday	Hours: H.E. 0100-2400 Pacific
including NERC holidays	Prevailing time ("PPT")
<pre>Price (S/MWh): \$48.00/MWh Quantity (MWh): 8 Magawatts Start date: November 1, 2002 Day(s) of week: Monday through Sunday including NERC holidays Price (S/MWh): \$36.00/MWh Quantity (MWh): 8 Magawatts Start date: November 1, 2003 Day(s) of week: Monday through Sunday including NERC holidays</pre>	Quantity (MWh): 28.992 MWh Type of energy: CAISO Firm End date: March 31, 2003 Hours: H.E. 0100-2400 Pacific Prevailing time ("PPT") Quantity (MWh): 28.184 MWh Type of energy: CAISO Firm End date: March 31, 2004 Hours: H.E. 0100-2400 Pacific Prevailing time ("PPT")
Delivery Point: SP15 (Inter SC Trade) Transmission Contingencies: None Ge	

Page 1 of 4

And

400	aacle West Capital Corporation N. 5th Street, M/S 9842 nix, Arizona 55004	630 1	nern California Water Company 3. Foothills Blvd. Dimas, CA 91773-9016
Confirm Administrator:	Tricie Henry (602) 250-2911 (phone) (602) 371-5256 (fax)	Confirm Administrator:	Raymond P. Juels (909) 394-3800 x 579 (phone) (909) 394-3758 (Fax)
Real Time:	(602) 250-4371 (602) 250-4470	Real Time:	
Price (S/MWh): Quantity (MWh): Start date:	\$74.65/MWh	Quantity (MWh): Type of energy: End date:	811.080 MWh
Quantity (MWh): Start date:	8 Magawatts November 1, 2002 Winter Periods	Type of energy: End date:	
	Period 1 - November 1, 2002 Period 2 - November 1, 2003 Period 3 - November 1, 2004 Period 4 - November 1, 2005 Period 5 - November 1, 2006 Period 6 - November 1, 2007 Period 7 - November 1, 2008		March 31, 2003 March 31, 2004 March 31, 2005 March 31, 2006 March 31, 2007 March 31, 2008 December 31, 2008
Day(s) of week:	Monday through Sunday including NERC holidays	Hours:	H.E. 0100-2400 Pacific Prevailing time ("PPT")

\_\_\_\_\_

Delivery Point: SP15 (Inter SC Trade)	
Transmission Contingencies: None	Generation Contingencies: None

PWMT end SCWC enter mm this Confirmation Agreement (hereinafter the "Confirmation") pursuant to and in accordance with the Western Systems Power Pool Agreement, effective as of July 27, 1991 and as further amended on September 1, 2002 (hereinafter the "Agreement"), to which Pinnacle West Capital Corporation and Southern California Water Company are Parties, Terms used but not defined herein shall have the meanings set forth in the Agreement.

In consideration of the premises and the agreements contained herein, the Parties agree as follows:

 PWMT shall supply firm energy in accordance with Service Schedule C of the Agreement utilizing available generation or purchased power resources at the point of delivery. If, in order to maintain firm energy deliveries. PWMT is required to obtain additional generation or transmission resources, PWMT shall absorb all additions costs incurred, Including any charges for generation, transmission or ancillary services.

Page 2 of 4

2. Section C-3M (c) of Service Schedule C is deleted in its entirety and replaced with the following:

"where applicable, to meet Seller's or its Affiliate's public utility or statutory obligations to its customers"

"Affiliate" means, with respect to Seller, any other entity that, directly or indirectly, through one or more intermediaries, controls, or is controlled by, or is under common control with such entity. For this purpose, "control" means the dir3ct or indirect ownership of fifty percent (50%) or more of the outstanding capital stock or other equity interests having ordinary voting power.

- 3. Deliveries will be mad except during interruptions or reductions which are due to uncontrollable forces as defined in Section 10 of the Agreement, in which case the obligations of both Parties will be reduced for the duraiton of the interruption or reduction.
- 4. Preschedules shall be exchanged for all deliveries of energy, including indentifications of receiving and generating control areas under this Confirmation by 11:00 a.m. Pacific prevailing Time on the last work day observed by both Parties prior to the scheduled date of delivery. Interchange scheduling shall be conducted in accordance with WSPP Operating Procedure No. 1.
- Neither Party shall transfer or assign all or any part of the Agreement or 5. its rights or obligations hereunder or otherwise dispose of any right, title or interest herein without the prior written consent of the other Party, which consent shall not be unreasonably withheld or delayed. Notwithstanding the foregoing, either Party may, without the need for consent from the other Party, (a) transfer, pledge, or assign the Agreement as security for any financing, (b) transfer, assign or delegate the Agreement to any person or entity succeeding to all or substantially all of the assets of such party;p provided, however, that any such assignee shall agree to be bound by the terms and conditions hereof and, provided further, that any transfer, assignment or delegation that does not require consent hereunder shall not, in any way, release the assignor from liability for full performance of any obligations (and only those obligations) arising under the Agreement prior to the effective date of the transfer, assignment or delegation. To the extent a transfer does not required consent, the transferring Party shall provide prompt notice to the other Party of the transfer and the effect date thereof. Any transfer in violation of this section shall be deemed null and void. "Affiliate" means, with respect to a party, any other entity that, directly or indirectly, through one or more intermediaries, controls, or is controlled by, or is under common control with such entity. For this purpose, control" means the direct or indirect ownership of fifty percent (50%) or more of the outstanding capital stock or other equity interests having ordinary voting power.
- 6. "CAISO" Firm Energy" means with respect to a Transaction, a product under which the Seller shall sell and the FUrchaser shall purchase a quantity of energy equal to the hourly quantity without Ancillary Services (as defined in the Tariff) that is or will be scheduled as a schedule coordinator to schedule coordinator transaction pursuant to the applicable tariff and protocol provisions of the California Independent System Operator ("CAISO") (as amended from time to time, the "Tariff") for which the only excuse for failure to deliver or receive is an "Uncontrollable Force". A CAISO "Schedule Adjustment" (defined as a schedule change implement by the CAISO that is neither caused by, or within the control of, either Party) shall not constitute an Uncommtrollable Force.

If the above accurately reflects the terms and conditions of the agreement between PWMT and SCWC on September 3, 2002, please sign a copy of this Confirmation and return it via fax to (602) 371-5256 the PWMT Confirm Administrator listed above.

Page 3 of 4

Counterparty's failure to notify PWMT of any exception to these terms or, the failure to return an executed Confirmation within five (5) business days of receipt of this Confirmation will constitute Counterparty's agreement to the terms set forth herein.

# PINNACLEWEST CAPITAL CORPORATION Signature: /s/ Steven McAdams

## SOUTHERN CALIFORNIA WATER COMPANY

Print Name: Joel A. Dickson
Title: Vice President
Date: 9/11/02
-

WM&T Contracts Department By: [ILLEGIBLE] 9/5/02 -----

Page 4 of 4

\_\_\_\_\_

CREDIT AGREEMENT

Dated as of June 6, 2002

## among

AMERICAN STATES WATER COMPANY,

as Borrower,

THE LENDERS NAMED HEREIN

and

WELLS FARGO BANK, NATIONAL ASSOCIATION,

as Administrative Agent and Lead Arranger

-1

\_\_\_\_\_

## \$75,000,000 CREDIT FACILITIES

AMERICAN STATES WATER COMPANY

As Borrower

## WELLS FARGO BANK, NATIONAL ASSOCIATION

As Administrative Agent

June 6, 2002

## Prepared by

Sheppard, Mullin, Richter & Hampton LLP 333 South Hope Street, 48th Floor Los Angeles, California 90071 (213) 620-1780

\_\_\_\_\_

Our File Number

0794-089374

Julie Cravitz (213) 617-5493 jcravitz@sheppardmullin.com

July 3, 2002

VIA MESSENGER / FEDERAL EXPRESS

To the Parties on the Attached Distribution List

Re: Wells Fargo Bank - American States Water

Ladies and Gentlemen:

Enclosed please find the closing book for the credit facilities provided to American States Water Company, dated June 6, 2002, with Wells Fargo Bank, N.A., as Administrative Agent.

Also enclosed in the packages for each Lender are their original Note and fee letter as appropriate. And in the package for Anthony Turner, the Swing Line Note.

Please do not hesitate to call if you have any questions regarding the above, or if we can be of further assistance.

Sincerely,

/s/ Julie M. Cravitz

Julie M. Cravitz Legal Assistant to Mark A. Spitzer

Enclosures cc: Mark Spitzer, Esq. (w/o encl)

LOS ANGELES O SAN FRANCISCO O ORANGE COUNTY O SAN DIEGO O SANTA BARBARA O WEST LOS ANGELES O DEL MAR HEIGHTS

June 6, 2002

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Re: Wells Fargo Bank, National Association, as Administrative Agent/American States Water Company

This is one of the fee letters referred to in Section 3.3(a) of that certain Credit Agreement dated as of June 6, 2002 (the "Agreement"), executed by and among American States Water Company, a California corporation ("Borrower"), the Lenders party thereto from time to time, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders ("Agent"). Unless otherwise indicated, all initially capitalized terms used in this letter without definition shall have the respective meanings set forth in the Agreement.

Borrower agrees to pay on the Closing Date to Agent, for the account of Northern Trust Bank of California, N.A. ("Northern Trust"), an upfront fee of ..200% of Northern Trust's Pro Rata Share as of the Closing Date, which fee equals \$6,000.00.

Borrower acknowledges and agrees that we shall effect payment of the foregoing fee by charging the full amount thereof, when due, to the Revolving Facility.

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Anthony Turner Anthony Turner Vice President American States Water Company June 6, 2002 Page 2

Agreed:

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

- -----
- Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

June 6, 2002

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Re: Wells Fargo Bank, National Association, as Administrative Agent/American States Water Company

This is one of the fee letters referred to in Section 3.3(a) of that certain Credit Agreement dated as of June 6, 2002 (the "Agreement"), executed by and among American States Water Company, a California corporation ("Borrower"), the Lenders party thereto from time to time, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders ("Agent"). Unless otherwise indicated, all initially capitalized terms used in this letter without definition shall have the respective meanings set forth in the Agreement.

Borrower agrees to pay on the Closing Date to Agent, for the account of Comerica West Incorporated ("Comerica"), an upfront fee of .200% of Comerica's Pro Rata Share as of the Closing Date, which fee equals \$20,000.00.

Borrower acknowledges and agrees that we shall effect payment of the foregoing fee by charging the full amount thereof, when due, to the Revolving Facility.

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Anthony Turner

Anthony Turner Vice President American States Water Company June 6, 2002 Page 2

Agreed:

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

- -----
- Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

June 6, 2002

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Re: Wells Fargo Bank, National Association, as Administrative Agent/American States Water Company

This is one of the fee letters referred to in Section 3.3(a) of that certain Credit Agreement dated as of June 6, 2002 (the "Agreement"), executed by and among American States Water Company, a California corporation ("Borrower"), the Lenders party thereto from time to time, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders ("Agent"). Unless otherwise indicated, all initially capitalized terms used in this letter without definition shall have the respective meanings set forth in the Agreement.

Borrower agrees to pay on the Closing Date to Agent, for the account of Union Bank of California, N.A. ("UBOC"), an upfront fee of .250% of UBOC's Pro Rata Share as of the Closing Date, which fee equals \$37,500.00.

Borrower acknowledges and agrees that we shall effect payment of the foregoing fee by charging the full amount thereof, when due, to the Revolving Facility.

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Anthony Turner

Anthony Turner Vice President American States Water Company June 6, 2002 Page 2

Agreed:

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

- -----
- Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

Wells Fargo Bank, National Association 333 South Grand Avenue, Third Floor Los Angeles, California 90071

June 6, 2002

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Re: Wells Fargo Bank, National Association, as Administrative Agent/American States Water Company

This is one of the fee letters referred to in Section 3.3(a) of that certain Credit Agreement dated as of June 6, 2002 (the "Agreement"), executed by and among American States Water Company, a California corporation ("Borrower"), the Lenders party thereto from time to time, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders ("Agent"). Unless otherwise indicated, all initially capitalized terms used in this letter without definition shall have the respective meanings set forth in the Agreement.

Borrower agrees to pay on the Closing Date to Agent, for the account of CoBank, ACB ("CoBank"), an upfront fee of .300% of CoBank's Pro Rata Share as of the Closing Date, which fee equals \$66,000.00.

Borrower acknowledges and agrees that we shall effect payment of the foregoing fee by charging the full amount thereof, when due, to the Revolving Facility.

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Anthony Turner

Anthony Turner Vice President American States Water Company June 6, 2002 Page 2

Agreed:

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

- -----
- Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

June 6, 2002

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Re: Wells Fargo Bank, National Association, as Administrative Agent/American States Water Company

This is one of the fee letters referred to in Section 3.3(a) of that certain Credit Agreement dated as of June 6, 2002 (the "Agreement"), executed by and among American States Water Company, a California corporation ("Borrower"), the Lenders party thereto from time to time, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders ("Wells Fargo" and in its capacity as the Administrative Agent for the Lenders, the "Agent"). Unless otherwise indicated, all initially capitalized terms used in this letter without definition shall have the respective meanings set forth in the Agreement.

Borrower agrees to pay on the Closing Date to Agent, for the account of Wells Fargo, an upfront fee of .325% of Wells Fargo's Pro Rata Share as of the Closing Date, which fee equals \$81,250.00.

Borrower acknowledges and agrees that we shall effect payment of the foregoing fee by charging the full amount thereof, when due, to the Revolving Facility.

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Anthony Turner

Anthony Turner Vice President American States Water Company June 6, 2002 Page 2

Agreed:

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

- -----
- Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

June 6, 2002

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016

Re: Wells Fargo Bank, National Association, as Administrative Agent/American States Water Company

This is the fee letter referred to in Sections 3.3(b) and (c) of that certain Credit Agreement dated as of June 6, 2002 (the "Agreement"), executed by and among American States Water Company, a California corporation ("Borrower"), the Lenders party thereto from time to time, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders ("Administrative Agent"). This letter, and the related letters dated today executed by the parties regarding upfront fees payable to each lender, supersede the Fee Letter dated December 19, 2001 executed by the parties, and the parties agree that no fees, except for the portion of the Arrangement Fee paid prior to the date hereof that is referenced below, are payable pursuant to such Fee Letter. Unless otherwise indicated, all initially capitalized terms used in this letter without definition shall have the respective meanings set forth in the Agreement.

Borrower hereby agrees to pay the following fees to Administrative Agent with respect to the credit facility under the Credit Agreement, each of which shall be deemed fully-earned and shall be nonrefundable once paid:

Arrangement Fee. On the Closing Date, for the account of the Lead Arranger, an arrangement fee of \$65,000. Administrative Agent acknowledges that it has heretofore received, on behalf of the Lead Arranger, \$25,000 toward the payment of such Arrangement Fee.

Agency Fee. On the Closing Date (and on each anniversary thereof), for the account of the Administrative Agent, an annual agency fee equal to \$25,000.

Borrower acknowledges and agrees that we shall effect payment of the foregoing fees by charging the full amount thereof, when due, to Borrower's Revolving Facility under the Credit Agreement.

American States Water Company June 6, 2002 Page 2

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By: /s/ Anthony Turner Anthony Turner Vice President

Agreed:

AMERICAN STATES WATER COMPANY

By:

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary American States Water Company June 6, 2002 Page 2

Please countersign below to confirm the foregoing.

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

Ву:\_\_\_\_ Anthony Turner Vice President

Agreed:

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

\_\_\_\_\_ Name: McClellan Harris III

Title: Vice President-Finance, Treasurer and Corporate Secretary

Page

	DEFINITIONS AND ACCOUNTING TERMS
1.1	Defined Terms1
1.2	Use of Defined Terms
1.3	Accounting Terms; Covenant Calculations
1.4	Rounding
1.5	Exhibits and Schedules27
1.6	References to "Borrower and its Subsidiaries"
1.7	Miscellaneous Terms
Article 2. A	ADVANCES AND LETTERS OF CREDIT
2.1	Advances-General
2.2	Alternate Base Rate Advances
2.3	Eurodollar Rate Advances
2.4	Conversion and Continuation of Advances
(a)	Optional Conversion
(b)	Certain Mandatory Conversions
(c)	Continuations
2.5	Letters of Credit
2.6	Termination or Reduction of the Commitments
(a)	Optional
	Mandatory
( - )	Reduction Pro Rata: No Reinstatements
2.7	Administrative Agent's Right to Assume Funds Available for Advances
2.8	Swing Line
Article 3. H	2AYMENTS AND FEES
3.1	Principal and Interest
3.2	Unused Revolving Facility Commitment Fee40
3.3	Closing Fees; Arrangement Fee; Agency Fee etc40
3.4	Letter of Credit Fees
3.5	Increased Commitment Costs
3.6	Eurodollar Costs and Related Matters42
3.7	Late Payments and Default Rate45
3.8	Computation of Interest and Fees45
3.9	Non-Banking Days
3.10	Manner and Treatment of Payments
3.11	Funding Sources
3.12	Failure to Charge Not Subsequent Waiver
3.13	Administrative Agent's Right to Assume Payments Will be Made
3.14	Fee Determination Detail
3.15	Survivability
0.10	

-i

	REPRESENTATIONS AND WARRANTIES
4.1	Existence and Qualification; Power; Compliance With Laws
4.2	Authority; Compliance With Other Agreements and Instruments and Government Regulations49
4.3	No Governmental Approvals Required
4.4	Subsidiaries
4.5	Financial Statements
4.6	No Other Liabilities; No Material Adverse Changes
4.7	Title to and Location of Property
4.8	Intangible Assets
4.9	Litigation
4.10	Binding Obligations
4.11	No Default
4.12	ERISA
4.13	Regulation U; Investment Company Act53
4.14	Disclosure
4.15	Tax Liability
4.16	Projections
4.17	Hazardous Materials
4.18	Employee Matters
4.19	Fiscal Year54
4.20	Solvency
Article 5.	AFFIRMATIVE COVENANTS (OTHER THAN INFORMATION AND REPORTING REQUIREMENTS)
5.1	Payment of Taxes and Other Potential Liens
5.2	Preservation of Existence
5.3	Maintenance of Properties
5.4	Maintenance of Insurance
5.5	Compliance With Laws
5.6	Inspection Rights
5.7	Keeping of Records and Books of Account
5.8	Compliance With Agreements
5.9	Use of Proceeds
5.10	Hazardous Materials Laws
5.11	Minimum Debt Rating
5.12	Syndication Process
Duti dalla d	
	NEGATIVE COVENANTS
6.1	Prepayment of Indebtedness
6.2	Prepayment of Subordinated Obligations
6.3	Disposition of Property
6.4	Mergers
6.5	Hostile Tender Offers
6.6	Distributions
6.7	ERISA

-ii

6.8	Change in Nature of Business
6.9	Liens and Negative Pledges
6.10	Indebtedness and Guaranty Obligations60
6.11	Transactions with Affiliates60
6.12	Total Funded Debt Ratio60
6.13	Interest Coverage Ratio
6.14	Investments and Acquisitions61
6.15	Operating Leases
6.16	Amendments
6.17	Use of Lender's Name
6.18	Change of Fiscal Periods62
Article 7.	INFORMATION AND REPORTING REQUIREMENTS
7.1	Financial and Business Information63
7.2	Compliance Certificates
Article 8.	CONDITIONS
8.1	Initial Advances
8.2	Any Advance
	-
Article 9.	EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT
9.1	Events of Default
9.2	Remedies Upon Event of Default
Article 10.	THE ADMINISTRATIVE AGENT
10.1	Appointment and Authorization74
10.2	Administrative Agent and Affiliates74
10.3	Proportionate Interest in any Collateral74
10.4	Lenders' Credit Decisions
10.5	Action by Administrative Agent75
10.6	Liability of Administrative Agent
10.7	Indemnification
10.8	Successor Administrative Agent
10.9	No Obligations of Borrower
Article 11	MISCELLANEOUS
11.1	Cumulative Remedies: No Waiver
11.2	Amendments: Consents
11.3	Costs and Expenses
11.4	Nature of Lenders' Obligations
11.5	Survival of Representations and Warranties
11.6	Notices
11.7	Execution of Loan Documents
11.8	Bioding Effect; Assignment
11.9	Right of Setoff
11.10	Sharing of Setoffs
±±•±0	Sharing of Second

-iii

11.11	Indemnity by Borrower	
11.12	Nonliability of the Lenders	
11.13	No Third Parties Benefited	
11.14	Confidentiality	
11.15	Further Assurances	
11.16	Integration	88
11.17	Governing Law	88
11.18	Severability of Provisions	
11.19	Headings	
11.20	Time of the Essence	
11.21	Foreign Lenders and Participants	
11.22	Waiver of Right to Trial by Jury	
11.23	Purported Oral Amendments	

# Exhibits

- A Assignment and Acceptance
   B Compliance Certificate
   C Letter of Credit Agreement
   D Note
- E Opinion of Counsel
  F Request for Borrowing
- G Request for Continuation/Conversion
   H Request for Letter of Credit

Schedules

- 1.1 1.2 Lender Commitments/Pro Rata Shares
- Existing Wells Fargo Bank L/Cs
- 1.3 Material Contracts
- 4.4 4.7 Subsidiaries
- Existing Liens, Negative Pledges and Rights of Others
- 4.8 Intangible Assets
- 4.9 Material Litigation
- Hazardous Materials Matters
- 4.17 6.14 Existing Investments

-iv

## CREDIT AGREEMENT

### Dated as of June 6, 2002

This CREDIT AGREEMENT (as amended, supplemented or otherwise modified from time to time, this "Agreement") is entered into by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), each lender whose name is set forth on the signature pages of this Agreement and each lender that may hereafter become a party to this Agreement pursuant to Section 11.8 (each a "Lender" and collectively, "Lenders"), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent and Lead Arranger.

In consideration of the mutual covenants and agreements herein contained, the parties hereto covenant and agree as follows:

#### Article 1. DEFINITIONS AND ACCOUNTING TERMS

1.1 Defined Terms. As used in this Agreement, the following terms shall have the meanings set forth below:

"Acquired Person" means (a) any Person that is the subject of an Acquisition after the Closing Date and (b) any assets constituting a discrete business or operation unit that is the subject of an Acquisition after the Closing Date.

"Acquisition" means any transaction, or any series of related transactions, consummated after the Closing Date, by which Borrower or any of its Subsidiaries directly or indirectly (a) acquires any ongoing business or all or substantially all of the assets of any firm, partnership, joint venture, limited liability company, corporation or division thereof, whether through purchase of assets, merger or otherwise, (b) acquires in one transaction or as the most recent transaction in a series of transactions control of Securities of a Person engaged in an ongoing business representing more than 50% of the ordinary voting power for the election of directors or other governing position if the business affairs of such Person are managed by a board of directors or other governing body or (c) acquires control of more than 50% of the ownership interest in any partnership, joint venture, limited liability company, business trust or other Person that is not managed by a board of directors or other governing body.

"Administrative Agent" means Wells Fargo when acting in its capacity as the Administrative Agent under any of the Loan Documents, or any successor Administrative Agent.

"Administrative Agent's Office" means the Administrative Agent's address as set forth on the signature pages of this Agreement, or such other address as the Administrative Agent hereafter may designate by written notice to Borrower and the Lenders.

-1-

"Advance" means any advance made or to be made by any Lender to Borrower as provided in Section 2.1(a).

"Affiliate" means, as to any Person, any other Person which directly or indirectly controls, or is under common control with, or is controlled by, such Person. As used in this definition, "control" (and the correlative terms, "controlled by" and "under common control with") shall mean possession, directly or indirectly, of power to direct or cause the direction of management or policies (whether through ownership of Securities or partnership or other ownership interests, by contract or otherwise); provided that, in any event, any Person that owns, directly or indirectly, 10% or more of the Securities having ordinary voting power for the election of directors or other governing body of a corporation that has more than 100 record holders of such Securities, or 10% or more of the partnership or other ownership interests, will be deemed to be an Affiliate of such corporation, partnership or other Person.

"Aggregate Effective Amount" means, as of any date of determination and with respect to all Letters of Credit then outstanding, the sum of (a) the aggregate effective face amounts of all such Letters of Credit not then paid by Issuing Lender plus (b) the aggregate amounts paid by Issuing Lender under such Letters of Credit not then reimbursed to Issuing Lender by Borrower pursuant to Section 2.5(d) and not the subject of one or more Advances made pursuant to Section 2.5(e) or (f).

"Alternate Base Rate" means, as of any date of determination, the rate per annum (rounded upwards, if necessary, to the next 1/100 of 1%) equal to the higher of (a) the Prime Rate in effect on such date and (b) the Federal Funds Rate in effect on such date plus 1/2 of 1% (50 basis points).

"Alternate Base Rate Advance" means an Advance that bears interest in relation to the Alternate Base Rate as provided in Section  $3.1\,(b)\,.$ 

"Applicable Alternate Base Rate Margin" means, with respect to any Alternate Base Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

-2-

Applicable Pricing	
Level	Margin
I	0
II	0
III	0
IV	0

"Applicable Commitment Fee Margin" means, for each Pricing Period, the margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	10.0
II	12.5
III	15.0
IV	17.5

"Applicable Eurodollar Rate Margin" means, with respect to any Eurodollar Rate Advance, for each Pricing Period, the interest rate margin set forth below (expressed in basis points per annum) opposite the Applicable Pricing Level for that Pricing Period:

Applicable Pricing	
Level	Margin
I	50.0
II	62.5
III	75.0
IV	87.5

"Applicable Letter of Credit Fee Rate" means, as of any date of determination, the then effective Applicable Eurodollar Rate Margin.

-3-

"Applicable Pricing Level" means, for each Pricing Period the pricing level set forth below opposite the Debt Rating achieved by Borrower as of the first day of that Pricing Period:

Debt Rating		
At least A1 or A+		
A2 or A		
A3 or A-		
Baal or BBB+		

provided that in the event that the then prevailing Debt Ratings are "split ratings", Borrower will receive the benefit of the higher Debt Rating, unless the split is a "double split rating" (in which case the intermediate pricing level will apply) or a "triple split rating" (in which case the pricing level above that applicable to the lowest Debt Rating will apply).

"Assignment and Acceptance" means an assignment and acceptance agreement substantially in the form of Exhibit A.

"Banking Day" means any Monday, Tuesday, Wednesday, Thursday or Friday, other than a day on which banks are authorized or required to be closed in California or New York.

"Borrowing" means a borrowing consisting of simultaneous Advances of the same type.

"Capital Expenditure" means any expenditure that is treated as a capital expenditure under GAAP, including any expenditure that is required to be capitalized in accordance with GAAP that relates to an asset subject to a Capital Lease.

"Capital Lease" means, as to any Person, a lease of any Property by that Person as lessee that is, or should be in accordance with GAAP (including Financial Accounting Standards Board Statement No. 13, as amended or superseded from time to time), recorded as a "capital lease" on the balance sheet of that Person prepared in accordance with GAAP.

"Capital Lease Obligations" means all monetary obligations of a Person under any Capital Lease.

"Cash" means, when used in connection with any Person, all monetary and non-monetary items owned by that Person that are treated as cash in accordance with GAAP.

-4-

"Cash Equivalents" means, when used in connection with any Person, that Person's Investments in:

(a) Government Securities due within one year after the date of the making of the Investment;

(b) readily marketable direct obligations of any State of the United States of America or any political subdivision of any such State or any public agency or instrumentality thereof given on the date of such Investment a credit rating of at least Aa by Moody's or AA by S&P, in each case due within one year from the making of the Investment;

(c) certificates of deposit issued by, bank deposits in, eurodollar deposits through, bankers' acceptances of, and repurchase agreements covering Government Securities executed by any Lender or any bank incorporated under the Laws of the United States of America, any State thereof or the District of Columbia and having on the date of such Investment combined capital, surplus and undivided profits of at least \$250,000,000, or total assets of at least \$5,000,000, in each case due within one year after the date of the making of the Investment;

(d) certificates of deposit issued by, bank deposits in, eurodollar deposits through, bankers' acceptances of, and repurchase agreements covering Government Securities executed by any Lender or any branch or office located in the United States of America of a bank incorporated under the Laws of any jurisdiction outside the United States of America having on the date of such Investment combined capital, surplus and undivided profits of at least \$500,000,000, or total assets of at least \$15,000,000,000, in each case due within one year after the date of the making of the Investment;

(e) repurchase agreements covering Government Securities executed by a broker or dealer registered under Section 15(b) of the Securities Exchange Act of 1934, as amended, having on the date of the Investment capital of at least \$50,000,000, due within 90 days after the date of the making of the Investment; provided that the maker of the Investment receives written confirmation of the transfer to it of record ownership of the Government Securities or the books of a "primary dealer" in such Government Securities or on the books of such registered broker or dealer, as soon as practicable after the making of the Investment;

(f) readily marketable commercial paper or other debt Securities issued by corporations doing business in and incorporated under the Laws of the United States of America or any State thereof or of any corporation that is the holding company for a bank described in clause (c) or (d) above given on the date of such Investment a credit rating of at least P-1 by Moody's or A-1 by S&P, in each case due within one year after the date of the making of the Investment;

-5-

(g) "money market preferred stock" issued by a corporation incorporated under the Laws of the United States of America or any State thereof (i) given on the date of such Investment a credit rating of at least Aa by Moody's and AA by S&P, in each case having an investment period not exceeding 50 days or (ii) to the extent that investors therein have the benefit of a standby letter of credit issued by Lender or a bank described in clauses (c) or (d) above; provided that (y) the amount of all such Investments issued by the same issuer does not exceed \$5,000,000 and (z) the aggregate amount of all such Investments does not exceed \$10,000,000;

(h) a readily redeemable "money market mutual fund" sponsored by a bank described in clause (c) or (d) hereof, or a registered broker or dealer described in clause (e) hereof, that has and maintains an investment policy limiting its investments primarily to instruments of the types described in clauses (a) through (g) hereof and given on the date of such Investment a credit rating of at least Aa by Moody's and AA by S&P; and

(i) corporate notes or bonds having an original term to maturity of not more than one year issued by a corporation incorporated under the Laws of the United States of America, or a participation interest therein; provided that (i) commercial paper issued by such corporation is given on the date of such Investment a credit rating of at least Aa by Moody's and AA by S&P, (ii) the amount of all such Investments issued by the same issuer does not exceed \$5,000,000 and (iii) the aggregate amount of all such Investments does not exceed \$10,000,000.

"Certificate" means a certificate signed by a Senior Officer or Responsible Official (as applicable) of the Person providing the certificate.

"Change in Control" means any of the following events: (a) the sale, lease, transfer or other disposition (other than by way of merger or consolidation), in one or a series of related transactions, of all or substantially all of the assets of Borrower and its Subsidiaries taken as a whole to any "person" or "group" (within the meaning of Sections 13(d) and 14(d)(2) of the Securities Exchange Act), (b) Borrower shall fail to own, directly or indirectly, 100% of the outstanding capital stock or other equity interests of any Closing Date Subsidiary, (c) any Person or two or more Persons acting in concert shall have acquired beneficial ownership, directly or indirectly, of, or shall have acquired by contract or otherwise, or shall have entered into a contract or arrangement that, upon consummation, will result in its or their acquisition of, control over, 20% or more of the capital stock or other equity interests of Borrower, (d) during any period of up to 24 consecutive months, commencing after the Closing Date, individuals who at the beginning of such 24-month period were directors of Borrower (together with any new director whose election by Borrower's board of directors or whose nomination for election by Borrower's shareholders was approved by a vote of at least two-thirds of the directors then still in office who either were directors at the beginning of such period or whose election or nomination for election was previously so approved) cease for any reason to constitute a majority of the directors of Borrower then in office or (e) any transaction or series of related transactions constituting a

-6-

"change in control" or similar occurrence under documentation evidencing or governing Indebtedness of Borrower and/or any of its Subsidiaries of \$1,000,000 or more, which gives the holder(s) of such Indebtedness the right to accelerate or otherwise require payment of such Indebtedness prior to the maturity date thereof. As used herein, "beneficial ownership" shall have the meaning provided in Rule 13d-3 of the Securities and Exchange Commission under the Securities Exchange Act.

"Closing Date" means the time and Banking Day on which the conditions set forth in Section 8.1 are satisfied or waived. The Administrative Agent shall notify Borrower and the Closing Date Lenders of the date that is the Closing Date.

"Closing Date Lenders" means Wells Fargo, CoBank, ACB, Union Bank of California, N.A., Comerica West Incorporated, Northern Trust Bank of California N.A., and any other lender party to this Agreement as of the Closing Date.

"Closing Date Subsidiaries" means SCW, American States Utility Services, Inc., a California corporation, and Chaparral City Water Company, an Arizona corporation.

"Code" means the Internal Revenue Code of 1986, as amended or replaced and as in effect from time to time.

"Commitment" means with respect to each Lender, the commitment of such Lender to make Advances (expressed as the maximum aggregate amount of the Advances to be made by such Lender hereunder), as such commitment may be (a) reduced from time to time pursuant to Section 2.6 and (b) reduced or increased from time to time pursuant to assignments by or to such Lender pursuant to Section 11.8. The initial amount of each Lender's Commitment is set forth on Schedule 1.1 or in the Assignment and Acceptance pursuant to which such Lender shall have assumed its Commitment, as applicable. The initial aggregate amount of the Lenders' Commitments is \$75,000,000.

"Compliance Certificate" means a certificate in the form of Exhibit B, properly completed and signed by the president or chief financial officer of Borrower.

"Continuation," "Continue" and "Continued" each refers to a continuation of Eurodollar Rate Advances from one Eurodollar Period to the next Eurodollar Period pursuant to Section 2.4(c).

"Contractual Obligation" means, as to any Person, any provision of any outstanding Security issued by that Person or of any material agreement, instrument or undertaking to which that Person is a party or by which it or any of its Property is bound.

"Conversion," "Convert" and "Converted" each refers to a conversion of Advances of one Type into Advances of the other Type pursuant to Section 2.4(a) or 2.4(b).

-7-

"Debtor Relief Laws" means the Bankruptcy Code of the United States of America, as amended from time to time, and all other applicable liquidation, conservatorship, bankruptcy, moratorium, rearrangement, receivership, insolvency, reorganization, or similar debtor relief Laws from time to time in effect affecting the rights of creditors generally.

"Debt Ratings" means, as of each date of determination, (a) in the event such a credit rating is issued by either Moody's or S&P, the bank debt credit rating assigned to the Indebtedness evidenced by this Agreement by that credit reporting agency, or (b) if no bank debt credit rating is assigned, the most creditworthy credit rating, actual or implicit, assigned to senior unsecured Indebtedness of Borrower by that credit rating agency.

"Default" means any event that, with the giving of any applicable notice or passage of time specified in Section 9.1, or both, would be an Event of Default.

"Default Rate" means the interest rate prescribed in Section 3.7.

"Designated Deposit Account" means a deposit account to be maintained by Borrower with Wells Fargo or one of its Affiliates, as from time to time designated by Borrower by written notification to the Administrative Agent.

"Designated Eurodollar Market" means, with respect to any Eurodollar Rate Advance, the London Eurodollar Market.

"Disqualified Stock" means any capital stock, warrants, options or other rights to acquire capital stock (but excluding any debt Security which is convertible, or exchangeable, for capital stock), which, by its terms (or by the terms of any Security into which it is convertible or for which it is exchangeable), or upon the happening of any event, matures or is mandatorily redeemable, pursuant to a sinking fund obligation or otherwise, or is redeemable at the option of the holder thereof, in whole or in part, on or prior to the Maturity Date.

"Disposition" means the sale, transfer or other disposition (each, a "Transfer") in any single transaction or series of related transactions of any asset, or group of related assets, of Borrower or any Subsidiary other than (a) a Transfer of Cash, Cash Equivalents, Investments (other than Investments in a Subsidiary), Inventory or other assets sold or otherwise disposed of in the ordinary course of business of Borrower or any Subsidiary, (b) a Transfer of equipment sold or otherwise disposed of where substantially similar equipment in replacement thereof has theretofore been acquired, or thereafter within 90 days is acquired, by Borrower or any Subsidiary and (c) a Transfer of obsolete assets no longer useful in the business of Borrower or any Subsidiary whose carrying value on the books of Borrower or such Subsidiary is less than \$1,000,000 and (d) a Transfer to Borrower or a wholly-owned Subsidiary of Borrower.

-8-

"Distribution" means, with respect to any equity interest or Security issued by a Person, or any warrant or right to acquire any equity interest or Security of a Person, (a) the retirement, redemption, purchase, or other acquisition for value by such Person of any such equity interest or Security, (b) the declaration or (without duplication) payment by such Person of any dividend in Cash or in Property (other than in common stock or an equivalent equity interest of such Person) on or with respect to any such equity interest or Security, (c) any Investment by such Person in the holder of any such equity interest or Security, and (d) any other payment by such Person constituting a distribution under applicable Laws with respect to such equity interest or Security.

"Dollars" or "\$" means United States of America dollars.

"EBITDA" means, with respect to any fiscal period, the sum of (a) Net Income for that period, plus (b) any extraordinary loss reflected in such Net Income, minus (c) any extraordinary gain reflected in such Net Income, plus (d) Interest Expense of Borrower and its Subsidiaries for that period, plus (e) the aggregate amount of federal and state taxes on or measured by income of Borrower and its Subsidiaries for that period (whether or not payable during that period), plus (f) depreciation and amortization expense of Borrower and its Subsidiaries for that period, plus (g) all other non-cash, extraordinary expenses of Borrower and its Subsidiaries for that period acceptable to the Requisite Lenders, in each case as determined in accordance with GAAP, consistently applied and, in the case of items (d), (e), (f), and (g) only to the extent reflected in the determination of Net Income for that period.

"Eligible Assignee" means (a) another Lender, (b) with respect to any Lender, any Affiliate of that Lender, (c) any commercial bank having total assets of \$250,000,000 or more, (d) any (i) savings bank, savings and loan association, finance company or similar financial institution or entity or (ii) insurance company engaged in the business of writing insurance which, in either case (A) has total assets of \$250,000,000 or more, (B) is engaged in the business of lending money and extending credit under credit facilities similar to those extended under this Agreement and (C) is operationally and procedurally able to meet the obligations of a Lender hereunder to the same degree as a commercial bank (as reasonably determined by the assigning Lender) and (e) any other financial institution (including a mutual fund or other fund) having total assets of \$250,000,000 or more which meets the requirements set forth in subclauses (B) and (C) of clause (d) above; provided that each Eligible Assignee must either (x) be organized under the Laws of the United States of America, any State thereof or the District of Columbia or (y) be organized under the Laws of the Cayman Islands or any country which is a member of the Organization for Economic Cooperation and Development, or a political subdivision of such a country, and (1) act hereunder through a branch, agency or funding office located in the United States of America and (2) be exempt from withholding of tax on interest and deliver the documents related thereto pursuant to Section 11.21.

-9-

"ERISA" means the Employee Retirement Income Security Act of 1974, and any regulations issued pursuant thereto, as amended or replaced and as in effect from time to time.

"ERISA Affiliate" means, with respect to any Person, any Person (or any trade or business, whether or not incorporated) that is under common control with that Person within the meaning of Section 414 of the Code. "Eurodollar Banking Day" means any Banking Day on which dealings in Dollar deposits are conducted by and among banks in the Designated Eurodollar Market.

"Eurodollar Base Rate" means with respect to any Eurodollar Rate Advance comprising part of the same Borrowing, the interest rate per annum (rounded upward, if necessary, to the nearest 1/100th of 1%) at which deposits in Dollars are offered by the Eurodollar Reference Lender to prime banks in the Designated Eurodollar Market at or about 10:00 a.m. local time in the Designated Eurodollar Market, two (2) Eurodollar Banking Days before the first day of the applicable Eurodollar Period in an aggregate amount approximately equal to the amount of the Advance to be made by the Eurodollar Reference Lender comprising part of such Borrowing and for a period of time comparable to the number of days in the applicable Eurodollar Period. The determination of the Eurodollar Base Rate by the Administrative Agent shall be conclusive in the absence of manifest error.

"Eurodollar Lending Office" means, as to each Lender, its office or branch so designated by written notice to Borrower and the Administrative Agent as its Eurodollar Lending Office. If no Eurodollar Lending Office is designated by a Lender, its Eurodollar Lending Office shall be its office at its address for purposes of notices hereunder.

"Eurodollar Market" means a regular established market located outside the United States of America by and among banks for the solicitation, offer and acceptance of Dollar deposits in such banks.

"Eurodollar Obligations" means eurocurrency liabilities, as defined in Regulation D or any comparable regulation of any Governmental Agency having jurisdiction over any Lender.

"Eurodollar Period" means, as to each Eurodollar Rate Advance comprising part of the same Borrowing, the period commencing on the date specified by Borrower pursuant to Section 2.1(b) and ending 1, 2, 3 or 6 months (or, if available to all Lenders, 9 or 12 months) thereafter, as specified by Borrower in the applicable Request for Borrowing or Request for Continuation/Conversion provided that:

(a) The first day of any Eurodollar Period shall be a Eurodollar Banking Day;

-10-

(b) Any Eurodollar Period that would otherwise end on a day that is not a Eurodollar Banking Day shall be extended to the immediately succeeding Eurodollar Banking Day unless such Eurodollar Banking Day falls in another calendar month, in which case such Eurodollar Period shall end on the immediately preceding Eurodollar Banking Day; and

(c) No Eurodollar Period for any Eurodollar Rate Advance shall extend beyond the Maturity Date.

"Eurodollar Rate" means, with respect to any Eurodollar Rate Advance comprising part of the same Borrowing, an interest rate per annum (rounded upward, if necessary, to the nearest 1/100th of one percent) determined pursuant to the following formula:

"Eurodollar Rate Advance" means an Advance that bears interest in relation to the Eurodollar Rate as provided in Section 3.1(c).

"Eurodollar Reference Lender" means Wells Fargo or the Administrative Agent if Wells Fargo is no longer the Administrative Agent.

"Eurodollar Reserve Percentage" means, with respect to any Eurodollar Rate Advance comprising part of the same Borrowing, the maximum reserve percentage (expressed as a decimal, rounded upward, if necessary, to the nearest 1/100th of one percent) in effect on the date the Eurodollar Base Rate for the Borrowing of which such Eurodollar Rate Advance is a part is determined (whether or not such reserve percentage is applicable to any Lender) under regulations issued from time to time by the Federal Reserve Board for determining the maximum reserve requirement (including any emergency, supplemental or other marginal reserve requirement) with respect to eurocurrency funding (currently referred to as "eurocurrency liabilities") having a term comparable to the Eurodollar Period for such Eurodollar Rate Advance. The determination by the Administrative Agent of any applicable Eurodollar Reserve Percentage shall be conclusive in the absence of manifest error.

"Event of Default" shall have the meaning provided in Section 9.1.

"Existing Credit Facilities" means, collectively, the credit facilities provided to SCW pursuant to a credit agreement with Union Bank of California, N.A. terminating on May 31, 2002; a credit agreement with Bank of America, N.A. terminating on July 31, 2002; and a credit agreement with Wells Fargo terminating on August 31, 2002 (the "Wells Fargo Existing Credit Facility").

-11-

"Existing Wells Fargo Bank L/Cs" means, collectively, the three (3) letters of credit issued by Wells Fargo for the account of SCW pursuant to the Wells Fargo Existing Credit Facility and further described in Schedule 1.2.

"Federal Funds Rate" means, as of any date of determination, the rate set forth in the weekly statistical release designated as H.15(519), or any successor publication, published by the Federal Reserve Board (including any such successor, "H.15(519)") for such date opposite the caption "Federal Funds (Effective)". If for any relevant date such rate is not yet published in H.15(519), the rate for such date will be the rate set forth in the daily statistical release designated as the Composite 3:30 p.m. Quotations for U.S. Government Securities, or any successor publication, published by the Federal Reserve Bank of New York (including any such successor, the "Composite 3:30 p.m. Quotation") for such date under the caption "Federal Funds Effective Rate". If on any relevant date the appropriate rate for such date is not yet published in either H.15(519) or the Composite 3:30 p.m. Quotations, the rate for such date will be the arithmetic mean of the rates for the last transaction in overnight Federal funds arranged prior to 9:00 a.m. (New York City time) on that date by each of three leading brokers of Federal funds transactions in New York City selected by the Administrative Agent. For purposes of this Agreement, any change in the Alternate Base Rate due to a change in the Federal Funds Rate shall be effective as of the opening of business on the effective date of such change.

"Fiscal Quarter" means any fiscal quarter of Borrower and its Subsidiaries ending on each March 31, June 30, September 30 and December 31.

"Fiscal Year" means the fiscal year of Borrower and its Subsidiaries ending on each December 31.

"GAAP" means, as of any date of determination, accounting principles (a) set forth as generally accepted in then currently effective Opinions of the Accounting Principles Board of the American Institute of Certified Public Accountants, (b) set forth as generally accepted in then currently effective Statements of the Financial Accounting Standards Board or (c) that are then approved by such other entity as may be approved by a significant segment of the accounting profession in the United States of America. The term "consistently applied," as used in connection therewith, means that the accounting principles applied are consistent in all material respects with those applied at prior dates or for prior periods.

"Government Securities" means readily marketable (a) direct full faith and credit obligations of the United States of America or obligations guaranteed by the full faith and credit of the United States of America and (b) obligations of an agency or instrumentality of, or corporation owned, controlled or sponsored by, the United States of America that are generally considered in the securities industry to be implicit obligations of the United States of America.

-12-

"Governmental Agency" means (a) any international, foreign, federal, state, county or municipal government, or political subdivision thereof, (b) any governmental or quasi-governmental agency, authority, board, bureau, commission, department, instrumentality or public body or (c) any court or administrative tribunal of competent jurisdiction.

"Guaranty Obligation" means, as to any Person, any (a) guarantee by that Person of Indebtedness of, or other obligation performable by, any other Person or (b) assurance given by that Person to an obligee of any other Person with respect to the performance of an obligation by, or the financial condition of, such other Person, whether direct, indirect or contingent, including any purchase or repurchase agreement covering such obligation or any collateral security therefor, any agreement to provide funds (by means of loans, capital contributions or otherwise) to such other Person, any agreement to support the solvency or level of any balance sheet item of such other Person or any "keep-well" or other arrangement of whatever nature given for the purpose of assuring or holding harmless such obligee against loss with respect to any obligation of such other Person; provided, however, that the term Guaranty Obligation shall not include endorsements of instruments for deposit or collection or similar arrangements in the ordinary course of business. The amount of any Guaranty Obligation in respect of Indebtedness shall be deemed to be an amount equal to the stated or determinable amount of the related Indebtedness (unless the Guaranty Obligation is limited by its terms to a lesser amount, in which case to the extent of such amount) or, if not stated or determinable, the maximum reasonably anticipated liability in respect thereof as determined by the Person in good faith. The amount of any other Guaranty Obligation shall be deemed to be zero unless and until the amount thereof has been (or in accordance with Financial Accounting Standards Board Statement No. 5 should be) quantified and reflected or disclosed in the consolidated financial statements (or notes thereto) of Borrower and its Subsidiaries.

"Hazardous Materials" means oil or petrochemical products, poly-chlorinated biphenyls, asbestos, urea formaldehyde, flammable explosives, radioactive materials, hazardous wastes, toxic substances or related materials, including any substances considered "hazardous substances," "hazardous wastes," "hazardous materials," "infectious wastes", "pollutant substances", "solid waste" or "toxic substances" under any Hazardous Materials Laws.

"Hazardous Materials Laws" means all Laws pertaining to the treatment, transportation or disposal of Hazardous Materials on or about any Real Property owned or leased by Borrower or any Subsidiary thereof, or any portion thereof, including without limitation the following: the Federal Water Pollution Control Act (33 U.S.C. Section 1251, et seq.), the Federal Resource Conservation and Recovery Act of 1976 (42 U.S.C. Section 6901, et seq.), the Comprehensive Environmental Response, Compensation and Liability Act of 1980, as amended (42 U.S.C. Section 9601, et seq.) and the Superfund Amendments and Reauthorization Act of 1986, the Hazardous Materials Transportation Act, as amended (44 U.S.C. Section 1801, et seq.), the Toxic Substances Control Act, 15 U.S.C. Section 2601 et seq., the California Health and Safety

-13-

Code (Section 25100, et seq.), the California Water Code and the California Administrative Code, in each case as such Laws are amended from time to time.

"Indebtedness" means, as to any Person (without duplication), (a) indebtedness of such Person for borrowed money or for the deferred purchase price of Property (excluding trade and other accounts payable in the ordinary course of business in accordance with ordinary trade terms), including any Guaranty Obligation for any such indebtedness, (b) indebtedness of such Person of the nature described in clause (a) that is non-recourse to the credit of such Person but is secured by assets of such Person, to the extent of the fair market value of such assets as determined in good faith by such Person, (c) Capital Lease Obligations of such Person, (d) indebtedness of such Person arising under bankers' acceptance facilities or under facilities for the discount of accounts receivable of such Person, (e) any direct or contingent obligations of such Person under letters of credit issued for the account of such Person and (f) any net obligations of such Person under Interest Rate Protection Agreements.

"Intangible Assets" means assets that are considered intangible assets under GAAP, including customer lists, goodwill, covenants not to compete, copyrights, trade names, trademarks and patents.

"Interest Coverage Ratio" means, as of the last day of any Fiscal Quarter, the ratio of (a) EBITDA for the Rolling Period ending on that date, to (b) Interest Expense of Borrower and its Subsidiaries for such Rolling Period.

"Interest Expense" means, with respect to any Person and as of the last day of any fiscal period, the sum of (a) all interest, fees, charges and related expenses (in each case as such expenses are calculated according to GAAP) paid or payable (without duplication) for that fiscal period by that Person to a lender in connection with borrowed money (including any obligations for fees, charges and related expenses payable to the issuer of any letter of credit) or the deferred purchase price of assets that are considered "interest expense" under GAAP plus (b) the portion of rent paid or payable (without duplication) for that fiscal period by that Person under Capital Lease Obligations that should be treated as interest in accordance with Financial Accounting Standards Board Statement No. 13.

"Interest Rate Protection Agreement" means a written agreement between Borrower and one or more financial institutions providing for "swap", "cap", "collar" or other interest rate protection with respect to any Indebtedness.

"Investment" means, when used in connection with any Person, any investment by or of that Person, whether by means of purchase or other acquisition of stock or other Securities of any other Person or by means of a loan, advance creating a debt, capital contribution, guaranty or other debt or equity participation or interest in any other Person, including any partnership, limited liability company and joint venture interests of such Person. The amount of any Investment shall be the amount actually invested (minus any return of capital with respect to such Investment which has

-14-

actually been received in Cash or has been converted into Cash), without adjustment for subsequent increases or decreases in the value of such Investment.

"Issuing Lender" means Wells Fargo, when acting in its capacity as Issuing Lender under any of the Loan Documents (including such other Persons that may act as agent for and on behalf of Wells Fargo) or any successor Issuing Lender.

"Laws" means, collectively, all international, foreign, federal, state and local statutes, treaties, rules, regulations, ordinances, codes and administrative or judicial precedents.

"Lead Arranger" means Wells Fargo Bank, National Association.

"Lender" means each Closing Date Lender and each lender that may hereafter become a party to this Agreement pursuant to Section 11.8.

"Letter of Credit" means any of the standby letters of credit issued by the Issuing Lender under the Revolving Facility pursuant to Section 2.5, either as originally issued or as the same may be supplemented, modified, amended, extended, restated or supplanted.

"Letter of Credit Agreement" means the standby letter of credit agreement to be executed by Borrower, in the form of Exhibit C, either as originally executed or as it may from time to time be supplemented, modified, amended, extended, restated or supplanted.

"Lien" means any mortgage, deed of trust, pledge, hypothecation, assignment for security, security interest, encumbrance, lien or charge of any kind, whether voluntarily incurred or arising by operation of Law or otherwise, affecting any Property, including any conditional sale or other title retention agreement, any lease in the nature of a security interest, and/or the filing of any financing statement (other than a precautionary financing statement with respect to a lease that is not in the nature of a security interest) under the UCC or comparable Law of any jurisdiction with respect to any Property.

"Loan Documents" means, collectively, this Agreement, the Notes, the Letter of Credit Agreement, any Request for Borrowing, any Request for Letter of Credit (and any corresponding application and/or reimbursement agreement with respect to any Letter of Credit), any Compliance Certificate, and any other agreements of any type or nature hereafter executed and delivered by Borrower or any other Party to the Administrative Agent or to any Lender in any way relating to or in furtherance of this Agreement, in each case either as originally executed or as the same may from time to time be supplemented, modified, amended, restated, extended or supplanted.

"Margin Stock" means "margin stock" as such term is defined in Regulation U.  $% \left[ {{\left[ {{{\rm{ST}}_{\rm{T}}} \right]}_{\rm{T}}} \right]$ 

-15-

"Material Adverse Effect" means any set of circumstances or events which (a) has had or would reasonably be expected to have any material adverse effect whatsoever upon the validity or enforceability of any Loan Document, (b) has been or would reasonably be expected to be material and adverse to the business, condition (financial or otherwise), prospects or operations of Borrower and its Subsidiaries, taken as a whole, (c) has materially impaired or would reasonably be expected to materially impair the ability of Borrower to perform the Obligations, or (d) has materially impaired or would reasonably be expected to materially of Borrower to perform its Obligations under the Loan Documents.

"Material Contracts" means, collectively, (a) the agreements identified on Schedule 1.3 attached hereto and (b) any other agreement that would, if terminated, materially affect the business, condition (financial or otherwise), prospects or operations of Borrower and its Subsidiaries, taken as a whole.

"Maturity Date" means the earlier of (a) June 5, 2005 and (b) the termination or cancellation of the Revolving Facility (and all of the Commitments pertaining thereto) pursuant to the terms of this Agreement.

"Maximum Revolving Credit Amount" means \$75,000,000.00.

"Monthly Payment Date" means the last Banking Day of each calendar month.

"Moody's" means Moody's Investor Service, Inc. and its successors.

"Multiemployer Plan" means any employee benefit plan of the type described in Section 4001(a)(3) of ERISA to which Borrower or any of its ERISA Affiliates contributes, is obligated to contribute or has had an obligation to contribute.

"Negative Pledge" means a Contractual Obligation which contains a covenant binding on Borrower or any Subsidiary that prohibits Liens on any of its Property, other than (a) any such covenant contained in a Contractual Obligation granting or relating to a particular Lien which affects only the Property that is the subject of such Lien and (b) any such covenant that does not apply to Liens securing the Obligations.

"Net Income" means, with respect to any fiscal period, the consolidated net income of Borrower and its Subsidiaries for that period, determined in accordance with GAAP, consistently applied.

"Note" means any of the promissory notes made by Borrower to a Lender evidencing Advances under that Lender's Commitment, substantially in the form of Exhibit D, either as originally executed or as the same may from time to time be supplemented, modified, amended, renewed, extended or supplanted.

"Obligations" means all present and future obligations of every kind or nature of Borrower at any time and from time to time owed to the Lenders, the Administrative Agent and/or the Issuing Lender, under any one or more of the Loan

-16-

Documents, whether due or to become due, matured or unmatured, liquidated or unliquidated, or contingent or noncontingent, including obligations of performance as well as obligations of payment, and including interest that accrues after the commencement of any proceeding under any Debtor Relief Law by or against Borrower.

"Opinion of Counsel" means the favorable written legal opinion of O'Melveny & Myers LLP, special counsel to Borrower, substantially in the form of Exhibit E.

"Party" means any Person other than Lenders and/or Administrative Agent, which now or hereafter is a party to any of the Loan Documents.

"PBGC" means the Pension Benefit Guaranty Corporation or any successor thereof established under ERISA.

"Pension Plan" means any "pension plan" (as such term is defined in Section 3(2) of ERISA), other than a Multiemployer Plan, which is maintained by Borrower or to which Borrower contributes or has or has had an obligation to contribute.

"Permitted Acquisition" means an Acquisition by Borrower or any whollyowned Subsidiary of Borrower of all or substantially all of the assets of, or all of the capital stock or other equity interests of, an Acquired Person engaged in similar or related line(s) of business as Borrower or any of its Subsidiaries, provided, that:

(a) if such Acquisition is of all of the capital stock or other equity interests of the Acquired Person, such Acquired Person is merged with and into Borrower or such Subsidiary substantially simultaneously with such party's acquisition of such capital stock or other equity interests or becomes a wholly-owned Subsidiary of Borrower or such Subsidiary;

(b) in the case of the Acquisition of the capital stock or other equity interest of an Acquired Person, the board of directors (or comparable governing body) of such Acquired Person shall have duly approved such Acquisition;

(c) Borrower shall have delivered a pro-forma Compliance Certificate for the most recently completed Rolling Period, demonstrating that, upon giving effect to the proposed Acquisition as of the last day of such Rolling Period, Borrower and its Subsidiaries shall be in compliance with the covenants set forth in Sections 6.12 and 6.13;

(d) such Acquired Person shall have had a positive "EBITDA" for the twelve-month fiscal period immediately preceding the date of such Acquisition (with EBITDA calculated for such Acquired Person in a manner consistent with the calculation of EBITDA for Borrower and its Subsidiaries specified herein);

-17-

(e) at the time of such Acquisition, each of the representations and warranties contained in the Loan Documents shall be true and correct in all material respects (except to the extent such representations and warranties expressly relate to an earlier date), no Default or Event of Default shall have occurred and remain in effect and after giving effect to such Acquisition, on a pro forma combined basis, (i) no Default of Event of Default would have occurred at any time during the twelve-month fiscal period immediately preceding the date of such Acquisition assuming that such Acquisition had occurred on the first day of such period and (ii) Borrower and its Subsidiaries, on a projected basis, will be in compliance with Section 6.12 and 6.13, as of each of the four Fiscal Quarters ending after the date of the Acquisition, as reflected in updated projections provided by Borrower to the Administrative Agent and the Lenders prior to the effective date of such Acquisition;

(f) if such Acquisition involves the purchase of an interest in a partnership between Borrower (or a Subsidiary of Borrower) as a general partner and entities unaffiliated with Borrower or such Subsidiary as the other partners, such transaction shall be effected by having such equity interest acquired by a corporate holding company directly or indirectly wholly-owned by Borrower newly formed for the sole purpose of effecting such transaction;

(g) the Indebtedness assumed or consideration paid or payable in cash in connection with such Acquisition shall not exceed the lesser of (x) 15,000,000 and (y) the fair market value thereof; and

(h) the Indebtedness assumed or consideration paid or payable in cash in connection with such Acquisition, when taken together with each other Permitted Acquisitions consummated since the Closing Date shall not exceed \$25,000,000 in the aggregate.

"Permitted Capital Asset Indebtedness" means Indebtedness of Borrower and its Subsidiaries consisting of Capital Lease Obligations, or otherwise incurred to finance the purchase or construction of capital assets (which shall be deemed to exist if the Indebtedness is incurred at or within 90 days before or after the purchase or construction of the capital asset), or to refinance any such Indebtedness.

#### "Permitted Encumbrances" means:

(a) Inchoate Liens incident to construction on or maintenance of Property; or Liens incident to construction on or maintenance of Property now or hereafter filed of record for which adequate reserves have been set aside (or deposits made pursuant to applicable Law) and which are being contested in good faith by appropriate proceedings and have not proceeded to judgment, provided that, by reason of nonpayment of the obligations secured by such Liens, no such Property is subject to an impending risk of loss or forfeiture;

-18-

(b) Liens for taxes and assessments on Property which are not yet past due; or Liens for taxes and assessments on Property for which adequate reserves have been set aside and are being contested in good faith by appropriate proceedings and have not proceeded to judgment, provided that, by reason of nonpayment of the obligations secured by such Liens, no such Property is subject to an impending risk of loss or forfeiture;

(c) defects and irregularities in title to any Property which in the aggregate do not materially impair the fair market value or use of the Property for the purposes for which it is or may reasonably be expected to be held;

(d) easements, exceptions, reservations, or other agreements for the purpose of pipelines, conduits, cables, wire communication lines, power lines and substations, streets, trails, walkways, drainage, irrigation, water, and sewerage purposes, dikes, canals, ditches, the removal of oil, gas, coal, or other minerals, and other like purposes affecting Property which in the aggregate do not materially burden or impair the fair market value or use of such Property for the purposes for which it is or may reasonably be expected to be held;

(e) easements, exceptions, reservations, or other agreements for the purpose of facilitating the joint or common use of Property in or adjacent to a shopping center or similar project affecting Property which in the aggregate do not materially burden or impair the fair market value or use of such Property for the purposes for which it is or may reasonably be expected to be held;

(f) rights reserved to or vested in any Governmental Agency to control or regulate, or obligations or duties to any Governmental Agency with respect to, the use of any Property;

(g) rights reserved to or vested in any Governmental Agency to control or regulate, or obligations or duties to any Governmental Agency with respect to, any right, power, franchise, grant, license, or permit;

(h) present or future zoning laws and ordinances or other laws and ordinances restricting the occupancy, use, or enjoyment of Property;

(i) statutory Liens, other than those described in clauses (a) or (b) above, arising in the ordinary course of business with respect to obligations which are not delinquent or are being contested in good faith, provided that, if delinquent, adequate reserves have been set aside with respect thereto and, by reason of nonpayment, no Property is subject to an impending risk of loss or forfeiture;

(j) covenants, conditions, and restrictions affecting the use of Property which in the aggregate do not materially impair the fair market value or use of the Property for the purposes for which it is or may reasonably be expected to be held;

-19-

 (k) rights of tenants under leases and rental agreements covering Property entered into in the ordinary course of business of the Person owning such Property;

(1) Liens consisting of pledges or deposits to secure obligations under workers' compensation laws or similar legislation, including Liens of judgments thereunder which are not currently dischargeable;

(m) Liens consisting of pledges or deposits of Property to secure performance in connection with operating leases made in the ordinary course of business, provided the aggregate value of all such pledges and deposits (excluding the property subject to such lease) in connection with any such lease does not at any time exceed 10% of the annual fixed rentals payable under such lease;

(n) Liens consisting of deposits of Property to secure bids made with respect to, or performance of, contracts (other than contracts creating or evidencing an extension of credit to the depositor);

(0) Liens consisting of any right of offset, or statutory bankers' lien, on bank deposit accounts maintained in the ordinary course of business so long as such bank deposit accounts are not established or maintained for the purpose of providing such right of offset or bankers' lien;

(p) any (i) interest or title of a lessor or sublessor under any lease not prohibited by this Agreement, but only to the extent such interest or title pertains solely to the property leased, (ii) Lien or restriction that the interest or title of such lessor or sublessor may be subject to, or (iii) subordination of the interest of the lessee or sublessee under such lease to any Lien or restriction referred to in the preceding clause (ii), so long as the holder of such Lien or restriction agrees to recognize the rights of such lessee or sublessee under such lease;

(q) Liens consisting of cash collateralization of letters of credit issued pursuant to the Existing Credit Facilities (other than the Wells Fargo Existing Credit Facility) to the extent required by banks party thereto;

 $% \left( r\right) ^{\prime }$  Liens consisting of deposits of Property to secure statutory obligations of Borrower; and

(s) Liens consisting of deposits of Property to secure (or in lieu of) surety, appeal or customs bonds.

"Permitted Right of Others" means a Right of Others consisting of (a) an interest (other than a legal or equitable co-ownership interest, an option or right to acquire a legal or equitable co-ownership interest and any interest of a ground lessor under a ground lease), that does not materially impair the fair market value or use of Property for the purposes for which it is or may reasonably be expected to be held, (b) an option or right to acquire a Lien that would be a Permitted Encumbrance or

-20-

other encumbrance permitted pursuant to Section 6.9, (c) the subordination of a lease or sublease in favor of a financing entity and (d) a license, or similar right, of or to Intangible Assets or other similar Property granted in the ordinary course of business.

"Person" means any individual or entity, including a trustee, corporation, limited liability company, general partnership, limited partnership, joint stock company, trust, estate, unincorporated organization, business association, firm, joint venture, Governmental Agency, or other entity.

"Pricing Occurrence" means with respect to any change in the Debt Rating which results in a change in the Applicable Pricing Level, the date which is five (5) Banking Days after the Administrative Agent has received evidence reasonably satisfactory to it of such change.

"Pricing Period" means (a) the period commencing on the Closing Date and ending on the first Pricing Occurrence to occur thereafter and (b) each subsequent period commencing on the date of a Pricing Occurrence and ending on the next Pricing Occurrence to occur.

"Prime Rate" means the rate of interest most recently announced within Wells Fargo, at its principal office in San Francisco, California, as its "prime rate." The "prime rate" is one of several base rates used by Wells Fargo and serves as the basis upon which effective rates of interest are calculated for loans and other credits making reference thereto. The "prime rate" is evidenced by the recording thereof after its announcement in such internal publication or publications as Wells Fargo may designate. Any change in the Prime Rate shall take effect on the day the change is announced within Wells Fargo.

"Projections" means the financial projections of Borrower and its Subsidiaries heretofore distributed by or on behalf of Borrower to the Administrative Agent, a true and correct copy of which is attached to the Certificate of Senior Officer described in Section 8.1(a) (7).

"Property" means any interest in any kind of property or asset, whether real, personal or mixed, or tangible or intangible.

"Pro Rata Share" of any amount means, with respect to any Lender at any time, the product of (a) a fraction the numerator of which is the amount of such Lender's Commitment (or, if such Commitment shall have expired or been terminated, the amount of such Lender's Advances), and the denominator of which is the aggregate Commitments or Advances, as the case may be, at such time, multiplied by (b) such amount. Schedule 1.1 sets forth the Pro Rata Shares of the Closing Date Lenders as of the Closing Date.

"Quarterly Payment Date" means each March 31, June 30, September 30 and December 31.

-21-

"Real Property" means, as of any date of determination, all real property then or theretofore owned, leased or occupied by Borrower or any Subsidiary.

"Regulation D" means Regulation D, as at any time amended, of the Board of Governors of the Federal Reserve System, or any other regulation in substance substituted therefor.

"Regulation U" means Regulation U, as at any time amended, of the Board of Governors of the Federal Reserve System, or any other regulation in substance substituted therefor.

"Request for Borrowing" means a written request for a Borrowing substantially in the form of Exhibit F, signed by a Responsible Official of Borrower, and properly completed to provide all information required to be included therein.

"Request for Continuation/Conversion" means a written request to Continue or Convert a Borrowing substantially in the form of Exhibit G, signed by a Responsible Official of Borrower, and properly completed to provide all information required to be included therein.

"Request for Letter of Credit" means a written request for a Letter of Credit substantially in the form of Exhibit H, signed by a Responsible Official of Borrower, and properly completed to provide all information required to be included therein.

"Requirement of Law" means, as to any Person, the articles or certificate of incorporation and by-laws or other organizational or governing documents of such Person, and any Law, or judgment, award, decree, writ or determination of a Governmental Agency, in each case applicable to or binding upon such Person or any of its Property or to which such Person or any of its Property is subject.

"Requisite Lenders" means, (a) as of any date of determination if the Commitments are then in effect, Lenders having in the aggregate more than 50% of such aggregate Commitments, and (b) as of any date of determination if the Commitments have then been terminated and there is then any outstanding Indebtedness evidenced by the Notes, the Swing Line Documents and/or Letters of Credit, Lenders owed or holding in the aggregate more than 50% of then applicable Revolving Credit Facility Usage.

"Responsible Official" means, as to any Person, (a) when used with reference to a Person other than an individual, a corporate officer of such Person, general partner of such Person, corporate officer of a corporate general partner of such Person, corporate officer of a corporate general partner of a partnership that is a general partner of such Person, manager or managing member (in the case of a Person that is a limited liability company), or any other responsible official thereof duly acting on behalf thereof, and (b) when used with reference to a Person who is an individual, such Person. The Lenders shall be entitled to conclusively rely upon any document or

-22-

certificate that is signed or executed by a Responsible Official of Borrower or any Subsidiary as having been authorized by all necessary corporate, partnership, limited liability company and/or other action on the part of Borrower or such Subsidiary.

"Revolving Credit Facility Usage" means, as of any date of determination, the sum of (a) the aggregate principal amount of funded Indebtedness then outstanding under the Notes plus (b) the Aggregate Effective Amount under all outstanding Letters of Credit plus (c) the Swing Line Outstandings.

"Revolving Facility" means the revolving credit facility provided hereunder in respect of the aggregate Commitments.

"Right of Others" means, as to any Property in which a Person has an interest, any legal or equitable right, title or other interest (other than a Lien) held by any other Person in that Property, and any option or right held by any other Person to acquire any such right, title or other interest in that Property, including any option or right to acquire a Lien; provided, however, that (a) no covenant restricting the use or disposition of Property of such Person contained in any Contractual Obligation of such Person and (b) no provision contained in a contract creating a right of payment or performance in favor of a Person that conditions, limits, restricts, diminishes, transfers or terminates such right shall be deemed to constitute a Right of Others.

"Rolling Period" means any period of four consecutive Fiscal Quarters of Borrower and its Subsidiaries.

"S&P" means Standard & Poor's Ratings Group, a division of McGraw Hill, Inc. and its successors.

"Sale and Leaseback" means, with respect to any Person, the sale of Property owned by that Person (the "Seller") to another Person (the "Buyer"), together with the substantially concurrent leasing of such Property by the Buyer to the Seller.

"SCW" means Southern California Water Company, a California corporation and wholly-owned Subsidiary of Borrower.

"Security" means any capital stock, share, voting trust certificate, bond, debenture, note or other evidence of Indebtedness, limited partnership interest, member interest, or any warrant, option or other right to purchase or acquire any of the foregoing.

"Senior Officer" means (a) the chief executive officer, (b) the president, (c) any executive vice president, (d) the chief financial officer, (e) the treasurer, or (f) any assistant treasurer, in each case of any Person.

"Solvent" means, as of any date of determination, and as to any Person, that on such date: (a) the fair valuation of the assets of such Person is greater than the fair valuation of such Person's probable liability in respect of existing debts; (b) such

-23-

Person does not intend to, and does not believe that it will, incur debts beyond such Person's ability to pay as such debts mature; (c) such Person is not engaged in a business or transaction, and is not about to engage in a business or transaction, which would leave such Person with assets remaining which would constitute unreasonably small capital after giving effect to the nature of the particular business or transaction (including, in the case of Borrower, the transactions occurring on the Closing Date); and (d) such Person is generally paying its debts as they become due. For purposes of the foregoing (1) the "fair valuation" of any assets means the amount realizable within a reasonable time, either through collection or sale, of such assets at their regular market value, which is the amount obtainable by a capable and diligent businessman from an interested buyer willing to purchase such assets within a reasonable time under ordinary circumstances; and (2) the term "debts" includes any legal liability whether matured or unmatured, liquidated or unliquidated, absolute, fixed, or contingent.

"Special Eurodollar Circumstance" means the application or adoption after the Closing Date of any Law or interpretation, or any change therein or thereof, or any change in the interpretation or administration thereof by any Governmental Agency, central bank or comparable authority charged with the interpretation or administration thereof, or compliance by any Lender or its Eurodollar Lending Office with any request or directive (whether or not having the force of Law) of any such Governmental Agency, central bank or comparable authority.

"Stockholders' Equity" means, as of any date of determination and with respect to Borrower and its Subsidiaries, the consolidated stockholders' equity of Borrower and its Subsidiaries as of that date determined in accordance with GAAP; provided that there shall be excluded from Stockholders' Equity any amount attributable to Disgualified Stock.

"Swing Line" means the revolving line of credit established by the Swing Line Lender in favor of Borrower pursuant to Section 2.8.

"Swing Line Documents" means the promissory note and any other documents executed by Borrower in favor of the Swing Line Lender in connection with the Swing Line.

"Swing Line Lender" means Wells Fargo.

"Swing Line Loans" means loans made by the Swing Line Lender to Borrower pursuant to Section 2.8.

"Swing Line Outstandings" means, as of any date of determination, the aggregate principal Indebtedness of Borrower on all Swing Line Loans then outstanding.

-24-

"Subordinated Obligations" means, as of any date of determination (without duplication), any Indebtedness of Borrower or any Subsidiary on that date which has been subordinated in right of payment to the Obligations in a manner reasonably satisfactory to the Administrative Agent and the Requisite Lenders and contains such other protective terms with respect to senior debt (such as payment blockage) as the Administrative Agent and the Requisite Lenders may reasonably require.

"Subsidiary" means, as of any date of determination and with respect to any Person, any corporation, limited liability company or partnership (whether or not, in any case, characterized as such or as a "joint venture"), whether now existing or hereafter organized or acquired: (a) in the case of a corporation or limited liability company, of which a majority of the Securities having ordinary voting power for the election of directors or other governing body (other than Securities having such power only by reason of the happening of a contingency) are at the time beneficially owned by such Person and/or one or more Subsidiaries of such Person, or (b) in the case of a partnership, of which a majority of the partnership or other ownership interests are at the time beneficially owned by such Person and/or one or more of its Subsidiaries. Any reference to a "Subsidiary" or "Subsidiaries" shall, unless otherwise provided, be deemed to be a reference to a Subsidiary (or Subsidiaries, as the case may be) of Borrower.

"Termination Date" means the date on which the Advances and all other Obligations under this Agreement and the other Loan Documents are indefeasibly paid in full, in Cash, and Borrower shall have no further right to borrow any moneys or obtain other credit extensions or financial accommodations under this Agreement or any of the other Loan Documents.

"to the best knowledge of" means, when modifying a representation, warranty or other statement of any Person, that the fact or situation described therein is known by the Person (or, in the case of a Person other than a natural Person, known by a Responsible Official of that Person) making the representation, warranty or other statement, or with the exercise of reasonable due diligence under the circumstances (in accordance with the standard of what a reasonable Person in similar circumstances would have done) would have been known by the Person (or, in the case of a Person other than a natural Person, would have been known by a Responsible Official of that Person).

"Total Funded Debt" means, as of any date of determination, without duplication, the sum of (a) all principal Indebtedness of Borrower and its Subsidiaries for borrowed money (including Subordinated Obligations and any other subordinated indebtedness, debt Securities issued by Borrower and any of its Subsidiaries, the aggregate principal Indebtedness outstanding under the Notes and the Aggregate Effective Amount of all outstanding Letters of Credit) on that date plus (b) the aggregate amount of the principal portion of all Capital Lease Obligations of Borrower and its Subsidiaries plus (c) any Guaranty Obligations of Borrower and its

-25-

Subsidiaries with respect to the Indebtedness of others of the types referred to in (a) and (b) above.

"Total Funded Debt Ratio" means, as of the last day of any Fiscal Quarter, the ratio of (a) Total Funded Debt as of such date to (b) the sum of (i) Total Funded Debt as of such date plus Stockholders' Equity as of such date.

"Type" refers to the distinction between Advances bearing interest at the Alternate Base Rate and Advances bearing interest at the Eurodollar Rate.

"UCC" means the Uniform Commercial Code as the same may from time to time be enacted and in effect in the State of California; provided that, in the event by reason of mandatory provisions of law, any or all of the attachment, perfection or priority of the Administrative Agent's or any Lender's Lien on any collateral is governed by the Uniform Commercial Code as enacted and in effect in a jurisdiction other than the State of California, the term "UCC" shall mean the Uniform Commercial Code as enacted and in effect in such other jurisdiction solely for purposes of the provisions hereof relating to such attachment, perfection or priority and for purposes of definitions related to such provisions.

"Wells Fargo" means Wells Fargo Bank, National Association.

"Wells Fargo Existing Credit Facility" has the meaning specified for such term in the definition of Existing Credit Facilities.

1.2 Use of Defined Terms. Any defined term used in the plural shall refer to all members of the relevant class, and any defined term used in the singular shall refer to any one or more of the members of the relevant class.

1.3 Accounting Terms; Covenant Calculations. All accounting terms not specifically defined in this Agreement shall be construed in conformity with, and all financial data required to be submitted by this Agreement shall be prepared in conformity with, GAAP applied on a consistent basis, except as otherwise specifically prescribed herein. In the event that GAAP changes during the term of this Agreement such that the covenants contained in Sections 6.12 and 6.13 would then be calculated in a different manner or with different components, (i) Borrower and the Lenders agree to amend this Agreement in such respects as are necessary to conform those covenants as criteria for evaluating Borrower's financial condition to substantially the same criteria as were effective prior to such change in GAAP and (ii) Borrower shall be deemed to be in compliance with the covenants contained in the aforesaid Sections if and to the extent that Borrower would have been in compliance therewith under GAAP as in effect immediately prior to such change, but shall have the obligation to deliver each of the materials described in Article 7 to the Administrative Agent and the Lenders, on the dates therein specified, with financial data presented in a manner which conforms with GAAP as in effect immediately prior to such change.

-26-

1.4 Rounding. Any financial ratios required to be maintained by Borrower pursuant to this Agreement shall be calculated by dividing the appropriate component by the other component, carrying the result to one place more than the number of places by which such ratio is expressed in this Agreement and rounding the result up or down to the nearest number (with a round-up if there is no nearest number) to the number of places by which such ratio is expressed in this Agreement.

1.5 Exhibits and Schedules. All Exhibits and Schedules to this Agreement, either as originally existing or as the same may from time to time be supplemented, modified or amended, are incorporated herein by this reference. A matter disclosed on any Schedule shall be deemed disclosed on all Schedules.

1.6 References to "Borrower and its Subsidiaries". Any reference herein to "Borrower and its Subsidiaries" or the like shall refer solely to Borrower during such times, if any, as Borrower shall have no Subsidiaries.

1.7 Miscellaneous Terms. The term "or" is disjunctive; the term "and" is conjunctive. The term "shall" is mandatory; the term "may" is permissive. Masculine terms also apply to females; feminine terms also apply to males. The term "including" is by way of example and not limitation.

-27-

#### Article 2. ADVANCES AND LETTERS OF CREDIT

#### 2.1 Advances-General.

(a) Subject to the terms and conditions set forth in this Agreement, from time to time on any Banking Day during the period from the Closing Date through the Maturity Date, each Lender severally agrees to make Advances ("Advances") to Borrower under the Revolving Facility in such amounts as Borrower may request provided that, after giving effect to such Advances, (i) Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount and (ii) as to each Lender, such Lender's Pro Rata Share of Revolving Credit Facility Usage does not exceed such Lender's Commitment. All Advances shall be made by the Lenders ratably according to their respective Commitments.

Within the limits of each Lender's Commitment in effect from time to time and subject to the foregoing, Borrower may borrow under this Section 2.1(a), prepay Advances pursuant to Section 3.1 and reborrow under this Section 2.1(a).

(b) Subject to the next sentence, each Borrowing shall be made pursuant to a Request for Borrowing which shall specify (i) the date of such requested Borrowing, (ii) the Type of Advances comprising such Borrowing, (iii) the amount of such Borrowing, and (iv) in the case of a Borrowing consisting of Eurodollar Rate Advances, the Eurodollar Period therefor. Unless the Administrative Agent has notified, in its sole and absolute discretion, Borrower to the contrary not less than three (3) days prior to the date of any Borrowing, a Borrowing may be requested by telephone by a Responsible Official of Borrower, in which case Borrower shall confirm such request by promptly delivering a Request for Borrowing (conforming to the preceding sentence) in person or by telecopier to the Administrative Agent. The Administrative Agent shall incur no liability whatsoever hereunder in acting upon any telephonic request for a Borrowing purportedly made by a Responsible Official of Borrower, and Borrower hereby agrees to indemnify the Administrative Agent from any loss, cost, expense or liability as a result of so acting.

(c) Promptly following receipt of a Request for Borrowing, the Administrative Agent shall notify each Lender by telephone or telecopier (and if by telephone, promptly confirmed by telecopier) of the date of the requested Borrowing, the Type of Advances comprising such Borrowing, the Eurodollar Period (if applicable), and the amount corresponding to that Lender's ratable share of the Borrowing. Not later than 1:00 p.m., California time, on the date specified for any Borrowing (which must be a Banking Day), each Lender shall make its ratable share of the Borrowing in immediately available funds available to the Administrative Agent at the Administrative Agent's Office. Upon satisfaction or waiver of the applicable conditions set forth in Article 8, all Advances shall be credited on that date in immediately available funds to the Designated Deposit Account.

-28-

(d) Anything in paragraph (b) above to the contrary notwithstanding, Borrower may not (a) request Alternate Base Rate Advances for any Borrowing if the aggregate amount of such Borrowing is less than \$500,000 (and any such Borrowing exceeding such minimum amount shall be in an integral multiple of \$100,000), provided that the foregoing minimum amount shall not apply to an Alternate Base Rate Advance that causes the aggregate amount borrowed under the Revolving Facility to equal the full amount available for Advances hereunder or Advances pursuant to Section 2.5, or (b) elect Eurodollar Rate Advances for any Borrowing (i) if the aggregate amount of such Borrowing is less than \$1,000,000 (and any such Borrowing exceeding such minimum amount shall be in an integral multiple of \$500,000) or (ii) if the obligation of the Lenders to make Eurodollar Rate Advances shall then be suspended pursuant to Section 2.4, 3.5 or 3.6.

(e) The Advances made by each Lender under its Commitment shall be evidenced by that Lender's Note.

(f) A Request for Borrowing shall be irrevocable upon the Administrative Agent's first notification thereof.

(g) The Administrative Agent, on behalf of the Lenders, is hereby authorized to make Borrowings available to Borrower upon fulfillment of the applicable conditions set forth in Article 8. Upon fulfillment of such applicable conditions, the proceeds of Borrowings shall either be credited in immediately available funds to the Designated Deposit Account or remitted directly to one or more third parties, as directed by Borrower and approved by the Administrative Agent. The proceeds of any Borrowing consisting of Eurodollar Rate Advances shall be so credited or remitted on the first day of the applicable Eurodollar Period for such Borrowings.

2.2 Alternate Base Rate Advances. Each request by Borrower for a Borrowing comprised of Alternate Base Rate Advances shall be made pursuant to a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of Section 2.1(b), if applicable) received by the Administrative Agent, at the Administrative Agent's Office, not later than 11:00 a.m. California time, at least one (1) Banking Day before the requested Borrowing. All Advances shall constitute Alternate Base Rate Advances unless properly designated as a Eurodollar Rate Advance pursuant to Section 2.3 or 2.4.

2.3 Eurodollar Rate Advances.

(a) Each request by Borrower for a Borrowing comprised of Eurodollar Rate Advances shall be made pursuant to a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of Section 2.1(b), if applicable) received by the Administrative Agent, at the Administrative Agent's Office, not later than 9:00 a.m., California time, at least three (3) Eurodollar Banking Days before the first day of the applicable Eurodollar Period.

-29-

(b) On the date which is two (2) Eurodollar Banking Days before the first day of the applicable Eurodollar Period, the Administrative Agent shall confirm its determination of the applicable Eurodollar Rate (which determination shall be conclusive in the absence of manifest error) and promptly shall give notice of the same to Borrower and the Lenders by telephone or telecopier (and if by telephone, promptly confirmed by telecopier).

(c) Unless the Administrative Agent and the Requisite Lenders otherwise consent, Eurodollar Rate Advances may not be outstanding under more than five (5) separate Eurodollar Periods at any one time.

(d) No Borrowing comprised of Eurodollar Rate Advances may be requested during the continuation of a Default or  ${\tt Event}$  of Default.

(e) Nothing contained herein shall require any Lender to fund any Eurodollar Rate Advance in the Designated Eurodollar Market.

2.4 Conversion and Continuation of Advances.

(a) Optional Conversion. Borrower may on any Banking Day, upon notice given to the Administrative Agent not later than 9:00 a.m. (California time) on the third Eurodollar Banking Day prior to the date of a proposed Conversion if the Conversion is into Eurodollar Rate Advances, or one Banking Day prior to the date of a proposed Conversion if the Conversion is into Alternate Base Rate Advances, and subject to the provisions of Sections 3.5 and 3.6, Convert all or any portion of the Advances of one Type outstanding under the Revolving Facility (and, in the case of Eurodollar Rate Advances, having the same Eurodollar Period) into Advances of the other Type under the Revolving Facility; provided that any Conversion of Eurodollar Rate Advances into Alternate Base Rate Advances on other than the last day of a Eurodollar Period for such Eurodollar Rate Advances shall be subject to Section 3.6(e), any Conversion of Alternate Base Rate Advances into Eurodollar Rate Advances shall be in an amount not less than \$1,000,000 or integral multiples of \$500,000 in excess thereof and no Conversion of any Advances shall result in more than five (5) separate Eurodollar Periods being outstanding under the Revolving Facility. Each such notice of Conversion shall be made pursuant to a Request for Continuation/Conversion and shall, within the restrictions specified above, specify (i) the date of such Conversion, (ii) the aggregate amount and Type of the Advances (and, in the case of Eurodollar Rate Advances, the Eurodollar Period therefor) to be Converted and (iii) if such Conversion is into Eurodollar Rate Advances, the duration of the initial Eurodollar Period for such Advances. Each request for Conversion shall be irrevocable and binding on Borrower.

(b) Certain Mandatory Conversions.

(i) On the date on which the aggregate unpaid principal amount of Eurodollar Rate Advances comprising any Borrowing shall be reduced, by payment or prepayment or otherwise, to less than \$1,000,000 such Advances shall automatically Convert into Alternate Base Rate Advances.

-30-

(ii) If Borrower shall fail to select the duration of any Eurodollar Period for any outstanding Eurodollar Rate Advances in accordance with the provisions contained in Section 2.1(b) and in clause (a) or (c) of this Section 2.4, each such Eurodollar Rate Advance will automatically, on the last day of the then existing Eurodollar Period therefor, Convert into an Alternate Base Rate Advance.

(iii) Upon the occurrence and during the continuance of any Event of Default and upon notice from the Administrative Agent to Borrower at the request of the Requisite Lenders, (x) each Eurodollar Rate Advance will automatically, on the last day of the then existing Eurodollar Period therefor, Convert into an Alternate Base Rate Advance and (y) the obligation of the Lenders to make, or to Convert Advances into, or to Continue, Eurodollar Rate Advances shall be suspended.

(c) Continuations. Borrower may, on any Eurodollar Banking Day, upon notice given to the Administrative Agent not later than 9:00 a.m. (California time) on the third Eurodollar Banking Day prior to the date of the proposed Continuation and subject to the provisions of Sections 3.5 and 3.6. Continue all or any portion of the Eurodollar Rate Advances outstanding under a Facility having the same Eurodollar Period; provided that any such Continuation shall be made only on the last day of a Eurodollar Period for such Eurodollar Rate Advances, no Continuation of Eurodollar Rate Advances shall be in an amount less than \$1,000,000 and no Continuation of any Eurodollar Rate Advances shall result in more than five (5) separate Eurodollar Periods being outstanding under the Revolving Facility. Each such notice of Continuation shall be made pursuant to a Request for Continuation/Conversion and shall, within the restrictions specified above, specify (i) the date of such Continuation, (ii) the aggregate amount and category of, and the Eurodollar Period for, the Advances being Continued and (iii) the duration of the initial Eurodollar Period for the Eurodollar Rate Advances subject to such Continuation. Each notice of Continuation shall be irrevocable and binding on Borrower.

2.5 Letters of Credit.

(a) As of the Closing Date, the Existing Wells Fargo Bank L/Cs shall be Letters of Credit for all purposes under this Agreement. Subject to the terms and conditions hereof, at any time and from time to time from the Closing Date through the Maturity Date, the Issuing Lender shall issue such Letters of Credit under the Revolving Facility for the benefit of Borrower and/or its wholly-owned Subsidiaries as Borrower may request by a Request for Letter of Credit; provided that giving effect to all such Letters of Credit, (i) Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount and (ii) the Aggregate Effective Amount under all outstanding Letters of Credit shall not exceed \$15,000,000. Each Letter of Credit shall be in a form reasonably acceptable to the Issuing Lender. Unless the Issuing Lender and the Requisite Lenders otherwise consent, the term of any Letter of Credit shall not exceed 365 days. Unless all the Lenders otherwise consent in a writing delivered to the Administrative Agent, the term of any Letter of Credit shall

-31-

not extend beyond the Maturity Date. A Request for Letter of Credit shall be irrevocable absent the consent of the Issuing Lender.

(b) Each Request for Letter of Credit shall be submitted to the Issuing Lender, with a copy to the Administrative Agent, at least three (3) Banking Days prior to the date upon which the related Letter of Credit is proposed to be issued. The Administrative Agent shall promptly notify the Issuing Lender whether such request, and the issuance of a Letter of Credit pursuant thereto, conforms to the requirements of this Agreement. Upon issuance of a Letter of Credit, the Issuing Lender shall promptly notify the Administrative Agent of the amount and terms thereof. Unless the Issuing Lender has notified, in its sole and absolute discretion, Borrower to the contrary not less than three (3) days prior to the date of any Request for Letter of Credit, a Request for Letter of Credit may be delivered to the Issuing Lender by facsimile by a Responsible Official of Borrower, in which case Borrower shall confirm such request by promptly delivering a Request for Letter of Credit (conforming to the preceding sentence) in person to the Issuing Lender. The Issuing Lender shall incur no liability whatsoever hereunder in acting upon any Reguest for Letter of Credit received by facsimile purportedly made by a Responsible Official of Borrower, and Borrower hereby agrees to indemnify the Issuing Lender from any loss, cost, expense or liability as a result of so acting.

(c) Upon issuance of a Letter of Credit, each Lender shall be deemed to have purchased a pro rata participation in such Letter of Credit from the Issuing Lender in proportion to that Lender's Pro Rata Share of the Revolving Facility. Without limiting the scope and nature of each Lender's participation in any Letter of Credit, to the extent that the Issuing Lender has not been reimbursed by Borrower for any payment required to be made by the Issuing Lender under any Letter of Credit, each Lender shall, pro rata according to its Pro Rata Share of the Revolving Facility, reimburse the Issuing Lender through the Administrative Agent promptly upon demand for the amount of such payment. The obligation of each Lender to so reimburse the Issuing Lender shall be absolute and unconditional and shall not be affected by the occurrence of an Event of Default or any other occurrence or event. Any such reimbursement shall not relieve or otherwise impair the obligation of Borrower to reimburse the Issuing Lender for the amount of any payment made by the Issuing Lender under any Letter of Credit together with interest as hereinafter provided.

(d) Borrower agrees to pay to the Issuing Lender through the Administrative Agent an amount equal to any payment made by the Issuing Lender with respect to each Letter of Credit within one (1) Banking Day after demand made by the Issuing Lender therefor, together with interest on such amount from the date of any payment made by the Issuing Lender at the rate applicable to Alternate Base Rate Advances under the Revolving Facility for the period commencing on the date of any such payment and continuing through the first Banking Day following such demand and thereafter at the Default Rate. The principal amount of any such payment shall be used to reimburse the Issuing Lender for the payment made by it under the Letter of

-32-

Credit. Each Lender that has reimbursed the Issuing Lender pursuant to Section 2.5(c) for its Pro Rata Share of any payment made by the Issuing Lender under a Letter of Credit shall thereupon acquire a pro rata participation, to the extent of such reimbursement, in the claim of the Issuing Lender against Borrower under this Section 2.5(d) and shall share, in accordance with that pro rata participation, in any payment made by Borrower with respect to such claim. Upon receipt of any such reimbursement from Borrower, the Issuing Lender shall pay to the Administrative Agent, for the ratable benefit of those Lenders that had reimbursed the Issuing Lender pursuant to Section 2.5(c) for their respective Pro Rata Shares of any payment made by the Issuing Lender under a Letter of Credit to which such reimbursement applies, the amount of such reimbursement.

(e) Borrower may, pursuant to a Request for Borrowing, request that Advances be made pursuant to Section 2.1(a) to provide funds for the payment required by Section 2.5(d). The proceeds of such Advances shall be paid directly to the Issuing Lender to reimburse it for the payment made by it under the Letter of Credit.

(f) If Borrower fails to make the payment required by Section 2.5(d) within the time period therein set forth, in lieu of the reimbursement to the Issuing Lender under Section 2.5(c) the Issuing Lender may (but is not required to), without notice to or the consent of Borrower, instruct the Administrative Agent to cause Advances to be made by the Lenders under the Revolving Facility in an aggregate amount equal to the amount paid by the Issuing Lender with respect to that Letter of Credit and, for this purpose, the conditions precedent set forth in Article 8 shall not apply. The proceeds of such Advances shall be paid to the Issuing Lender to reimburse it for the payment made by it under the Letter of Credit.

(g) The issuance of any supplement, modification, amendment, renewal, or extension to or of any Letter of Credit shall be treated in all respects the same as the issuance of a new Letter of Credit.

(h) The obligation of Borrower to pay to the Issuing Lender the amount of any payment made by the Issuing Lender under any Letter of Credit shall be absolute, unconditional, and irrevocable, subject only to performance by the Issuing Lender of its obligations to Borrower under Section 5108 of the UCC. Without limiting the foregoing, Borrower's obligations shall not be affected by any of the following circumstances:

(i) any lack of validity or enforceability of the Letter of Credit, this Agreement, or any other agreement or instrument relating thereto;

(ii) any amendment or waiver of or any consent to departure from the Letter of Credit, this Agreement, or any other agreement or instrument relating thereto, with the written consent of Borrower executed by a Responsible Official of Borrower;

-33-

(iii) the existence of any claim, setoff, defense, or other rights that Borrower may have at any time against the Issuing Lender, the Administrative Agent or any Lender, any beneficiary of the Letter of Credit (or any Persons for whom any such beneficiary may be acting) or any other Person, whether in connection with the Letter of Credit, this Agreement, or any other agreement or instrument relating thereto, or any unrelated transactions;

(iv) any demand, statement, or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid, or insufficient in any respect or any statement therein being untrue or inaccurate in any respect whatsoever so long as any such document reasonably appeared to comply with the terms of the Letter of Credit;

 $(\nu)$  payment by the Issuing Lender in good faith under the Letter of Credit against presentation of a draft or any accompanying document which does not strictly comply with the terms of the Letter of Credit;

(vi) the existence, character, quality, quantity, condition, packing, value or delivery of any Property purported to be represented by documents presented in connection with any Letter of Credit or for any difference between any such Property and the character, quality, quantity, condition, or value of such Property as described in such documents;

(vii) the time, place, manner, order or contents of shipments or deliveries of Property as described in documents presented in connection with any Letter of Credit or the existence, nature and extent of any insurance relative thereto;

(ix) any failure or delay in notice of shipments or arrival of any Property;

 $({\rm x})$  any error in the transmission of any message relating to a Letter of Credit not caused by the Issuing Lender, or any delay or interruption in any such message;

(xi) any error, neglect or default of any correspondent of the Issuing Lender in connection with a Letter of Credit;

 $(\rm xii)$  any consequence arising from acts of God, war, insurrection, civil unrest, disturbances, labor disputes, emergency conditions or other causes beyond the control of the Issuing Lender; and

 $% \left( xiii\right)$  so long as the Issuing Lender in good faith determines that the contract or document appears to comply with the terms of the Letter of

-34-

Credit, the form, accuracy, genuineness or legal effect of any contract or document referred to in any document submitted to the Issuing Lender in connection with a Letter of Credit.

Notwithstanding anything to the contrary contained in this Section 2.5(h), Borrower shall retain any and all rights it may have against the Issuing Lender for any liability arising solely out of the gross negligence or willful misconduct of the Issuing Lender, as determined by a final judgment of a court of competent jurisdiction.

(i) The Issuing Lender shall be entitled to the protection accorded to the Administrative Agent pursuant to Section 10.6 (subject to the standards set forth therein), mutatis mutandis.

(j) The Uniform Customs and Practice for Documentary Credits, as published in its most current version by the International Chamber of Commerce, shall be deemed a part of this Section and shall apply to all Letters of Credit to the extent not inconsistent with applicable Law.

2.6 Termination or Reduction of the Commitments.

(a) Optional. Borrower may at any time or from time to time, upon not less than three (3) Banking Days' notice to the Administrative Agent, terminate in whole or reduce in part the Commitments, provided that each partial reduction of the Commitments shall be in an aggregate amount of \$1,000,000 or an integral multiple of \$500,000 in excess thereof.

(b) Mandatory. The Commitments shall be automatically and permanently reduced to zero on the Maturity Date.

(c) Reduction Pro Rata; No Reinstatements. Each reduction of the Commitments shall be applied to the respective Commitments of the Lenders according to their respective Pro Rata Shares. Commitments once terminated or reduced may not be reinstated.

2.7 Administrative Agent's Right to Assume Funds Available for Advances. Unless the Administrative Agent shall have been notified by any relevant Lender no later than 10:00 a.m., California time, on the Banking Day of the proposed funding by the Administrative Agent of any Borrowing that such Lender does not intend to make available to the Administrative Agent such Lender's portion of the total amount of such Borrowing, the Administrative Agent may assume that such Lender has made such amount available to the Administrative Agent on the date of the Borrowing and the Administrative Agent may, in reliance upon such assumption, make available to Borrower a corresponding amount. If the Administrative Agent has made funds available to Borrower based on such assumption and such corresponding amount is not in fact made available to the Administrative Agent by such Lender, the Administrative Agent shall be entitled to recover such corresponding amount on

-35-

demand from such Lender. If such Lender does not pay such corresponding amount forthwith upon the Administrative Agent's demand therefor, the Administrative Agent promptly shall notify Borrower and Borrower shall pay such corresponding amount to the Administrative Agent. The Administrative Agent also shall be entitled to recover from such Lender interest on such corresponding amount in respect of each day from the date such corresponding amount was made available by the Administrative Agent to Borrower to the date such corresponding amount is recovered by the Administrative Agent, at a rate per annum equal to the daily Federal Funds Rate. Nothing herein shall be deemed to relieve any Lender from its obligation to fulfill its Commitments or to prejudice any rights which the Administrative Agent or Borrower may have against any Lender as a result of any default by such Lender hereunder.

2.8 Swing Line.

(a) The Swing Line Lender shall from time to time from the Closing Date through the day prior to the Maturity Date make Swing Line Loans to Borrower in such amounts as Borrower may request, provided that (a) after giving effect to such Swing Line Loan, Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount, (b) after giving effect to such Swing Line Loan, the Swing Line Outstandings do not exceed \$5,000,000 and/or, (c) without the consent of all of the Lenders, no Swing Line Loan may be made during the continuation of an Event of Default if written notice of such Event of Default shall have been provided to Swing Line Lender by the Administrative Agent or a Lender sufficiently in advance of the making of such Swing Line Loan. Borrower may borrow, repay and reborrow under this Section. Borrowings under the Swing Line may be made in amounts which are integral multiples of \$100,000 (or the remaining availability under the Swing Line) upon telephonic request by a Responsible Official of Borrower made to the Administrative Agent not later than 2:00 p.m., California time, on the Banking Day of the requested borrowing (which telephonic request shall be promptly confirmed in writing by telecopier or electronic mail). Promptly after receipt of such a request for borrowing, the Administrative Agent shall provide telephonic verification to the Swing Line Lender that, after giving effect to such request, availability for Loans will exist under Section 2.1(a) (and such verification shall be promptly confirmed in writing by telecopier or electronic mail). Each repayment of a Swing Line Loan shall be in an amount which is an integral multiple of \$100,000 (or the Swing Line Outstandings). Borrower shall notify the Swing Line Lender of its intention to make a repayment of a Swing Line Loan not later than 1:00 p.m. California time on the date of repayment. If Borrower instructs the Swing Line Lender to debit its demand deposit account at the Swing Line Lender in the amount of any payment with respect to a Swing Line Loan, or the Swing Line Lender otherwise receives repayment, after 3:00 p.m., California time, on a Banking Day, such payment shall be deemed received on the next Banking Day. The Swing Line Lender shall promptly notify the Administrative Agent of the Swing Loan Outstandings each time there is a change therein.

-36-

(b) Swing Line Loans shall bear interest at a fluctuating rate per annum equal to the Alternate Base Rate plus the Applicable Alternate Base Rate Margin. Interest shall be payable on such dates, not more frequent than monthly, as may be specified by the Swing Line Lender and in any event on the Maturity Date. The Swing Line Lender shall be responsible for invoicing Borrower for such interest. The interest payable on Swing Line Loans is solely for the account of the Swing Line Lender (subject to clause (d) below).

(c) Subject to subsection (e) below, the principal amount of all Swing Line Loans shall be due and payable on the earlier of (i) the maturity date agreed to by the Swing Line Lender and Borrower with respect to such loan (which maturity date shall not be a date more than ten (10) consecutive Banking Days from the date of advance thereof) or (ii) the Maturity Date.

(d) Upon the making of a Swing Line Loan, each Lender shall be deemed to have purchased from the Swing Line Lender a participation therein in an amount equal to that Lender's Pro Rata Share of the Revolving Facility times the amount of the Swing Line Loan. Upon demand made by the Swing Line Lender, each Lender shall, according to its Pro Rata Share of the Revolving Facility, promptly provide to the Swing Line Lender its purchase price therefor in an amount equal to its participation therein. The obligation of each Lender to so provide its purchase price to the Swing Line Lender shall be absolute and unconditional (except only demand made by the Swing Line Lender) and shall not be affected by the occurrence of a Default or Event of Default; provided that no Lender shall be obligated to purchase its Pro Rata Share of (i) Swing Line Loans to the extent that, after giving effect to such Swing Line Loan, Revolving Credit Facility Usage exceeds the Maximum Revolving Credit Amount, (ii) Swing Line Loans to the extent that, after giving effect to such Swing Line Loan, Swing Line Outstandings exceed \$5,000,000 and (iii) any Swing Line Loan made (absent the consent of all of the Lenders) during the continuation of an Event of Default if written notice of such Event of Default shall have been provided to Swing Line Lender by the Administrative Agent or a Lender sufficiently in advance of the making of such Swing Line Loan. Each Lender that has provided to the Swing Line Lender the purchase price due for its participation in Swing Line Loans shall thereupon acquire a pro rata participation, to the extent of such payment, in the claim of the Swing Line Lender against Borrower for principal and interest and shall share, in accordance with that pro rata participation, in any principal payment made by Borrower with respect to such claim and in any interest payment made by Borrower (but only with respect to periods subsequent to the date such Lender paid the Swing Line Lender its purchase price) with respect to such claim.

(e) In the event that any Swing Line Loan remains outstanding for ten (10) consecutive Banking Days, then on the next Banking Day (unless Borrower has made other arrangements acceptable to the Swing Line Lender to repay such Swing Line Loan, in full), Borrower shall request a Loan pursuant to Section 2.1(a) sufficient to repay the aggregate principal amount of such Swing Line Loan together with any and all accrued and unpaid interest with respect thereto. In addition, the

-37-

Swing Line Lender may, at any time, in its sole discretion, by written notice to Borrower and the Lenders, demand payment of the Swing Line Loans by way of a Advance in the full amount or any portion of the Swing Line Outstandings. In each case, the Administrative Agent shall automatically provide the responsive Advances made by each Lender to the Swing Line Lender (which the Swing Line Lender shall then apply to the Swing Line Outstandings). In the event that Borrower fails to request a Loan within the time specified by Section 2.2 on any such date, the Administrative Agent may, but is not required to, without notice to or the consent of Borrower, cause Alternate Base Rate Advances to be made by the Lenders under the Revolving Facility in amounts which are sufficient to reduce the Swing Line Outstandings as required above. The proceeds of such Advances shall be paid directly to the Swing Line Lender for application to the Swing Line Outstandings.

-38-

3.1 Principal and Interest.

(a) Interest shall be payable on the outstanding daily unpaid principal amount of each Advance from the date thereof until payment in full is made and shall accrue and be payable at the rates set forth or provided for herein before and after Default, before and after maturity, before and after judgment, and before and after the commencement of any proceeding under any Debtor Relief Law, with interest on overdue interest at the Default Rate to the fullest extent permitted by applicable Laws.

(b) Interest accrued on each Alternate Base Rate Advance shall be due and payable on each Monthly Payment Date. Except as otherwise provided in Section 3.7, the unpaid principal amount of any Alternate Base Rate Advance shall bear interest at a fluctuating rate per annum equal to the Alternate Base Rate plus the Applicable Alternate Base Rate Margin. Each change in the interest rate under this Section 3.1(b) due to a change in the Alternate Base Rate shall take effect simultaneously with the corresponding change in the Alternate Base Rate.

(c) Interest accrued on each Eurodollar Rate Advance which is for a term of three months or less shall be due and payable on the last day of the related Eurodollar Period. Interest accrued on each other Eurodollar Rate Advance shall be due and payable on the date which is three months after the date such Eurodollar Rate Advance was made (and, in the event that all of the Lenders have approved a Eurodollar Period of longer than six months, every three months thereafter through the last day of the Eurodollar Period) and on the last day of the related Eurodollar Period. Except as otherwise provided in Section 3.7, the unpaid principal amount of any Eurodollar Rate Advance shall bear interest at a rate per annum equal to the Eurodollar Rate for that Eurodollar Rate Advance plus the Applicable Eurodollar Rate Margin.

(d) If not sooner paid, the principal Indebtedness evidenced by the Notes shall be payable as follows:

(i) the amount, if any, by which the principal Indebtedness evidenced by the Notes at any time exceeds the Maximum Revolving Credit Amount shall be payable immediately; and

(ii) the principal Indebtedness evidenced by the Notes shall in any event be payable on the Maturity Date.

(e) The principal Indebtedness evidenced by the Notes may, at any time and from time to time, voluntarily be paid or prepaid in whole or in part without premium or penalty, except that with respect to any voluntary prepayment under this

-39-

subsection, (i) any partial prepayment shall be not less than \$1,000,000 and shall be an integral multiple of \$500,000, except as provided in Section 2.8(a), (ii) the Administrative Agent shall have received written notice of any prepayment by 9:00 a.m. California time on the date that is (x) in the case of a Eurodollar Rate Advance three (3) Banking Days before the date of prepayment unless the prepayment is of a Eurodollar Rate Advance to be made at the end of its applicable Eurodollar Period and (y) in the case of an Alternate Base Rate Advance or a prepayment of a Eurodollar Rate Advance made at the end of its applicable Eurodollar Period, one (1) Banking Day before the date of prepayment, which notice shall identify the date and amount of the prepayment and the Advance(s) being prepaid, (iii) each prepayment of principal on any Eurodollar Rate Advance shall be accompanied by payment of interest accrued to the date of payment on the amount of principal paid, and (iv) any payment or prepayment of all or any part of any Eurodollar Rate Advance on a day other than the last day of the applicable Eurodollar Period shall be subject to Section 3.6(e).

3.2 Unused Revolving Facility Commitment Fee. From the Closing Date through the Maturity Date, Borrower shall pay to the Administrative Agent, for the ratable accounts of the applicable Lenders in accordance with their respective Pro Rata Shares, a commitment fee equal to the Applicable Commitment Fee Margin times the average daily amount by which the Maximum Revolving Credit Amount exceeds the sum of (a) the aggregate principal amount of funded Indebtedness then outstanding under the Notes plus (b) the Aggregate Effective Amount under all outstanding Letters of Credit. The commitment fee shall be payable quarterly in arrears on each Quarterly Payment Date.

3.3 Closing Fees; Arrangement Fee; Agency Fee etc.

(a) On the Closing Date, Borrower shall pay to the Closing Date Lenders, through the Administrative Agent, the closing fees in the amount heretofore agreed upon by letter agreement among Borrower and each Closing Date Lender. All such fees shall be fully earned when paid and shall be non-refundable.

(b) On the date of the execution hereof, Borrower shall pay to the Administrative Agent, for the sole account of the Lead Arranger, an arrangement fee in the amount heretofore agreed upon by letter agreement between Borrower and the Lead Arranger. Such arrangement fee is for the services of the Lead Arranger in arranging the credit facility under this Agreement, is fully earned as of the date hereof and is nonrefundable.

(c) Borrower shall pay to the Administrative Agent an annual agency fee in such amounts and at such times as heretofore agreed upon by letter agreement between Borrower and the Administrative Agent. The agency fee is for the services to be performed by the Administrative Agent in acting as Administrative Agent and is fully earned on the date paid. The agency fee paid to the Administrative Agent is solely for its own account and is nonrefundable.

-40-

3.4 Letter of Credit Fees. With respect to each Letter of Credit, Borrower shall pay the following fees:

(a) concurrently with the issuance of each Letter of Credit and on each Quarterly Payment Date thereafter so long as such Letter of Credit shall remain outstanding, to the Administrative Agent for the ratable accounts of the Lenders in accordance with their respective Pro Rata Shares, a standby letter of credit fee in an amount equal to the product of the then Applicable Letter of Credit Fee Rate times the then outstanding undrawn amount of such Letter of Credit, for the period commencing on such payment date and ending on the next succeeding Quarterly Payment Date or for the remaining term of such Letter of Credit, whichever is shorter; provided, however, that the applicable standby letter of credit fee payable in connection with the original issuance of any Letter of Credit (and on each anniversary date thereof if such Letter of Credit is renewed or extended) shall be no less than \$350; and

(b) concurrently with the issuance of each Letter of Credit, and on each Quarterly Payment Date thereafter so long as such Letter of Credit shall remain outstanding, to the Issuing Lender for its own account, a fronting fee equal to 0.125% (12.5 basis points) per annum on the daily average stated amount of such Letter of Credit.

In addition to the foregoing, in connection with a Letter of Credit and activity relating thereto, Borrower also shall pay amendment, transfer, issuance, negotiation and such other fees as the Issuing Lender normally charges, in the amounts set forth from time to time as the Issuing Lender's published scheduled fees for such services. Each of the fees payable with respect to Letters of Credit under this Section is earned when due and is nonrefundable.

3.5 Increased Commitment Costs. If any Lender shall determine in good faith that the introduction after the Closing Date of any applicable law, rule, regulation or guideline regarding capital adequacy, or any change therein or any change in the interpretation or administration thereof by any central bank or other Governmental Agency charged with the interpretation or administration thereof, or compliance by such Lender (or its Eurodollar Lending Office) or any corporation controlling such Lender, with any request, guideline or directive regarding capital adequacy (whether or not having the force of Law) of any such central bank or other authority not imposed as a result of such Lender's or such corporation's failure to comply with any other Laws, affects or would affect the amount of capital required or expected to be maintained by such Lender or any corporation controlling such Lender and (taking into consideration such Lender's or such corporation's policies with respect to capital adequacy and such Lender's desired return on capital) determines in good faith that the amount of such capital is increased, or the rate of return on capital is reduced, as a consequence of its obligations under this Agreement, then, within five (5) days after demand of such Lender, Borrower shall pay to such Lender, from time to time as specified in good faith by such Lender, additional amounts sufficient to compensate such Lender in light of such circumstances, to the extent reasonably allocable to such obligations under this Agreement, provided that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand or is attributable

-41-

to periods prior to the date which is 180 days preceding the date of such demand. Each Lender's determination of such amounts shall be conclusive in the absence of manifest error.

3.6 Eurodollar Costs and Related Matters.

(a) In the event that any Governmental Agency imposes on any Lender any reserve or comparable requirement (including any emergency, supplemental or other reserve) with respect to the Eurodollar Obligations of that Lender, other than the Eurodollar Reserve Percentage, Borrower shall pay that Lender within five (5) days after demand all amounts necessary to compensate such Lender (determined as though such Lender's Eurodollar Lending Office had funded 100% of its Eurodollar Rate Advances in the Designated Eurodollar Market) in respect of the imposition of such reserve requirements (provided that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand or is attributable to periods prior to the date which is 180 days preceding the date of such demand). Any Lender's determination of such amount shall be conclusive in the absence of manifest error.

(b) If, after the date hereof, the existence or occurrence of any Special Eurodollar Circumstance:

(i) shall subject any Lender or its Eurodollar Lending Office to any tax, duty or other charge or cost with respect to any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances, or shall change the basis of taxation of payments to any Lender attributable to the principal of or interest on any Eurodollar Rate Advance or any other amounts due under this Agreement in respect of any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances, excluding (A) taxes imposed on or measured in whole or in part by its overall net income by (1) any jurisdiction (or political subdivision thereof) in which it is organized or maintains its principal office or Eurodollar Lending Office or (2) any jurisdiction (or political subdivision thereof) in which it is "doing business" and (B) any withholding taxes or other taxes based on gross income imposed by the United States of America for any period with respect to which it has failed to provide Borrower with the appropriate form or forms required by Section 11.21, to the extent such forms are then required by applicable Laws;

(ii) shall impose, modify or deem applicable any reserve not applicable or deemed applicable on the date hereof (including any reserve imposed by the Board of Governors of the Federal Reserve System (other than the Eurodollar Reserve Percentage), special deposit, capital or similar requirements against assets of, deposits with or for the account of, or credit extended by, any Lender or its Eurodollar Lending Office); or

-42-

(iii) shall impose on any Lender or its Eurodollar Lending Office or the Designated Eurodollar Market any other condition affecting any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances, its obligation to make Eurodollar Rate Advances or this Agreement, or shall otherwise affect any of the same;

and the result of any of the foregoing, as determined in good faith by such Lender, increases the cost to such Lender or its Eurodollar Lending Office of making or maintaining any Eurodollar Rate Advance or in respect of any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances or reduces the amount of any sum received or receivable by such Lender or its Eurodollar Lending Office with respect to any Eurodollar Rate Advance, any of its Notes evidencing Eurodollar Rate Advances or its obligation to make Eurodollar Rate Advances (assuming such Lender's Eurodollar Lending Office had funded 100% of its Eurodollar Rate Advances in the Designated Eurodollar Market), then, within five (5) Banking Days after demand by such Lender (with a copy to the Administrative Agent), Borrower shall pay to such Lender such additional amount or amounts as will compensate such Lender for such increased cost or reduction (determined as though such Lender's Eurodollar Lending Office had funded 100% of its Eurodollar Rate Advances in the Designated Eurodollar Market); provided that Borrower shall not be obligated to pay any such amount which arose prior to the date which is 180 days preceding the date of such demand or is attributable to periods prior to the date which is 180 days preceding the date of such demand. A statement of any Lender claiming compensation under this subsection shall be conclusive in the absence of manifest error.

(c) If, after the date hereof, the existence or occurrence of any Special Eurodollar Circumstance shall, in the good faith opinion of any Lender, make it unlawful or impossible for such Lender or its Eurodollar Lending Office to make, maintain or fund its portion of any Borrowing consisting of Eurodollar Rate Advances, or materially restrict the authority of such Lender to purchase or sell, or to take deposits of, Dollars in the Designated Eurodollar Market, or to determine or charge interest rates based upon the Eurodollar Rate, then such Lender's obligation to make Eurodollar Rate Advances shall be suspended for the duration of such illegality or impossibility and the Administrative Agent forthwith shall give notice thereof to the other Lenders and Borrower. Upon receipt of such notice, the outstanding principal amount of such Lender's affected Eurodollar Rate Advances, together with accrued interest thereon, automatically shall be converted to Alternate Base Rate Advances on either (i) the last day of the Eurodollar Period(s) applicable to such Eurodollar Rate Advances if such Lender may lawfully continue to maintain and fund such Eurodollar Rate Advances to such day(s) or (ii) immediately if such Lender may not lawfully continue to fund and maintain such Eurodollar Rate Advances to such day(s). Each Lender agrees to endeavor promptly to notify Borrower of any event of which it has actual knowledge, occurring after the Closing Date, which will cause such Lender to notify Borrower as set forth in the first sentence of this Section, and agrees to designate a different Eurodollar Lending Office if such designation will avoid the need

-43-

for such notice and will not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender. In the event that any Lender is unable, for the reasons set forth above, to make, maintain or fund any Eurodollar Rate Advance, such Lender shall fund such Eurodollar Dollar Rate Advance as an Alternate Base Rate Advance for the same period of time, and such amount shall be treated in all respects as an Alternate Base Rate Advance. In the event that any Lender's obligation to make Eurodollar Rate Advances has been suspended under this Section, such Lender shall promptly notify the Administrative Agent and Borrower of the cessation of the Special Eurodollar Circumstance which gave rise to such suspension.

(d) If, with respect to any proposed Borrowing comprised of Eurodollar Rate Advances:

(i) the Administrative Agent reasonably determines that, by reason of circumstances affecting the Designated Eurodollar Market generally that are beyond the reasonable control of the Lenders, deposits in Dollars (in the applicable amounts) are not being offered to any Lender in the Designated Eurodollar Market for the applicable Eurodollar Period; or

(ii) the Requisite Lenders advise the Administrative Agent that the Eurodollar Rate as determined by the Administrative Agent (A) does not represent the effective pricing to such Lenders for deposits in Dollars in the Designated Eurodollar Market in the relevant amount for the applicable Eurodollar Period, or (B) will not adequately and fairly reflect the cost to such Lenders of making the applicable Eurodollar Rate Advances;

then the Administrative Agent forthwith shall give notice thereof to Borrower and the Lenders, whereupon until the Administrative Agent notifies Borrower that the circumstances giving rise to such suspension no longer exist, the obligation of the Lenders to make any future Eurodollar Rate Advances shall be suspended.

(e) Upon payment or prepayment of any Eurodollar Rate Advance on a day other than the last day in the applicable Eurodollar Period (whether voluntarily, involuntarily, by reason of acceleration, or otherwise), or upon the failure of Borrower (for a reason other than the breach by a Lender of its obligation pursuant to Section 2.1(a)) to borrow on the date or in the amount specified for a Borrowing comprised of Eurodollar Rate Advances in any Request for Borrowing, Borrower shall pay to the appropriate Lender within five (5) Banking Days after demand a prepayment fee or failure to borrow fee, as the case may be (determined as though 100% of the Eurodollar Rate Advance had been funded in the Designated Eurodollar Market) equal to the sum of:

(1) the amount, if any, by which (i) the additional interest would have accrued on the amount prepaid or not borrowed at the Eurodollar Rate plus the Applicable Eurodollar Rate Margin if that amount had remained or been outstanding through the last day of the applicable Eurodollar Period

-44-

exceeds (ii) the interest that such Lender could recover by placing such amount on deposit in the Designated Eurodollar Market for a period beginning on the date of the prepayment or failure to borrow and ending on the last day of the applicable Eurodollar Period (or, if no deposit rate quotation is available for such period, for the most comparable period for which a deposit rate quotation may be obtained); plus

(2) all out-of-pocket expenses incurred by such Lender reasonably attributable to such payment, prepayment or failure to borrow.

Each Lender's determination of the amount of any prepayment fee payable under this Section shall be conclusive in the absence of manifest error.

(f) Each Lender agrees to endeavor promptly to notify Borrower of any event of which it has actual knowledge, occurring after the Closing Date, which will entitle such Lender to compensation pursuant to clause (a) or clause (b) of this Section, and agrees to designate a different Eurodollar Lending Office if such designation will avoid the need for or reduce the amount of such compensation and will not, in the good faith judgment of such Lender, otherwise be materially disadvantageous to such Lender. Any request for compensation by a Lender under this Section shall set forth the basis upon which it has been determined that such an amount is due from Borrower, a calculation of the amount due, and a certification that the corresponding costs have been incurred by such Lender.

3.7 Late Payments and Default Rate. If any installment of principal or interest or any fee or cost or other amount payable under any Loan Document to the Administrative Agent or any Lender is not paid when due, it shall thereafter bear interest at a fluctuating interest rate per annum at all times equal to the sum of the interest rate otherwise applicable thereto hereunder (or, if no interest rate is otherwise applicable thereto hereunder, the Alternate Base Rate) plus 2.00% (the "Default Rate"), to the fullest extent permitted by applicable Laws. While any Event of Default exists or after acceleration, at the option of the Requisite Lenders, Borrower shall pay interest (after as well as before entry of judgment thereon to the extent permitted by Law) on the principal amount of all outstanding Obligations, at the Default Rate, to the fullest extent permitted by Law. Accrued and unpaid interest on past due amounts (including interest on past due interest) shall be compounded monthly, on the last day of each calendar month, to the fullest extent permitted by applicable Laws.

3.8 Computation of Interest and Fees. Computation of interest on Alternate Base Rate Advances calculated with reference to the Prime Rate shall be calculated on the basis of a year of 365 or 366 days, as the case may be, and the actual number of days elapsed; computation of interest on Alternate Base Rate Advances calculated by reference to the Federal Funds Rate, and on Eurodollar Rate Advances and all fees under this Agreement shall be calculated on the basis of a year of 360 days and the actual number of days elapsed. Borrower acknowledges that such latter calculation method will result in a higher yield to the Lenders than a method based on a year of 365 or 366 days. Interest shall accrue on each

-45-

Advance for the day on which the Advance is made; interest shall not accrue on an Advance, or any portion thereof, for the day on which the Advance or such portion is paid. Any Advance that is repaid on the same day on which it is made shall bear interest for one day. Notwithstanding anything in this Agreement to the contrary, interest in excess of the maximum amount permitted by applicable Laws shall not accrue or be payable hereunder or under the Notes, and any amount paid as interest hereunder or under the Notes which would otherwise be in excess of such maximum permitted amount shall instead be treated as a payment of principal.

3.9 Non-Banking Days. If any payment to be made by Borrower or any other Party under any Loan Document shall come due on a day other than a Banking Day, payment shall instead be considered due on the next succeeding Banking Day and the extension of time shall be reflected in computing interest and fees.

# 3.10 Manner and Treatment of Payments.

(a) Each payment hereunder (except payments pursuant to Sections 3.4, 3.5, 11.3, 11.11 and 11.21) or on the Notes or under any other Loan Document shall be made to the Administrative Agent at the Administrative Agent's Office, in immediately available funds not later than 11:00 a.m. California time, on the day of payment (which must be a Banking Day). All payments received after such time, on any Banking Day, shall be deemed received on the next succeeding Banking Day. The amount of all payments received by the Administrative Agent for the account of each Lender shall be immediately paid by the Administrative Agent to the applicable Lender in immediately available funds and, if such payment was received by the Administrative Agent of a Lender on that Banking Day, the Administrative Agent shall reimburse that Lender for the cost to such Lender of funding the amount of such payment at the Federal Funds Rate. All payments shall be made in lawful money of the United States of America.

(b) Borrower hereby authorizes the Administrative Agent to debit the Designated Deposit Account to effect any payment due to the Lenders or the Administrative Agent pursuant to this Agreement. Any resulting overdraft in the Designated Deposit Account shall be payable by Borrower to the Administrative Agent on the next following Banking Day.

(c) Each payment or prepayment on account of any Borrowing shall be applied pro rata according to the outstanding Advances made by each Lender comprising such Borrowing.

(d) Each Lender shall use its best efforts to keep a record (in writing or by an electronic data entry system) of Advances made by it and payments received by it with respect to its Note and, subject to Section 10.6(g), such record shall, as against Borrower, be presumptive evidence of the amounts owing.

Notwithstanding the foregoing sentence, the failure by any Lender to keep such a record shall not affect Borrower's obligation to pay the Obligations.

(e) Each payment of any amount payable by Borrower or any other Party to any Lender under this Agreement or any other Loan Document shall be made free and clear of, and without reduction by reason of, any taxes, assessments or other charges imposed by any Governmental Agency, central bank or comparable authority, excluding (i) taxes imposed on or measured in whole or in part by its overall net income and franchise taxes imposed in lieu of net income taxes by (A) any jurisdiction (or political subdivision thereof) in which it is organized or maintains its principal office or Eurodollar Lending Office or (B) any jurisdiction (or political subdivision thereof) in which it is "doing business" and (ii) any withholding taxes or other taxes based on gross income imposed by the United States of America for any period with respect to which it has failed to provide Borrower with the appropriate form or forms required by Section 11.21, to the extent such forms are then required by applicable Laws (all such non-excluded taxes, assessments or other charges being hereinafter referred to as "Taxes"). To the extent that Borrower is obligated by applicable Laws to make any deduction or withholding on account of Taxes from any amount payable to any Lender under this Agreement, Borrower shall (1) make such deduction or withholding and pay the same to the relevant Governmental Agency and (2) pay such additional amount to that Lender as is necessary to result in that Lender's receiving a net after-Tax amount equal to the amount to which that Lender would have been entitled under this Agreement absent such deduction or withholding. If and when receipt of such payment results in an excess payment or credit to that Lender on account of such Taxes, that Lender shall promptly refund such excess to Borrower.

3.11 Funding Sources. Nothing in this Agreement shall be deemed to obligate any Lender to obtain the funds for any Advance in any particular place or manner or to constitute a representation by any Lender that it has obtained or will obtain the funds for any Advance in any particular place or manner.

3.12 Failure to Charge Not Subsequent Waiver. Any decision by the Administrative Agent or any Lender not to require payment of any interest (including interest arising under Section 3.7), fee, cost or other amount payable under any Loan Document, or to calculate any amount payable by a particular method, on any occasion shall in no way limit or be deemed a waiver of the Administrative Agent's or such Lender's right to require full payment of any interest (including interest arising under Section 3.7), fee, cost or other amount payable under any Loan Document, or to calculate an amount payable by another method that is not inconsistent with this Agreement, on any other or subsequent occasion, except as provided in Sections 3.5 and 3.6.

3.13 Administrative Agent's Right to Assume Payments Will be Made. Unless the Administrative Agent shall have been notified by Borrower prior to the date on which any payment to be made by Borrower hereunder is due that Borrower does not intend to remit such payment (or otherwise cause sufficient funds to be available in the Designated Deposit Account for debit pursuant to Section 3.10(b)), the Administrative Agent may, in its

-47-

discretion, assume that Borrower has remitted such payment (or caused funds sufficient to make such payment to be available) when so due and the Administrative Agent may, in its discretion and in reliance upon such assumption, make available to each Lender on such payment date, an amount equal to such Lender's share of such assumed payment. If Borrower has not in fact remitted such payment (or caused funds sufficient to make such payment to be available) to the Administrative Agent, each Lender shall forthwith on demand repay to the Administrative Agent the amount of such assumed payment made available to such Lender, together with interest thereon in respect of each day from and including the date such amount was made available by the Administrative Agent to such Lender to the date such amount is repaid to the Administrative Agent at the Federal Funds Rate.

3.14 Fee Determination Detail. The Administrative Agent, and any Lender, shall provide reasonable detail to Borrower regarding the manner in which the amount of any payment to the Administrative Agent and the Lenders, or that Lender, under Article 3 has been determined, concurrently with demand for such payment.

3.15 Survivability. All of Borrower's obligations under Sections 3.4 and 3.5 shall survive for the one year period following the Termination Date, and Borrower shall remain obligated thereunder for all claims under such Sections made by any Lender to Borrower prior to the expiration of such period.

-48-

### Article 4. REPRESENTATIONS AND WARRANTIES

Borrower represents and warrants to the Administrative Agent and each of the Lenders that:

4.1 Existence and Qualification; Power; Compliance With Laws. Borrower is a corporation duly formed, validly existing and in good standing under the Laws of the State of California. Borrower is duly qualified or registered to transact business and is in good standing in the State of California, and each other jurisdiction in which the conduct of its business or the ownership or leasing of its Properties makes such qualification or registration necessary, except where the failure so to qualify or register and to be in good standing could not reasonably be expected to have a Material Adverse Effect. Borrower has all requisite power and authority to conduct its business, to own and lease its Properties and to execute and deliver each Loan Document to which it is a Party and to perform its Obligations. The chief executive offices of Borrower are located in San Dimas, California. All outstanding capital stock of Borrower is duly authorized, validly issued, fully paid and non-assessable, and no holder thereof has any enforceable right of rescission under any applicable state or federal securities or other Laws. Borrower is in compliance with all Laws and other legal requirements applicable to its business, has obtained all authorizations, consents, approvals, orders, licenses and permits from, and has accomplished all filings, registrations and qualifications with, or obtained exemptions from any of the foregoing from, any Governmental Agency that are necessary for the transaction of its business, except where the failure so to comply with Laws and other legal requirements applicable to its business, obtain authorizations, etc., file, register, qualify or obtain exemptions could not reasonably be expected to have a Material Adverse Effect.

4.2 Authority; Compliance With Other Agreements and Instruments and Government Regulations. The execution and delivery by Borrower of the Loan Documents to which it is a Party and payment of the Obligations have been duly authorized by all necessary corporate or company action, as applicable, and do not and will not:

(a) Require any consent or approval not heretofore obtained of any partner, director, stockholder, member, security holder or creditor of Borrower;

(b) Violate or conflict with any provision of Borrower's charter, certificate of incorporation, bylaws, or other organizational documents, as applicable;

(c) Result in or require the creation or imposition of any Lien (other than pursuant to the Loan Documents) or Right of Others upon or with respect to any Property now owned or leased or hereafter acquired by Borrower;

(d) Violate any Requirement of Law applicable to Borrower;

(e) Result in a breach of or constitute a default under, or cause or permit the acceleration of any obligation owed under, any indenture or loan or credit

-49-

agreement or any other Contractual Obligation to which Borrower is a party or by which Borrower or any of its Property is bound or affected;

and Borrower is not in violation of, or default under, any Requirement of Law or Contractual Obligation, or any indenture, loan or credit agreement described in Section 4.2(e), in any respect that could reasonably be expected to have a Material Adverse Effect.

4.3 No Governmental Approvals Required. Except as previously obtained or made, no authorization, consent, approval, order, license or permit from, or filing, registration or qualification with, any Governmental Agency is or will be required to authorize or permit under applicable Laws the execution and delivery by Borrower of the Loan Documents to which it is a Party and payment of the Obligations.

4.4 Subsidiaries.

(a) Schedule 4.4 hereto correctly sets forth, as of the Closing Date, the names, form of legal entity, number of shares of capital stock or membership or other equity interests, as applicable, issued and outstanding, number of shares of capital stock or membership or other equity interests, as applicable, owned by Borrower or any Subsidiary of Borrower (specifying such owner) and jurisdictions of organization of all Subsidiaries of Borrower. Except as described in Schedule 4.4, as of the Closing Date, Borrower does not own any capital stock, membership interest, other equity interest or debt Security which is convertible, or exchangeable, for capital stock, membership interests or other equity interests in any Person. Unless otherwise indicated in Schedule 4.4, as of the Closing Date, all of the outstanding shares of capital stock, all of the outstanding membership interests or all of the units of other equity interest, as the case may be, of each Subsidiary are owned of record and beneficially by Borrower, there are no outstanding options, warrants or other rights to purchase capital stock of any such Subsidiary, and all such shares, membership interests or other equity interests so owned are duly authorized, validly issued, fully paid and non-assessable, and were issued in compliance with all applicable state and federal securities and other Laws, and are free and clear of all Liens, except for Permitted Encumbrances and other encumbrances permitted pursuant to Section 6.9.

(b) As of the Closing Date, each Subsidiary is a legal entity of the type described in Schedule 4.4 duly formed, validly existing and, if such concept is legally recognized in such Subsidiary's jurisdiction of organization, in "good standing" under the Laws of its jurisdiction of organization, is duly qualified to do business as a foreign organization and, if such concept is legally recognized in any applicable jurisdiction, is in "good standing" as such in each jurisdiction in which the conduct of its business or the ownership or leasing of its Properties makes such qualification necessary (except where the failure to be so duly qualified and in good standing could not reasonably be expected to have a Material Adverse Effect), and has all requisite power and authority to conduct its business and to own and lease its Properties.

-50-

(c) Each Subsidiary is in compliance with all Laws and other requirements applicable to its business and has obtained all authorizations, consents, approvals, orders, licenses, and permits from, and each such Subsidiary has accomplished all filings, registrations, and qualifications with, or obtained exemptions from any of the foregoing from, any Governmental Agency that are necessary for the transaction of its business, except where the failure to be in such compliance, obtain such authorizations, consents, approvals, orders, licenses, and permits, accomplish such filings, registrations, and qualifications, or obtain such exemptions, could not reasonably be expected to have a Material Adverse Effect.

4.5 Financial Statements. Borrower has furnished to the Lenders (a) the audited consolidated financial statements of Borrower and its Subsidiaries for the Fiscal Year ended December 31, 2001 and (b) the consolidating and consolidated financial statements of Borrower and its Subsidiaries for the Fiscal Quarter ended March 31, 2002. Such financial statements fairly present in all material respects the financial condition, results of operations and changes in financial position as of such dates and for such periods in conformity with GAAP consistently applied.

4.6 No Other Liabilities; No Material Adverse Changes. As of the Closing Date, Borrower and its Subsidiaries do not have any material liability or material contingent liability required under GAAP to be reflected or disclosed, and not reflected or disclosed, in the financial statements described in Section 4.5, other than liabilities and contingent liabilities arising in the ordinary course of business since the date of such financial statements. As of the Closing Date, no circumstance or event has occurred that could reasonably be expected to have a Material Adverse Effect since December 31, 2001.

4.7 Title to and Location of Property. As of the Closing Date, except as set forth in Schedule 4.7, Borrower and its Subsidiaries have valid title to the Property (other than assets which are the subject of a Capital Lease Obligation) reflected in the financial statements described in Section 4.5, other than items of Property or exceptions to title which are in each case immaterial and Property subsequently sold or disposed of in the ordinary course of business. Such Property is free and clear of all Liens and Rights of Others, other than Liens or Rights of Others described in Schedule 4.7 and Permitted Encumbrances, other encumbrances permitted pursuant to Section 6.9, and Permitted Rights of Others.

4.8 Intangible Assets. Borrower and its Subsidiaries own, or possess the right to use to the extent necessary in their respective businesses, all material trademarks, trade names, copyrights, patents, patent rights, computer software, licenses and other Intangible Assets that are used in the conduct of their businesses as now operated, and no such Intangible Asset, to the best knowledge of Borrower, conflicts with the valid trademark, trade name, copyright, patent, patent right or Intangible Asset of any other Person to the extent that such conflict could reasonably be expected to have a Material Adverse Effect. Schedule 4.8 sets forth all patents, patent applications, trademarks, trade names and trade styles used by Borrower or any of its Subsidiaries at any time within the five (5) year period ending on the Closing Date.

-51-

4.9 Litigation. Except for (a) any matter fully covered as to subject matter and amount (subject to applicable deductibles and retentions) by insurance for which the insurance carrier has not asserted lack of subject matter coverage or reserved its right to do so, (b) any matter, or series of related matters, involving a claim against Borrower or any of its Subsidiaries of less than \$250,000, (c) matters of an administrative nature not involving a claim or charge against Borrower or any Subsidiary of Borrower and (d) matters set forth in Schedule 4.9, there are no actions, suits, proceedings or investigations pending as to which Borrower or any of its Subsidiaries have been served or have received notice or, to the best knowledge of Borrower threatened against or affecting Borrower or any of its Subsidiaries or any Property of any of them before any Governmental Agency. None of Borrower, its Subsidiaries, or, to the best knowledge of Borrower, any executive officer of any such Persons has been indicted or convicted in connection with or is engaging in any criminal conduct which constitutes a felony, or is currently subject to any lawsuit or proceeding or, to the best of Borrower's knowledge, under investigation in connection with any anti-racketeering or criminal conduct or activity which constitutes a felony.

4.10 Binding Obligations. Each of the Loan Documents to which Borrower is a Party will, when executed and delivered by Borrower, constitute the legal, valid and binding obligation of Borrower, enforceable against Borrower in accordance with its terms, except as enforcement may be limited by Debtor Relief Laws or equitable principles relating to the granting of specific performance and other equitable remedies as a matter of judicial discretion.

 $\ensuremath{4.11}$  No Default. No event has occurred and is continuing that is a Default or Event of Default.

4.12 ERISA.

(a) With respect to each Pension Plan:

(i) such Pension Plan complies in all material respects with ERISA and any other applicable Laws to the extent that noncompliance could reasonably be expected to have a Material Adverse Effect;

(ii) such Pension Plan has not incurred any "accumulated funding deficiency" (as defined in Section 302 of ERISA) that could reasonably be expected to have a Material Adverse Effect;

(iii) no "reportable event" (as defined in Section 4043 of ERISA, but excluding such events as to which the PBGC has by regulation waived the requirement therein contained that it be notified within thirty days of the occurrence of such event) has occurred that could reasonably be expected to have a Material Adverse Effect; and

-52-

(iv) neither Borrower nor any of its Subsidiaries has engaged in any non-exempt "prohibited transaction" (as defined in Section 4975 of the Code) that could reasonably be expected to have a Material Adverse Effect.

(b) Neither Borrower nor any of its Subsidiaries has incurred or expects to incur any withdrawal liability to any Multiemployer Plan that could reasonably be expected to have a Material Adverse Effect.

4.13 Regulation U; Investment Company Act. No part of the proceeds of any Advance hereunder will be used to purchase or carry, or to extend credit to others for the purpose of purchasing or carrying, any Margin Stock in violation of Regulation U. Neither Borrower nor any of its Subsidiaries is or is required to be registered as an "investment company" under the Investment Company Act of 1940.

4.14 Disclosure. No written statement made by a Senior Officer of Borrower to the Administrative Agent or any Lender pursuant to this Agreement, or in connection with any Advance, as of the date thereof contained any untrue statement of a material fact or omitted a material fact necessary to make the statement made not misleading in light of all the circumstances existing at the date the statement was made.

4.15 Tax Liability. Borrower and its Subsidiaries have filed all tax returns which are required to be filed, and have paid, or made provision for the payment of, all taxes with respect to the periods, Property or transactions covered by said returns, or pursuant to any assessment received by Borrower or any of its Subsidiaries, except (a) such taxes, if any, as are being contested in good faith by appropriate proceedings and as to which adequate reserves have been established and maintained and (b) immaterial taxes so long as no material Property of Borrower or any Subsidiary is at impending risk of being seized, levied upon or forfeited.

4.16 Projections. As of the Closing Date, to the best knowledge of Borrower the assumptions set forth in the Projections are reasonable and consistent with each other and with all facts known to Borrower, and the Projections are reasonably based on such assumptions. Nothing in this Section 4.16 shall be construed as a representation or covenant that the Projections in fact will be achieved.

4.17 Hazardous Materials. Except as described in Schedule 4.17, or as may subsequently be disclosed by Borrower in writing to the Administrative Agent, (a) neither Borrower nor any of its Subsidiaries at any time has disposed of, discharged, released or threatened the release of any Hazardous Materials on, from or under the Real Property in violation of any Hazardous Materials Law that would individually or in the aggregate constitute a Material Adverse Effect, (b) no condition exists that violates any Hazardous Material Law affecting any Real Property except for such violations that would not individually or in the aggregate constitute a Material Adverse Effect, (c) no Real Property or any portion thereof is or has been utilized by Borrower or any Subsidiary as a site for the manufacture of any Hazardous Materials and (d) to the extent that any Hazardous Materials are used, generated or stored by Borrower or any Subsidiary on any Real Property, or

-53-

transported to or from such Real Property by Borrower, or any Subsidiary, such use, generation, storage and transportation are in compliance in all material respects with all Hazardous Materials Laws.

4.18 Employee Matters. There is no strike, work stoppage or labor dispute with any union or group of employees pending or, to the best knowledge of Borrower overtly threatened involving Borrower or any Subsidiary that individually or in the aggregate could reasonably be expected to have a Material Adverse Effect.

 $4.19\ {\rm Fiscal}\ {\rm Year}.$  Borrower and its Subsidiaries each operate on a fiscal year ending on December 31.

4.20 Solvency. After giving effect to this Agreement and the other Loan Documents (including after giving effect to Advances under this Agreement as of the Closing Date), Borrower shall be Solvent.

-54-

# Article 5. AFFIRMATIVE COVENANTS (OTHER THAN INFORMATION AND REPORTING REQUIREMENTS)

So long as any Advance remains unpaid, or any other Obligation remains unpaid, or any portion of any of the Commitments remains in force, Borrower shall, and shall cause each of the Subsidiaries to, unless the Administrative Agent (with the written approval of the Requisite Lenders) otherwise consents:

5.1 Payment of Taxes and Other Potential Liens. Pay and discharge promptly all taxes, assessments and governmental charges or levies imposed upon any of them, upon their respective Property or any part thereof and upon their respective income or profits or any part thereof, except that Borrower and its Subsidiaries shall not be required to pay or cause to be paid (a) any tax, assessment, charge or levy that is not yet past due, or is being contested in good faith by appropriate proceedings so long as the relevant entity has established and maintains adequate reserves for the payment of the same or (b) any immaterial tax so long as no material Property of Borrower or any Subsidiary is at impending risk of being seized, levied upon or forfeited.

5.2 Preservation of Existence. Preserve and maintain their respective existences (except as permitted by Section 6.4) in their respective jurisdictions of formation and all material authorizations, rights, franchises, privileges, consents, approvals, orders, licenses, permits, or registrations from any Governmental Agency that are necessary for the transaction of their respective business and qualify and remain qualified to transact business in each jurisdiction in which such qualification is necessary in view of their respective business or the ownership or leasing of their respective Properties except where the failure to so qualify or remain qualified could not reasonably be expected to have a Material Adverse Effect.

5.3 Maintenance of Properties. Maintain, preserve and protect all of their respective Properties in good order and condition, subject to wear and tear in the ordinary course of business, and not permit any waste of their respective Properties, except (a) that the failure to maintain, preserve and protect a particular item of Property that is at the end of its useful life or that is not of significant value, either intrinsically or to the operations of Borrower and Subsidiaries, taken as a whole, shall not constitute a violation of this covenant, and (b) this covenant shall not be construed to prohibit any Disposition otherwise permitted pursuant to Section 6.3.

5.4 Maintenance of Insurance. Maintain, or cause to be maintained, liability, casualty and other insurance (subject to customary deductibles and retentions) with responsible insurance companies in such amounts and against such risks as is carried by responsible companies engaged in similar businesses and owning similar assets.

5.5 Compliance With Laws. Comply with all Requirements of Law noncompliance with which could reasonably be expected to have a Material Adverse Effect,

-55-

except that Borrower and its Subsidiaries need not comply with a Requirement of Law then being contested by any of them in good faith by appropriate proceedings.

5.6 Inspection Rights. Upon reasonable notice, at any time during regular business hours and, as requested by the Administrative Agent (but not so as to materially interfere with the business of Borrower or any of the Subsidiaries) permit the Administrative Agent, or any Lender, or any authorized employee, agent or representative thereof, to examine, audit and make copies and abstracts from the records and books of account of (including any software or CD Rom files relating thereto), and to visit and inspect the Properties of, Borrower and its Subsidiaries and to discuss the affairs, finances and accounts of Borrower and its Subsidiaries with any of their officers, key employees or accountants and, upon request, furnish promptly to the Administrative Agent or any Lender true copies of all financial information made available to the board of directors or audit committee of the board of directors of Borrower. If any of the Properties, books or records of Borrower or any of the Subsidiaries are in the possession of a third party, Borrower authorizes that third party to permit the Administrative Agent or any Lender or any agents thereof to have access to perform inspections or audits and to respond to the Administrative Agent's or any Lender's request for information concerning such Properties, books and records. Notwithstanding the foregoing, no prior notice of any such examination, audit, visit, inspection or discussion shall be required if an Event of Default has occurred and remains in effect.

5.7 Keeping of Records and Books of Account. Keep adequate records and books of account reflecting all financial transactions in conformity with GAAP, consistently applied, and in material conformity with all applicable requirements of any Governmental Agency having regulatory jurisdiction over Borrower and its Subsidiaries.

5.8 Compliance With Agreements. Promptly and fully comply with all Contractual Obligations to which any one or more of them is a party, except for any such Contractual Obligations (a) the performance of which would cause a Default or (b) then being contested by any of them in good faith by appropriate proceedings or (c) if the failure to comply could not reasonably be expected to have a Material Adverse Effect.

5.9 Use of Proceeds. Use the proceeds of all Advances to (a) refinance certain Indebtedness of Borrower and its Subsidiaries (including SCW), (ii) finance certain acquisitions and (c) provide for the working capital and general corporate purpose need of Borrower and its Subsidiaries.

5.10 Hazardous Materials Laws. Keep and maintain all Real Property and each portion thereof in compliance in all material respects with all applicable Hazardous Materials Laws and promptly notify the Administrative Agent in writing (attaching a copy of any pertinent written material) of (a) any and all material enforcement, cleanup, removal or other governmental or regulatory actions instituted, completed or threatened in writing by a Governmental Agency pursuant to any applicable Hazardous Materials Laws, (b) any and all material claims made or threatened in writing by any Person against Borrower or any Subsidiary relating to damage, contribution, cost recovery, compensation, loss or injury resulting from any Hazardous Materials and (c) discovery by any Senior Officer of Borrower

-56-

of any material occurrence or condition on any real property adjoining or in the vicinity of such Real Property that could reasonably be expected to cause such Real Property or any part thereof to be subject to any restrictions on the ownership, occupancy, transferability or use of such Real Property under any applicable Hazardous Materials Laws.

5.11 Minimum Debt Rating. Maintain at all times a Debt Rating (in the case of Borrower only) equal to (or better than) Baal or BBB+.

5.12 Syndication Process. Cooperate in such respects as may be reasonably requested by the Lead Arranger in connection with the syndication of the credit facility under this Agreement, including the provision of information (in form and substance acceptable to the Lead Arranger) for inclusion in written materials furnished to prospective syndicate members and the participation by Senior Officers of Borrower and its Subsidiaries in meetings with prospective syndicate members.

-57-

# Article 6. NEGATIVE COVENANTS

So long as any Advance remains unpaid, or any other Obligation remains unpaid, or any portion of any of the Commitments remains in force, Borrower shall not, and shall not permit any of its Subsidiaries to, unless the Administrative Agent (with the written approval of the Requisite Lenders or, if required by Section 11.2, all of the Lenders) otherwise consents:

6.1 Prepayment of Indebtedness. Prepay any principal or interest on any Indebtedness of Borrower or any Subsidiary prior to the date when due, or make any payment or deposit with any Person that has the effect of providing for the satisfaction of any Indebtedness of Borrower or any Subsidiary prior to the date when due, except (a) Indebtedness to the Administrative Agent and the Lenders under this Agreement and the other Loan Documents and (b) Indebtedness to other Persons the prepayment of which is approved in advance by the Requisite Lenders in writing.

6.2 Prepayment of Subordinated Obligations. Pay any (a) principal (including sinking fund payments) or any other amount (other than scheduled interest payments) with respect to any Subordinated Obligation, or purchase or redeem any Subordinated Obligation or deposit any monies, Securities or other Property with any trustee or other Person to provide assurance that the principal or any portion thereof of any Subordinated Obligation will be paid when due or otherwise to provide for the defeasance of any Subordinated Obligation or (b) scheduled interest on any Subordinated Obligation if the payment thereof is then prohibited under the terms of the subordination provisions governing such Subordinated Obligations.

6.3 Disposition of Property. Make any Disposition of its Property, whether now owned or hereafter acquired, except (a) Dispositions of obsolete Property or Property with no material remaining useful life and (b) Dispositions in an aggregate amount not to exceed \$1,000,000 in any Fiscal Year ending after the Closing Date or \$5,000,000 in the aggregate from and after the Closing Date to the Termination Date; provided that (i) at the time of any such Disposition pursuant to clause (b) only, no Default or Event of Default shall exist or shall result from such Disposition and (ii) the sales price relating to a Disposition (pursuant to clause (a) or (b)) shall be paid in Cash and/or Indebtedness or other evidence of an Investment permitted pursuant to Section 6.14(h).

6.4 Mergers. Merge or consolidate with or into any Person, except mergers and consolidations of a Subsidiary into Borrower (with Borrower as the surviving entity), mergers of Subsidiaries with each other or mergers entered into in connection with Permitted Acquisitions, provided that (a) no Default or Event of Default would result therefrom and (b) any such "surviving" entity shall have executed such amendments to the Loan Documents, if any, as the Administrative Agent may reasonably determine are appropriate as a result of such merger.

-58-

6.5 Hostile Tender Offers. Make any offer to purchase or acquire, or consummate a purchase or acquisition of, five percent (5%) or more of the voting interest in any corporation or other business entity if the board of directors or management of such corporation or business entity has notified Borrower that it opposes such offer or purchase and such notice has not been withdrawn or superseded.

6.6 Distributions. Declare or pay or make any form of Distribution, whether from capital, income or otherwise, and whether in Cash or other Property, except:

(a) Distributions by any Subsidiary to Borrower or to any wholly-owned Subsidiary of Borrower;

(b) Distributions consisting of dividends payable solely in capital stock or rights to purchase capital stock so long as no Default or Event of Default then exists; and

(c) Distributions consisting of (i) repurchases of preferred stock of Borrower in an amount not to exceed \$2,500,000 in the aggregate from and after the Closing Date and (ii) dividends paid in Cash in a manner reasonably consistent with Borrower's past practices made in any Fiscal Year if, in any such case, no Default or Event of Default then exists and, giving effect thereto on a pro-forma estimated basis, Borrower would be in compliance with Section 6.12 as of the end of the then current Fiscal Quarter.

6.7 ERISA. (a) At any time, permit any Pension Plan to: (i) engage in any non-exempt "prohibited transaction" (as defined in Section 4975 of the Code); (ii) fail to comply with ERISA or any other applicable Laws; (iii) incur any material "accumulated funding deficiency" (as defined in Section 302 of ERISA); or (iv) terminate in any manner, which, with respect to each event listed above, could reasonably be expected to result in a Material Adverse Effect or (b) withdraw, completely or partially, from any Multiemployer Plan if to do so could reasonably be expected to result in a Material Adverse Effect.

6.8 Change in Nature of Business. Make any change in the nature of the business of Borrower and its Subsidiaries, taken as a whole, as at present conducted; provided that a Permitted Acquisition shall not be deemed such a change.

6.9 Liens and Negative Pledges. Create, incur, assume or suffer to exist any Lien or Negative Pledge of any nature upon or with respect to any of their respective Properties, or engage in any Sale and Leaseback transaction with respect to any of their respective Properties, whether now owned or hereafter acquired, except:

(a) Liens and Negative Pledges existing on the Closing Date and disclosed in Schedule 4.7 and any renewals/extensions or amendments thereof, provided that the obligations secured or benefited thereby are not increased;

(b) Liens and Negative Pledges under the Loan Documents;

-59-

#### (c) Permitted Encumbrances;

(d) Liens on Property acquired by Borrower or any Subsidiary that were in existence at the time of the acquisition of such Property and were not created in contemplation of such acquisition provided that the aggregate amount of Indebtedness of Borrower and its subsidiaries secured by such liens shall not at any time exceed \$10,000,000; and

(e) Liens securing Permitted Capital Asset Indebtedness on and limited to the capital assets acquired, constructed or financed with the proceeds of such Permitted Capital Asset Indebtedness or with the proceeds of any Indebtedness directly or indirectly refinanced by such Indebtedness; provided that the aggregate principal amount of such Indebtedness secured by such Liens and incurred by Borrower and/or its Subsidiaries after the Closing Date shall not exceed \$10,000,000 at any one time outstanding (as determined in accordance with GAAP consistently applied).

6.10 Indebtedness and Guaranty Obligations. Create, incur or assume any Indebtedness or Guaranty Obligation if an Event of Default has occurred and is continuing or if, after giving effect thereto, Borrower would not be in compliance with the provisions of Section 6.12 or Section 6.13 or an Event of Default would otherwise occur. Notwithstanding the foregoing, Borrower shall not permit any Subsidiary to create, incur, assume or suffer to exist any Indebtedness or Guaranty Obligation, except (a) Indebtedness permitted under Sections 6.9(d) and 6.9(e), (b) Indebtedness owed to Borrower or a wholly-owned Subsidiary and (c) unsecured term Indebtedness (i.e., not revolving credit) that (i) has a longer weighted average life than the Borrowings hereunder, (ii) is permitted by any applicable Governmental Agency having regulatory jurisdiction over such Subsidiary, (iii) is incurred in the ordinary course of business of such Subsidiary in a manner substantially consistent with the prior practices of SCW and (iv) is provided by financial institutions other than commercial banks.

6.11 Transactions with Affiliates. Enter into any transaction of any kind with any Affiliate of Borrower other than (without duplication): (a) salary, bonus, employee stock option and other compensation arrangements with directors or officers in the ordinary course of business; (b) Investments permitted pursuant to Section 6.14(d), (c) Distributions permitted pursuant to Section 6.6; (d) transactions with wholly-owned Subsidiaries; and (e) transactions on overall terms at least as favorable to Borrower or its Subsidiaries as would be the case in an arm's-length transaction between unrelated parties of equal bargaining power.

6.12 Total Funded Debt Ratio. Permit the Total Funded Debt Ratio, as of the last day of any Fiscal Quarter, to be greater than 0.65 to 1.00.

6.13 Interest Coverage Ratio. Permit the Interest Coverage Ratio, as of the last day of any Fiscal Quarter, to be less than 3.25 to 1.00.

-60-

6.14 Investments and Acquisitions. Make any Acquisition or enter into any agreement to make any Acquisition unless approved in advance by the Administrative Agent and the Requisite Lenders in writing, or make or suffer to exist any Investment, other than:

(a) Permitted Acquisitions;

(b) Investments in existence on the Closing Date and disclosed on Schedule 6.14;

(c) Investments consisting of Cash Equivalents;

(d) Investments consisting of advances to officers, directors and employees of Borrower and its Subsidiaries for travel, entertainment, relocation, anticipated bonus, exercise of stock options and analogous ordinary business purposes provided that the aggregate amount of such Investments does not exceed \$1,000,000 at any time outstanding;

(e) Investments in a Subsidiary that is a wholly-owned Subsidiary of Borrower;

 (f) Investments consisting of the extension of credit to customers or suppliers of Borrower and its Subsidiaries in the ordinary course of business and any Investments received in satisfaction or partial satisfaction thereof;

(g) Investments received in connection with the settlement of a bona fide dispute with another Person provided that the aggregate amount of such Investments does not exceed \$500,000 at any time outstanding;

(h) Investments representing all or a portion of the sales price of Property sold or services provided to another Person provided that the aggregate amount of such Investments does not exceed 500,000 at any time outstanding; and

(i) Investments consisting of Indebtedness and Guaranty Obligations owed to Borrower or any of its Subsidiaries.

6.15 Operating Leases. Incur any obligation to pay rent under an operating lease in any Fiscal Year if to do so would result in the aggregate obligation of Borrower and its Subsidiaries to pay rent under all operating leases in that Fiscal Year to exceed \$2,000,000.

6.16 Amendments. Amend or modify any term or provision of (a) any indenture, agreement or instrument evidencing or governing any Subordinated Obligation or (b) any material provision of any Material Contract, if in any such case such amendment or modification in any respect will or may adversely affect the interest of the Lenders.

6.17 Use of Lender's Name. Use any Lender's name (or the name of any of any Lender's Affiliates) in connection with any of their business operations except to identify the existence of the Revolving Facility and the names of the Lenders in the ordinary course of

-61-

Borrower's business or to comply with Borrower's obligations under Law. Nothing contained in this Agreement is intended to permit or authorize Borrower to make any commitment or contract on behalf of any Lender or the Administrative Agent.

6.18 Change of Fiscal Periods. Change its Fiscal Year or any other fiscal period with respect to which it reports financial results hereunder or otherwise.

## Article 7. INFORMATION AND REPORTING REQUIREMENTS

7.1 Financial and Business Information. So long as any Advance remains unpaid, or any other Obligation remains unpaid, or any portion of any of the Commitments remains in force, Borrower shall, unless the Administrative Agent (with the written approval of the Requisite Lenders) otherwise consents, at Borrower's sole expense, deliver to the Administrative Agent for distribution by it to the Lenders, a sufficient number of copies for all of the Lenders of the following:

(a) (i) As soon as practicable, and in any event within fifty (50) days after the end of each Fiscal Quarter ending March 31, June 30 and September 30 (commencing with the Fiscal Quarter ending June 30, 2002), the consolidating and consolidated balance sheet of Borrower and its Subsidiaries as at the end of such Fiscal Quarter and the consolidating and consolidated statements of income, operations and cash flows for such Fiscal Quarter, and the portion of the Fiscal Year ended with such Fiscal Quarter, together with a statement of Stockholders' Equity as of the last day of such Fiscal Quarter, all in reasonable detail, (ii) such financial statements shall be certified by the president or chief financial officer of Borrower as fairly presenting the financial condition, results of operations and cash flows of Borrower and its Subsidiaries in accordance with GAAP (other than footnote disclosures), consistently applied, as at such date and for such periods, subject only to normal year-end accruals and audit adjustments.

(b) (i) As soon as practicable, and in any event within one hundred (100) days after the end of each Fiscal Year, the consolidating and consolidated balance sheet of Borrower and its Subsidiaries as at the end of such Fiscal Year and the consolidating and consolidated statements of income, operations, stockholders' equity and cash flows, in each case of Borrower and its Subsidiaries for such Fiscal Year, all in reasonable detail. Such financial statements shall be prepared in accordance with GAAP, consistently applied, and such consolidated financial statements shall be accompanied by a report of PricewaterhouseCoopers LLP or other independent public accountants of recognized standing selected by Borrower and reasonably satisfactory to the Requisite Lenders, which report shall be prepared in accordance with generally accepted auditing standards as at such date, and shall not be subject to any qualifications or exceptions, and which report shall specifically disclose any changes discovered by such accountants in Borrower's or its Subsidiaries' applicable process of management of accounts, (ii) such accountants' report shall be accompanied by a certificate stating that, in making the examination pursuant to generally accepted auditing standards necessary for the certification of such financial statements and such report, such accountants have obtained no knowledge of any Default then existing or, if, in the opinion of such accountants, any such Default shall exist, stating the nature and status of such Default, and stating that such accountants have reviewed Borrower's financial calculations as at the end of such Fiscal Year (which shall accompany such certificate) under Sections 6.12 and 6.13, have read such Sections (including the definitions of all defined terms used therein) and that nothing

-63-

has come to the attention of such accountants in the course of such examination that would cause them to believe that the same were not calculated by Borrower in the manner prescribed by this Agreement, and (iii) in addition, Borrower shall deliver to the Administrative Agent a copy of (A) any "management letter" prepared by such accountants in conjunction with preparation of the foregoing report and (B) a separate report prepared by such accountants in conjunction with preparation of the foregoing report, pursuant to which separate report such accountants shall be required to disclose any material changes discovered by such accountants in the then current account management process (including the determination of returns and reserves, inventory management practices, and accounts receivable management practices).

(c) As soon as practicable, and in any event not later than thirty (30) days subsequent to the commencement of each Fiscal Year, a budget and projection of Borrower and its Subsidiaries setting forth (i) by Fiscal Quarter for the four (4) Fiscal Quarters of that Fiscal Year and (ii) on an annual basis for each succeeding Fiscal Year thereafter through the Maturity Date, projected balance sheets, statements of operations and statements of cash flow, all in reasonable detail;

(d) Promptly after request by the Administrative Agent or any Lender, copies of any detailed audit reports, management letters or recommendations submitted to Borrower (or its board of directors) by independent accountants in connection with the accounts or books of Borrower, or any of its Subsidiaries, or any audit of any of them;

(e) Promptly after the same are available, and in any event within five (5) Banking Days after filing with the Securities and Exchange Commission, copies of each annual report, proxy or financial statement or other report or communication sent to the stockholders of Borrower or any of its Subsidiaries, and copies of all annual, regular, periodic and special reports and registration statements which Borrower or any of its Subsidiaries may file or be required to file with the Securities and Exchange Commission under Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended, and not otherwise required to be delivered to the Lenders pursuant to other provisions of this Section 7.1;

(f) Promptly after request by Lender, subject to confidentiality requirements of any Governmental Agency, copies of any other report or other document that was filed by Borrower, with any Governmental Agency;

(g) Promptly upon a Senior Officer of Borrower, becoming aware, and in any event within five (5) Banking Days after becoming aware, of the occurrence of any (i) "reportable event" (as such term is defined in Section 4043 of ERISA, but excluding such events as to which the PBGC has by regulation waived the requirement therein contained that it be notified within thirty days of the occurrence of such event) or (ii) non-exempt "prohibited transaction" (as such term is defined in Section 406 of ERISA or Section 4975 of the Code) involving any Pension Plan or any trust created thereunder, telephonic notice specifying the nature thereof, and, no

-64-

more than two (2) Banking Days after such telephonic notice, written notice again specifying the nature thereof and specifying what action Borrower is taking or proposes to take with respect thereto, and, when known, any action taken by the Internal Revenue Service with respect thereto; provided that no such notice shall be required pursuant to this Section 7.1(g) if the anticipated liability is less than \$100,000;

(h) As soon as practicable, and in any event within two (2) Banking Days after a Senior Officer of Borrower becomes aware of the existence of any condition or event which constitutes a Default or Event of Default, telephonic notice specifying the nature and period of existence thereof, and, no more than two (2) Banking Days after such telephonic notice, written notice again specifying the nature and period of existence thereof and specifying what action Borrower is taking or proposes to take with respect thereto;

(i) Promptly upon a Senior Officer of Borrower becoming aware that (i) any Person has commenced a legal proceeding with respect to a claim against Borrower or any of its Subsidiaries that is \$250,000 or more in excess of the amount thereof that is fully covered by insurance, (ii) any creditor under a credit agreement involving Indebtedness of \$250,000 or more or any lessor under a lease involving aggregate rent of \$250,000 or more has asserted a default thereunder on the part of Borrower or any of its Subsidiaries or, (iii) any Person has commenced a legal proceeding with respect to a claim against Borrower or any of its Subsidiaries under a contract that is not a credit agreement or material lease with respect to a claim of in excess of \$250,000 or which otherwise may reasonably be expected to result in a Material Adverse Effect, a written notice describing the pertinent facts relating thereto and what action Borrower and/or its applicable Subsidiaries are taking or propose to take with respect thereto; and

(j) Such other data and information as from time to time may be reasonably requested by the Administrative Agent or the Requisite Lenders.

7.2 Compliance Certificates. So long as any Advance remains unpaid, or any other Obligation remains unpaid or unperformed, or any portion of any of the Commitments remain outstanding, Borrower shall, at Borrower's sole expense, deliver to the Administrative Agent for distribution by it to the Lenders concurrently with the financial statements required pursuant to Sections 7.1(a) and 7.1(b), a Compliance Certificate signed by the president or chief financial officer of Borrower.

-65-

### Article 8. CONDITIONS

8.1 Initial Advances. The obligation of each Closing Date Lender to make the initial Advance to be made by it, and the obligation of the Issuing Lender to issue the initial Letter of Credit (as applicable), is subject to the following conditions precedent, each of which shall be satisfied prior to the making of the initial Advances or the issuance of the initial Letter of Credit (as applicable) (unless all of the Closing Date Lenders, in their sole and absolute discretion, shall agree otherwise):

(a) The Administrative Agent shall have received all of the following, each of which shall be originals unless otherwise specified, each properly executed by a Responsible Official of each party thereto, each dated as of the Closing Date and each in form and substance satisfactory to the Administrative Agent and its legal counsel (unless otherwise specified or, in the case of the date of any of the following, unless the Administrative Agent otherwise agrees or directs):

(1) at least one (1) executed counterpart of this Agreement, together with arrangements satisfactory to the Administrative Agent for additional executed counterparts, sufficient in number for distribution to the Closing Date Lenders and Borrower;

(2) Notes executed by Borrower in favor of each Closing Date Lender, each in a principal amount equal to that Lender's Commitment;

- (3) the Letter of Credit Agreement executed by Borrower;
- (4) the Swing Line Documents executed by Borrower;

(5) with respect to Borrower, such documentation as the Administrative Agent may reasonably require to establish the due organization, valid existence and good standing of Borrower, its qualification to engage in business in each material jurisdiction in which it is engaged in business or required to be so qualified, its authority to execute, deliver and perform the Loan Documents to which it is a Party, the identity, authority and capacity of each Responsible Official thereof authorized to act on its behalf, including certified copies of articles of incorporation and amendments thereto, bylaws and amendments thereto, certificates of good standing and/or qualification to engage in business, tax clearance certificates, certificates of corporate resolutions or other applicable authorization documents, incumbency certificates, Certificates of Responsible Officials, and the like;

(6) the Opinion of Counsel;

(7) a Certificate of the president or chief financial officer of Borrower, certifying that attached thereto is a true and correct copy of the

-66-

Projections, and further certifying that the representation contained in Section 4.16 is, to the best of his or her knowledge, true and correct;

(8) one or more Requests for Borrowing or Requests for Letters of Credit;

(9) a Certificate signed by a Senior Officer of Borrower certifying that the conditions specified in Sections 8.1(d) and 8.1(e) have been satisfied; and

(10) such other assurances, certificates, documents, consents or opinions as the Administrative Agent and/or any Closing Date Lender reasonably may require.

(b) The fees payable on or before the Closing Date pursuant to Section 3.3 shall have been paid.

(c) The reasonable costs and expenses of the Administrative Agent in connection with the preparation of the Loan Documents payable pursuant to Section 11.3, and invoiced to Borrower prior to the Closing Date, shall have been paid.

(d) The representations and warranties of Borrower contained in Article 4 shall be true and correct in all material respects.

(e) Borrower and any other Parties shall be in compliance with all the terms and provisions of the Loan Documents, and giving effect to the initial Advances or initial Letter of Credit (as applicable), no Default or Event of Default shall have occurred and be continuing.

(f) Borrower shall have delivered to the Administrative Agent, evidence, in form and substance satisfactory to the Lenders, that the Existing Credit Facilities have been or will be concurrently terminated and that all Liens, if any, securing any part of the Existing Credit Facilities have been or will be concurrently reconveyed, released and/or terminated, as the case may be.

(g) All legal matters relating to the Loan Documents shall be reasonably satisfactory to Sheppard, Mullin, Richter & Hampton LLP, special counsel to Lender.

 $% \left( h\right) ^{2}$  (h) The Closing Date shall have occurred on or before June 6, 2002.

8.2 Any Advance. The obligation of each Lender to make any Advance, and the obligation of the Issuing Lender to issue any Letter of Credit, is subject to the following conditions precedent (unless the Requisite Lenders or, in any case where the approval of all of the Lenders is required pursuant to Section 11.2, all of the Lenders, in their sole and absolute discretion, shall agree otherwise):

-67-

(a) except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change which is permitted by this Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, the representations and warranties contained in Article 4 (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) shall be true and correct in all material respects on and as of the date of the Advance or the Letter of Credit as though made on that date;

(b) no circumstance or event shall have occurred that constitutes a Material Adverse Effect since the Closing Date;

(c) other than matters described in Schedule 4.9 or not required as of the Closing Date to be therein described, there shall not be then pending or threatened any action, suit, proceeding or investigation against or affecting Borrower or any Subsidiary of Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect;

(d) the Administrative Agent shall have timely received a Request for Borrowing (or telephonic or other request for Borrowing referred to in the second sentence of Section 2.1(b), if applicable), or the Issuing Lender shall have received a Request for Letter of Credit, as the case may be, in compliance with Article 2; and

(e) the Administrative Agent shall have received, in form and substance reasonably satisfactory to the Administrative Agent, such other certificates, documents or consents related to the foregoing as the Administrative Agent or Requisite Lenders reasonably may require.

-68-

## Article 9. EVENTS OF DEFAULT AND REMEDIES UPON EVENT OF DEFAULT

9.1 Events of Default. The existence or occurrence of any one or more of the following events, whatever the reason therefor and under any circumstances whatsoever, shall constitute an Event of Default:

(a) Borrower fails to pay any principal on any of the Notes, or any portion thereof, on the date when due; or

(b) Borrower fails to pay any interest on any of the Notes, or any fees under Sections 3.2 or 3.4, or any portion thereof, within five (5) Banking Days after the date when due; or fails to pay any other fee or amount payable to the Lenders or the Administrative Agent under any Loan Document, or any portion thereof, within five (5) Banking Days after written demand therefor; or

(c) Borrower fails to comply with, or cause or permit any of its Subsidiaries to fail to comply with, any of the covenants contained in Article 6; or

(d) (i) Borrower fails to comply with Section 7.1(i) in the manner stated therein or (ii) Borrower fails to perform any other reporting requirement set forth in Article 7 within five (5) Banking Days of the date specified for performance therein; or

(e) Borrower or any other Party fails to perform or observe any other covenant or agreement (not specified in clause (a), (b), (c) or (d) above) contained in any Loan Document on its part to be performed or observed and such default shall continue unremedied for twenty (20) days after the giving of notice by the Administrative Agent on behalf of the Requisite Lenders of such Default; or

(f) Any representation or warranty of Borrower or any other Party made in any Loan Document, or in any certificate or other writing delivered by Borrower or such other Party pursuant to any Loan Document, proves to have been incorrect when made or reaffirmed in any material respect; or

(g) Borrower or any of its Subsidiaries (i) fails to pay the principal, or any principal installment, of any present or future Indebtedness of \$1,000,000 or more, or any guaranty of present or future Indebtedness of \$1,000,000 or more, on its part to be paid, when due (or within any stated grace period), whether at the stated maturity, upon acceleration, by reason of required prepayment or otherwise or (ii) fails to perform or observe any other term, covenant or agreement on its part to be performed or observed, or suffers any event of default to occur, in connection with any present or future Indebtedness of \$1,000,000 or more, or of any guaranty of present or future Indebtedness of \$1,000,000 or more, if as a result of such failure or sufferance any holder or holders thereof (or an agent or trustee on its or their behalf) has the right to declare such Indebtedness due before the date on which it otherwise would become

-69-

due or the right to require Borrower or any such Subsidiary to redeem or purchase, or offer to redeem or purchase, all or any portion of such Indebtedness; or

(h) Any Loan Document, at any time after its execution and delivery and for any reason other than the agreement or action (or omission to act) of the Administrative Agent or satisfaction in full of all the Obligations, ceases to be in full force and effect or is declared by a court of competent jurisdiction to be null and void, invalid or unenforceable in any respect which is materially adverse to the interests of the Lenders; or any Party thereto denies in writing that it has any or further liability or obligation under any Loan Document, or purports to revoke, terminate or rescind same; or

(i) A final judgment against Borrower or any of its Subsidiaries is entered for the payment of money in excess of \$1,000,000 (not covered by insurance or for which an insurer has reserved its rights) and, absent procurement of a stay of execution, such judgment remains unsatisfied for thirty (30) calendar days after the date of entry of judgment, or in any event later than five (5) days prior to the date of any proposed sale thereunder; or any writ or warrant of attachment or execution or similar process is issued or levied against all or any material part of the Property of Borrower or any of its Subsidiaries and is not released, vacated or fully bonded within thirty (30) calendar days after its issue or levy; or

(j) Borrower or any of its Subsidiaries institutes or consents to the institution of any proceeding under a Debtor Relief Law relating to it or to all or any material part of its Property, or is unable or admits in writing its inability to pay its debts as they mature, or makes an assignment for the benefit of creditors; or applies for or consents to the appointment of any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer for it or for all or any material part of its Property; or any receiver, trustee, custodian, conservator, liquidator, rehabilitator or similar officer is appointed without the application or consent of that Person and the appointment continues undischarged or unstayed for sixty (60) calendar days; or any proceeding under a Debtor Relief Law relating to any such Person or to all or any part of its Property is instituted without the consent of that Person and continues undismissed or unstayed for sixty (60) calendar days; or

# (k) A Change in Control occurs; or

(1) The dissolution or liquidation of Borrower, any of SCW or Chapparal City Water Company, or any other Subsidiary that has, immediately prior to the commencement of such dissolution or liquidation, assets having a fair market value of more than \$20,000,000 or Borrower or any such Subsidiary, or any of their respective partners, members, directors or stockholders, as the case may be, shall take action seeking to effect the dissolution or liquidation of Borrower or such Subsidiary; or

-70-

(m) The occurrence of an Event of Default (as such term is or may hereafter be specifically defined in any other Loan Document) under any other Loan Document; or

(n) Any Pension Plan maintained by Borrower is finally determined by the PBGC to have a material "accumulated funding deficiency" as that term is defined in Section 302 of ERISA in excess of an amount equal to 5% of the consolidated total assets of Borrower as of the most-recently ended Fiscal Quarter; or

(o) Any holder of a Subordinated Obligation asserts in writing that such Subordinated Obligation is not subordinated to the Obligations in accordance with its terms and Borrower does not promptly deny in writing such assertion and contest any attempt by such holder to take action based on such assertion; or

(p) Any event occurs which gives the holder or holders of any Subordinated Obligation (or an agent or trustee on its or their behalf) the right to declare such Subordinated Obligation due before the date on which it otherwise would become due, or the right (other than by reason of a Change in Control) to require the issuer thereof, to redeem or purchase, or offer to redeem or purchase, all or any portion of any Subordinated Obligation, or a final judgment is entered by a court of competent jurisdiction that any Subordinated Obligation is not subordinated in accordance with its terms to the Obligations.

9.2 Remedies Upon Event of Default. Without limiting any other rights or remedies of the Administrative Agent or the Lenders provided for elsewhere in this Agreement, or the other Loan Documents, or by applicable Law, or in equity, or otherwise:

(a) Upon the occurrence, and during the continuance, of any Event of Default other than an Event of Default described in Section 9.1(j):

(1) the commitments to make Advances and all other obligations of the Administrative Agent or the Lenders and all rights of Borrower and any other Parties under the Loan Documents shall be suspended without notice to or demand upon Borrower, which are expressly waived by Borrower except that all of the Lenders or the Requisite Lenders (as the case may be, in accordance with Section 11.2) may waive an Event of Default or, without waiving, determine, upon terms and conditions satisfactory to the Lenders or Requisite Lenders, as the case may be, to reinstate the Commitments and such other obligations and rights and make further Advances, which waiver or determination shall apply equally to, and shall be binding upon, all the Lenders;

(2) the Issuing Lender may demand immediate payment by Borrower of an amount equal to the Aggregate Effective Amount of all outstanding Letters of Credit to be held by the Administrative Agent, on behalf

-71-

of the Lenders, in an interest-bearing cash collateral account as collateral for all of the Obligations; and

(3) the Requisite Lenders may request the Administrative Agent to, and the Administrative Agent thereupon shall, terminate the Commitments and/or declare all or any part of the unpaid principal of all Notes, all interest accrued and unpaid thereon and all other amounts payable under the Loan Documents to be forthwith due and payable, whereupon the same shall become and be forthwith due and payable, without protest, presentment, notice of dishonor, demand or further notice of any kind, all of which are expressly waived by Borrower.

(b) Upon the occurrence of any Event of Default described in Section 9.1(j):

(1) the Commitments shall terminate without notice to or demand upon Borrower, which are expressly waived by Borrower, except that all of the Lenders may waive the Event of Default or, without waiving, determine, upon terms and conditions satisfactory to all the Lenders, to reinstate the Commitments and make further Advances, which determination shall apply equally to, and shall be binding upon, all the Lenders;

(2) an amount equal to the Aggregate Effective Amount of all outstanding Letters of Credit shall be immediately due and payable to the Issuing Lender without notice to or demand upon Borrower, which are expressly waived by Borrower, to be held by the Administrative Agent, on behalf of the Lenders, in an interest-bearing cash collateral account as collateral for all of the Obligations; and

(3) the unpaid principal of all Notes, all interest accrued and unpaid thereon and all other amounts payable under the Loan Documents shall be forthwith due and payable, without protest, presentment, notice of dishonor, demand or further notice of any kind, all of which are expressly waived by Borrower.

(c) Upon the occurrence of any Event of Default, the Lenders and the Administrative Agent, or any of them, without notice to (except as expressly provided for in any Loan Document) or demand upon Borrower, which are expressly waived by Borrower (except as to notices expressly provided for in any Loan Document), may proceed (but only with the consent of the Requisite Lenders) to protect, exercise and enforce their rights and remedies under the Loan Documents against Borrower and any other Party and such other rights and remedies as are provided by Law or equity.

(d) The order and manner in which the Lenders' rights and remedies are to be exercised shall be determined by the Requisite Lenders in their sole

-72-

discretion, and all payments received by the Administrative Agent and the Lenders, or any of them, shall be applied first to the costs and expenses (including reasonable attorneys' fees and disbursements and the reasonably allocated costs of attorneys employed by the Administrative Agent or by any Lender) of the Administrative Agent and of the Lenders, and thereafter paid pro rata to the Lenders in the same proportions that the aggregate Obligations owed to each Lender under the Loan Documents bear to the aggregate Obligations owed under the Loan Documents to all the Lenders, without priority or preference among the Lenders. Regardless of how each Lender may treat payments for the purpose of its own accounting, for the purpose of computing Borrower's Obligations hereunder and under the Notes, payments shall be applied first, to the costs and expenses of the Administrative Agent and the Lenders, as set forth above, second, to the payment of accrued and unpaid interest due under any Loan Documents to and including the date of such application (ratably, and without duplication, according to the accrued and unpaid interest due under each of the Loan Documents), and third, to the payment of all other amounts (including principal and fees) then owing to the Administrative Agent or the Lenders under the Loan Documents. No application of payments will cure any Event of Default, or prevent acceleration, or continued acceleration, of amounts payable under the Loan Documents, or prevent the exercise, or continued exercise, of rights or remedies of the Lenders hereunder or thereunder or at Law or in equity.

-73-

#### Article 10. THE ADMINISTRATIVE AGENT

10.1 Appointment and Authorization. Subject to Section 10.8, each Lender hereby irrevocably appoints and authorizes the Administrative Agent to take such action as agent on its behalf and to exercise such powers under the Loan Documents as are delegated to the Administrative Agent by the terms thereof or are reasonably incidental, as determined by the Administrative Agent, thereto. This appointment and authorization is intended solely for the purpose of facilitating the servicing of the Revolving Facility and does not constitute appointment of the Administrative Agent as trustee for any Lender or as representative of any Lender for any other purpose and, except as specifically set forth in the Loan Documents to the contrary, the Administrative Agent shall take such action and exercise such powers only in an administrative and ministerial capacity.

10.2 Administrative Agent and Affiliates. Wells Fargo (and each successor Administrative Agent) has the same rights and powers under the Loan Documents as any other Lender and may exercise the same as though it were not the Administrative Agent, and the term "Lender" or "Lenders" includes Wells Fargo in its individual capacity. Wells Fargo (and each successor Administrative Agent) and its Affiliates may accept deposits from, lend money to and generally engage in any kind of banking, trust or other business with Borrower, any Subsidiary thereof, or any Affiliate of Borrower or any Subsidiary thereof, as if it were not the Administrative Agent and without any duty to account therefor to the Lenders. Wells Fargo (and each successor Administrative Agent) need not account to any other Lender for any monies received by it for reimbursement of its costs and expenses as Administrative Agent hereunder, or (subject to Section 11.10) for any monies received by it in its capacity as a Lender hereunder. The Administrative Agent shall not be deemed to hold a fiduciary relationship with any Lender and no implied covenants, functions, responsibilities, duties, obligations or liabilities shall be read into this Agreement or otherwise exist against the Administrative Agent.

10.3 Proportionate Interest in any Collateral. The Administrative Agent, on behalf of all the Lenders, shall hold in accordance with the Loan Documents all items of collateral (if any) or interests therein received or held by the Administrative Agent. Subject to the Administrative Agent's and the Lenders' rights to reimbursement for their costs and expenses hereunder (including reasonable attorneys' fees and disbursements and other professional services and the reasonably allocated costs of attorneys employed by the Administrative Agent or a Lender) and subject to the application of payments in accordance with Section 9.2(d), each Lender shall have an interest in the Lenders' interest in such collateral or interests therein in the same proportions that the aggregate Obligations owed such Lender the Loan Documents bear to the aggregate Obligations owed under the Loan Documents to all the Lenders, without priority or preference among the Lenders.

10.4 Lenders' Credit Decisions. Each Lender agrees that it has, independently and without reliance upon the Administrative Agent, any other Lender or the directors, officers, agents, employees or attorneys of the Administrative Agent or of any other Lender, and instead in reliance upon information supplied to it by or on behalf of Borrower

-74-

and upon such other information as it has deemed appropriate, made its own independent credit analysis and decision to enter into this Agreement. Each Lender also agrees that it shall, independently and without reliance upon the Administrative Agent, any other Lender or the directors, officers, agents, employees or attorneys of the Administrative Agent or of any other Lender, continue to make its own independent credit analyses and decisions in acting or not acting under the Loan Documents.

#### 10.5 Action by Administrative Agent.

(a) The Administrative Agent may assume that no Default has occurred and is continuing, unless the Administrative Agent (or the Lender that is then the Administrative Agent) has received notice from Borrower stating the nature of the Default or has received notice from a Lender stating the nature of the Default and that such Lender considers the Default to have occurred and to be continuing.

(b) The Administrative Agent has only those obligations under the Loan Documents as are expressly set forth therein.

(c) Except for any obligation expressly set forth in the Loan Documents and as long as the Administrative Agent may assume that no Event of Default has occurred and is continuing, the Administrative Agent may, but shall not be required to, exercise its discretion to act or not act, except that the Administrative Agent shall be required to act or not act upon the instructions of the Requisite Lenders (or of all the Lenders, to the extent required by Section 11.2) and those instructions shall be binding upon the Administrative Agent and all the Lenders, provided that the Administrative Agent shall not be required to act or not act if to do so would be contrary to any Loan Document or to applicable Law or would result, in the reasonable judgment of the Administrative Agent, in substantial risk of liability to the Administrative Agent.

(d) If the Administrative Agent has received a notice specified in clause (a), the Administrative Agent shall immediately give notice thereof to the Lenders and shall act or not act upon the instructions of the Requisite Lenders (or of all the Lenders, to the extent required by Section 11.2), provided that the Administrative Agent shall not be required to act or not act if to do so would be contrary to any Loan Document or to applicable Law or would result, in the reasonable judgment of the Administrative Agent, in substantial risk of liability to the Administrative Agent, and except that if the Requisite Lenders (or all the Lenders, if required under Section 11.2) fail, for five (5) Banking Days after the receipt of notice from the Administrative Agent, to instruct the Administrative Agent, then the Administrative Agent, in its sole discretion, may act or not act as it deems advisable for the protection of the interests of the Lenders.

(e) The Administrative Agent shall have no liability to any Lender for acting, or not acting, as instructed by the Requisite Lenders (or all the Lenders, if required under Section 11.2), notwithstanding any other provision hereof.

-75-

10.6 Liability of Administrative Agent. Neither the Administrative Agent nor any of its directors, officers, agents, employees or attorneys shall be liable for any action taken or not taken by them under or in connection with the Loan Documents, except for their own gross negligence or willful misconduct. Without limitation on the foregoing, the Administrative Agent and its directors, officers, agents, employees and attorneys:

(a) May treat the payee of any Note as the holder thereof until the Administrative Agent receives notice of the assignment or transfer thereof, in form satisfactory to the Administrative Agent, signed by the payee, and may treat each Lender as the owner of that Lender's interest in the Obligations for all purposes of this Agreement until the Administrative Agent receives notice of the assignment or transfer thereof, in form satisfactory to the Administrative Agent, signed by that Lender;

(b) May consult with legal counsel (including in-house legal counsel), accountants (including in-house accountants) and other professionals or experts selected by it, or with legal counsel, accountants or other professionals or experts for Borrower or any Subsidiary of Borrower and/or any of their Affiliates or the Lenders, and shall not be liable for any action taken or not taken by it in good faith in accordance with any advice of such legal counsel, accountants or other professionals or experts;

(c) Shall not be responsible to any Lender for any statement, warranty or representation made in any of the Loan Documents or in any notice, certificate, report, request or other statement (written or oral) given or made in connection with any of the Loan Documents;

(d) Except to the extent expressly set forth in the Loan Documents, shall have no duty to ask or inquire as to the performance or observance by Borrower of any of the terms, conditions or covenants of any of the Loan Documents or to inspect any Property, books or records of Borrower or any Subsidiary of Borrower;

(e) Will not be responsible to any Lender for the due execution, legality, validity, enforceability, genuineness, effectiveness, sufficiency or value of any Loan Document, any other instrument or writing furnished pursuant thereto or in connection therewith;

(f) Will not incur any liability by acting or not acting in reliance upon any Loan Document, notice, consent, certificate, statement, request or other instrument or writing believed in good faith by it to be genuine and signed or sent by the proper party or parties; and

(g) Will not incur any liability for any arithmetical error in computing any amount paid or payable by Borrower or paid or payable to or received or receivable from any Lender under any Loan Document, including principal, interest, commitment fees, Advances and other amounts; provided that, promptly upon

-76-

discovery of such an error in computation, the Administrative Agent, the Lenders and (to the extent applicable) Borrower shall make such adjustments as are necessary to correct such error and to restore the parties to the position that they would have occupied had the error not occurred.

10.7 Indemnification. Each Lender shall, ratably in accordance with its Pro Rata Share of all of the then applicable Commitments (if any of the Commitments are then in effect) and/or in accordance with its proportion of the aggregate Indebtedness then evidenced by the Notes (if all of the Commitments have then been terminated), indemnify and hold the Administrative Agent and its directors, officers, agents, employees and attorneys harmless against any and all liabilities, obligations, losses, damages, penalties, actions, judgments, suits, costs, expenses or disbursements of any kind or nature whatsoever (including reasonable attorneys' fees and disbursements and allocated costs of attorneys employed by the Administrative Agent) that may be imposed on, incurred by or asserted against it or them in any way relating to or arising out of the Loan Documents (other than losses incurred by reason of the failure of Borrower to pay the Indebtedness represented by the Notes) or any action taken or not taken by it as Administrative Agent thereunder, except such as result from its own gross negligence or willful misconduct. Without limitation on the foregoing, each Lender shall reimburse the Administrative Agent upon demand for that Lender's Pro Rata Share of any out-of-pocket cost or expense incurred by the Administrative Agent in connection with the negotiation, preparation, execution, delivery, amendment, waiver, restructuring, reorganization (including a bankruptcy reorganization), enforcement or attempted enforcement of the Loan Documents, to the extent that Borrower or any other Party are required by Section 11.3 to pay that cost or expense but fails to do so upon demand. Nothing in this Section 10.7 shall entitle the Administrative Agent or any indemnitee referred to above to recover any amount from the Lenders if and to the extent that such amount has theretofore been recovered from Borrower. To the extent that the Administrative Agent or any indemnitee referred to above is later reimbursed such amount by Borrower, it shall return the amounts paid to it by the Lenders in respect of such amount.

10.8 Successor Administrative Agent. The Administrative Agent may, and at the request of the Requisite Lenders shall, resign as Administrative Agent upon reasonable notice to the Lenders and Borrower effective upon acceptance of appointment by a successor Administrative Agent. If the Administrative Agent shall resign as Administrative Agent under this Agreement, the Requisite Lenders shall appoint from among the Lenders a successor Administrative Agent for the Lenders, which successor Administrative Agent shall be approved by Borrower (and such approval shall not be unreasonably withheld or delayed). If no successor Administrative Agent is appointed prior to the effective date of the resignation of the Administrative Agent, the Administrative Agent may appoint, after consulting with the Lenders and Borrower, a successor Administrative Agent from among the Lenders. Upon the acceptance of its appointment as successor Administrative Agent hereunder, such successor Administrative Agent shall succeed to all the rights, powers and duties of the retiring Administrative Agent and the term "Administrative Agent" shall mean such successor Administrative Agent and the retiring Administrative Agent's appointment, powers and duties as Administrative Agent shall be terminated. After any retiring Administrative Agent's resignation hereunder as Administrative Agent, the provisions of this Article 10, and

-77-

Sections 11.3, 11.11 and 11.21, shall inure to its benefit as to any actions taken or omitted to be taken by it while it was Administrative Agent under this Agreement. Notwithstanding the foregoing, if no successor Administrative Agent has accepted appointment as Administrative Agent by the date which is thirty (30) days following a retiring Administrative Agent's notice of resignation, the retiring Administrative Agent's resignation shall nevertheless thereupon become effective and the Lenders shall perform all of the duties of the Administrative Agent a successor Administrative Agent as provided for above.

10.9 No Obligations of Borrower. Nothing contained in this Article 10 shall be deemed to impose upon Borrower any obligation in respect of the due and punctual performance by the Administrative Agent of its obligations to the Lenders under any provision of this Agreement, and Borrower shall have no liability to the Administrative Agent or any of the Lenders in respect of any failure by the Administrative Agent or any Lender to perform any of its obligations to the Administrative Agent or the Lenders under this Agreement. Without limiting the generality of the foregoing, where any provision of this Agreement relating to the payment of any amounts due and owing under the Loan Documents provides that such payments shall be made by Borrower to the Administrative Agent for the account of the Lenders, Borrower's obligations to the Lenders in respect of such payments shall be deemed to be satisfied upon the making of such payments to the Administrative Agent in the manner provided by this Agreement. In addition, Borrower may rely on a written statement by the Administrative Agent to the effect that it has obtained the written consent of the Requisite Lenders or all of the Lenders, as applicable under Section 11.2, in connection with a waiver, amendment, consent, approval or other action by the Lenders hereunder, and shall have no obligation to verify or confirm the same.

-78-

#### Article 11. MISCELLANEOUS

11.1 Cumulative Remedies; No Waiver. The rights, powers, privileges and remedies of the Administrative Agent and the Lenders provided herein or in any Note or other Loan Document are cumulative and not exclusive of any right, power, privilege or remedy provided by Law or equity. No failure or delay on the part of the Administrative Agent or any Lender in exercising any right, power, privilege or remedy may be, or may be deemed to be, a waiver thereof; nor may any single or partial exercise of any right, power, privilege or remedy preclude any other or further exercise of the same or any other right, power, privilege or remedy. The terms and conditions of Article 8 hereof are inserted for the sole benefit of the Administrative Agent and the Lenders; the same may be waived in whole or in part, with or without terms or conditions, in respect of any Borrowing without prejudicing the Administrative Agent's or the Lenders' rights to assert them in whole or in part in respect of any other Borrowing.

11.2 Amendments; Consents. No amendment, modification, supplement, extension, termination or waiver of any provision of this Agreement or any other Loan Document, no approval or consent thereunder, and no consent to any departure by Borrower or any other Party therefrom, may in any event be effective unless in writing signed by the Administrative Agent with the written approval of the Requisite Lenders (and, in the case of any amendment, modification or supplement of or to any Loan Document to which Borrower is a Party, signed by Borrower, and, in the case of any amendment, modification or supplement to Article 10, signed by the Administrative Agent), and then only in the specific instance and for the specific purpose given; and, without the approval in writing of all the Lenders, no amendment, modification, supplement, termination, waiver or consent may be effective:

(a) To amend or modify the principal of, or the amount of principal, principal prepayments or the rate of interest payable on, any Note, or the amount of the Revolving Facility, or the Pro Rata Share of any Lender or the amount of any commitment fee payable to any Lender, or any other fee or amount payable to any Lender (in its capacity as a Lender) under the Loan Documents or to waive an Event of Default consisting of the failure of Borrower to pay when due principal, interest or any fee, or to provide for additional extensions of credit to Borrower by the Lenders pursuant to the Loan Documents;

(b) To postpone any date fixed for any payment of principal of, prepayment of principal of or any installment of interest on, any Note or any installment of any fee, or to extend the term of the Revolving Facility;

(c) To amend the provisions of the definition of "Requisite Lenders" or "Maturity Date";

(d) To amend or waive Article 8 or this Section 11.2; or

-79-

(e) To amend any provision of this Agreement that expressly requires the consent or approval of all the Lenders.

Any amendment, modification, supplement, termination, waiver or consent pursuant to this Section 11.2 shall apply equally to, and shall be binding upon, all the Lenders and the Administrative Agent.

11.3 Costs and Expenses. Borrower agrees to pay within five (5) Banking Days after demand, accompanied by an invoice therefor, all reasonable, out-of-pocket expenses (except in the case of the Administrative Agent's allocated in-house counsel costs described below, which shall not be required to be "out-of-pocket") of the Administrative Agent (including the reasonable fees and out-of-pocket expenses of counsel to the Administrative Agent (including reasonable allocated costs of in-house counsel employed by the Administrative Agent) and of local counsel, if any, who may be retained by counsel to the Administrative Agent) in connection with:

(a) The negotiation, preparation, execution and delivery of this Agreement and of each other Loan Document, including schedules and exhibits, and any amendments, waivers, consents, supplements or other modifications to this Agreement or any other Loan Document as may from time to time hereafter be required, whether or not the transactions contemplated hereby (or thereby) are consummated, provided that, Borrower's maximum liability for fees and expenses of counsel to the Administrative Agent in connection with the Closing Date shall not exceed \$50,000;

(b) The preparation and review of the form of any document or instrument relevant to this Agreement or any other Loan Document; and

(c) The preparation of any information or response required with respect to any investigative request or inquiry, approval, findings of suitability or any other response or communication involving a Governmental Agency arising out of this Agreement, any other Loan Document or any Obligation evidenced by the Loan Documents or the participation in any public or investigatory hearing or meeting.

Borrower further agrees to pay, and to save the Administrative Agent, the Issuing Lender and the Lenders harmless from all liability for, any stamp and similar taxes that may be payable in connection with the execution or delivery of this Agreement, the credit extensions made hereunder, or the issuance of the Notes, the Letters of Credit or any other Loan Document. Borrower also agrees to reimburse the Administrative Agent and, after the occurrence and during the continuance of an Event of Default, the Issuing Lender and each Lender upon demand for all reasonable out-of-pocket expenses (including reasonable attorneys' fees and legal expenses of counsel and fees and expenses of consultants to the Administrative Agent, the Issuing Lender or such Lenders) incurred by the Administrative Agent, the Issuing Lender or such Lenders in connection with (i) the negotiation of any restructuring or "work-out" with Borrower whether or not consummated, of any Obligations, (ii) the enforcement or attempted enforcement of any Obligations and any matter related thereto and (iii) any bankruptcy of

-80-

Borrower or any of its Subsidiaries. Any amount payable to the Administrative Agent or any Lender under this Section 11.3 shall bear interest from the fifth Banking Day following the date of demand, if not then paid, for payment at the Default Rate.

11.4 Nature of Lenders' Obligations. The obligations of the Lenders hereunder are several and not joint or joint and several. Nothing contained in this Agreement or any other Loan Document and no action taken by the Administrative Agent or the Lenders or any of them pursuant hereto or thereto may, or may be deemed to, make the Lenders a partnership, an association, a joint venture or other entity, either among themselves or with Borrower or any Subsidiary or Affiliate of Borrower. A default by any Lender will not increase the Commitment of any other Lender or the Pro Rata Share of the Revolving Facility attributable to any other Lender. Any Lender not in default may, if it desires, assume (in such proportion as the nondefaulting Lenders agree) the obligations of any Lender in default, but no Lender is obligated to do so.

11.5 Survival of Representations and Warranties. All representations and warranties contained herein or in any other Loan Document, or in any certificate or other writing delivered by or on behalf of any one or more of the Parties to any Loan Document, will survive the making of the Advances hereunder and the execution and delivery of the Notes, and have been or will be relied upon by the Administrative Agent and each Lender, notwithstanding any investigation made by the Administrative Agent or any Lender or on their behalf.

11.6 Notices. Except as otherwise expressly provided in the Loan Documents, all notices, requests, demands, directions and other communications provided for hereunder or under any other Loan Document must be in writing and must be mailed, telegraphed, telecopied, dispatched by commercial courier or delivered to the appropriate party at the address set forth on the signature pages of this Agreement or other applicable Loan Document or, as to any party to any Loan Document, at any other address as may be designated by it in a written notice sent to all other parties to such Loan Document in accordance with this Section. Except as otherwise expressly provided in any Loan Document, if any notice, request, demand, direction or other communication required or permitted by any Loan Document is given by mail it will be effective on the earlier of receipt or the fourth Banking Day after deposit in the United States mail with first class or airmail postage prepaid; if given by telegraph or cable, when delivered to the telegraph company with charges prepaid; if given by telecopier, when sent; if dispatched by commercial courier, on the scheduled delivery date; or if given by personal delivery, when delivered.

11.7 Execution of Loan Documents. Unless the Administrative Agent otherwise specifies with respect to any Loan Document, (a) this Agreement and any other Loan Document may be executed in any number of counterparts and any party hereto or thereto may execute any counterpart, each of which when executed and delivered will be deemed to be an original and all of which counterparts of this Agreement or any other Loan Document, as the case may be, when taken together will be deemed to be but one and the same instrument and (b) execution of any such counterpart may be evidenced by a telecopier transmission of the signature of such party. The execution of this Agreement or any other

-81-

Loan Document by any party hereto or thereto will not become effective until counterparts hereof or thereof, as the case may be, have been executed by all the parties hereto or thereto.

11.8 Binding Effect; Assignment.

(a) This Agreement and the other Loan Documents to which Borrower is a Party will be binding upon and inure to the benefit of Borrower, the Administrative Agent, each of the Lenders, and their respective successors and assigns, except that Borrower may not assign its rights hereunder or thereunder or any interest herein or therein without the prior written consent of all the Lenders. Each Lender represents that it is not acquiring its Note with a view to the distribution thereof within the meaning of the Securities Act of 1933, as amended (subject to any requirement that disposition of such Note must be within the control of such Lender). Any Lender may at any time pledge its Note or any other instrument evidencing its rights as a Lender under this Agreement to a Federal Reserve Bank, but no such pledge shall release that Lender from its obligations hereunder or grant to such Federal Reserve Bank the rights of a Lender hereunder absent foreclosure of such pledge.

(b) From time to time following the Closing Date, each Lender may assign to one or more Eligible Assignees all or any portion of its rights and obligations under this Agreement (including all or a portion of its Commitment, the Advances owing to it and the Note or Notes held by it); provided that, subject to subsection (f) below, (i) such Eligible Assignee, if not then a Lender or an Affiliate of the assigning Lender, shall be approved by the Administrative Agent and Borrower (neither of which approvals shall be unreasonably withheld or delayed), (ii) such assignment shall be evidenced by an Assignment and Acceptance, a copy of which shall be furnished to the Administrative Agent as hereinbelow provided, (iii) except in the case of an assignment to an Affiliate of the assigning Lender, to another Lender or of the entire remaining rights and obligations of the assigning Lender under this Agreement, the assignment shall not assign a portion of such assigning Lender's Commitments and/or Advances owing to such assigning Lender that is equivalent to less than \$3,000,000, and (iv) the effective date of any such assignment shall be as specified in the Assignment and Acceptance, but not earlier than the date which is five (5) Banking Days after the date the Administrative Agent has received the Assignment and Acceptance. Upon the effective date of such Assignment and Acceptance, the Eligible Assignee named therein shall be a Lender for all purposes of this Agreement, with the Commitments and/or Advances therein set forth and, to the extent of such Commitments and/or Advances, the assigning Lender shall be released from its further obligations under this Agreement. Borrower agrees that it shall execute and deliver (against delivery by the assigning Lender to Borrower of such Lender's Notes) to such assignee Lender, Notes evidencing that assignee Lender's Commitments and/or Advances, and to the assigning Lender, Notes evidencing the remaining balance of the Commitments and/or Advances retained by the assigning Lender.

(c) By executing and delivering an Assignment and Acceptance, the Eligible Assignee thereunder acknowledges and agrees that: (i) other than the

-82-

representation and warranty that it is the legal and beneficial owner of the rights and obligations hereunder being assigned thereby free and clear of any adverse claim, the assigning Lender has made no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with this Agreement or the execution, legality, validity, enforceability, genuineness or sufficiency of this Agreement or any other Loan Document; (ii) the assigning Lender has made no representation or warranty and assumes no responsibility with respect to the financial condition of Borrower or the performance by Borrower of the Obligations; (iii) it has received a copy of this Agreement, together with copies of the most recent financial statements delivered pursuant to Section 7.1 and such other documents and information as it has deemed appropriate to make its own credit analysis and decision to enter into such Assignment and Acceptance; (iv) it will, independently and without reliance upon the Administrative Agent or any Lender and based on such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under this Agreement; (v) it appoints and authorizes the Administrative Agent to take such action and to exercise such powers under this Agreement as are delegated to the Administrative Agent by this Agreement; and (vi) it will perform in accordance with their terms all of the obligations which by the terms of this Agreement are required to be performed by it as a Lender.

(d) The Administrative Agent shall maintain at the Administrative Agent's Office a copy of each Assignment and Acceptance delivered to it and a register (the "Register") of the names and address of each of the Lenders and the Pro Rata Share of the Commitments held by each Lender, giving effect to each Assignment and Acceptance. The Register shall be available during normal business hours for inspection by Borrower or any Lender upon reasonable prior notice to the Administrative Agent. After receipt of a completed Assignment and Acceptance executed by any Lender and an Eligible Assignee, and receipt of an assignment fee of \$3,500 from such Lender or Eligible Assignee, the Administrative Agent shall, promptly following the effective date thereof, provide to Borrower and the Lenders a revised Schedule 1.1 giving effect thereto. Borrower, the Administrative Agent and the Lenders shall deem and treat the Persons listed as Lenders in the Register as the holders and owners of the Pro Rata Shares of the Revolving Facility listed therein for all purposes hereof, and no assignment or transfer of any Lender's rights and obligations hereunder shall be effective, in each case unless and until an Assignment and Acceptance effecting the assignment or transfer thereof shall have been accepted by the Administrative Agent and recorded in the Register as provided above. Prior to such recordation, all amounts owed with respect to the applicable Pro Rata Share of the Revolving Facility shall be owed to the Lender listed in the Register as the owner thereof, and any request, authority or consent of any Person who, at the time of making such request or giving such authority or consent, is listed in the Register as a Lender shall be conclusive and binding on any subsequent holder, assignee or transferee of the corresponding Pro Rata Share of the Revolving Facility.

-83-

(e) Each Lender may from time to time grant participations to one or more banks or other financial institutions in or to all or a portion of its rights and/or obligations under this Agreement; provided, however, that (i) such Lender's obligations under this Agreement shall remain unchanged, (ii) such Lender shall remain solely responsible to the other parties hereto for the performance of such obligations, (iii) the participating banks or other financial institutions shall not be a Lender hereunder for any purpose except, if the participation agreement so provides, for the purposes of Sections 3.5, 3.6, 11.11 and 11.21 but only to the extent that the cost of such benefits to Borrower does not exceed the cost which Borrower would have incurred in respect of the Lender granting such participation absent the participation, (iv) Borrower, the Administrative Agent and the other Lenders shall continue to deal solely and directly with such Lender in connection with such Lender's rights and obligations under this Agreement, (v) the participation interest shall be expressed as a percentage of the granting Lender's Pro Rata Share of the Revolving Facility as it then exists and shall not restrict an increase in the Revolving Facility (or the aggregate Commitments pertaining thereto), or in the granting Lender's rights and obligations hereunder, so long as the amount of the participation interest is not affected thereby and (vi) the consent of the holder of such participation interest shall not be required for amendments or waivers of provisions of the Loan Documents other than those which (A) extend any Amortization Date, any applicable Maturity Date or any other date upon which any payment of money is due to the Lenders, (B) reduce the rate of interest on the Notes, any fee or any other monetary amount payable to the Lenders, (C) reduce the amount of any installment of principal due under the Notes, or (D) release any Guarantor from its Guaranty.

(f) Borrower agrees that upon the occurrence and during the continuance of any Event of Default, each Lender shall be entitled to assign its rights hereunder and under the Loan Documents, or grant participation interests in its rights under this Agreement and the Loan Documents, to any Person, in whole or in any part thereof, notwithstanding any provisions contained herein (including those set forth in subsection (b) above) or in any other Loan Document to the contrary, except that, other than (i) assignments by a Lender to an Affiliate of such Lender or to another Lender or (ii) pledges described in the last sentence of subsection (a) above, no assignment shall be made without the approval of the Administrative Agent.

11.9 Right of Setoff. If an Event of Default has occurred and is continuing, the Administrative Agent or any Lender (but in each case only with the consent of the Requisite Lenders) may exercise its rights under applicable Laws and, to the extent permitted by applicable Laws, apply any funds in any deposit account maintained with it by Borrower and/or any Property of Borrower in its possession against the Obligations.

11.10 Sharing of Setoffs. Each Lender severally agrees that if it, through the exercise of any right of setoff, banker's lien or counterclaim against Borrower, or otherwise, receives payment of the Obligations held by it that is ratably more than any other Lender, through any means, receives in payment of the Obligations held by that Lender, then, subject to applicable Laws: (a) the Lender exercising the right of setoff, banker's lien or counterclaim

-84-

or otherwise receiving such payment shall purchase, and shall be deemed to have simultaneously purchased, from each of the other Lenders a participation in the Obligations held by the other Lenders and shall pay to the other Lenders a purchase price in an amount so that the share of the Obligations held by each Lender after the exercise of the right of setoff, banker's lien or counterclaim or receipt of payment shall be in the same proportion that existed prior to the exercise of the right of setoff, banker's lien or counterclaim or receipt of payment; and (b) such other adjustments and purchases of participations shall be made from time to time as shall be equitable to ensure that all of the Lenders share any payment obtained in respect of the Obligations ratably in accordance with each Lender's share of the Obligations immediately prior to, and without taking into account, the payment; provided that, if all or any portion of a disproportionate payment obtained as a result of the exercise of the right of setoff, banker's lien, counterclaim or otherwise is thereafter recovered from the purchasing Lender by Borrower or any Person claiming through or succeeding to the rights of Borrower, the purchase of a participation shall be rescinded and the purchase price thereof shall be restored to the extent of the recovery, but without interest. Each Lender that purchases a participation in the Obligations pursuant to this Section 11.10 shall from and after the purchase have the right to give all notices, requests, demands, directions and other communications under this Agreement with respect to the portion of the Obligations purchased to the same extent as though the purchasing Lender were the original owner of the Obligations purchased. Borrower expressly consents to the foregoing arrangements and agrees that any Lender holding a participation in an Obligation so purchased pursuant to this Section 11.10 may exercise any and all rights of setoff, banker's lien or counterclaim with respect to the participation as fully as if the Lender were the original owner of the Obligation purchased.

11.11 Indemnity by Borrower. Borrower agrees to indemnify, save and hold harmless the Administrative Agent and each Lender and their respective directors, officers, agents, attorneys and employees (collectively the "Indemnitees") from and against: (a) any and all claims, demands, actions or causes of action (except a claim, demand, action, or cause of action for any amount excluded from the definition of "Taxes" in Section 3.10(e)) if the claim, demand, action or cause of action arises out of or relates to (i) any act or omission (or alleged act or omission) of Borrower, any Subsidiary or other Affiliate of Borrower or any partner, officer, director, stockholder, or other equity interest holder of Borrower relating to the Revolving Facility, (ii) the use or contemplated use of proceeds of any Borrowing, (iii) the relationship of Borrower and the Lenders under this Agreement, or (iv) the Loan Documents or the Revolving Facility in any other manner or aspect; (b) any administrative or investigative proceeding by any Governmental Agency arising out of or related to a claim, demand, action or cause of action described in clause (a) above; and (c) any and all liabilities, losses, reasonable costs or expenses (including reasonable attorneys' fees and the reasonably allocated costs of attorneys employed by any Indemnitee and disbursements of such attorneys and other professional services) that any Indemnitee suffers or incurs as a result of the assertion of any foregoing claim, demand, action or cause of action; provided that no Indemnitee shall be entitled to indemnification for any liability, loss, cost or expense caused by its own gross negligence or willful misconduct or for any liability, loss, cost or expense asserted against it by another Indemnitee. If any claim, demand, action or cause of action is asserted against any Indemnitee, such Indemnitee shall promptly notify Borrower, but the failure to so promptly notify Borrower shall not affect Borrower's obligations under this

-85-

Section unless such failure materially prejudices Borrower's right to participate in the contest of such claim, demand, action or cause of action, as hereinafter provided. Such Indemnitee may (and shall, if requested by Borrower in writing) contest the validity, applicability and amount of such claim, demand, action or cause of action and shall permit Borrower to participate in such contest. Any Indemnitee that proposes to settle or compromise any claim or proceeding for which Borrower may be liable for payment of indemnity hereunder shall give Borrower written notice of the terms of such proposed settlement or compromise reasonably in advance of settling or compromising such claim or proceeding and shall obtain Borrower's prior consent (which shall not be unreasonably withheld or delayed). In connection with any claim, demand, action or cause of action covered by this Section 11.11 against more than one Indemnitee, all such Indemnitees shall be represented by the same legal counsel (which may be a law firm engaged by the Indemnitees or attorneys employed by an Indemnitee or a combination of the foregoing) selected by the Indemnitees, provided, that if such legal counsel determines in good faith that representing all such Indemnitees would or could result in a conflict of interest under Laws or ethical principles applicable to such legal counsel or that a defense or counterclaim is available to an Indemnitee that is not available to all such Indemnitees, then to the extent reasonably necessary to avoid such a conflict of interest or to permit unqualified assertion of such a defense or counterclaim, each affected Indemnitee shall be entitled to separate representation by legal counsel selected by that Indemnitee, with all such legal counsel using reasonable efforts to avoid unnecessary duplication of effort by counsel for all Indemnitees; and further provided that the Administrative Agent (as an Indemnitee) shall at all times be entitled to representation by separate legal counsel (which may be a law firm or attorneys employed by the Administrative Agent or a combination of the foregoing). Any obligation or liability of Borrower to any Indemnitee under this Section 11.11 shall survive the expiration or termination of this Agreement and the repayment of all Borrowings and the payment and performance of all other Obligations owed to the Lenders.

11.12 Nonliability of the Lenders. Borrower acknowledges and agrees that:

(a) Any inspections of any Property of Borrower or any Subsidiary of Borrower made by or through the Administrative Agent or the Lenders are for purposes of administration of the Revolving Facility only and Borrower is not entitled to rely upon the same (whether or not such inspections are at the expense of Borrower);

(b) By accepting or approving anything required to be observed, performed, fulfilled or given to the Administrative Agent or the Lenders pursuant to the Loan Documents, neither the Administrative Agent nor the Lenders shall be deemed to have warranted or represented the sufficiency, legality, effectiveness or legal effect of the same, or of any term, provision or condition thereof, and such acceptance or approval thereof shall not constitute a warranty or representation to anyone with respect thereto by the Administrative Agent or the Lenders;

(c) The relationship between Borrower and the Administrative Agent and the Lenders is, and shall at all times remain, solely that of borrower and

-86-

lenders; neither the Administrative Agent nor the Lenders shall under any circumstance be construed to be partners or joint venturers of Borrower. any Subsidiary of Borrower or any of their respective Affiliates, neither the Administrative Agent nor the Lenders shall under any circumstance be deemed to be in a relationship of confidence or trust or a fiduciary relationship with Borrower, any Subsidiary of Borrower or any of their respective Affiliates, or to owe any fiduciary duty to Borrower, any Subsidiary of Borrower or any of their respective Affiliates; neither the Administrative Agent nor the Lenders undertake or assume any responsibility or duty to Borrower, any Subsidiary of Borrower or any of their respective Affiliates to select, review, inspect, supervise, pass judgment upon or inform Borrower, any Subsidiary of Borrower or any of their respective Affiliates of any matter in connection with their Property or the operations of Borrower, any Subsidiary of Borrower or any of their respective Affiliates; Borrower, Subsidiary of Borrower and their respective Affiliates shall rely entirely upon their own judgment with respect to such matters; and any review, inspection, supervision, exercise of judgment or supply of information undertaken or assumed by the Administrative Agent or the Lenders in connection with such matters is solely for the protection of the Administrative Agent and the Lenders and neither Borrower nor any other Person is entitled to rely thereon; and

(d) The Administrative Agent and the Lenders shall not be responsible or liable to any Person for any loss, damage, liability or claim of any kind relating to injury or death to Persons or damage to Property caused by the actions, inaction or negligence of Borrower, any Subsidiary of Borrower and/or any of their respective Affiliates and Borrower hereby indemnifies and holds the Administrative Agent and the Lenders harmless on the terms set forth in Section 11.11 from any such loss, damage, liability or claim.

11.13 No Third Parties Benefited. This Agreement is made for the purpose of defining and setting forth certain obligations, rights and duties of Borrower, the Administrative Agent and the Lenders in connection with the Revolving Facility, and is made for the sole benefit of Borrower, the Administrative Agent and the Lenders, and the Administrative Agent's and the Lenders' successors and assigns. Except as provided in Sections 11.8 and 11.11, no other Person shall have any rights of any nature hereunder or by reason hereof.

11.14 Confidentiality. Each Lender agrees to hold any confidential information that it may receive from Borrower pursuant to this Agreement in confidence, except for disclosure: (a) to other Lenders or Affiliates of a Lender that have agreed to keep such information confidential to the same extent as if they were a party hereto; (b) to legal counsel and accountants for Borrower, any Subsidiary of Borrower or any Lender; (c) to other professional advisors to Borrower or any Subsidiary of Borrower or any Lender, provided that the recipient has accepted such information subject to a confidentiality agreement substantially similar to this Section 11.14; (d) to regulatory officials having jurisdiction over that Lender; (e) as required by Law or legal process, provided that each Lender agrees to notify Borrower of any such disclosures unless prohibited by applicable Laws, or in connection with any legal proceeding to which that Lender and Borrower or any Subsidiary of

-87-

Borrower are adverse parties; and (f) to another financial institution in connection with a disposition or proposed disposition to that financial institution of all or part of that Lender's interests hereunder or a participation interest in its Note(s), provided that the recipient has accepted such information subject to a confidentiality agreement substantially similar to this Section 11.14. For purposes of the foregoing, "confidential information" shall mean any information respecting Borrower or any Subsidiary of Borrower reasonably considered by Borrower to be confidential, other than (i) information previously filed with any Governmental Agency and available to the public, (ii) information previously published in any public medium from a source other than, directly or indirectly, that Lender, and (iii) information previously disclosed by Borrower or such Subsidiary of Borrower to any Person not associated with Borrower or such Subsidiary of Borrower which does not owe a professional duty of confidentiality to Borrower or such Subsidiary of Borrower or which has not executed an appropriate confidentiality agreement with Borrower or such Subsidiary of Borrower. Nothing in this Section shall be construed to create or give rise to any fiduciary duty on the part of the Administrative Agent or the Lenders to Borrower or any Subsidiary of Borrower.

11.15 Further Assurances. Borrower shall, at its expense and without expense to the Lenders or the Administrative Agent, do, execute and deliver such further acts and documents as the Requisite Lenders or the Administrative Agent from time to time reasonably require for the assuring and confirming unto the Lenders or the Administrative Agent of the rights hereby created or intended now or hereafter so to be, or for carrying out the intention or facilitating the performance of the terms of any Loan Document.

11.16 Integration. This Agreement, together with the other Loan Documents and the letter agreements referred to in Section 3.3, comprises the complete and integrated agreement of the parties on the subject matter hereof and supersedes all prior agreements, written or oral, on the subject matter hereof. In the event of any conflict between the provisions of this Agreement and those of any other Loan Document, the provisions of this Agreement shall control and govern; provided that the inclusion of supplemental rights or remedies in favor of the Administrative Agent or the Lenders in any other Loan Document shall not be deemed a conflict with this Agreement. Each Loan Document was drafted with the joint participation of the respective parties thereto and shall be construed neither against nor in favor of any party, but rather in accordance with the fair meaning thereof.

11.17 Governing Law. EXCEPT TO THE EXTENT OTHERWISE PROVIDED THEREIN, EACH LOAN DOCUMENT SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS OF CALIFORNIA APPLICABLE TO CONTRACTS MADE AND PERFORMED IN CALIFORNIA. THE PARTIES AGREE THAT ALL ACTIONS OR PROCEEDINGS ARISING IN CONNECTION WITH THIS AGREEMENT SHALL BE TRIED AND LITIGATED ONLY IN A STATE OR FEDERAL COURT LOCATED IN THE STATE OF CALIFORNIA. THE PARTIES EXPRESSLY SUBMIT AND CONSENT IN ADVANCE TO SUCH JURISDICTION IN ANY ACTION OR PROCEEDING COMMENCED IN ANY SUCH COURT, AND THE PARTIES HEREBY WAIVE ANY OBJECTION THEY MAY HAVE BASED UPON LACK OF PERSONAL JURISDICTION AND HEREBY CONSENT

-88-

TO THE GRANTING OF SUCH LEGAL OR EQUITABLE RELIEF AS IS DEEMED APPROPRIATE BY ANY SUCH COURT. FURTHERMORE, THE PARTIES HEREBY WAIVE, TO THE EXTENT PERMITTED UNDER APPLICABLE LAW, ANY RIGHT THEY MAY HAVE TO ASSERT THE DOCTRINE OF "FORUM NON CONVENIENS" OR TO OBJECT TO VENUE TO THE EXTENT ANY PROCEEDING IS BROUGHT IN ACCORDANCE WITH THIS SECTION 11.16.

11.18 Severability of Provisions. Any provision in any Loan Document that is held to be inoperative, unenforceable or invalid as to any party or in any jurisdiction shall, as to that party or jurisdiction, be inoperative, unenforceable or invalid without affecting the remaining provisions or the operation, enforceability or validity of that provision as to any other party or in any other jurisdiction, and to this end the provisions of all Loan Documents are declared to be severable.

11.19 Headings. Article and Section headings in this Agreement and the other Loan Documents are included for convenience of reference only and are not part of this Agreement or the other Loan Documents for any other purpose.

11.20 Time of the Essence. Time is of the essence of the Loan Documents.

11.21 Foreign Lenders and Participants. Each Lender, and each holder of a participation interest herein, that is incorporated or otherwise organized under the Laws of a jurisdiction other than the United States of America or any State thereof or the District of Columbia shall deliver to Borrower (with a copy to the Administrative Agent), on or before the Closing Date (or on or before accepting an assignment or receiving a participation interest herein pursuant to Section 11.8, if applicable) two duly completed copies, signed by a Responsible Official, of Form W-8BEN or W-8ECI (or other equivalent successor form) satisfactory to Borrower and the Administrative Agent that no withholding under the federal income tax laws is required with respect to such Person. Thereafter and from time to time, each such Person shall (a) promptly submit to Borrower (with a copy to the Administrative Agent), such additional duly completed and signed copies of such form (or such successor form(s) as shall be adopted from time to time by the relevant United States taxing authorities) as may then be available under then current United States laws and regulations to avoid, or such evidence as is satisfactory to Borrower and the Administrative Agent of any available exemption from, United States withholding taxes in respect of all payments to be made to such Person by Borrower pursuant to this Agreement and (b) take such steps as shall not be materially disadvantageous to it, in the reasonable judgment of such Person, and as may be reasonably necessary (including the re-designation of its Eurodollar Lending Office, if any) to avoid any requirement of applicable Laws that Borrower make any deduction or withholding for taxes from amounts payable to such Person. In the event that Borrower or the Administrative Agent becomes aware that a participation has been granted pursuant to Section 11.8(e) to a financial institution that is incorporated or otherwise organized under the Laws of a jurisdiction other than the United States of America, any State thereof or the District of Columbia, then, upon request made by Borrower or the Administrative Agent to the Lender that granted such participation, such Lender shall cause such participant financial

-89-

institution to deliver the same documents and information to Borrower and the Administrative Agent as would be required under this Section if such financial institution were a Lender.

11.22 Waiver of Right to Trial by Jury. EACH PARTY TO THIS AGREEMENT HEREBY EXPRESSLY WAIVES ANY RIGHT TO TRIAL BY JURY OF ANY CLAIM, DEMAND, ACTION OR CAUSE OF ACTION ARISING UNDER ANY LOAN DOCUMENT OR IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTY HERETO OR ANY OF THEM WITH RESPECT TO ANY LOAN DOCUMENT, OR THE TRANSACTIONS RELATED THERETO, IN EACH CASE WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER SOUNDING IN CONTRACT OR TORT OR OTHERWISE; AND EACH PARTY HEREBY AGREES AND CONSENTS THAT ANY SUCH CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY, AND THAT ANY PARTY TO THIS AGREEMENT MAY FILE AN ORIGINAL COUNTERPART OR A COPY OF THIS SECTION WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE SIGNATORIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY.

11.23 Purported Oral Amendments. BORROWER EXPRESSLY ACKNOWLEDGES THAT THIS AGREEMENT AND THE OTHER LOAN DOCUMENTS MAY ONLY BE AMENDED OR MODIFIED, OR THE PROVISIONS HEREOF OR THEREOF WAIVED OR SUPPLEMENTED, BY AN INSTRUMENT IN WRITING THAT COMPLIES WITH SECTION 11.2. BORROWER AGREES THAT IT WILL NOT RELY ON ANY COURSE OF DEALING, COURSE OF PERFORMANCE, OR ORAL OR WRITTEN STATEMENTS BY ANY REPRESENTATIVE OF THE ADMINISTRATIVE AGENT OR ANY LENDER THAT DOES NOT COMPLY WITH SECTION 11.2 TO EFFECT AN AMENDMENT, MODIFICATION, WAIVER OR SUPPLEMENT TO THIS AGREEMENT OR THE OTHER LOAN DOCUMENTS.

> [THIS SPACE INTENTIONALLY LEFT BLANK -SIGNATURE PAGES TO FOLLOW]

> > -90-

AMERICAN STATES WATER COMPANY, a California corporation

By /s/ Floyd E. Wicks ------Name: Floyd E. Wicks Title: President and Chief Executive Officer

Address for Borrower:

630 East Foothill Boulevard San Dimas, California 91773-9016 Attn: McClellan Harris III Telecopier: (909) 394-1382 Telephone: (909) 394-3600

With a copy to:

American States Water Company 630 East Foothill Boulevard San Dimas, California 91773-9016 Attn: Eva Tang Telecopier: (909) 394-1382 Telephone: (909) 394-3600

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By /s/ Anthony Turner

Name: Anthony D. Turner Title: Vice President

Address for notices to Administrative Agent for borrowings and payments:

Wells Fargo Bank, National Association 201 Third Street, 8th Floor San Francisco, California 94103 Attn: Deborah Moore

Telecopier: (415) 512-9408 Telephone: (415) 477-5379 E-mail: mooredj@wellsfargo.com

With a copy to:

333 South Grand Avenue, Third Floor Los Angeles, California 90071 Attn: Carmen Badescu

Telecopier: (213) 628-1415 Telephone: (213) 253-6146 E-mail: badescuc@wellsfargo.com

s-2

WELLS FARGO BANK, NATIONAL ASSOCIATION, as a Lender  $% \left( {{\left( {{{\left( {{{\left( {{{}_{{\rm{T}}}}} \right)}} \right)}_{\rm{T}}}} \right)}_{\rm{T}}} \right)$ 

By /s/ Anthony Tuner Anthony Turner Vice President

Address:

333 South Grand Avenue, Third Floor Los Angeles, California 90071 Attn: American States Water Account Officer

Telecopier: (213) 687-3501 Telephone: (213) 253-6208 E-mail: turnerad@wellsfargo.com

s-3

COBANK, ACB, as a Lender

Address:

5500 South Quebec St. Greenwood Village, Colorado 80111 Attn: Jerry Bucholz, Vice President

Telecopier: (303) 224-2537 Telephone: (303) 740-4314 E-mail: jbucholz@cobank.com

UNION BANK OF CALIFORNIA, N.A., as a Lender

By /s/ Susan K. Johnson

Name: Susan K. Johnson Title: Vice President

Address:

Energy Capital Services 445 South Figueroa St., 15th Floor Los Angeles, California 90071 Attn: Kevin Zitar, Vice President

Telecopier: (213) 236-4096 Telephone: (213) 236-5503 E-mail: Kevin.Zitar@uboc.com

COMERICA WEST INCORPORATED, as a Lender

By /s/ Elise Walker Elise M. Walker Assistant Vice President

Address:

1920 Main Street, Suite 1150 Irvine, California 92614 Attn: Elise M. Walker, Asst. Vice President

Telecopier: (949) 476-1222 Telephone: (949) 798-7244 E-mail: elise\_m\_walker@comerica.com

NORTHERN TRUST BANK OF CALIFORNIA, N.A., as a Lender  $% \left( {{\left( {{{\left( {{{\left( {{K_{\rm{T}}}} \right)}} \right.}} \right)}} \right)$ 

By /s/ Stephen W. Lind ------Name: Stephen W. Lind Title: Senior Vice President

Address:

355 South Grand Ave., Suite 2600 Los Angeles, California 90071 Attn: Stephen W. Lind

Telecopier: (213) 617-4360 Telephone: (213) 346-1368 E-mail: swl2@ntrs.com

s-7

## EXHIBIT A

# ASSIGNMENT AND ACCEPTANCE

THIS ASSIGNMENT AND ACCEPTANCE ("Agreement") dated as of \_\_\_\_\_, is made with reference to that certain Credit Agreement dated as of June 6, 2002 (as amended, extended, renewed, supplemented or otherwise modified from time to time, the "Credit Agreement") by and among AMERICAN STATES WATER COMPANY, a California corporation (the "Borrower"), the lenders from time to time party thereto (the "Lenders"), and WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent for the Lenders, and is entered into between the "Assigner" described below, in its capacity as a Lender under the Credit Agreement, and the "Assignee" described below.

Assignor and Assignee hereby represent, warrant and agree as follows:

1. Definitions. Capitalized terms defined in the Credit Agreement are used herein with the meanings set forth for such terms in the Credit Agreement. As used in this Agreement, the following capitalized terms shall have the meanings set forth below:

"Assignee" means

"Aggregate Commitments" means the aggregate amount of the Lenders' Commitments under the Credit Agreement as of the date hereof.

"Assigned Share" means \_\_\_\_% of the Aggregate Commitments of the Lenders under the Credit Agreement which equals  $\$ \_\_\_\_\_.

"Assignor" means \_\_\_\_\_

"Effective Date" means \_\_\_\_\_, the effective date of this Agreement determined in accordance with Section 11.8 of the Credit Agreement.

2. Representations and Warranties of the Assignor. The Assignor represents and warrants to the Assignee as follows:

a. As of the date hereof, the Pro Rata Share of the Assignor of the Revolving Facility is \_\_\_\_\_\_% of the Revolving Facility (without giving effect to assignments thereof which have not yet become effective). The Assignor is the legal and beneficial owner of the Assigned Pro Rata Share and the Assigned Pro Rata Share is free and clear of any adverse claim.

b. As of the date hereof, the outstanding principal balance of Advances made by the Assignor under the Assignor's Note is \$\_\_\_\_\_\_. Unless otherwise agreed among Assignor and Assignee, amounts outstanding under Assignor's Note, if any, shall not be assigned.

-1-

c. The Assignor has full power and authority, and has taken all action necessary, to execute and deliver this Agreement and any and all other documents required or permitted to be executed or delivered by it in connection with this Agreement and to fulfill its obligations under, and to consummate the transactions contemplated by, this Agreement, and no governmental authorizations or other authorizations are required in connection therewith; and

d. This Agreement constitutes the legal, valid and binding obligation of the Assignor.

The Assignor makes no representation or warranty and assumes no responsibility with respect to the financial condition of Borrower or the performance by Borrower of the Obligations, and the Assignor makes no representation or warranty and assumes no responsibility with respect to any statements, warranties or representations made in or in connection with the Credit Agreement or the execution, legality, validity, enforceability, genuineness, or sufficiency of the Credit Agreement or any other Loan Document other than as expressly set forth above.

3. Representations and Warranties of the Assignee. The Assignee hereby represents and warrants to the Assignor as follows:

(a) The Assignee has full power and authority, and has taken all action necessary, to execute and deliver this Agreement, and any and all other documents required or permitted to be executed or delivered by it in connection with this Agreement and to fulfill its obligations under, and to consummate the transactions contemplated by, this Agreement, and no governmental authorizations or other authorizations are required in connection therewith;

(b) This Agreement constitutes the legal, valid and binding obligation of the Assignee;

(c) The Assignee has independently and without reliance upon the Administrative Agent or Assignor and based on such documents and information as the Assignee has deemed appropriate, made its own credit analysis and decision to enter into this Agreement. The Assignee will, independently and without reliance upon the Administrative Agent or any Lender, and based upon such documents and information as it shall deem appropriate at the time, continue to make its own credit decisions in taking or not taking action under the Credit Agreement;

(d) The Assignee has received copies of such of the Loan Documents delivered pursuant to Section 8.1 of the Credit Agreement as it has requested, together with copies of the most recent financial statements delivered pursuant to Section 7.1 of the Credit Agreement;

(e) The Assignee will perform in accordance with their respective terms all of the obligations which by the terms of the Credit Agreement are required to be performed by it as a Lender; and

(f) The Assignee is an Eligible Assignee.

-2-

4. Assignment. On the terms set forth herein, the Assignor, as of the Effective Date, hereby irrevocably sells, assigns and transfers to the Assignee all of the rights and obligations of the Assignor under the Credit Agreement, the other Loan Documents and the Assignor's Note to the extent of the Assigned Share, and the Assignee irrevocably accepts such assignment of rights and assumes such obligations from the Assignor on such terms and effective as of the Effective Date. As of the Effective Date, the Assignee shall have the rights and obligations of a "Lender" under the Loan Documents, except to the extent of any arrangements with respect to payments referred to in Section 5 hereof. Assignee hereby appoints and authorizes the Administrative Agent, to take such action and to exercise such powers under the Credit Agreement as are delegated to the Administrative Agent by the Credit Agreement.

5. Payment. On the Effective Date, the Assignee shall pay to the Assignor, in immediately available funds, an amount equal to the purchase price of the Assigned Share, as agreed between the Assignor and the Assignee pursuant to a letter agreement of even date herewith. Such letter agreement also sets forth the agreement between the Assignor and the Assignee with respect to the amount of interest, fees, and other payments with respect to the Assigned Share which are to be retained by the Assignor. Assignee shall also pay to the Administrative Agent, as a condition to the effectiveness of this Agreement, an assignment fee of \$3,500 in accordance with Section 11.8(d) of the Credit Agreement.

The Assignor and the Assignee hereby agree that if either receives any payment of interest, principal, fees or any other amount under the Credit Agreement, their respective Note or any other Loan Documents which is for the account of the other, it shall hold the same in trust for such party to the extent of such party's interest therein and shall promptly pay the same to such party.

6. Principal, Interest, Fees, etc. Any principal that would be payable and any interest, fees and other amounts that would accrue from and after the Effective Date to or for the account of the Assignor pursuant to the Credit Agreement and its Note shall be payable to or for the account of the Assignor and the Assignee, in accordance with their respective interests as adjusted pursuant to this Agreement.

7. Notes. The Assignor and the Assignee shall make appropriate arrangements with the Borrower concurrently with the execution and delivery hereof so that a replacement Note is issued to the Assignor and a new Note, is issued to the Assignee in principal amounts reflecting their Pro Rata Shares of the Revolving Facility and their outstanding Advances under the Revolving Facility (as adjusted pursuant to this Agreement).

8. Further Assurances. Concurrently with the execution of this Agreement, the Assignor shall execute two counterpart original Requests for Registration, in the form of Exhibit A to this Agreement, to be forwarded to the Administrative Agent. The Assignor and the Assignee further agree to execute and deliver such other instruments, and take such other action, as either party may reasonably request in connection with the transactions contemplated by this Agreement, and the Assignor specifically agrees to cause the delivery of (i) two original counterparts of this Agreement and (ii) the Request for

-3-

Registration, to the Administrative Agent for the purpose of registration of the Assignee as a "Lender" pursuant to Section 11.8 of the Credit Agreement.

9. Governing Law. THIS AGREEMENT SHALL BE DEEMED TO BE A CONTRACTUAL OBLIGATION UNDER, AND SHALL BE GOVERNED BY AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF CALIFORNIA. FOR ANY DISPUTE ARISING IN CONNECTION WITH THIS AGREEMENT, THE ASSIGNEE HEREBY IRREVOCABLY SUBMITS TO THE JURISDICTION OF THE COURTS OF THE STATE OF CALIFORNIA.

10. Notices. All communications among the parties or notices in connection herewith shall be in writing, hand delivered or sent by U.S. registered mail, postage prepaid, or by telecopy, addressed to the appropriate party at its address set forth on the signature pages hereof. All such communications and notices shall be effective upon receipt.

11. Binding Effect. This Agreement shall be binding upon and inure to the benefit of the parties and their respective successors and assigns; provided, however, that the Assignee shall not assign its rights or obligations without the prior written consent of the Assignor and any purported assignment, absent such consent, shall be void. Nothing contained in this Section shall restrict the assignment by Assignee of its rights under the Loan Document following the Effective Date.

12. Interpretation. The headings of the various sections hereof are for convenience of reference only and shall not affect the meaning or construction of any provision hereof.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed and delivered by their respective officials, officers or agents thereunto duly authorized as of the date first above written.

"Assignor"
Ву:
[Printed Name and Title]
Address:
Telephone
Telecopier

-4-

# "Assignee"

By: \_\_\_\_\_\_ [Printed Name and Title] Address: \_\_\_\_\_\_ Telephone \_\_\_\_\_\_ Telecopier \_\_\_\_\_\_

-5-

#### REQUEST FOR REGISTRATION

To: Wells Fargo Bank, National Association, as Administrative Agent, and the Borrower (described below).

THIS REQUEST FOR REGISTRATION is made as of the date of the enclosed Assignment and Acceptance with reference to that certain Credit Agreement, dated as of June 6, 2002 by and among AMERICAN STATES WATER COMPANY, a California corporation (the "Borrower"), the lenders from time to time party thereto (the "Lenders"), and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as amended as of the date hereof, the "Credit Agreement").

The Assignor and Assignee described below hereby request that Administrative Agent register the Assignee as a Lender pursuant to Section 11.8 of the Credit Agreement effective as of the Effective Date described in the Assignment and Acceptance. Assignee hereby confirms the representations and warranties made in Section 3 of the Assignment and Acceptance for the benefit of Administrative Agent and the Borrower. Concurrently herewith a similar portion of the interests of Assignor under the other Loan Documents is being assigned to Assignee.

Enclosed with this Request are two counterpart originals of the Assignment and Acceptance as well as the original Note of Borrower in favor of the Assigner in the principal amount of \$\_\_\_\_\_\_. The Assignor and Assignee hereby jointly request that Administrative Agent cause Borrower to issue a replacement Note, dated as of the Effective Date, pursuant to Section 11.8 of the Credit Agreement in favor of Assignor in the principal amount of the remainder of its Pro Rata Share of the Revolving Facility and a Note in favor of the Assignee in the amount of the Assignee Pro Rata Share, dated as of the Effective Date, substantially in the form of Exhibit D to the Credit Agreement.

-6-

IN WITNESS WHEREOF, the Assigne	or and Assignee have executed this
Request for Registration by their duly at	uthorized officers as of this day of
'·	
"Assignor"	"Assignee"
By:	By:
[Printed Name and Title]	[Printed Name and Title]

-7-

CONSENT OF ADMINISTRATIVE AGENT AND BORROWER

[When Required Pursuant to Credit Agreement]

TO: The Assignor and Assignee referred to in the Request for Registration to which this Consent is attached

When countersigned by Administrative Agent and (when applicable) Borrower below, this document shall certify that:

[\_] [CHECK HERE IF BORROWER'S SIGNATURE IS REQUIRED PURSUANT TO SECTION 11.8(b)(i) OF THE CREDIT AGREEMENT:]

1. Borrower has consented, pursuant to the terms of the Loan Documents, to the assignment by the Assignor to the Assignee of the Assigned Share.

2. Administrative Agent has registered the Assignee as a Lender under the Credit Agreement, effective as of the Effective Date described above, with a Share of the Revolving Facility corresponding to the Assigned Pro Rata Share and has adjusted the registered Share of the Revolving Facility of the Assignor to reflect the assignment of the Assigned Share.

AMERICAN STATES WATER COMPANY, a California corporation

By:

Name: Title:

WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

By:

Name: Title:

-8-

# EXHIBIT B

### COMPLIANCE CERTIFICATE

#### TO: WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

Reference is made to the Credit Agreement dated as of June 6, 2002, entered into by and among AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), the lenders party thereto from time to time, and WELL FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent (the "Agent") (as amended from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined in this Compliance Certificate ("Certificate") shall have the meanings defined for them in the Credit Agreement. Section references herein relate to the Credit Agreement unless stated otherwise.

## COMPLIANCE WITH FINANCIAL COVENANTS

This Certificate is delivered in accordance with Section 7.2 of the Credit Agreement by the president or chief financial officer of Borrower. This certificate relates to the financial statements of Borrower and its Subsidiaries delivered concurrently herewith for the fiscal period ended \_\_\_\_\_\_, (the "Determination Date"). Computations indicating compliance with respect to the covenants specified in 6.3, 6.9(d) and (e), 6.12, 6.13, 6.14(d), (g) and (h), and 6.15 of the Loan Agreement are as follows. All calculations are as determined in accordance with GAAP, consistently applied except to the extent modified by the terms of the Credit Agreement.

A. 6.3 Disposition of Property. As of the Determination Date:

1. The aggregate amount of Dispositions of Property (other than obsolete Property or Property with no remaining useful life) made after the Closing Date by Borrower and its Subsidiaries during the Fiscal Year (or portion thereof) ending on the Determination Date was \_\_\_\_\_.

#### Maximum Permitted:

\$ 1,000,000

2. The aggregate amount of Dispositions of Property (other than obsolete Property or Property with no remaining useful life) made after the Closing Date by Borrower and its Subsidiaries as of the Determination Date was \$\_\_\_\_\_.

Maximum Permitted:

\$ 5,000,000

B. 6.9(d) and (e) Liens and Negative Pledges.

1. The aggregate amount of Indebtedness secured by Liens on Property acquired by Borrower or any Subsidiary that were in existence at the time of the acquisition of such

-1-

Property and were not created in contemplation of such acquisition as of the Determination Date was  $\qquad$ 

Maximum Permitu	Maximum	Permitted:
-----------------	---------	------------

\$10,000,000

2. As of the Determination Date, the aggregate amount of Permitted Capital Asset Indebtedness secured by Liens on capital assets acquired, constructed or financed with the proceeds of such Permitted Capital Asset Indebtedness or with the proceeds of any Indebtedness directly or indirectly refinanced by such Indebtedness was \$\_\_\_\_\_.

Maximum Permitted:

\$10,000,000

\$

Ś

C. 6.12; Total Funded Debt Ratio. As of the last day of the Fiscal Quarter ending on the Determination Date, the Funded Debt Ratio was \_\_\_:1.00.

Maximum Permitted for Determination Date: 0.65:1.00

1. Total Funded Debt Ratio Calculation. As of the Determination Date, without duplication:

(a) Total Funded Debt (item 2)	\$
divided by (b)(ii) Total Funded Debt	\$
plus (iii) Stockholders' Equity	\$
equals Total Funded Debt Ratio [(a)/(b)]	\$

2. Total Funded Debt Calculation as of the Determination Date:

(a) (i) All principal Indebtedness of Borrower and its Subsidiaries for borrowed money (including Subordinated Obligations and any other subordinated indebtedness, debt Securities issued by Borrower and any of its Subsidiaries, the aggregate principal Indebtedness outstanding under the Notes and the Aggregate Effective Amount of all Outstanding Letters of Credit).

plus (ii) the aggregate amount of the principal portion of all Capital Lease Obligations of Borrower and its Subsidiaries.

plus (iii) any Guaranty Obligations of Borrower and its Subsidiaries with respect to the Indebtedness of others of the types referred to in (i) and (ii) above.

equals (a) Total Funded Debt [(i)+(ii)+(iii)]

-2-

D. 6.13; Interest Coverage Ratio. As of the last day of the Fiscal Quarter ending on the Determination Date, the Interest Coverage Ratio was:1:00.					
	Minimum Permitted for such fiscal period:	3.25:1.00			
	Interest Coverage Ratio for the Rolling Period ending as of nation Date:	the			
	Without duplication, the sum of:				
	(a) EBITDA for the Rolling Period ending on the Determination Date (item 2 below)	\$			
	divided by (b) Interest Expense for such Rolling Period ending on the Determination Date	\$			
	equals Interest Coverage Ratio $[(a)/(b)]$	:1.00			
2. Date:	EBITDA Calculation for the Rolling Period ending as of the D	etermination			
	(a) Net Income for such fiscal period	\$			
	plus (b) any extraordinary loss reflected in such Net Income	\$			
	minus (c) any extraordinary gain reflected in such Net Income	\$			
	plus (d) Interest Expense of Borrower and its Subsidiaries for that Fiscal period	\$			
	plus (e) the aggregate amount of federal and state taxes on or measured by income of Borrower or its Subsidiaries for that fiscal period (whether or not payable during that fiscal period)	\$			
	plus (f) depreciation and amortization expenses of Borrower and its Subsidiaries for that fiscal period	\$			

-3-

plus (g) all other non-cash extraordinary expenses of Borrower and its Subsidiaries for that fiscal period acceptable to the Requisite Lenders(1)

\$		

\$

equals EBITDA [(a) + (b) - (c) + (d) + (e) + (f) + (g)]

E. 6.14 (d), (g) and (h) Investments and Acquisitions.

1. The aggregate amount of Investments consisting of advances to officers, directors and employees of Borrower and its Subsidiaries for travel, entertainment, relocation, anticipated bonus and analogous ordinary business purposes as of the Determination Date was \$\_\_\_\_\_\_.

Maximum Permitted: \$1,000,000

2. The aggregate amount of Investments received in connection with the settlement of a bona fide dispute with another Person as of the Determination Date was  $\qquad$ .

## Maximum Permitted:

3. The aggregate amount of Investments representing all or a portion of the sale price of Property sold or services provided to another Person as of the Determinate Date was \$ .

Maximum Permitted: \$ 500,000

F. 6.15 Operating Leases. The aggregate obligations of Borrower and its Subsidiaries to pay rent under all operating leases during the Fiscal Year (or portion thereof) ending on the Determination Date was \$\_\_\_\_\_.

Maximum Permitted:

\$ 2,000,000

\$ 500,000

#### PERFORMANCE OF OBLIGATIONS

A review of the activities of Borrower and its Subsidiaries during the fiscal period covered by the attached financial statements has been made under the supervision of the undersigned with a view to determining whether during such fiscal period Borrower and its Subsidiaries performed and observed all of their Obligations under the Loan Documents. Except as described in an attached document (which includes the response thereto which Borrower and its Subsidiaries have taken or propose to take) or in an earlier Compliance Certificate, to the best knowledge of the undersigned, after the exercise of reasonable due

- -----

(1) Items (d), (e), (f) and (g) should be added only to the extent reflected in the determination of Net Income for that fiscal period.

-4-

diligence, as of the date of this Certificate no Default or  $\ensuremath{\mathsf{Event}}$  of Default has occurred and is continuing.

# BOOKS AND RECORDS

The undersigned certifies, on behalf of Borrower, that the calculations made and the information contained herein are derived from the books and records of Borrower and its Subsidiaries, as applicable, and that each and every matter contained herein correctly reflects those books and records.

## NO ADVERSE MATERIAL CHANGE

To the best knowledge of the undersigned, after the exercise of reasonable due diligence, except as described in an attached document or in an earlier Compliance Certificate, no Material Adverse Effect has occurred since the date of the most recent Compliance Certificate delivered to the Agent.

Date: \_\_\_\_\_, \_\_\_\_

By: Name: Title:

-5-

# STANDBY LETTER OF CREDIT AGREEMENT

#### To: WELLS FARGO BANK, NATIONAL ASSOCIATION

Applicant hereby requests that you, Wells Fargo Bank, National Association ("Wells Fargo"), issue in your name one or more standby letters of credit pursuant to Applications for the issuance of such Credits and the terms and conditions of this Agreement. Each Credit will be issued at Applicant's request and for its account, and, unless otherwise specifically provided in any Loan Document, at your option. Applicant agrees that the terms and conditions in this Agreement shall apply to each Application and the Credit issued pursuant to each Application, and to transactions under each Application, each Credit and this Agreement.

SECTION 1. DEFINITIONS. As used in this Agreement, the following terms shall have the meanings set forth after each term: "Agreement" means this Standby Letter of Credit Agreement as it may be revised or amended from time to time. "Applicant" means collectively each person and/or entity signing this Agreement as Applicant. "Application" means your printed form titled "Application For Standby Letter of Credit" or any other form acceptable to you on which Applicant applies for the issuance by you of a Credit and/or an application for amendment of a Credit or any combination of such applications, as the context may require. "Beneficiary" means the person or entity named on an Application as the beneficiary or any transferee of such beneficiary. 'Collateral" means the Property, together with the proceeds of such Property, securing any or all of Applicant's obligations and liabilities at any time existing under or in connection with any L/C Document and/or any Loan Document. "Commission Fee" means the fee, computed at the commission fee rate specified by you or specified in any Loan Document, charged by you at the time or times specified by you on the amount of each Credit and on the amount of each increase in a Credit for the time period each Credit is outstanding. "Credit" means an instrument or document titled "Irrevocable Standby Letter of Credit" or "Standby Letter of Credit", or any instrument or document whatever it is titled or whether or not it is titled functioning as a standby letter of credit, issued under or pursuant to an Application, and all renewals, extensions and amendments of such instrument or document. "Demand" means any sight draft, electronic or telegraphic transmission or other written demand drawn or made, or purported to be drawn or made, under or in connection with any Credit. "Document" means any instrument, statement, certificate, or other document referred to in or related to any Credit or required by any Credit to be presented with any Demand. "Dollars" means the lawful currency at any time for the payment of public or private debts in the United States of America. "Event of Default" means any of the events set forth in Section 13 of this Agreement. "Expiration Date" means the date any Credit expires. "Guarantor" means any person or entity guaranteeing the payment and/or performance of any or all of Applicant's obligations under or in connection with any L/C Document and/or any Loan Document. "Holding Company" means any company or other entity directly or indirectly controlling you. "L/C Document" means this Agreement, each Application, each Credit, and each Demand. "Loan Document" means each and any promissory note, loan agreement, security agreement, pledge agreement, guarantee or other agreement or document executed in connection with, or relating to, any extension of credit under which any Credit is issued. "Maximum Rate" means the maximum amount of interest (as defined by applicable laws), if any, permitted to be paid, taken, reserved, received, collected or charged under applicable laws, as the same may be amended or modified from time to time. "Negotiation Fee" means the fee, computed at the negotiation fee rate specified by you or specified in any Loan Document, charged by you on the amount of each Demand paid by you or any other bank specified by you when each Demand is paid. "Payment Office" means the office specified by you or specified in any Loan Document as the office where reimbursements and other payments under or in connection with any L/C Document are to be made by Applicant. "Prime Rate" means the rate of interest most recently announced within Wells Fargo at its principal office as its Prime Rate, with the understanding that the Prime Rate is one of Wells Fargo's base rates and serves as the basis upon which effective rates of interest are calculated for those loans making reference thereto, and is evidenced by the recording thereof after its announcement in such internal publication or publications as Wells Fargo may designate. "Property" means all forms of property, whether tangible or intangible, real, personal or mixed. "Rate of Exchange" means Wells Fargo's then current selling rate of exchange in San Francisco, California for sales of the currency of payment of any Demand, or of any fees or expenses or other amounts payable under this Agreement, for cable transfer to the country of which such currency is the legal tender. "UCP" means the Uniform Customs and Practice for Documentary Credits, an International Chamber of Commerce publication, or any substitution therefor or replacement thereof. "Unpaid and Undrawn Balance" means at any time the entire amount which has not been paid by you under all the Credits issued for Applicant's account, including, without limitation, the amount of each Demand on which you have not yet effected payment as well as the amount undrawn under all such Credits. "Wells Fargo & Company" means Wells Fargo & Company, a Delaware corporation.

SECTION 2. HONORING DEMANDS AND DOCUMENTS. You may receive, accept and honor, as complying with the terms of any Credit, any Demand and any Documents accompanying such Demand; provided, however, that such Demand and accompanying Documents appear on their face to comply substantially with the provisions of such Credit and are, or appear on their face to be, signed or issued by (a) a person or entity authorized under such Credit to draw, sign or issue such Demand and accompanying Documents, or (b) an administrator, executor, trustee in bankruptcy, debtor in possession, assignee for the benefit of creditors, liquidator, receiver or other legal representative or successor in interest by operation of law of any such person or entity.

SECTION 3. REIMBURSEMENT FOR PAYMENT OF DEMANDS. Applicant shall reimburse you for all amounts paid by you on each Demand, including, without limitation, all such amounts paid by you to any paying, negotiating or other bank. If in connection with the issuance of any Credit, you agree to pay any other bank the amount of any payment or negotiation made by such other bank under such Credit upon your receipt of a cable, telex or other written telecommunication advising you of such payment or negotiation, or authorizes any other bank to debit your account for the amount of such payment or negotiation, Applicant agrees to reimburse you for all such amounts paid by you, or debited to your account with such other bank, even if any Demand or Document specified in such Credit fails to arrive in whole or in part or if, upon the arrival of any such Demand or Document, the terms of such Credit have not been complied with or such Demand or Document does not conform to the requirements of such Credit or is not otherwise in order.

SECTION 4. FEES AND EXPENSES. Applicant agrees to pay to you (a) all Commission Fees, Negotiation Fees, cable fees, amendment fees, non-usance fees, and cancellation fees of, and all out-of-pocket expenses incurred by, you under or in connection with any L/C Document, and (b) all fees and charges of banks or other entities other than you under or in connection with any L/C Document if any Application (i) does not indicate who will pay such fees and charges, (ii) indicates that such fees and charges are to be paid by Applicant, or (iii) indicates that such fees and charges are to be paid by the Beneficiary and the Beneficiary does not, for any reason whatsoever, pay such fees or charges. There shall be no refund of any portion of any Commission Fee in the event any Credit is used, reduced, amended modified or terminated before its Expiration Date.

SECTION 5. DEFAULT INTEREST. Unless otherwise specified in any Loan Document, or on an Application and agreed to by you, all amounts to be reimbursed by Applicant to you, and all fees and expenses to be paid by Applicant to you, and all other amounts due from Applicant to you under or in connection with any L/C Documents, will bear interest (to the extent permitted by law), payable on demand, from the date you paid the amounts to be reimbursed or the date such fees, expenses and other amounts were due until such amounts are paid in full, at a rate per annum (computed on the basis of a 360-day year, actual days elapsed) which is the lesser of (a) two percent (2%) above the Prime Rate in effect from time to time, or (b) the Maximum Rate.

SECTION 6. TIME AND METHOD OF REIMBURSEMENT AND PAYMENT. Unless otherwise specified in this Section, in any Loan Document, or on an Application and agreed to by you, all amounts to be reimbursed by Applicant to you, all fees and expenses to be paid by Applicant to you, and all interest and other amounts due to you from Applicant under or in connection with any L/C Documents will be reimbursed or paid at the Payment Office in Dollars in immediately available funds without setoff or counterclaim (i) on demand or, (ii) at your option by your debiting any of Applicant's accounts with you, with each such debit being made without presentment, protest, demand for reimbursement or payment, notice of dishonor or any other notice whatsoever, all of which are hereby expressly waived by Applicant. Each such debit will be made at the time each Demand is paid by you or, if earlier, at the time each amount is paid by you to any paying, negotiating or other bank, or at the time each fee and expense is to be paid or any interest or other amount is due under or in connection with any L/C Documents. If any Demand or any fee, expense, interest or other amount payable under or in connection with any L/C Documents is payable in a currency other than Dollars, Applicant agrees to reimburse you for all amounts paid by you on such Demand, and/or to pay you all such fees, expenses, interest and other amounts, in one of the three following ways, as determined by you in your sole discretion in each case: (a) at such place as you shall direct, in such other currency; or (b) at the Payment Office in the Dollar equivalent of the amount of such other currency calculated at the Rate of Exchange on the date determined by you in your sole discretion; or (c) at the Payment Office in the Dollar equivalent, as determined by you (which determination shall be deemed correct absent manifest error), of such fees, expenses, interest or other amounts or of the actual cost to you of paying such Demand. Applicant assumes all political, economic and other risks of disruptions or interruptions in any currency exchange.

SECTION 7. AGREEMENTS OF APPLICANT. Applicant agrees that (a) unless otherwise specifically provided in any Loan Document, you shall not be obligated at any time to issue any Credit for Applicant's account; (b) unless otherwise specifically provided in any Loan Document, if any Credit is issued by you for Applicant's account, you shall not be obligated to issue any further Credit for Applicant's account or to make other extensions of credit to Applicant or in any other manner to extend any financial consideration to Applicant; (c) you have not given Applicant any legal or other advice with regard to any  $\ensuremath{\mathrm{L/C}}$  Document or Loan Document; (d) if you at any time discuss with Applicant the wording for any Credit, any such discussion will not constitute legal or other advice by you or any representation or warranty by you that any wording or Credit will satisfy Applicant's needs; (e) Applicant is responsible for the wording of each Credit, including, without limitation, any drawing conditions, and will not rely on you in any way in connection with the wording of any Credit or the structuring of any transaction related to any Credit; (f) Applicant, and not you, is responsible for entering into the contracts relating to the Credits between Applicant and the Beneficiaries and for causing Credits to be issued; (g) you may, as you deem appropriate, modify or altar and use in any Credit the terminology contained on the Application for such Credit; (h) unless the Application for a Credit specifies whether the Documents to be presented with a Demand under such Credit must be sent to you in one parcel or in two parcels or may be sent to you in any number of parcels, you may, if you so desire, make such determination and specify in the Credit whether such

Documents must be sent in one parcel or two parcels or may be sent in any number of parcels; (i) you shall not be deemed Applicant's agent or the agent of any Beneficiary or any other user of any Credit, and neither Applicant, nor any Beneficiary nor any other user of any Credit shall be deemed your agent; (j) Applicant will promptly examine all Documents and each Credit if and when they are delivered to Applicant and, in the event of any claim of noncompliance of any Documents or any Credit with Applicant's instructions or any Application, or in the event of any other irregularity, Applicant will promptly notify you in writing of such noncompliance or irregularity, Applicant being conclusively deemed to have waived any such claim of noncompliance or irregularity unless such notice is promptly given; (k) all directions and correspondence relating to any L/C Document are to be sent at Applicant's risk; (1) if any Credit has a provision concerning the automatic extension of its Expiration Date, you may, at your sole option, give notice of nonrenewal of such Credit and if Applicant does not at any time want such Credit to be renewed Applicant will so notify you at least fifteen (15) calendar days before you are to notify the Beneficiary of such Credit or any advising bank of such nonrenewal pursuant to the terms of such Credit; (m) Applicant will not seek to obtain, apply for, or acquiesce in any temporary or permanent restraining order, preliminary or permanent injunction, permanent injunction or any other pretrial or permanent injunctive or similar relief, restraining, prohibiting or enjoining you, any of your correspondents or any advising, confirming, negotiating, paying or other bank from paying or negotiating any Demand or honoring any other obligation under or in connection with any Credit; and (n) except for Applicant's obligations specifically affected by those actions or failures to act referred to in subsections (ii) and (vii) of this Section 7(n) which you have performed or approved or accepted, Applicant's obligations under or in connection with each L/C Document and Loan Document shall be absolute, unconditional and irrevocable, and shall be performed strictly in accordance with the terms of each such  $\mbox{\rm L/C}$ Document and Loan Document under all circumstances whatsoever, including, without limitation, the following circumstances, the circumstances listed in Section 12(b) through (u) of this Agreement, and any other event or circumstance similar to such circumstances: (i) any lack of validity or enforceability of any L/C Document, any Loan Document, any Document or any agreement relating to any of the foregoing; (ii) any amendment of or waiver relating to, or any consent to or departure from, any L/C Document, any Loan Document or any Document; (iii) any release or substitution at any time of any Property held as Collateral; (iv) your failure to deliver to Applicant any Document you have received with a drawing under a Credit because doing so would, or is likely to, violate any law, rule or regulation of any government authority; (v) the existence of any claim, set-off, defense or other right which Applicant may have at any time against you or any Beneficiary (or any person or entity for whom any Beneficiary may be acting) or any other person or entity, whether under or in connection with any L/C Document, any Loan Document, any Document or any Property referred to in or related to any of the foregoing or under or in connection with any unrelated transaction; (vi) any breach of contract or other dispute between or among any two or more of you, Applicant, any Beneficiary, any transferee of any Beneficiary, any person or entity for whom any Beneficiary or any transferee of any Beneficiary may be acting, or any other person or entity; or (vii) any delay, extension of time, renewal, compromise or other indulgence granted or agreed to by you with or without notice to Applicant, or Applicant's approval, in respect of any of Applicant's indebtedness or other obligations to you under or in connection with any L/C Document or any Loan Document.

SECTION 8. COMPLIANCE WITH LAWS AND REGULATIONS. Applicant represents and warrants to you that no Application, Credit or transaction under any Application and/or Credit will contravene any law or regulation of the government of the United States or any state thereof. Applicant agrees (a) to comply with all federal, state and foreign exchange regulations and other government laws and regulations now or hereafter applicable to any L/C Document, to any payments under or in connection with any L/C Document, to each transaction under or in connection with any L/C Document, export, shipping or financing of the Property referred to in or shipped under or in connection with any Credit, and (b) to reimburse you for such amounts as you may be required to expend as a result of such laws or regulations, or any change therein or in the interpretation thereof by any court or administrative or government authority charged with the administration of such laws or regulations.

SECTION 9. TAXES, RESERVES AND CAPITAL ADEQUACY REQUIREMENTS. In addition to, and notwithstanding any other provision of any L/C Document or any Loan Document, in the event that any law, treaty, rule, regulation, guideline, request, order, directive or determination (whether or not having the force of law) of or from any government authority, including, without limitation, any court, central bank or government regulatory authority, or any change therein or in the interpretation or application thereof, (a) does or shall subject you to any tax of any kind whatsoever with respect to the L/C Documents, or change the basis of taxation of payments to you of any amount payable thereunder (except for changes in the rate of tax on your net income); (b) does or shall impose, modify or hold applicable any reserve, special deposit, assessment, compulsory loan, Federal Deposit Insurance Corporation insurance or similar requirement against assets held by, deposits or other liabilities in or for the account of, advances or loans, by, other credit extended by or any other acquisition of funds by, any of your offices; (c) does or shall impose, modify or hold applicable any capital adequacy requirements (whether or not having the force of law); or (d) does or shall impose on you any other condition; and the result of any of the foregoing is (i) to increase the cost to you of issuing or maintaining any Credit or of performing any transaction under any L/C Document, (ii) to reduce any amount receivable by you under any L/C Document, or (iii) to reduce the rate of return on your capital or the capital of the Holding Company to a level below that which you or the Holding Company could have achieved but for any imposition, modification or application of any capital adequacy requirement (taking into consideration your policy and the policy of the Holding Company, as the case may be, with respect to capital adequacy), and any such increase or reduction is material (as determined by you or the Holding Company, as the case may be, in your or the Holding Company's sole discretion); then, in any such case, Applicant agrees to pay to you or the Holding Company, as the case may be, such amount or amounts as may be necessary to compensate you or the Holding Company for (A) any such additional cost, (B) any reduction in the amount received by you under any L/C Document, or (C) to the extent allocable (as determined by you or the Holding Company, as the case may be, in your or the Holding Company's sole discretion) to any L/C Document, any reduction in the rate of return on your capital or the capital of the Holding Company.

SECTION 10. COLLATERAL. In addition to, and not in substitution for, any Property delivered, conveyed, transferred or assigned to you under any Loan Document as security for any or all of Applicant's obligations and liabilities to you at any time existing under or in connection with any L/C Document or any Loan Document, Applicant grants to you a security interest in and to the following Collateral, whether or not any such Collateral is in your possession or control or the possession or control of your agents or correspondents or in transit to or set apart for, you or your agents or correspondents, until such time as all Applicant's obligations and liabilities to you at any time existing under or in connection with each L/C Document and each Loan Document have been fully paid and discharged, all as security for such obligations and liabilities, (a) all Applicant's property, claims, demands, right, title and interest in and to the balance of each of Applicant's deposit accounts with you now or at any time hereafter existing, and all evidences of such deposit accounts, (b) all Property belonging to Applicant or in which it may have an interest, now or at any time hereafter delivered, conveyed, transferred, assigned, pledged or paid to you or your agents or correspondents in any manner whatsoever, whether as security or for safekeeping or otherwise, including, without limitation, any items received for collection or transmission, and the proceeds of such items, whether or not such Property is in whole or in part released to Applicant on trust or bailee receipt or otherwise, and (c) where Applicant is more than one person or entity, all right, title and interest of each of Applicants in and to all the Property which any of Applicants may now or hereafter obtain as security for the obligations of any one or more of Applicants to one or more of the others of Applicants arising under or in connection with the transaction to which any Credit relates. Further, in addition to, and not in substitution for, any Property delivered, conveyed, transferred or assigned to you under any Loan Document as security for any or all of Applicant's obligations and liabilities to you at any time existing under or in connection with any L/C Document or any Loan Document, Applicant agrees to deliver, convey, transfer and assign to you on demand, as security, Property of a value and character satisfactory to you, (i) if you at any time feel insecure about Applicant's ability or willingness to repay any amounts which you have paid or may pay in the future on any Demand or in honoring any other of your obligations under or in connection with any Credit, or (ii) without limiting the generality of the foregoing, if any temporary or permanent restraining order, preliminary or permanent injunction, or any other pretrial or permanent injunctive or similar relief is obtained restraining, prohibiting or enjoining you, any of your correspondents, or any advising, confirming, negotiating, paying or other bank from paying or negotiating any Demand or honoring any other obligation under or in connection with any Credit. Applicant agrees that the receipt by you or any of your agents or correspondents at any time of any kind of security, including, without limitation, cash, shall not be deemed a waiver of any of your rights or powers under this Agreement. Applicant agrees to sign and deliver to you on demand, all such deeds of trust, security agreements financing statements and other documents as you shall at any time request which are necessary or desirable (in your sole opinion) to grant to you an effective and perfected security interest in and to any or all of the Collateral. Applicant agrees to pay all filing and recording fees related to the perfection of any security interest granted to you in accordance with this Section. Applicant hereby agrees that any or all of the Collateral may be held and disposed of as provided in this Agreement by you. Upon any transfer, sale, delivery, surrender or endorsement of any Document or Property which is or was part of the Collateral, Applicant will indemnify and hold you and your agents and correspondents harmless from and against each and every claim, demand, action or suit which may arise against you or any of your agents or correspondents by reason of such transfer, sale, delivery, surrender or endorsement.

SECTION 11. INDEMNIFICATION. Except to the extent caused by your lack of good faith, and notwithstanding any other provision of this Agreement, Applicant agrees to reimburse and indemnify you for (a) all amounts paid by you to any Beneficiary under or in connection with any guarantee or similar undertaking issued by such Beneficiary to a third party at Applicant's request, whether such request is communicated directly by Applicant or through you to such Beneficiary; and (b) all damages, losses, liabilities, actions, claims, suits, penalties, judgments, obligations, costs or expenses, of any kind whatsoever and howsoever caused, including, without limitation, attorneys' fees and interest, paid, suffered or incurred by, or imposed upon, you directly or indirectly arising out of or in connection with (i) any L/C Document, any Loan Document, any Document or any Property referred to in or related to any Credit; (ii) Applicant's failure to comply with any of its obligations under this Agreement; (iii) the issuance of any Credit; (iv) the transfer of any Credit; (v) any guarantee or similar undertaking, or any transactions thereunder, issued by any Beneficiary to a third party at Applicant's request, whether such request is communicated directly by Applicant or through you to such Beneficiary; (vi) any communication made by you, on Applicant's instructions, to any Beneficiary requesting that such Beneficiary issue a guarantee or similar undertaking to a third party or the issuance of any such guarantee or similar undertaking; (vii) the collection of any amounts Applicant owes to you under or in connection with any L/C Document or any Loan Document; (viii) the foreclosure against, or other enforcement of, any Collateral; (ix) the protection, exercise or enforcement of your rights and remedies under or in connection with any L/C Document or any Loan Document; (x) any court decrees or orders, including, without limitation, temporary or permanent restraining orders, preliminary or permanent injunctions, or any other pretrial or permanent injunctive or similar relief, restraining, prohibiting or enjoining or seeking to restrain, prohibit or enjoin you, any of your correspondents or any advising, confirming, negotiating, paying or other bank from paying or negotiating any Demand

or [ILLEGIBLE] any other obligation under or in connection with any credit; or (xi) any Credit being governed by laws or rules other than the UCP in effect on the date such Credit is issued. The indemnity provided in this Section will survive the termination of this Agreement and the expiration or cancellation of any or all the Credits.

SECTION 12. LIMITATION OF LIABILITY. Notwithstanding any other provision of this Agreement, neither you nor any of your agents or correspondents will have any liability to Applicant for any action, neglect or omission, if done in good faith, under or in connection with any L/C Document, Loan Document or Credit, including, without limitation, the issuance or any amendment of any Credit, the failure to issue or amend any Credit, or the honoring or dishonoring of any Demand under any Credit, and such good faith action, neglect or omission will bind Applicant. Notwithstanding any other provision of any L/C Document, in no event shall you or your officers or directors be liable or responsible, regardless of whether any claim is based on contract or tort, for (a) any special, consequential, indirect or incidental damages, including, without limitation, lost profits, arising out of or in connection with the issuance of any Credit or any action taken or not taken by you in connection with any L/C Document, any Loan Document, or any Document or Property referred to in or related to any Credit; (b) the honoring of any Demand in accordance with any order or directive of any court or government or regulatory body or entity requiring such honor despite any temporary restraining order, restraining order, preliminary injunction, permanent injunction or any type of pretrial or permanent injunctive relief or any similar relief, however named, restraining, prohibiting or enjoining such honor; (c) the dishonoring of any Demand in accordance with any legal or other restriction in force at the time and in the place of presentment or payment; (d) verifying the existence or reasonableness of any act or condition referenced, or any statement made, in connection with any drawing or presentment under any Credit; (e) the use which may be made of any Credit; (f) the validity of any purported transfer of any Credit or the identity of any purported transferee of any Beneficiary; (g) any acts or omissions of any Beneficiary or any other user of any Credit; (h) the form, validity, sufficiency, correctness, genuineness or legal effect of any Demand or any Document, or of any signatures or endorsements on any Demand or Document, even if any Demand or any Document should in fact prove to be in any or all respects invalid, insufficient, fraudulent or forged; (i) payment by you of any Demand when the Demand and any accompanying Documents appear on their face to comply substantially with the terms of the Credit to which they relate or dishonor by you of any Demand when the Demand and any accompanying Documents do not strictly comply on their face with the terms of the Credit to which they relate; (i) the failure of any Demand or Document to bear any reference or adequate reference to the Credit to which it relates; (k) the failure of any Document to accompany any Demand; (1) the failure of any person or entity to note the amount of any Demand on the Credit to which it relates or on any Document; (m) the failure of any person or entity to surrender or take up any Credit; (n) the failure of any Beneficiary to comply with the terms of any Credit or to meet the obligations of such Beneficiary to Applicant; (o) the failure of any person or entity to send or forward Documents if and as required by the terms of any Credit; (p) any errors, inaccuracies, omissions, interruptions or delays in transmission or delivery of any messages, directions or correspondence by mail, cable, telegraph, wireless or otherwise, whether or not they are in cipher; (q) any notice of nonrenewal of a Credit sent by you not being received on time or at any time by the Beneficiary of such Credit; (r) any inaccuracies in the translation of any messages, directions or correspondence; (s) any Beneficiary's use of the proceeds of any Demand; (t) any Beneficiary's failure to repay to you or Applicant the proceeds of any Demand if the terms of any Credit require such repayment; or (u) any act, error, neglect, default, negligence, gross negligence, omission, willful misconduct, lack of good faith, insolvency or failure in business of any of your agents or correspondents or of any advising, confirming, negotiating, paying or other bank. The occurrence of any one or more of the contingencies referred to in the preceding sentence shall not affect, impair or prevent the vesting of your rights or powers under this Agreement or any Loan Document or Applicant's obligation to make reimbursement or payment to you under this Agreement or any Loan Document. The provisions of this Section will survive the termination of this Agreement and any Loan Documents and the expiration or cancellation of any or all the Credits.

SECTION 13. EVENTS OF DEFAULT. Each of the following shall constitute an Event of Default under this Agreement: (a) Applicant's or any Guarantor's failure to pay any principal, interest, fee or other amount when due under or in connection with any L/C Document or any Loan Document; (b) Applicants failure to deliver to you Property of a value and character satisfactory to you at any time you have demanded security from Applicant pursuant to Section 10 of this Agreement; (c) the occurrence and continuance of any default or defined event of default under any Loan Document or any other agreement, document or instrument signed or made by Applicant or any Guarantor in your favor; (d) Applicant's or any Guarantor's failure to perform or observe any term, covenant or agreement contained in this Agreement or any Loan Document (other than those referred to in subsections (a), (b) and (c) of this Section, or the breach of any other obligation owed by Applicant or any Guarantor to you, and any such failure or breach shall be impossible to remedy or shall remain unremedied for thirty (30) calendar days after such failure or breach occurs; (e) any representation, warranty or certification made or furnished by Applicant or any Guarantor under or in connection with any L/C Document, any Loan Document or any Collateral, or as an inducement to you to enter into any L/C Document or Loan Document or to accept any Collateral, shall be materially false, incorrect or incomplete when made; (f) any material provision of this Agreement or any Loan Document shall at any time for any reason cease to be valid and binding on Applicant or any Guarantor or shall be declared to be null and void, or the validity or enforceability thereof shall be contested by Applicant, any Guarantor or any government agency or authority or Applicant or any Guarantor shall deny that it has any or further liability or obligation under this Agreement or any Loan Document; (g) Applicant's or any Guarantor's failure to pay or perform when due any indebtedness or other obligation Applicant or such Guarantor has to any person or entity other than you if such failure gives the payee of such indebtedness or the beneficiary of the performance of such obligation the right

to accelerate the time of payment of such indebtedness or the performance of such obligation; (h) any guarantee of, or any security covering, any of Applicant's indebtedness to you arising under or in connection with any L/C Document or any Loan Document fails to be in full force and effect at any time; (i) any material adverse change in Applicant's or any Guarantor's financial condition; (j) Applicant or any Guarantor suspends the transaction of its usual business or is expelled or suspended from any exchange; (k) Applicant or any Guarantor dies or is incapacitated; (1) Applicant or any Guarantor dissolves or liquidates; (m) Applicant or any Guarantor is not generally paying its debts as they become due; (n) Applicant or any Guarantor becomes insolvent, however such insolvency may be evidenced, or Applicant or any Guarantor makes any general assignment for the benefit of creditors; (o) a petition is filed by or against Applicant or any Guarantor seeking Applicant's or such Guarantor's liquidation or reorganization under the Bankruptcy Reform Act. Title 11 of the United States Code, as amended or recodified from time to time, or a similar action is brought by or against Applicant or any Guarantor under any federal, state or foreign law; (p) a proceeding is instituted by or against Applicant or any Guarantor for any relief under any bankruptcy, insolvency or other law relating to the relief of debtors, reorganization, readjustment or extension of indebtedness or composition with creditors; (q) a custodian or a receiver is appointed for, or a writ or order of attachment, execution or garnishment is issued, levied or made against, any of Applicant's or any Guarantor's Property or assets; (r) an application is made by any of Applicant's or any Guarantor's judgment creditors for an order directing you to pay over money or to deliver other of Applicant's or such Guarantor's Property; or (s) any government authority or any court takes possession of any substantial part of Applicant's or any Guarantor's Property or assets or assumes control over Applicant's or any Guarantor's affairs.

SECTION 14. REMEDIES. Upon the occurrence and continuance of any  $\operatorname{Event}$  of Default all amounts paid by you on any Demand which have not previously been repaid to you, together with all interest on such amounts, and the Unpaid and Undrawn Balance, if any, shall automatically be owing by Applicant to you and shall be due and payable by Applicant on demand without presentment or any other notice of any kind, including, without limitation, notice of nonperformance, notice of protest, protest, notice of dishonor, notice of intention to accelerate, or notice of acceleration, all of which are expressly waived by Applicant. Upon payment of the Unpaid and Undrawn Balance to you Applicant shall have no further legal or equitable interest therein, and you will not be required to segregate on your books or records the Unpaid and Undrawn Balance paid by Applicant. After you receive the Unpaid and Undrawn Balance, you agree to pay to Applicant, upon termination of all of your liability under all the Credits and Demands, a sum equal to the amount which has not been drawn under all the Credits less all amounts due and owing to you from Applicant under or in connection with the L/C Documents and the Loan Documents. Further, upon the occurrence and continuance of any Event of Default, you may sell immediately without demand for payment, advertisement or notice to Applicant, all of which are hereby expressly waived, any and all Collateral, received or to be received, at private sale or public auction or at brokers' board or upon any exchange or otherwise, at your option, in such parcel or parcels, at such times and places, for such prices and upon such terms and conditions as you may deem proper, and you may apply the net proceeds of each sale, together with any sums due from you to Applicant, to the payment of any and all obligations and liabilities due from Applicant to you under or in connection with the L/C Documents and the Loan Documents, all without prejudice to your rights against Applicant with respect to any and all such obligations and liabilities which may be or remain unpaid. If any such sale be at brokers' board or at public auction or upon any exchange, you may yourself be a purchaser at such sale free from any right of redemption, which Applicant hereby expressly waive and release. All your rights and remedies existing under the L/C Documents and the Loan Documents are in addition to, and not exclusive of, any rights or remedies otherwise available to you under applicable law. In addition to any rights now or hereafter granted under applicable law, and not by way of limitation of any such rights, upon the occurrence and continuance of any Event of Default, Applicant hereby authorizes you at any time or from time to time, without notice to Applicant or to any other person (any such notice being hereby expressly waived by Applicant) and to the extent permitted by law, to appropriate and to apply any and all Applicant's deposits (general or special, including, without limitation, indebtedness evidenced by certificates of deposit) with you or elsewhere, whether matured or unmatured, and any other indebtedness at any time held or owing by you to or for Applicant's credit or its account, against and on account of Applicant's obligations and liabilities to you under or in connection with any of the  $\mbox{L/C}$ Documents or the Loan Documents, irrespective of whether or not you shall have made any demand for payment of any or all such obligations and liabilities or declared any or all such obligations and liabilities to be due and payable, and although any or all such obligations and liabilities shall be contingent or unmatured.

SECTION 15. WAIVERS. No delay, extension of time, renewal, compromise or other indulgence which may occur or be granted by you under any L/C Document or any Loan Document shall impair your rights or powers under this Agreement or any Application. You shall not be deemed to have waived any of your rights under this Agreement or any Application unless such waiver is in writing signed by your authorized representative. No such waiver, unless expressly provided therein, shall be effective as to any transactions which occur subsequent to the date of such waiver or as to the continuance of any Event of Default after such waiver. No amendment or modification of this Agreement shall be effective unless it is in writing signed by Applicant's and your authorized representative(s).

SECTION 16. AMENDMENTS AND MODIFICATIONS TO CREDITS. At Applicant's verbal or written request, or with Applicant's verbal or written consent, and without extinguishing or otherwise affecting Applicant's obligations under this Agreement or any Loan Document, you may with respect to any Credit, in writing or by any other action, but you will not be obligated to, (a) increase the amount of such Credit, (b) extend the time for, and amend or modify the terms and conditions governing, the making and honoring of any Demand or Document or any other terms and conditions of such Credit, or (c) waive the failure of any Demand or Document to comply with the terms of such Credit, and any Collateral pledged or granted to you in connection with such Credit will secure Applicant's obligations to you with respect to such Credit as amended, modified or waived. No amendment to, or modification of, the terms of any Credit will become effective if the Beneficiary of such Credit or any confirming bank objects to such amendment or modification. If any Credit is amended or modified in accordance with this Section, Applicant shall be bound by, and obligated under, the provisions of this Agreement with respect to such Credit as so amended or modified, and any action taken by you or any advising, confirming, negotiating, paying or other bank in accordance with such amendment or modification.

SECTION 17. SUCCESSORS AND ASSIGNS. The terms and conditions of this Agreement and each Application shall bind Applicant's heirs, executors, administrators, successors and assigns, and all rights, benefits and privileges conferred on you under or in connection with each L/C Document and each Loan Document shall be and hereby are extended to, conferred upon and may be enforced by your successors and assigns. Applicant will not assign this Agreement or Applicant's obligations or liabilities to you under or in connection with any L/C Document or Loan Document to any person or entity without your prior written approval.

SECTION 18. GOVERNING LAW. This Agreement and each Application, and Applicant's and your performance under this Agreement and each Application, shall be governed by and be construed in accordance with the laws of the State of California. Unless you otherwise specifically agree in writing, each Credit, the opening of each Credit, the performance by you under each Credit, and the performance by the Beneficiary and any advising, confirming, negotiating, paying or other bank under each Credit, shall be governed by and be construed in accordance with the UCP in force on the date of the issuance of each Credit.

SECTION 19. JURISDICTION AND SERVICE OF PROCESS. Any suit, action or proceeding against Applicant under or with respect to any L/C Document may, at your sole option, be brought in (a) the courts of the State of California, (b) the United States District Courts in California, (c) the courts of Applicant's jurisdiction of incorporation or principal office, or (d) the courts of the jurisdiction where any Beneficiary, any advising, confirming, negotiating, paying or other bank or any other person or entity has brought any suit, action or proceeding against you with respect to any Credit or any Demand, and Applicant hereby submits to the nonexclusive jurisdiction of such courts for the purpose of any such suit, action, proceeding or judgment and waives any other preferential jurisdiction by reason of domicile. Applicant will accept joinder in any suit, action or proceeding brought in any court or jurisdiction against you by any Beneficiary, any advising, confirming, negotiating, paying or other bank or any other person or entity with respect to any Credit or any Demand. Applicant irrevocably waives trial by jury and any objection, including, without limitation, any objection of the laying of venue or any objection based on the grounds of forum non conveniens, which Applicant may now or hereafter have to the bringing of any such action or proceeding. Applicant further waives any right to transfer or change the venue of any suit, action or proceeding brought against Applicant by you under or in connection with any L/C Document. Applicant irrevocably consents to the service of process in any action or proceeding in any court by the mailing of copies thereof by registered or certified mail, postage prepaid, to Applicant at its address specified next to its signature on this Agreement or at such other address as Applicant shall have notified to you in writing, such service to be effective ten (10) days after such mailing.

SECTION 20. JOINT APPLICANTS. If this Agreement is signed by more than one person and/or entity as an Applicant, this Agreement and the Applications shall be the joint and several agreement of all such persons and/or entities and that all references to "Applicant" or "Applicant's" in this Agreement and the Applications shall refer to all such persons and/or entities jointly and severally.

SECTION 21. SEVERABILITY. Any provision of any L/C Document which is prohibited or unenforceable in any jurisdiction shall be, only as to such jurisdiction, ineffective to the extent of such prohibition or unenforceability, but all the remaining provisions of such L/C Document and all the other L/C Documents shall remain valid.

SECTION 22. HEADINGS. The headings used in this Agreement are for convenience of reference only and shall not define or limit the provisions of this Agreement.

SECTION 23. COMPLETE AGREEMENT. The entire agreement with respect to each Credit will consist of this Agreement and the Application for such Credit, together with any written document or instrument signed by Applicant and you, or signed by Applicant and approved by you, which specifically references such Credit, the Application for such Credit, or this Agreement. Except as specifically provided in this Agreement, in any Application or in any written document or instrument referred to in the preceding sentence, no statements or representations not contained in this Agreement, such Application or such written document or instrument shall have any force or effect on this Agreement, such Application or such written document or instrument.

### ADDITIONAL PROVISIONS APPLICABLE IF THE APPLICANT IS LOCATED IN OREGON

Section Oregon 1. UNDER OREGON LAW, MOST AGREEMENTS, PROMISES AND COMMITMENTS MADE BY A LENDER AFTER OCTOBER 3,1989 CONCERNING LOANS AND OTHER CREDIT EXTENSIONS WHICH ARE NOT FOR PERSONAL, FAMILY, OR HOUSEHOLD PURPOSES OR SECURED SOLELY BY THE BORROWER'S RESIDENCE MUST BE IN WRITING, EXPRESS CONSIDERATION, AND BE SIGNED BY THE LENDER TO BE ENFORCEABLE.

ADDITIONAL PROVISIONS APPLICABLE IF THE APPLICANT IS LOCATED IN WASHINGTON

Section Washington 1. ORAL AGREEMENTS OR ORAL COMMITMENTS TO LOAN MONEY, EXTEND CREDIT OR FOREBEAR FROM ENFORCING REPAYMENT OF A DEBT ARE NOT ENFORCEABLE UNDER WASHINGTON LAW. ADDITIONAL PROVISIONS APPLICABLE IF THE APPLICANT IS LOCATED IN NEBRASKA

Section Nebraska 1. ENFORCEABILITY OF WRITTEN TERMS ONLY. A CREDIT AGREEMENT MUST BE IN WRITING TO BE ENFORCEABLE UNDER NEBRASKA LAW. TO PROTECT THE PARTIES FROM ANY MISUNDERSTANDINGS OR DISAPPOINTMENTS, ANY CONTRACT, PROMISE, UNDERTAKING OR OFFER TO FOREBEAR REPAYMENT OF MONEY OR TO MAKE ANY OTHER FINANCIAL ACCOMMODATION IN CONNECTION WITH THIS LOAN OF MONEY OR GRANT OR EXTENSION OF CREDIT, OR ANY AMENDMENT OF, CANCELLATION OF, WAIVER OF, OR SUBSTITUTION FOR ANY OR ALL OF THE TERMS OR PROVISIONS OF ANY INSTRUMENT OR DOCUMENT EXECUTED IN CONNECTION WITH THIS LOAN OF MONEY OR GRANT OR EXTENSION OF CREDIT, MUST BE IN WRITING TO BE EFFECTIVE.

ADDITIONAL PROVISIONS APPLICABLE IF THE APPLICANT IS LOCATED IN IOWA

Section Iowa 1. IMPORTANT: READ BEFORE SIGNING. THE TERMS OF THIS AGREEMENT SHOULD BE READ CAREFULLY BECAUSE ONLY THOSE TERMS IN WRITING ARE ENFORCEABLE. NO OTHER TERMS OR ORAL PROMISES NOT CONTAINED IN THIS WRITTEN CONTRACT MAY BE LEGALLY ENFORCED. YOU MAY CHANGE THE TERMS OF THIS AGREEMENT ONLY BY ANOTHER WRITTEN AGREEMENT.

Section Iowa 2. By signing this Agreement, Applicant acknowledges receipt of a copy of this Agreement.

This Agreement is signed by Applicant's duly authorized representative or representatives on the date specified below.

[APPLICANT'S NAME]	[APPLICANT'S NAME]
By:	By:
Title:	Title:
[ADDRESS]	[ADDRESS]
DATE:	DATE:

4 of 4

EXHIBIT C

APPLICATION FOR STANDBY LETTER OF CREDIT

TO: WELLS FARGO BANK, NATIONAL ASSOCIATION	
DATE FOR WELLS FARGO'S LETTER OF CREDIT NO. DOC USE ONLY	UMENT TRACK NO
APPLICANT SIGNING BELOW HEREBY REQUESTS THAT WELLS FARGO BANK, NA ASSOCIATION ("WELLS FARGO") ISSUE IN WELL FARGO'S NAME AN IRREVOC LETTER OF CREDIT (THE "CREDIT") ON SUBSTANTIALLY THE TERMS BELOW OTHERWISE SPECIFIED BELOW IN SPECIAL INSTRUCTIONS, FORWARD THE CR FOLLOWING MEANS TO THE BENEFICIARY DIRECTLY OR THROUGH A BANK SEL FARGO:	CABLE STANDBY AND, UNLESS REDIT BY THE
_  FULL CABLE/TELEX  _  COURIER  _  MAIL WITH BRIEF ADVICE BY  _  MAIL  _  OTHER	CABLE/TELEX
ADVISING BANK: (If left blank, Wells Fargo may select)	BENEFICIARY: (Name and Address)
PARTY TO BE NAMED AS REQUESTING THE CREDIT: (Name and Address)	AMOUNT: (In words)
	(In figures) (Currency)
AVAILABILITY: Unless otherwise specified herein, the Credit is to be available with Wells Fargo's issuing office by payment of draft(s) drawn at sight on Wells Fargo or, at Wells Fargo's option, with any bank(s) or with a bank nominated by Wells Fargo by negotiation of draft(s) drawn at sight on Wells Fargo.	EXPIRATION DATE: PLACE OF EXPIRATION: Unless otherwise specified herein, the Credit is to expire at Wells Fargo's issuing office or, if the Credit is available with any bank(s) or with a specific bank other than Wells Fargo's issuing office, at such place as Wells Fargo shall elect.
DRAWING(S):  _  Partial drawings are permitted. (More than one d	
$ \_ $ the draft must be for the full amount of the	of the Credit.
SPECIAL INSTRUCTIONS: (Attach additional signed sheet(s), if nece	essary, and label as attachments to this Application.)
	able.)
	ges for:     Applicant's account     Beneficiary's account
INQUIRIES: Direct to:	Telephone Number:
APPLICANT'S AGREEMENT AND SIGNATURE: Applicant's signature here is agreement to all the terms and conditions on this Application and agreement that the Credit and its issuance will be governed by (1 and conditions of the Standby Letter of Credit Agreement between Wells Fargo and/or (2) any other agreement signed by Applicant pu the Credit is to be issued. This Application is signed by Applica authorized representative(s) on the date specified above.	Indicates Applicant's ) the terms Applicant and ursuant to which
APPLICANT ADDRESS	
AUTHORIZED SIGNATURE TITLE ADDRESS	
AUTHORIZED SIGNATURE TITLE ADDRESS	

# CREDIT ISSUANCE HAS BEEN APPROVED IN ACCORDANCE WITH WELLS FARGO'S CREDIT POLICIES AND PROCEDURES

APPROVING OFF	FICER'S SIGNATURE	APPROVING OFFIC	CER'S NAME (Print)	APPROVING O	FFICER'S OFFICE	(Print) AU	J MAC COM	MITMENT NO.
								=====
PHONE	AFS INTERFACE REQ	UIRED:	STANDALONE TRANSA	ACTION:	COLLATERAL COD	E I	PURPOSE CODE	DATE
	YES     NO		YES     NO					
	=							=====
SPECIAL INSTRUCTIONS: (Indicate any provisions applicable to the Credit different from those on Applicant's Relationship Management Instructions Form)								

### NOTE

June 6, 2002

pursuant to the Lender's Pro Rata Share of the Revolving Facility under the Credit Agreement referred to below, payable as hereinafter set forth. Borrower promises to pay interest on the principal amount hereof remaining unpaid from time to time from the date hereof until the date of payment in full, payable as hereinafter set forth.

Reference is made to the Credit Agreement dated as of June 6, 2002, by and among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as the same may be amended, renewed, extended or otherwise modified from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein with the meanings given those terms in the Credit Agreement. This is one of the Notes referred to in the Credit Agreement, and any holder hereof is entitled to all of the rights, remedies, benefits and privileges provided for in the Credit Agreement as originally executed or as it may from time to time be supplemented, modified or amended. The Credit Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events upon the terms and conditions therein specified.

The principal indebtedness evidenced by this Note shall be payable as provided in the Credit Agreement and in any event on the Maturity Date.

Interest shall be payable on the outstanding daily unpaid principal amount of each Advance hereunder from the date thereof until payment in full and shall accrue and be payable at the rates and on the dates set forth in the Credit Agreement, both before and after default and before and after maturity and judgment, with interest on overdue principal and interest to bear interest at the rate set forth in Section 3.7 of the Credit Agreement, to the fullest extent permitted by applicable Law.

The amount of each payment hereunder shall be made to the Administrative Agent at the Administrative Agent's Office for the account of the Lender in immediately available funds not later than 11:00 a.m. (California time) on the day of payment (which must be a Banking Day). All payments received after 11:00 a.m. (California time) on any particular Banking Day shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

-1-

\$

The Lender shall use its best efforts to keep a record of Advances made by it and payments of principal received by it with respect to this Note, and such record shall be presumptive evidence of the amounts owing under this Note.

Borrower hereby promises to pay all costs and expenses of any holder hereof incurred in collecting Borrower's obligations hereunder or in enforcing or attempting to enforce any of such holder's rights hereunder, including reasonable attorneys' fees and disbursements (including allocated costs of legal counsel employed by the Administrative Agent or the holder), whether or not an action is filed in connection therewith.

Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest, notice of protest and any other notice or formality, to the fullest extent permitted by applicable Laws.

THIS NOTE SHALL BE DELIVERED TO AND ACCEPTED BY THE LENDER, OR BY THE ADMINISTRATIVE AGENT ON ITS BEHALF, IN THE STATE OF CALIFORNIA, AND SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS THEREOF.

AMERICAN STATES WATER COMPANY, a California corporation

-2-

## ADVANCES AND PAYMENTS OF PRINCIPAL (Alternate Base Rate Advances)

Da	ate	Amount of Advance	Amount Principal	of Paid	Unpaid Principal Balance	Notation Made By
-						
-						
-						
-						
-						
_						
-						
-						
_						
-						
-						
_						

-3-

## ADVANCES AND PAYMENTS OF PRINCIPAL (Eurodollar Rate Advances)

-		Amount of	Amount		Unpaid Principal	Notation
Dá	ate	Advance	Principal	Paid	Balance	Made By
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
-						
_						

-4-

June 6, 2002

Wells Fargo Bank, National Association, as Administrative Agent 333 South Grand Avenue, Third Floor Los Angeles, California 90071 and

The Lenders set forth on Exhibit A attached hereto

Re: Credit Agreement by and among American States Water Company, the Lenders named therein, and Wells Fargo Bank, National Association

### Ladies and Gentlemen:

We have acted as special counsel to American States Water Company, a California corporation (the "Borrower"), in connection with the Credit Agreement dated as of June 6, 2002 (the "Credit Agreement") by and among the Borrower, the Lenders named therein, and Wells Fargo Bank, National Association, as Administrative Agent (in such capacity, the "Administrative Agent"). Capitalized terms not otherwise defined herein shall have the meanings ascribed to them in the Credit Agreement. For the purpose of this opinion, "Loan Documents" means those Loan Documents (as defined in the Credit Agreement) in existence as of the Closing Date and as further described on Schedule I attached hereto and incorporated herein.

In our capacity as special counsel, we have examined originals or copies of the Credit Agreement and the other Loan Documents, as well as originals or copies of such records and documents we considered appropriate. As to relevant factual matters, we have relied upon, among other things, the representations made by the Borrower in the Credit Agreement and other Loan Documents and the Borrower's factual representations in a certificate of a Responsible Official delivered by the Borrower in connection with this opinion letter (the "Certificate"), a copy of which has been delivered to you. In addition, we have obtained and relied upon those certificates of public officials we considered appropriate.

June 6, 2002 - Page 2

We have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals and the conformity with originals of all documents submitted to us as copies. To the extent the Borrower's obligations depend on the due authorization, execution and delivery of the Credit Agreement and other Loan Documents by the other parties thereto, we have assumed that each of the Credit Agreement and other Loan Documents has each been so authorized, executed and delivered by the other parties thereto and that each constitutes the legally valid and binding obligation of the other parties enforceable in accordance with its terms. We have also assumed that each Lender is (a) a bank incorporated or organized under, or a foreign bank licensed to conduct a banking business through an agency located in the United States of America pursuant to the laws of the United States of America or the State of California within the meaning of Section 1 of Article XV of the California Constitution and Section 1716 of the California Financial Code, or (b) a finance lender licensed under the California Finance Lenders Law (California Financial Code Section 22000 et seq.) and, as a result thereof, is exempt from the usury restrictions of Section 1 of Article XV of the Constitution of the State of California

On the basis of such examination, our reliance upon the assumptions contained herein and our consideration of those questions of law we considered relevant, and subject to the limitations and qualifications in this opinion, we are of the opinion that:

1. The Borrower has been duly incorporated, and is validly existing as a corporation in good standing under the laws of the State of California, with corporate power to own its properties and assets, to enter into the Loan Documents and perform its obligations under the Loan Documents.

2. The execution, delivery and performance of each of the Loan Documents have been duly authorized by all necessary corporate action on the part of the Borrower, and each of the Loan Documents has been duly executed and delivered by the Borrower. Each of the Loan Documents constitutes the legally valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or affecting creditors' rights generally (including, without limitation, fraudulent conveyance laws), and by general principles of equity, including, without limitation, concepts of materiality, reasonableness, good faith and fair dealing, and the possible unavailability of specific performance or injunctive relief, regardless of whether considered in a proceeding in equity or at law.

3. The execution and delivery by the Borrower of the Loan Documents do not, and the Borrower's payment of its obligations under the Loan Documents will not, violate any current California or federal statute, rule or regulation that we have, in the exercise of customary professional diligence, recognized as applicable to the Borrower or to transactions of the type contemplated by Loan Documents.

 $\rm 4.$  The execution and delivery by the Borrower of the Loan Documents do not, and the Borrower's payment of its obligations under the Loan Documents will not, (i)

June 6, 2002 - Page 3

violate the Borrower's Articles of Incorporation or Bylaws, (ii) violate, breach, or result in a default, or result in the creation or imposition of any lien, charge or encumbrance upon any of the assets of the Borrower, under any existing obligation of or restriction on the Borrower under any other agreement (the "Other Agreements") identified in the Borrower's most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission (the "Annual Report"), or (iii) breach or otherwise violate any existing obligation of or restriction on the Borrower under any order, judgment or decree of any California or federal court or governmental authority binding on the Borrower identified in the Certificate. We express no opinion as to the effect of the Borrower's performance of its obligations in the Credit Agreement or other Loan Documents on the Borrower's compliance with financial covenants in the Other Agreements.

5. No order, consent, permit or approval of any California or federal governmental authority that we have, in the exercise of customary professional diligence, recognized as applicable to the Borrower or to transactions of the type contemplated by the Loan Documents is required on the part of the Borrower for the execution and delivery of, and payment of its obligations under, the Loan Documents.

 $\,$  6. The Borrower is not an investment company required to register under the Investment Company Act of 1940, as amended.

To our knowledge, there are no actions, suits, investigations or proceedings pending or threatened against the Borrower in any court or by or before any arbitrator or governmental authority or agency that (i) seek to affect the enforceability of the Credit Agreement or any of the other Loan Documents or (ii), except for the matters described in, or incorporated by reference into, the Annual Report, are required by the Securities Exchange Act of 1934 to be disclosed therein.

The law covered by this opinion is limited to the present federal law of the United States and the present law of the State of California. We express no opinion as to the law of any other jurisdiction and no opinion regarding the statutes, administrative decisions, rules, regulations or requirements of any county, municipality, subdivision or local authority of any jurisdiction.

Our opinion in paragraph 2 above as to the enforceability of the Loan Documents is subject to:

- (a.) public policy considerations, statutes or court decisions that may limit the rights of a party to obtain indemnification against its own gross negligence, willful misconduct or unlawful conduct;
- (b.) the unenforceability under certain circumstances of broadly or vaguely stated waivers or waivers of rights granted by law where the waivers are against public policy or prohibited by law; and

June 6, 2002 - Page 4

(c.) the unenforceability under certain circumstances of provisions waiving the right to a jury trial.

We express no opinion as to:

- i. your ability to collect attorneys' fees and costs in an action involving the Loan Documents if you are not the prevailing party in that action (we call your attention to the effect of Section 1717 of the California Civil Code, which provides that where a contract permits one party thereto to recover attorneys' fees, the prevailing party in any action to enforce any provision of the contract shall be entitled to recover its reasonable attorneys' fees);
- ii. any provision of the Loan Documents requiring written amendments or waivers of the Loan Documents insofar as it suggests that oral or other modifications, amendments or waivers could not be effectively agreed upon by the parties or that the doctrine of promissory estoppel might not apply;
- iii. the unenforceability under certain circumstances of provisions imposing penalties, liquidated damages, acceleration of future amounts due (other than principal) without appropriate discount to present value, late charges, prepayment charges and increased interest rates upon default;
- iv. the effect of non-compliance by you with any state or federal laws or regulations applicable to the transactions contemplated by the Loan Documents because of the nature of your business; and
- any provision of the Loan Documents insofar as it purports to grant a right of setoff in respect of the Borrower's assets
   (a) to any person other than a creditor of the Borrower or (b) to the Administrative Agent or any Lender in an amount greater than the amount owing by the Borrower to the Administrative Agent or such Lender.

We advise you that Section 11.17 of the Credit Agreement, which provides for exclusive jurisdiction of the courts of the State of California and federal courts sitting in that state, may not be binding on the courts in the forums excluded, and, as to federal courts, in the forums selected.

Our use of the terms "to our knowledge" or any similar phrase to qualify a statement in this opinion means (a) the knowledge obtained by us from the Credit Agreement and the Certificates and (b) that those attorneys in this firm who have given substantive attention to the representation described in the introductory paragraph of this opinion do not have current actual knowledge that the statement is inaccurate. Such terms do not include any knowledge of other attorneys within our firm or any constructive or imputed notice of any matters or items of information. Except as otherwise expressly indicated, we have not undertaken any independent

June 6, 2002 - Page 5

investigation to determine the accuracy of the statement, and any limited inquiry undertaken by us during the preparation of this opinion letter should not be regarded as such an investigation. No inference as to our knowledge of any matters bearing on the accuracy of any such statement should be drawn from the fact of our representation of the Borrower in connection with this opinion letter or in other matters.

This letter is furnished by us as special counsel for the Borrower and may be relied by you only in connection with the transactions described herein. It may not be used or relied upon by any other person, nor may copies be delivered to any other person, without in each instance our prior written consent. You may, however, deliver a copy of this opinion to your accountants, attorneys, and other professional advisors, to governmental regulatory agencies having jurisdiction over you and to permitted transferees of the Loan Documents in connection with such transfer, and such transferees may rely on this opinion as if it were addressed and had been delivered to them on the date of this opinion, unless statements in this opinion would be affected by the status of the transferee. This opinion is expressly limited to the matters set forth above and we render no opinion, whether by implication or otherwise, as to any other matters. We assume no obligation to update or supplement this opinion to reflect any facts or circumstances that arise after the date of this opinion and come to our attention, or any future changes in laws.

Respectfully submitted,

/s/ O'MELVENY & MYERS LLP

Schedule 1

# LOAN DOCUMENTS(1)

1. Credit Agreement (with Schedules and Exhibits thereto)

- 2. Notes issued in favor of each of the financial institutions listed on Exhibit A.
- 3. Swing Line Note issued in favor of Wells Fargo Bank, National Association.

- -----

 Unless otherwise indicated, the Loan Documents are dated as of June 6, 2002, and are designated above as defined in the Credit Agreement.

Exhibit A

Wells Fargo Bank, National Association

CoBank, ACB

Union Bank of California, N.A.

Comerica West Incorporated

Northern Trust Bank of California, N.A.

#### AMERICAN STATES WATER COMPANY BACKUP CERTIFICATE

I, McClellan Harris III, hereby certify that I am Vice President--Finance, Treasurer, and Corporate Secretary of American States Water Company, a California corporation (the "Company"), and as follows:

(1) The representations and warranties of the Company in the Credit Agreement (the "Agreement"), dated as of June 6, 2002, by and among the Company, the Lenders thereto, and Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent") are true and correct as of the date hereof in all material respects.

(2) In connection with its opinion (the "Opinion"), dated as of the date hereof, delivered pursuant to Section 8.1(a)(6) of the Agreement, O'Melveny & Myers LLP is hereby authorized to rely on the representations made by the Company in the certificates delivered to the Administrative Agent on the date hereof in connection with the execution of the Agreement to the same extent as if such certificates were delivered to O'Melveny & Myers LLP.

(3) There are no orders, judgments or decrees of any California or federal court or governmental authority binding on the Company.

(4) There are no actions, suits, investigations or proceedings pending or threatened against the Company in any court or by or before any arbitrator or governmental authority or agency that (a) seek to affect the enforceability of the Agreement or any of the other Loan Documents (as defined in the Opinion), or (b) except for the matters described in, or incorporated by reference into, the Company's most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission, are required by the Securities Exchange Act of 1934 to be disclosed therein.

(5) The Company (a) is not and does not hold itself out as being engaged primarily, nor does it propose to engage primarily, in the business of investing, reinvesting or trading in securities; (b) is not engaged in, and does not propose to engage in, the business of issuing face-amount certificates of the installment type, nor has it been engaged in such business and has any such certificate outstanding; and (c) does not own or propose to acquire investment securities having a value exceeding 40% of the value of the Company's total assets (exclusive of government securities and cash items) on an unconsolidated basis.

IN WITNESS WHEREOF, I have executed this Certificate on behalf of American States Water Company as of June 6, 2002.

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

## EXHIBIT F

# REQUEST FOR BORROWING

1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:

- (a) AMOUNT OF REQUESTED BORROWING: \$2,000,000
- (b) DATE OF REQUESTED BORROWING: June 6, 2002
- (c) TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one box only):
  - | | ALTERNATE BASE RATE ADVANCES
  - |X| EURODOLLAR RATE ADVANCES FOR A EURODOLLAR PERIOD OF one (1) MONTHS(1)
- 3. In connection with the request, Borrower certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of

- --

\_\_\_\_

(1) Specify whether 1, 2, 3, 6, 9 or 12-month Eurodollar Period.

-1-

Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

4. This Request for Borrowing is executed on June 6, 2002, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ McClellan Harris III McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

#### REQUEST FOR BORROWING

1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:

- (a) AMOUNT OF REQUESTED BORROWING: \$6,000,000
- (b) DATE OF REQUESTED BORROWING: June 6, 2002
- TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one (C)box only):
  - ALTERNATE BASE RATE ADVANCES |X|
  - EURODOLLAR RATE ADVANCES FOR A EURODOLLAR PERIOD OF \_\_\_\_\_ MONTHS (1)

3. In connection with the request, Borrower certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of

(1) Specify whether 1, 2, 3, 6, 9 or 12-month Eurodollar Period.

Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

4. This Request for Borrowing is executed on June 6, 2002, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ McClellan Harris III McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

#### REQUEST FOR BORROWING

1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:

- (a) AMOUNT OF REQUESTED BORROWING: \$6,000,000
- (b) DATE OF REQUESTED BORROWING: June 6, 2002
- (c) TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one box only):
  - | | ALTERNATE BASE RATE ADVANCES
  - |X| \*EURODOLLAR RATE ADVANCES FOR A EURODOLLAR PERIOD OF one (1) MONTHS(1) \*Roll from Base Rate Advance
- 3. In connection with the request, Borrower certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4; 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of

- ----

(1) Specify whether 1, 2, 3, 6, 9 or 12-month Eurodollar Period.

-1-

Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

4. This Request for Borrowing is executed on June 6, 2002, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ McClellan Harris III McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

# EXHIBIT G

# REQUEST FOR CONTINUATION/CONVERSION

1. This REQUEST FOR CONTINUATION/CONVERSION is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Pursuant to Section 2.4 of the Credit Agreement, Borrower hereby irrevocably requests the [continuation] [conversion] of the Advances specified herein, as follows:

(a) The date of [continuation] [conversion] is \_\_\_\_\_, which is a [Banking Day] [Eurodollar Banking Day which is the last day of the applicable Eurodollar Period].

(b) The amount of the Advances to be [converted] [continued] is

(c) Type of Conversion/Continuation being requested (check all boxes that apply):

\$

- [\_] Alternate Base Rate Advance of \$\_\_\_\_\_ converted to a Eurodollar Rate Advance for a Eurodollar Period of [one] [two] [three] [six] [nine] [twelve] month[s].
- [\_] Eurodollar Rate Advance of \$\_\_\_\_\_ converted to an Alternate Base Rate Advance.
- [\_] Eurodollar Rate Advance of \$\_\_\_\_\_ with a Eurodollar Period of [one] [two] [three] [six] [nine] [twelve] month[s] [converted to ] [continued as] a Eurodollar Rate Advance with a Eurodollar Period of [one] [two] [three] [six] [nine] [twelve] month[s].

3. In connection with the request, Borrower certifies that no event has occurred and is continuing that is a Default or Event of Default.

-1-

4. This Request for Continuation/Conversion is executed on \_\_\_\_\_, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By:

```
Name:
Title:
```

-2-

#### REQUEST FOR LETTER OF CREDIT

1. This Request for Letter of Credit is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation (the "Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as the Administrative Agent and the Issuing Lender pursuant to the Credit Agreement (as amended, modified or extended, the "Credit Agreement") dated as June 6, 2002, among the Borrower, the Lenders therein named, and Administrative Agent. Any terms used herein and not defined herein shall have the meanings defined in the Credit Agreement.

2. The Borrower hereby requests that the Issuing Lender issue a Letter of Credit as follows:

- (a) Amount of Letter of Credit: \$ \_\_\_\_\_.
- (b) Date of Issuance: \_\_\_\_\_,\_\_\_.
- (c) Beneficiary under Letter of Credit:

Name:

Address:

- (d) Expiry Date:
- \_\_\_\_\_′ \_\_\_\_·
- (e) Purpose of Letter of Credit:
- (f) Additional Information/Terms:

3. The requested Letter of Credit is (check one box only):

 $\left[ \ \right]$  a new Letter of Credit in addition to Letters of Credit already outstanding.

-1-

[\_] a supplement, modification, amendment, renewal, or extension to or of the following outstanding Letter(s) of Credit: [identify]

4. In connection with the issuance of the Letter of Credit requested herein, the Borrower represents, warrants and certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or other Obligor or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

(d) Giving effect to the issuance of the requested Letter of Credit, (i) the Revolving Credit Facility Usage does not exceed the Maximum Revolving Credit Amount, and (ii) the Aggregate Effective Amount of all outstanding Letters of Credit does not exceed \$15,000,000.

5. Attached hereto is a Letter of Credit Agreement on the form provided to the Borrower by the Issuing Lender.

6. This Request for Letter of Credit is executed on June 6, 2002, by a Responsible Official of the Borrower. The undersigned, in such capacity, hereby certifies each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By:

McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

# SCHEDULE 1.1 TO CREDIT AGREEMENT

# LENDER COMMITMENTS/PRO RATA SHARES

Lender 	Pro Rata Share	Amount of Revolving Commitment
Wells Fargo Bank, National Association	33.3333334%	\$25,000,000.00
CoBank, ACB	29.3333333%	\$22,000,000.00
Union Bank of California, N.A.	20.000000%	\$15,000,000.00
Comerica West Incorporated	13.3333333%	\$10,000,000.00
Northern Trust Bank of California, N.A.	4.000000%	\$ 3,000,000.00
Total	100%	\$75,000,000.00

-1-

#### NOTE

#### June 6, 2002

FOR VALUE RECEIVED, the undersigned, AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), promises to pay to the order of WELLS FARGO BANK, NATIONAL ASSOCIATION (the "Lender"), the principal amount of TWENTY-FIVE MILLION DOLLARS (\$25,000,000.00) or such lesser aggregate amount of Advances as may be made pursuant to the Lender's Pro Rata Share of the Revolving Facility under the Credit Agreement referred to below, payable as hereinafter set forth. Borrower promises to pay interest on the principal amount hereof remaining unpaid from time to time from the date hereof until the date of payment in full, payable as hereinafter set forth.

Reference is made to the Credit Agreement dated as of June 6, 2002, by and among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as the same may be amended, renewed, extended or otherwise modified from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein with the meanings given those terms in the Credit Agreement. This is one of the Notes referred to in the Credit Agreement, and any holder hereof is entitled to all of the rights, remedies, benefits and privileges provided for in the Credit Agreement as originally executed or as it may from time to time be supplemented, modified or amended. The Credit Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events upon the terms and conditions therein specified.

The principal indebtedness evidenced by this Note shall be payable as provided in the Credit Agreement and in any event on the Maturity Date.

Interest shall be payable on the outstanding daily unpaid principal amount of each Advance hereunder from the date thereof until payment in full and shall accrue and be payable at the rates and on the dates set forth in the Credit Agreement, both before and after default and before and after maturity and judgment, with interest on overdue principal and interest to bear interest at the rate set forth in Section 3.7 of the Credit Agreement, to the fullest extent permitted by applicable Law.

The amount of each payment hereunder shall be made to the Administrative Agent at the Administrative Agent's Office for the account of the Lender in immediately available funds not later than 11:00 a.m. (California time) on the day of payment (which must be a Banking Day). All payments received after 11:00 a.m. (California time) on any particular Banking Day shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

# \$25,000,000.00

The Lender shall use its best efforts to keep a record of Advances made by it and payments of principal received by it with respect to this Note, and such record shall be presumptive evidence of the amounts owing under this Note.

Borrower hereby promises to pay all costs and expenses of any holder hereof incurred in collecting Borrower's obligations hereunder or in enforcing or attempting to enforce any of such holder's rights hereunder, including reasonable attorneys' fees and disbursements (including allocated costs of legal counsel employed by the Administrative Agent or the holder), whether or not an action is filed in connection therewith.

Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest, notice of protest and any other notice or formality, to the fullest extent permitted by applicable Laws.

THIS NOTE SHALL BE DELIVERED TO AND ACCEPTED BY THE LENDER, OR BY THE ADMINISTRATIVE AGENT ON ITS BEHALF, IN THE STATE OF CALIFORNIA, AND SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS THEREOF.

AMERICAN STATES WATER COMPANY, a California corporation

By:

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

-2-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Alternate Base Rate Advances)

					_
Date	Amount of Advance	Amount of Principal Pai	Unpaid Principal d Balance	Notation Made By	
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					-
					_

-3-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Eurodollar Rate Advances)

				Unpaid	
		Amount of	Amount of	Principal	Notation
Date	5	Advance	Principal Paid	Balance	Made By

-4-

#### \$22,000,000.00

June 6, 2002

FOR VALUE RECEIVED, the undersigned, AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), promises to pay to the order of COBANK, ACB (the "Lender"), the principal amount of TWENTY-TWO MILLION DOLLARS (\$22,000,000.00) or such lesser aggregate amount of Advances as may be made pursuant to the Lender's Pro Rata Share of the Revolving Facility under the Credit Agreement referred to below, payable as hereinafter set forth. Borrower promises to pay interest on the principal amount hereof remaining unpaid from time to time from the date hereof until the date of payment in full, payable as hereinafter set forth.

NOTE

Reference is made to the Credit Agreement dated as of June 6, 2002, by and among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as the same may be amended, renewed, extended or otherwise modified from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein with the meanings given those terms in the Credit Agreement. This is one of the Notes referred to in the Credit Agreement, and any holder hereof is entitled to all of the rights, remedies, benefits and privileges provided for in the Credit Agreement as originally executed or as it may from time to time be supplemented, modified or amended. The Credit Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events upon the terms and conditions therein specified.

The principal indebtedness evidenced by this Note shall be payable as provided in the Credit Agreement and in any event on the Maturity Date.

Interest shall be payable on the outstanding daily unpaid principal amount of each Advance hereunder from the date thereof until payment in full and shall accrue and be payable at the rates and on the dates set forth in the Credit Agreement, both before and after default and before and after maturity and judgment, with interest on overdue principal and interest to bear interest at the rate set forth in Section 3.7 of the Credit Agreement, to the fullest extent permitted by applicable Law.

The amount of each payment hereunder shall be made to the Administrative Agent at the Administrative Agent's Office for the account of the Lender in immediately available funds not later than 11:00 a.m. (California time) on the day of payment (which must be a Banking Day). All payments received after 11:00 a.m. (California time) on any particular Banking Day shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

-1-

The Lender shall use its best efforts to keep a record of Advances made by it and payments of principal received by it with respect to this Note, and such record shall be presumptive evidence of the amounts owing under this Note.

Borrower hereby promises to pay all costs and expenses of any holder hereof incurred in collecting Borrower's obligations hereunder or in enforcing or attempting to enforce any of such holder's rights hereunder, including reasonable attorneys' fees and disbursements (including allocated costs of legal counsel employed by the Administrative Agent or the holder), whether or not an action is filed in connection therewith.

Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest, notice of protest and any other notice or formality, to the fullest extent permitted by applicable Laws.

THIS NOTE SHALL BE DELIVERED TO AND ACCEPTED BY THE LENDER, OR BY THE ADMINISTRATIVE AGENT ON ITS BEHALF, IN THE STATE OF CALIFORNIA, AND SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS THEREOF.

AMERICAN STATES WATER COMPANY, a California corporation

By:

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

-2-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Alternate Base Rate Advances)

Da	ite	Amount of Advance	Amount Principal	Unpaid Principal Balance	Notation Made By
_				 	
-				 	
-				 	
-				 	
-				 	
-				 	
-				 	
-				 	
_				 	
_				 	
_				 	
_				 	
_				 	
-				 	
_				 	

-3-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Eurodollar Rate Advances)

			Unpaid	
	Amount of	Amount of	Principal	Notation
Date	Advance	Principal Paid	Balance	Made By

-4-

#### June 6, 2002

FOR VALUE RECEIVED, the undersigned, AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), promises to pay to the order of UNION BANK OF CALIFORNIA, N.A. (the "Lender"), the principal amount of FIFTEEN MILLION DOLLARS (\$15,000,000.00) or such lesser aggregate amount of Advances as may be made pursuant to the Lender's Pro Rata Share of the Revolving Facility under the Credit Agreement referred to below, payable as hereinafter set forth. Borrower promises to pay interest on the principal amount hereof remaining unpaid from time to time from the date hereof until the date of payment in full, payable as hereinafter set forth.

NOTE

Reference is made to the Credit Agreement dated as of June 6, 2002, by and among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as the same may be amended, renewed, extended or otherwise modified from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein with the meanings given those terms in the Credit Agreement. This is one of the Notes referred to in the Credit Agreement, and any holder hereof is entitled to all of the rights, remedies, benefits and privileges provided for in the Credit Agreement as originally executed or as it may from time to time be supplemented, modified or amended. The Credit Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events upon the terms and conditions therein specified.

The principal indebtedness evidenced by this Note shall be payable as provided in the Credit Agreement and in any event on the Maturity Date.

Interest shall be payable on the outstanding daily unpaid principal amount of each Advance hereunder from the date thereof until payment in full and shall accrue and be payable at the rates and on the dates set forth in the Credit Agreement, both before and after default and before and after maturity and judgment, with interest on overdue principal and interest to bear interest at the rate set forth in Section 3.7 of the Credit Agreement, to the fullest extent permitted by applicable Law.

The amount of each payment hereunder shall be made to the Administrative Agent at the Administrative Agent's Office for the account of the Lender in immediately available funds not later than 11:00 a.m. (California time) on the day of payment (which must be a Banking Day). All payments received after 11:00 a.m. (California time) on any particular Banking Day shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

# \$15,000,000.00

The Lender shall use its best efforts to keep a record of Advances made by it and payments of principal received by it with respect to this Note, and such record shall be presumptive evidence of the amounts owing under this Note.

Borrower hereby promises to pay all costs and expenses of any holder hereof incurred in collecting Borrower's obligations hereunder or in enforcing or attempting to enforce any of such holder's rights hereunder, including reasonable attorneys' fees and disbursements (including allocated costs of legal counsel employed by the Administrative Agent or the holder), whether or not an action is filed in connection therewith.

Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest, notice of protest and any other notice or formality, to the fullest extent permitted by applicable Laws.

THIS NOTE SHALL BE DELIVERED TO AND ACCEPTED BY THE LENDER, OR BY THE ADMINISTRATIVE AGENT ON ITS BEHALF, IN THE STATE OF CALIFORNIA, AND SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS THEREOF.

AMERICAN STATES WATER COMPANY, a California corporation

By:

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

-2-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Alternate Base Rate Advances)

				Unpaid	
	Amount of	Amount o	of	Principal	Notation
Date	Advance	Principal H	Paid	Balance	Made By

-3-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Eurodollar Rate Advances)

Data	Amount of Advance	Amount Principal	of	Unpaid Principal Balance	Notation	
Date	Advance	Principal	Pala	Balance	Made By	

-4-

#### NOTE

#### June 6, 2002

FOR VALUE RECEIVED, the undersigned, AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), promises to pay to the order of COMERICA WEST INCORPORATED (the "Lender"), the principal amount of TEN MILLION DOLLARS (\$10,000,000.00) or such lesser aggregate amount of Advances as may be made pursuant to the Lender's Pro Rata Share of the Revolving Facility under the Credit Agreement referred to below, payable as hereinafter set forth. Borrower promises to pay interest on the principal amount hereof remaining unpaid from time to time from the date hereof until the date of payment in full, payable as hereinafter set forth.

Reference is made to the Credit Agreement dated as of June 6, 2002, by and among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as the same may be amended, renewed, extended or otherwise modified from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein with the meanings given those terms in the Credit Agreement. This is one of the Notes referred to in the Credit Agreement, and any holder hereof is entitled to all of the rights, remedies, benefits and privileges provided for in the Credit Agreement as originally executed or as it may from time to time be supplemented, modified or amended. The Credit Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events upon the terms and conditions therein specified.

The principal indebtedness evidenced by this Note shall be payable as provided in the Credit Agreement and in any event on the Maturity Date.

Interest shall be payable on the outstanding daily unpaid principal amount of each Advance hereunder from the date thereof until payment in full and shall accrue and be payable at the rates and on the dates set forth in the Credit Agreement, both before and after default and before and after maturity and judgment, with interest on overdue principal and interest to bear interest at the rate set forth in Section 3.7 of the Credit Agreement, to the fullest extent permitted by applicable Law.

The amount of each payment hereunder shall be made to the Administrative Agent at the Administrative Agent's Office for the account of the Lender in immediately available funds not later than 11:00 a.m. (California time) on the day of payment (which must be a Banking Day). All payments received after 11:00 a.m. (California time) on any particular Banking Day shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

# \$10,000,000.00

The Lender shall use its best efforts to keep a record of Advances made by it and payments of principal received by it with respect to this Note, and such record shall be presumptive evidence of the amounts owing under this Note.

Borrower hereby promises to pay all costs and expenses of any holder hereof incurred in collecting Borrower's obligations hereunder or in enforcing or attempting to enforce any of such holder's rights hereunder, including reasonable attorneys' fees and disbursements (including allocated costs of legal counsel employed by the Administrative Agent or the holder), whether or not an action is filed in connection therewith.

Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest, notice of protest and any other notice or formality, to the fullest extent permitted by applicable Laws.

THIS NOTE SHALL BE DELIVERED TO AND ACCEPTED BY THE LENDER, OR BY THE ADMINISTRATIVE AGENT ON ITS BEHALF, IN THE STATE OF CALIFORNIA, AND SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS THEREOF.

AMERICAN STATES WATER COMPANY, a California corporation

By:

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

-2-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Alternate Base Rate Advances)

- ------Unpaid Amount of Amount of Principal Paid Principal Notation Balance Date Made By \_\_\_\_\_ \_\_\_\_\_ \_ \_\_\_ \_ \_\_\_\_\_ - -----\_ \_\_\_\_\_ - -\_\_\_\_\_ - -----\_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_ \_ \_ \_\_\_\_\_ \_ \_\_\_\_\_ - ------

-3-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Eurodollar Rate Advances)

			Unpaid		
Amount of	Amount	of	Principal	Notation	
Advance	Principal	Paid	Balance	Made By	
	Amount of Advance	Amount of Amount Advance Principal		Amount of       Amount of       Principal         Advance       Principal Paid       Balance	Amount of Amount of Principal Notation

-4-

#### \$3,000,000.00

FOR VALUE RECEIVED, the undersigned, AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), promises to pay to the order of NORTHERN TRUST BANK OF CALIFORNIA, N.A. (the "Lender"), the principal amount of THREE MILLION DOLLARS (\$3,000,000.00) or such lesser aggregate amount of Advances as may be made pursuant to the Lender's Pro Rata Share of the Revolving Facility under the Credit Agreement referred to below, payable as hereinafter set forth. Borrower promises to pay interest on the principal amount hereof remaining unpaid from time to time from the date hereof until the date of payment in full, payable as hereinafter set forth.

NOTE

Reference is made to the Credit Agreement dated as of June 6, 2002, by and among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent for the Lenders (as the same may be amended, renewed, extended or otherwise modified from time to time, the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein with the meanings given those terms in the Credit Agreement. This is one of the Notes referred to in the Credit Agreement, and any holder hereof is entitled to all of the rights, remedies, benefits and privileges provided for in the Credit Agreement as originally executed or as it may from time to time be supplemented, modified or amended. The Credit Agreement, among other things, contains provisions for acceleration of the maturity hereof upon the happening of certain stated events upon the terms and conditions therein specified.

The principal indebtedness evidenced by this Note shall be payable as provided in the Credit Agreement and in any event on the Maturity Date.

Interest shall be payable on the outstanding daily unpaid principal amount of each Advance hereunder from the date thereof until payment in full and shall accrue and be payable at the rates and on the dates set forth in the Credit Agreement, both before and after default and before and after maturity and judgment, with interest on overdue principal and interest to bear interest at the rate set forth in Section 3.7 of the Credit Agreement, to the fullest extent permitted by applicable Law.

The amount of each payment hereunder shall be made to the Administrative Agent at the Administrative Agent's Office for the account of the Lender in immediately available funds not later than 11:00 a.m. (California time) on the day of payment (which must be a Banking Day). All payments received after 11:00 a.m. (California time) on any particular Banking Day shall be deemed received on the next succeeding Banking Day. All payments shall be made in lawful money of the United States of America.

The Lender shall use its best efforts to keep a record of Advances made by it and payments of principal received by it with respect to this Note, and such record shall be presumptive evidence of the amounts owing under this Note.

Borrower hereby promises to pay all costs and expenses of any holder hereof incurred in collecting Borrower's obligations hereunder or in enforcing or attempting to enforce any of such holder's rights hereunder, including reasonable attorneys' fees and disbursements (including allocated costs of legal counsel employed by the Administrative Agent or the holder), whether or not an action is filed in connection therewith.

Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest, notice of protest and any other notice or formality, to the fullest extent permitted by applicable Laws.

THIS NOTE SHALL BE DELIVERED TO AND ACCEPTED BY THE LENDER, OR BY THE ADMINISTRATIVE AGENT ON ITS BEHALF, IN THE STATE OF CALIFORNIA, AND SHALL BE GOVERNED BY, AND CONSTRUED AND ENFORCED IN ACCORDANCE WITH, THE LAWS THEREOF.

AMERICAN STATES WATER COMPANY, a California corporation

By:

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

-2-

#### ADVANCES AND PAYMENTS OF PRINCIPAL (Alternate Base Rate Advances)

- ------Unpaid Amount of Principal Amount of Amount of Principal Paid Notation Balance Date Made By \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_ \_ \_\_\_\_\_ - -----\_ \_\_\_\_\_ \_ -\_\_\_\_\_ \_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_ \_ \_ \_ \_\_\_\_\_ \_\_\_\_\_ \_\_\_\_\_

-3-

# ADVANCES AND PAYMENTS OF PRINCIPAL (Eurodollar Rate Advances)

Date	Amount of Advance	Amount of Principal Paid	Unpaid Principal Balance	Notation Made By

-4-

#### SWING LINE NOTE

#### \$5,000,000.00

June 6, 2002 Los Angeles, California

FOR VALUE RECEIVED, upon demand or, if no demand is made, on the Revolving Credit Facility Maturity Date, AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), hereby promises to pay to the order of WELLS FARGO BANK, NATIONAL ASSOCIATION (the "Swing Line Lender") the principal amount of FIVE MILLION AND NO/100 DOLLARS (\$5,000,000) or such lesser aggregate amount of Swing Line Outstandings under the Credit Agreement referred to below, together with interest at a fluctuating rate per annum equal to the Alternate Base Rate plus the Applicable Alternate Base Rate Margin, on the principal amount of each Swing Line Loan made hereunder and remaining unpaid from time to time from the date of each such Swing Line Loan until the date of payment in full, payable as hereinafter set forth.

A. General Conditions.

1. Definitions. This Note is delivered with reference to the Credit Agreement of even date herewith, entered into by and among Borrower, the Lenders therein named and Wells Fargo Bank, National Association, as Administrative Agent (the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used with the meanings given those terms in the Credit Agreement.

 $2.\ {\rm Loan}$  Procedures. Advances hereunder shall be made in accordance with the terms, provisions and procedures set forth in the Credit Agreement applicable to Swing Line Loans.

3. Payments. Interest on the Swing Line Loans hereunder shall accrue at the rate applicable to Alternate Base Rate Advances under the Credit Agreement and shall be due and payable on such dates, not more frequent than monthly, as may be specified by the Swing Line Lender, and, in any event, all accrued interest shall be due and payable on the Maturity Date or upon demand, whichever is earlier. Any principal or interest not paid when due hereunder shall be subject to the default rate of interest set forth in Section 3.7 of the Credit Agreement with respect to Swing Line Loans, to the fullest extent permitted by applicable Law. Each payment hereunder shall be made to the Swing Line Lender at its office located at 333 South Grand Avenue, Third Floor, Los Angeles, California 90071, in immediately available funds not later than 3:00 p.m. (California time) on the day of payment (which must be a Business Day). All payments received after 3:00 p.m. (California time) shall be deemed received on the next Business Day. All payments shall be made in lawful money of the United States of America.

4. Records. The Swing Line Lender shall use its best efforts to keep a record of each Swing Line Loan made by it and payments of principal received by it with respect to this Swing Line Note, and such record shall be presumptive evidence of the amounts owing under this Swing Line Note.

-1-

5. Setting and Notice of Rates. The applicable Alternate Base Rate shall be determined and notice thereof given in accordance with the Credit Agreement. Each determination of the applicable Alternate Base Rate shall be conclusive and binding upon Borrower, in the absence of manifest error.

 $\,$  6. Computation of Interest. Interest shall be computed for the actual number of days elapsed on the basis of a year of 365 or 366 days, as the case may be, and the actual number of days elapsed.

B. Default. The "Events of Default" provisions contained in Article 9 of the Credit Agreement are hereby incorporated by this reference as though fully set forth herein.

C. Waiver. Borrower hereby waives presentment, demand for payment, dishonor, notice of dishonor, protest and any other notice of formality, to the fullest extent permitted by applicable Laws.

D. Collection Costs. In the event of the occurrence of an Event of Default, Borrower agrees to pay all costs of collection, including reasonable attorney's fees, in addition to and at the time of the payment of such sum of money and/or the performance of such acts as may be required to cure such default. In the event legal action is commenced for the collection of any sums owing hereunder, Borrower agrees that any judgment issued as a consequence of such action against Borrower shall bear interest at a rate equal to the default rate provided in Section 3.7 of the Credit Agreement until fully paid.

E. Governing Law. This Swing Line Note has been delivered in the State of California, and shall be governed by, and construed and enforced in accordance with, the Laws of California applicable to contracts made and performed in California.

F. Partial Invalidity. If any provision of this Swing Line Note shall be prohibited by or invalid under an applicable law, such provision shall be ineffective only to the extent of such prohibition or invalidity, without invalidation the remainder of such provision or any other provision of this Swing Line Note.

IN WITNESS WHEREOF, this Swing Line Note has been executed as of the date first hereinabove written.

AMERICAN STATES WATER COMPANY

Ву

Name: McClellan Harris III Title: Vice President-Finance, Treasurer and Corporate Secretary

-2-

# SCHEDULE OF SWING LINE LOANS AND PAYMENTS OF PRINCIPAL

Date	Amount of Swing Line Loan	Amount of Principal Paid	Unpaid Principal	Notation Made By Balance

-3-

June 6, 2002

Wells Fargo Bank, National Association, as Administrative Agent 333 South Grand Avenue, Third Floor Los Angeles, California 90071 and The Lenders set forth on Exhibit A attached hereto

Re: Credit Agreement by and among American States Water Company, the Lenders named therein, and Wells Fargo Bank, National Association

Ladies and Gentlemen:

We have acted as special counsel to American States Water Company, a California corporation (the "Borrower"), in connection with the Credit Agreement dated as of June 6, 2002 (the "Credit Agreement") by and among the Borrower, the Lenders named therein, and Wells Fargo Bank, National Association, as Administrative Agent (in such capacity, the "Administrative Agent"). Capitalized terms not otherwise defined herein shall have the meanings ascribed to them in the Credit Agreement. For the purpose of this opinion, "Loan Documents" means those Loan Documents (as defined in the Credit Agreement) in existence as of the Closing Date and as further described on Schedule I attached hereto and incorporated herein.

In our capacity as special counsel, we have examined originals or copies of the Credit Agreement and the other Loan Documents, as well as originals or copies of such records and documents we considered appropriate. As to relevant factual matters, we have relied upon, among other things, the representations made by the Borrower in the Credit Agreement and other Loan Documents and the Borrower's factual representations in a certificate of a Responsible Official delivered by the Borrower in connection with this opinion letter (the "Certificate"), a copy of which has been delivered to you. In addition, we have obtained and relied upon those certificates of public officials we considered appropriate.

June 6, 2002 - Page 2

We have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals and the conformity with originals of all documents submitted to us as copies. To the extent the Borrower's obligations depend on the due authorization, execution and delivery of the Credit Agreement and other Loan Documents by the other parties thereto, we have assumed that each of the Credit Agreement and other Loan Documents has each been so authorized, executed and delivered by the other parties thereto and that each constitutes the legally valid and binding obligation of the other parties enforceable in accordance with its terms. We have also assumed that each Lender is (a) a bank incorporated or organized under, or a foreign bank licensed to conduct a banking business through an agency located in the United States of America pursuant to the laws of the United States of America or the State of California within the meaning of Section 1 of Article XV of the California Constitution and Section 1716 of the California Financial Code, or (b) a finance lender licensed under the California Finance Lenders Law (California Financial Code Section 22000 et seq.) and, as a result thereof, is exempt from the usury restrictions of Section 1 of Article XV of the Constitution of the State of California

On the basis of such examination, our reliance upon the assumptions contained herein and our consideration of those questions of law we considered relevant, and subject to the limitations and qualifications in this opinion, we are of the opinion that:

1. The Borrower has been duly incorporated, and is validly existing as a corporation in good standing under the laws of the State of California, with corporate power to own its properties and assets, to enter into the Loan Documents and perform its obligations under the Loan Documents.

2. The execution, delivery and performance of each of the Loan Documents have been duly authorized by all necessary corporate action on the part of the Borrower, and each of the Loan Documents has been duly executed and delivered by the Borrower. Each of the Loan Documents constitutes the legally valid and binding obligation of the Borrower, enforceable against the Borrower in accordance with its terms, except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or affecting creditors' rights generally (including, without limitation, fraudulent conveyance laws), and by general principles of equity, including, without limitation, concepts of materiality, reasonableness, good faith and fair dealing, and the possible unavailability of specific performance or injunctive relief, regardless of whether considered in a proceeding in equity or at law.

3. The execution and delivery by the Borrower of the Loan Documents do not, and the Borrower's payment of its obligations under the Loan Documents will not, violate any current California or federal statute, rule or regulation that we have, in the exercise of customary professional diligence, recognized as applicable to the Borrower or to transactions of the type contemplated by Loan Documents.

 $\rm 4.$  The execution and delivery by the Borrower of the Loan Documents do not, and the Borrower's payment of its obligations under the Loan Documents will not, (i)

June 6, 2002 - Page 3

violate the Borrower's Articles of Incorporation or Bylaws, (ii) violate, breach, or result in a default, or result in the creation or imposition of any lien, charge or encumbrance upon any of the assets of the Borrower, under any existing obligation of or restriction on the Borrower under any other agreement (the "Other Agreements") identified in the Borrower's most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission (the "Annual Report"), or (iii) breach or otherwise violate any existing obligation of or restriction on the Borrower under any order, judgment or decree of any California or federal court or governmental authority binding on the Borrower identified in the Certificate. We express no opinion as to the effect of the Borrower's performance of its obligations in the Credit Agreement or other Loan Documents on the Borrower's compliance with financial covenants in the Other Agreements.

5. No order, consent, permit or approval of any California or federal governmental authority that we have, in the exercise of customary professional diligence, recognized as applicable to the Borrower or to transactions of the type contemplated by the Loan Documents is required on the part of the Borrower for the execution and delivery of, and payment of its obligations under, the Loan Documents.

 $\,$  6. The Borrower is not an investment company required to register under the Investment Company Act of 1940, as amended.

To our knowledge, there are no actions, suits, investigations or proceedings pending or threatened against the Borrower in any court or by or before any arbitrator or governmental authority or agency that (i) seek to affect the enforceability of the Credit Agreement or any of the other Loan Documents or (ii), except for the matters described in, or incorporated by reference into, the Annual Report, are required by the Securities Exchange Act of 1934 to be disclosed therein.

The law covered by this opinion is limited to the present federal law of the United States and the present law of the State of California. We express no opinion as to the law of any other jurisdiction and no opinion regarding the statutes, administrative decisions, rules, regulations or requirements of any county, municipality, subdivision or local authority of any jurisdiction.

Our opinion in paragraph 2 above as to the enforceability of the Loan Documents is subject to:

- (a.) public policy considerations, statutes or court decisions that may limit the rights of a party to obtain indemnification against its own gross negligence, willful misconduct or unlawful conduct;
- (b.) the unenforceability under certain circumstances of broadly or vaguely stated waivers or waivers of rights granted by law where the waivers are against public policy or prohibited by law; and

June 6, 2002 - Page 4

(c.) the unenforceability under certain circumstances of provisions waiving the right to a jury trial.

We express no opinion as to:

- i. your ability to collect attorneys' fees and costs in an action involving the Loan Documents if you are not the prevailing party in that action (we call your attention to the effect of Section 1717 of the California Civil Code, which provides that where a contract permits one party thereto to recover attorneys' fees, the prevailing party in any action to enforce any provision of the contract shall be entitled to recover its reasonable attorneys' fees);
- ii. any provision of the Loan Documents requiring written amendments or waivers of the Loan Documents insofar as it suggests that oral or other modifications, amendments or waivers could not be effectively agreed upon by the parties or that the doctrine of promissory estoppel might not apply;
- iii. the unenforceability under certain circumstances of provisions imposing penalties, liquidated damages, acceleration of future amounts due (other than principal) without appropriate discount to present value, late charges, prepayment charges and increased interest rates upon default;
- iv. the effect of non-compliance by you with any state or federal laws or regulations applicable to the transactions contemplated by the Loan Documents because of the nature of your business; and
- any provision of the Loan Documents insofar as it purports to grant a right of setoff in respect of the Borrower's assets
   (a) to any person other than a creditor of the Borrower or (b) to the Administrative Agent or any Lender in an amount greater than the amount owing by the Borrower to the Administrative Agent or such Lender.

We advise you that Section 11.17 of the Credit Agreement, which provides for exclusive jurisdiction of the courts of the State of California and federal courts sitting in that state, may not be binding on the courts in the forums excluded, and, as to federal courts, in the forums selected.

Our use of the terms "to our knowledge" or any similar phrase to qualify a statement in this opinion means (a) the knowledge obtained by us from the Credit Agreement and the Certificates and (b) that those attorneys in this firm who have given substantive attention to the representation described in the introductory paragraph of this opinion do not have current actual knowledge that the statement is inaccurate. Such terms do not include any knowledge of other attorneys within our firm or any constructive or imputed notice of any matters or items of information. Except as otherwise expressly indicated, we have not undertaken any independent

# June 6, 2002 - Page 5

investigation to determine the accuracy of the statement, and any limited inquiry undertaken by us during the preparation of this opinion letter should not be regarded as such an investigation. No inference as to our knowledge of any matters bearing on the accuracy of any such statement should be drawn from the fact of our representation of the Borrower in connection with this opinion letter or in other matters.

This letter is furnished by us as special counsel for the Borrower and may be relied by you only in connection with the transactions described herein. It may not be used or relied upon by any other person, nor may copies be delivered to any other person, without in each instance our prior written consent. You may, however, deliver a copy of this opinion to your accountants, attorneys, and other professional advisors, to governmental regulatory agencies having jurisdiction over you and to permitted transferees of the Loan Documents in connection with such transfer, and such transferees may rely on this opinion as if it were addressed and had been delivered to them on the date of this opinion, unless statements in this opinion would be affected by the status of the transferee. This opinion is expressly limited to the matters set forth above and we render no opinion, whether by implication or otherwise, as to any other matters. We assume no obligation to update or supplement this opinion to reflect any facts or circumstances that arise after the date of this opinion and come to our attention, or any future changes in laws.

Respectfully submitted,

/s/ O'MELVENY & MYERS LLP

Schedule 1

# LOAN DOCUMENTS(1)

1. Credit Agreement (with Schedules and Exhibits thereto)

- 2. Notes issued in favor of each of the financial institutions listed on Exhibit A.
- 3. Swing Line Note issued in favor of Wells Fargo Bank, National Association.

- -----

 Unless otherwise indicated, the Loan Documents are dated as of June 6, 2002, and are designated above as defined in the Credit Agreement.

Exhibit A

Wells Fargo Bank, National Association

CoBank, ACB

Union Bank of California, N.A.

Comerica West Incorporated

Northern Trust Bank of California, N.A.

#### AMERICAN STATES WATER COMPANY BACKUP CERTIFICATE

I, McClellan Harris III, hereby certify that I am Vice President--Finance, Treasurer, and Corporate Secretary of American States Water Company, a California corporation (the "Company"), and as follows:

(1) The representations and warranties of the Company in the Credit Agreement (the "Agreement"), dated as of June 6, 2002, by and among the Company, the Lenders thereto, and Wells Fargo Bank, National Association, as Administrative Agent (the "Administrative Agent") are true and correct as of the date hereof in all material respects.

(2) In connection with its opinion (the "Opinion"), dated as of the date hereof, delivered pursuant to Section 8.1(a)(6) of the Agreement, O'Melveny & Myers LLP is hereby authorized to rely on the representations made by the Company in the certificates delivered to the Administrative Agent on the date hereof in connection with the execution of the Agreement to the same extent as if such certificates were delivered to O'Melveny & Myers LLP.

(3) There are no orders, judgments or decrees of any California or federal court or governmental authority binding on the Company.

(4) There are no actions, suits, investigations or proceedings pending or threatened against the Company in any court or by or before any arbitrator or governmental authority or agency that (a) seek to affect the enforceability of the Agreement or any of the other Loan Documents (as defined in the Opinion), or (b) except for the matters described in, or incorporated by reference into, the Company's most recent Annual Report on Form 10-K filed with the Securities and Exchange Commission, are required by the Securities Exchange Act of 1934 to be disclosed therein.

(5) The Company (a) is not and does not hold itself out as being engaged primarily, nor does it propose to engage primarily, in the business of investing, reinvesting or trading in securities; (b) is not engaged in, and does not propose to engage in, the business of issuing face-amount certificates of the installment type, nor has it been engaged in such business and has any such certificate outstanding; and (c) does not own or propose to acquire investment securities having a value exceeding 40% of the value of the Company's total assets (exclusive of government securities and cash items) on an unconsolidated basis.

IN WITNESS WHEREOF, I have executed this Certificate on behalf of American States Water Company as of June 6, 2002.

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris III

McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

#### REQUEST FOR BORROWING

1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:

- (a) AMOUNT OF REQUESTED BORROWING: \$2,000,000
- (b) DATE OF REQUESTED BORROWING: June 6, 2002
- (c) TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one box only):
  - | | ALTERNATE BASE RATE ADVANCES
  - |X| EURODOLLAR RATE ADVANCES FOR A EURODOLLAR PERIOD OF one (1) MONTHS(1)
- 3. In connection with the request, Borrower certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of

#### - ----

(1) Specify whether 1, 2, 3, 6, 9 or 12-month Eurodollar Period.

-1-

Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

4. This Request for Borrowing is executed on June 6, 2002, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ McClellan Harris III McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

#### REQUEST FOR BORROWING

1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:

- (a) AMOUNT OF REQUESTED BORROWING: \$6,000,000
- (b) DATE OF REQUESTED BORROWING: June 6, 2002
- TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one (C)box only):
  - ALTERNATE BASE RATE ADVANCES |X|
  - EURODOLLAR RATE ADVANCES FOR A EURODOLLAR PERIOD 1 1 OF \_\_\_\_\_ MONTHS (1)

3. In connection with the request, Borrower certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4, 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of

(1) Specify whether 1, 2, 3, 6, 9 or 12-month Eurodollar Period.

-1-

Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

4. This Request for Borrowing is executed on June 6, 2002, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ McClellan Harris III McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

#### REQUEST FOR BORROWING

1. This REQUEST FOR BORROWING is executed and delivered by AMERICAN STATES WATER COMPANY, a California corporation ("Borrower"), to WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent, pursuant to that certain Credit Agreement (as amended, modified or extended, the "Agreement") dated as of June 6, 2002, among Borrower, the Lenders that are parties thereto, and Wells Fargo Bank, National Association, as Administrative Agent and Arranger. Any terms used herein and not defined herein shall have the meanings set forth for such terms in the Agreement.

2. Borrower hereby requests a Borrowing pursuant to the Agreement as follows:

- (a) AMOUNT OF REQUESTED BORROWING: \$6,000,000
- (b) DATE OF REQUESTED BORROWING: June 6, 2002
- (c) TYPE OF ADVANCES COMPRISING REQUESTED BORROWINGS (Check one box only):
  - | | ALTERNATE BASE RATE ADVANCES
  - |X| \*EURODOLLAR RATE ADVANCES FOR A EURODOLLAR PERIOD OF one (1) MONTHS(1) \*Roll from Base Rate Advance
- 3. In connection with the request, Borrower certifies that:

(a) Now and as of the date of the requested Borrowing, except (i) for representations and warranties which expressly speak as of a particular date or are no longer true and correct as a result of a change permitted by the Agreement or (ii) as disclosed by Borrower and approved in writing by the Requisite Lenders, each representation and warranty made by Borrower in Article 4 of the Agreement (other than Sections 4.4; 4.6 (first sentence), 4.9 and 4.16) will be true and correct in all material respects, both immediately before and after giving effect to such Borrowing, as though such representations and warranties were made on and as of that date;

(b) No circumstance or event has occurred that constitutes a Material Adverse Effect since the Closing Date; and

(c) Other than matters described in Schedule 4.9 to the Agreement or not required as of the Closing Date to be described therein or disclosed by Borrower and approved in writing by the Requisite Lenders, there is no action, suit, proceeding or investigation pending or, to the best knowledge of Borrower (on behalf of Borrower and its Subsidiaries), threatened against or affecting Borrower or Subsidiary of

- -----

(1) Specify whether 1, 2, 3, 6, 9 or 12-month Eurodollar Period.

-1-

Borrower or any Property of any of them before any Governmental Agency that constitutes a Material Adverse Effect.

4. This Request for Borrowing is executed on June 6, 2002, by a Responsible Official of Borrower. The undersigned, in such capacity, hereby certifies, on behalf of Borrower, each and every matter contained herein to be true and correct.

AMERICAN STATES WATER COMPANY, a California corporation

By: /s/ McClellan Harris III McClellan Harris III Vice President - Finance, Treasurer and Corporate Secretary

-2-

#### CERTIFICATE OF THE CHIEF FINANCIAL OFFICER OF AMERICAN STATES WATER COMPANY

# (Projections)

TO: WELLS FARGO BANK, NATIONAL ASSOCIATION, as Administrative Agent

The undersigned is the chief financial officer of American States Water Company, a California corporation ("Borrower"). Reference is made to the Credit Agreement dated as of June 6, 2002, entered into by and among Borrower, the Lenders party thereto, and Wells Fargo Bank, National Association, as Administrative Agent (the "Credit Agreement"). Terms defined in the Credit Agreement and not otherwise defined herein are used herein as so defined.

I, McClellan Harris III, being the chief financial officer of Borrower, hereby certify that:

(a) Attached hereto as Exhibit  $\mbox{A}$  is a true and correct copy of the Projections.

(b) The representation contained in Section 4.16 is, to the best of my knowledge, true and correct in all material respects.

Dated as of June 6, 2002

/s/ McClellan Harris Name: McClellan Harris III Title: Vice President--Finance, Chief Financial Officer, Treasurer and Corporate Secretary

-1-

# American States Water Company Consolidated Balance Sheets

(\$ in thousands)	2002	2003	2004	2005
Assets				
Utility Plant, at cost Water Electric	\$ 703,510 39,552	\$ 756,210 40,852	42,152	43,452
Less Accumulated depreciation	743,062	797,062 (228,614)	853,562 (249,720)	912,562 (272,323)
Construction work in progress `		568,448	603,842	640,239 41,800
Net utility plant		609,148	645,142	682,039
Other Property and Investments	24,314	24,050	23,786	23,522
Current Assets Cash and cash equivalents Accounts receivable-Customers Other account receivable Intercompany receivable Unbilled revenue Materials and supplies, at average cost Supply cost balancing accounts Prepayments Accumulated deferred income taxes net Total current assets	13,600 5,250  11,700 1,285 25,325 4,250 178	\$ 62,971	14,000 2,750  11,750 1,295 16,085 4,350 6,891 \$ 61,842	14,200 2,750  11,650 1,300 11,465 4,400 10,294
Deferred Charges Unamortized debt expense and redemption Regulatory tax-related assets Other		11,486 8,501		7,340 8,417
Total deferred charges		39,125		
Total Assets	\$ 707 <b>,</b> 898	\$ 735,294	\$ 767,014	\$ 798,762

American States Water Company Consolidated Balance Sheets

(\$ in thousands)	2002	2003	2004	2005
Capitalization and Liabilities				
Capitalization Common shareholders' equity Preferred Shares Preferred Shares mandatory redemption Long-term debt	1,600 240 204,452	\$ 217,880 1,600 200 231,643	1,600 160 255,781	1,600 120 255,160
Total capitalization		451,322		
Current Liabilities Notes payable to banks Long-term debt and Preferred Shares current Accounts payable Intercompany payable Taxes payable Accrued interest Other Total current liabilities	13,331 15,100  6,850 1,860 14,910	7,450 2,140 15,420	933 14,100  7,800 2,686 15,930	694 14,600  8,150 2,951 16,440
Other Credits Advances for construction Contributions in aid of construction Accumulated deferred income taxes net Unamortized investment tax credits Regulatory tax-related liability Other Total other credits Total Capitalization and Liabilities	42,500 52,650 3,000 1,729 1,800  172,049 \$ 707,898	70,570 43,500 53,600 2,900 1,685 1,800  \$ 735,294	44,500 54,550 2,800 1,641 1,800  177,661 \$ 767,014	45,500 56,300 2,700 1,597 1,800  181,827 \$ 798,762

American States Water Company Consolidated Statements of Income

(\$ in thousands, except per share amounts)	2002	2003	2004	2005
Operating Revenues	\$ 209,225	\$ 220,247	\$ 230,621	\$ 240,605
Operating Expenses Supply Costs Other operating and administrative expenses		72,680 69,897	75,231 72,008	
Total Operating Expenses		142,577	147,239	
EBITDA	\$ 74,221	\$77 <b>,</b> 670	\$ 83,382	\$ 88,800
Depreciation and Amortization	18,570	19,937	21,370	22,868
EBIT	\$ 55,651	\$ 57 <b>,</b> 733	\$ 62,012	\$ 65,932
Interest Charges	17,512	18,687	19,921	21,194
Taxes on Income	17,051	17,463	18,829	20,017
Net Income	\$ 21,088	\$ 21,583	\$ 23,262	\$ 24,722
Dividends on Preferred Shares	(82)	(80)	(78)	(76)
Earnings Available For Common	\$ 21,006	\$ 21,503	\$ 23,184	\$ 24,646
Weighted Average Number of Common Shares Basic Earnings Per Common Share Increased by	10,107 \$ 2.08 2.6%	,	,	
Diluted Number of Common Shares Outstanding Diluted Earnings Per Common Share	10,245 \$ 2.05		10,722 \$ 2.16	11,125 \$ 2.22
Dividend per Common Share Payout Ratio	\$ 1.305 63%	\$ 1.320 62%	\$ 1.325 60%	\$ 1.33 59%
Weighted Rate Base Rate of Return	\$ 433,190 8.9%	\$ 473,236 8.5%	\$ 506,635 8.5%	\$ 539,791 8.5%
Return on Average Common Equity	10.3%	10.1%	9.9%	9.6%

# Amcrican States Water Company Consolicated Statements of Cash Flows

(\$ in thousands)	2002	2003	2004	2005
Cash Flows From Operating Activities: Net income	¢ 01 000	\$ 21,583	¢ 22.262	¢ 04 700
Adjustments for non-cash items:	Ş ∠1,088	\$ 21,383	⇒ ∠3,202	⇒ 24,122
Depreciation and amortization Deferred income taxes and investment tax credits	18,570	19,937	21,370	22,868
			339	1,094
Other net Changes in assets and liabilities:				
Customer receivables	(200)	(200)	(200)	(200)
Prepayments	(50)	(50)	(50)	(50)
Supply cost balancing accounts	300	4,620 (1,500)	4,620	4,620
Accounts payable	(500)	(1,500)	500	500
Taxes payable		600	350	350
Unbilled revenue	(100)	100 280	(150)	100 266
Accrued Interest				
Other net	505	3,505		
Net cash provided	\$ 26,048		\$ 50,591	\$ 54,774
Cash Flows From Investing Activities:				
Construction expenditures	\$ (52,050)	\$ (54,500)	\$ (57,100)	\$ (59,500)
Acquisition				
Net cash used		\$ (54,500)	\$ (57,100)	
Cash Flows from Financing Activities:				
Issuance of Securities	\$ (26,027)	\$ 28,030	\$ 50,031	\$ 32
Receipt of advances for and contributions, net	1,440	1,200	2,800	2,560
Retirement or repayments of long-term debt	(790)	(13,331) \$ 2,676	(880)	(933)
Net change in notes payable to banks	52,293	\$ 2,676	(31,752)	16,641
Common and preferred dividends paid	(13,308)	(13,438)	(14,454)	
Net cash provided	\$ 13,608	\$ 5,117		\$ 3,795
Net Increase (Decrease) in Cash and Cash Equivalents		\$ (610)		\$ (931)
Cash and Cash Equivalents, Beginning of Year	18,488	6,094		
Cash and Cash Equivalents, End-of Year		\$ 5,484		
		=======		

American States Water Company Consolidated Rate Base

(\$ in thousands)	2002	2003	2004	2005
Utility Plant	\$ 574,321	\$ 609,148	\$ 645,142	\$ 682,039
Acquisition adjustment	\$ 4,033	\$ 4,033	\$ 4,033	\$ 4,033
Material & Supplies	\$ 1,285	\$ 1,290	\$ 1,295	\$ 1,300
Contribution and Advances	\$(112,870)	\$(114,070)	\$(116,870)	\$(119,430)
Deferred Taxes	\$ (16,150)	\$ (16,046)	\$ (16,384)	\$ (17,475)
Working Cash	\$ 5,700	\$ 5,800	\$ 5,900	\$ 6,000
Total Rate Base	\$ 456,318	\$ 490,154	\$ 523 <b>,</b> 116	\$ 556,466

(\$ in thousands)	2002	2003		2005
Common Shareholders' Equity: Common Shares, \$2.50 stated value Outstanding 10,079,629 in 2000	\$ 25.199	\$ 25,199	\$ 25,199	\$ 25.199
Additional paid-in capital Earnings reinvested in the business	102,183 82,373	102,183 90,498	127,183 99,306	127,183 109,522
		217,880	251,688	261,904
Preferred Shares: \$25 par value				
Authorized 64,000 shares Outstanding 32,000 shares, 4% Series	800 800 		800	800
Outstanding 32,000 shares, 4 1/4% Series		1600	1600	1600
Redemption				
Requirements: \$25 par value 5% Series	280 -40	240 -40	-40	-40
within one year	240	200	160	120
Long-Term Debt	4.0 5.0.0	0 1,100 2,000 1,416 6,000 515 5,000 10,000 8,000 22,000 7,950 20,000 15,000 0 50,000 0 28,000 0 28,000 0 0 800 1,000		
5.82% notes due 2003	12,500	1 1 0 0	1 100	1 100
6.64% notes due 2013 6.80% notes due 2013	1,100	2,100	2,100	1,100
8.50% fixed rate obligation due 2013	1 524	2,000	2,000	2,000
Variable rate obligation due 2014	6,000	6.000	6,000	6.000
Variable rate obligation due 2018	551	515	480	444
6.87% notes due 2023	5,000	5,000	5,000	5,000
7.00% notes due 2023	10,000	10,000	10,000	10,000
7.55% notes due 2025	8,000	8,000	8,000	8,000
7.65% notes due 2025	22,000	22,000	22,000	22,000
5.50% notes due 2026	7,950	7,950	7,950	7,950
7.875% notes due 2030	20,000	20,000	20,000	20,000
6.81% notes due 2028	15,000	15,000	15,000	15,000
6.59% notes due 2029	40,000	40,000	40,000	40,000
9.56% notes due 2031	0	0	0	0
8.00% 8.00%	50,000	50,000	50,000	50,000
8.00%	0	28 000	28 000	28 000
8.00%	0	20,000	25,000	25,000
8.00%	0	0	20,000	20,000
4% to 4.85% serial bonds due 2007	1,035	800	555	295
5.20% term bonds due 2011	1,000	1,000	1,000	1,000
5.40% term bonds due 2022	4,610	4,610	4,610	4,610
4.65% term bonds due 2006	215	215	215	215
5.30% term bonds due 2022	1,015	1,015	1,015	1,015
3.34% repayment contract due 2006	1,014	723	408	70
State Water Project due 2035 Other	6,780 450	0 800 1,000 4,610 215 1,015 723 6,689 450	6,594 450	6,495 450
Less: Current maturities	-13,291	-840	-893	-654
	204,452	231,643	255,781	255,160
Total Capitalization	416,048	451,322		

(\$ in thousand)	2002	2003	2004	2005
Interest LT				
5.82% notes due 2003	728	364		
6.64% notes due 2013	73		73	73
6.80% notes due 2013	136		136	
8.50% fixed rate obligation due 2013	130			
Variable rate obligation due 2014	210		210	
Variable rate obligation due 2018	42	40	37	31
6.87% notes due 2023	344	344	344	344
7.00% notes due 2023	700	700	700	700
7.55% notes due 2025	604	604	604	604
7.65% notes due 2025	1,683	1,683	1,683	1,683
5.50% notes due 2026	437		437	
7.875% notes due 2030	1,575	1,575	1,575	1,575
6.81% notes due 2028	1,022		1,022	1,022
6.59% notes due 2029	2,636	2,636	2,636	2,636
9.56% notes due 2031	915			
	3,615	3,615	3,615	3,615
		1,120	2,240	2,240
			1,063	2,125
4% to 4.85% serial bonds due 2007	46	36	26	14
5.20% term bonds due 2011	52	52	52	52
5.40% term bonds due 2022	249	249	249	249
4.65% term bonds due 2006	10	10	10	10
5.30% term bonds due 2022	54	54	54	54
3.34% repayment contract due 2006	34	24	14	2
State Water Project due 2035	289	286	282	277
Other	187	32	32	32
	698	767	776	776
		16,187.39		18,995.24

#### AMERICAN STATES WATER COMPANY OFFICERS CERTIFICATE

To: Wells Fargo Bank, National Association, as Administrative Agent

This Certificate is delivered with reference to the Credit Agreement (the "Credit Agreement") dated as of June 6, 2002 among American States Water Company, a California corporation ("Borrower"), the Lenders thereto and Wells Fargo Bank, National Association, as Administrative Agent. Capitalized terms used herein are used with the meanings set forth for those terms in the Credit Agreements.

I, McClellan Harris III, hereby certify that I am the Vice President - Finance, Treasurer and Corporate Secretary of the Borrower and as follows:

(a) Floyd S. Wicks is the President and Chief Executive Officer of the Borrower and together with McClellan Harris III, is authorized to execute and deliver the Credit Agreement on behalf of the Borrower; the signature of Floyd E. Wicks appearing on the signature page of the Credit Agreement delivered to the Administrative Agent and the Lenders on the date hereof is his genuine signature;

(b) Floyd E. Wicks and each of the following named persons is authorized, in the name and on behalf of the Borrower to execute and deliver each other Loan Document to which the Borrower is a party and request the making of Advances and the issuance of Letters of Credit under the Credit Agreements, and the signatures of such persons set forth below are their genuine signatures;

Name	Title/Office	Signature
McClellan Harris III	CFO, V.P. Finance, Treasurer & Secretary	/s/ McClellan Harris III
Joel A. Dickson	V.P. Administration	/s/ Joel A. Dickson
Donald K. Saddoris	V.P. Chief of Operations	/s/ Donald K. Saddoris

(c) The execution, delivery and performance of the Credit Agreement and each Loan Document to which the Borrower is a party was duly authorized by resolutions duly adopted by the Board of Directors of the Borrower (the "Resolutions"), a true, current and complete copy of which is

-1-

attached hereto as Exhibit A and incorporated herein by this reference; the Resolutions are still in full force and effect.

(d) Attached hereto as Exhibit B and incorporated herein by this reference is a true, correct and complete copy of the articles of incorporation of Borrower.

(e) Attached hereto as Exhibit C and incorporated herein by this reference is a true, correct and complete copy of the corporate bylaws of Borrower.

(f) Attached hereto as Exhibit D and incorporated herein by this reference is a corporate and tax good standing certificate with respect to the Borrower from the Secretary of State of California.

(g) Each of the representations and warranties made by the Borrower in Article 4 of the Credit Agreement is true and correct in all material respects on and as of the date hereof as though made on and as of the date hereof.

(h) As of the date hereof, (i) the Borrower is in compliance with the terms and provisions of the Loan Documents to which it is a party and (ii) giving effect to the initial Advances or initial Letter of Credit (as applicable), no Default or Event of Default under the Credit Agreement has occurred or is continuing.

IN WITNESS WHEREOF, I have executed this Certificate on behalf of American States Water Company as of June 6, 2002.

AMERICAN STATES WATER COMPANY

By: /s/ McClellan Harris McClellan Harris III Vice President - Finance,

Treasurer and Corporate Secretary

-2-

## UNANIMOUS WRITTEN CONSENT OF CERTAIN MEMBERS OF THE AUDIT AND FINANCE COMMITFEE OF AMERICAN STATES WATER COMPANY

WHEREAS, the undersigned are members of the Audit and Finance Committee (the "Committee") of the Board of Directors of American States Water Company (the "Company").

WHEREAS, the Committee has recommended approval of management's plan for obtaining an a revolving credit facility in an amount not to exceed \$75,000,000.

WHEREAS, the Board of Directors has determined that it is advisable and in the best interests of the Company and its shareholders for the Company to enter into an unsecured revolving credit facility of no more than three years' duration, in an amount not to exceed \$75,000,000, with one or more commercial banks or other commercial lenders to be used for general corporate purposes, subject to the approval of at least two members of the Committee.

WHEREAS, each of the undersigned has reviewed a draft of the proposed 3-year, unsecured, \$75,000,000 credit facility (the "Credit Agreement") proposed by the Company's management.

NOW, THEREFORE, BE IT RESOLVED, that the execution and delivery of the Credit Agreement, in the form presented to this Committee, by Floyd E. Wicks, the President and Chief Executive Officer of the Company, and McClellan Harris, III, the Vice President-Finance, Chief Financial Officer, Treasurer and Secretary of the Company, are hereby approved, with such changes in the Credit Agreement as may be approved by such officers, such approval to be evidenced by the execution and delivery of the Credit Agreement.

RESOLVED FURTHER, that, as provided in the resolutions adopted by the Board of Directors, each of the officers of the Company is authorized to make all such arrangements, execute such other documents required to be executed in connection with the Credit Agreement (including promissory notes), and to do and perform all such acts and things such officer deems necessary or appropriate in order to consummate the execution and delivery of the Credit Agreement by such officers.

IN WITNESS WHEREOF, each of the undersigned has set his hand on this 22 day of March, 2002.

Name: /s/ Robert F. Kathol

Name:

\_\_\_\_\_

#### UNANIMOUS WRITTEN CONSENT OF CERTAIN MEMBERS OF THE AUDIT AND FINANCE COMMITTEE

#### OF AMERICAN STATES WATER COMPANY

WHEREAS, the undersigned are members of the Audit and Finance Committee (the "Committee") of the Board of Directors of American States Water Company (the "Company").

WHEREAS, the Committee has recommended approval of management's plan for obtaining an a revolving credit facility in an amount not to exceed \$75,000,000.

WHEREAS, the Board of Directors has determined that it is advisable and in the best interests of the Company and its shareholders for the Company to enter into an unsecured revolving credit facility of no more than three years' duration, in an amount not to exceed \$75,000,000, with one or more commercial banks or other commercial lenders to be used for general corporate purposes, subject to the approval of at least two members of the Committee.

WHEREAS, each of the undersigned has reviewed a draft of the proposed 3-year, unsecured, \$75,000,000 credit facility (the "Credit Agreement") proposed by the Company's management.

NOW, THEREFORE, BE IT RESOLVED, that the execution and delivery of the Credit Agreement, in the form presented to this Committee, by Floyd E. Wicks, the President and Chief Executive Officer of the Company, and McClellan Harris, III, the Vice President-Finance, Chief Financial Officer, Treasurer and Secretary of the Company, are hereby approved, with such changes in the Credit Agreement as may be approved by such officers, such approval to be evidenced by the execution and delivery of the Credit Agreement.

RESOLVED FURTHER, that, as provided in the resolutions adopted by the Board of Directors, each of the officers of the Company is authorized to make all such arrangements, execute such other documents required to be executed in connection with the Credit Agreement (including promissory notes), and to do and perform all such acts and things such officer deems necessary or appropriate in order to consummate the execution and delivery of the Credit Agreement by such officers.

IN WITNESS WHEREOF, each of the undersigned has set his hand on this 22 day of March, 2002.

Name:

Name: /s/ N.P. Dodge Jr.

-------

# RESOLUTION

RESOLVED, the Committee has reviewed the general terms, costs and conditions of the Corporation's intent to establish a \$75,000,000 syndicated revolving credit facility to be used for general corporate purposes and hereby recommends approval to the Board of Directors.

#### RESOLUTION

WHEREAS, the Audit and Finance Committee of this Board of Directors (this "Board") has reviewed management's \$75,000,000 proposed credit facility.

WHEREAS, the Audit and Finance Committee of this Board has recommended approval of management's plan for obtaining the credit facility.

WHEREAS, this Board has determined that it is advisable and in the best interests of the Corporation and its shareholders to obtain funds from a revolving credit facility with one or more commercial banks or other commercial lenders to be used for general corporate purposes.

NOW, THEREFORE, BE IT RESOLVED, that within the limitations specified in these Resolutions and with the approval of at least 2 members of the Audit and Finance Committee of the Corporation, Floyd E. Wicks, the President and Chief Executive Officer of the Corporation, and McClellan Harris, III, the Vice President-Finance, Chief Financial Officer, Treasurer and Secretary of the Corporation, be and hereby are authorized, in the name and on behalf of the Corporation, to negotiate, execute and deliver a credit agreement with one or more commercial banks or other commercial lenders (the "Credit Agreement");

BE IT RESOLVED FURTHER, that the Credit Agreement shall provide for a revolving credit facility in an amount not to exceed \$75,000,000;

BE IT RESOLVED FURTHER, that the term of the Credit Agreement shall not exceed 3 years;

BE IT RESOLVED FURTHER, that all loans made under the Credit Agreement shall be unsecured;

BE IT RESOLVED FURTHER, that the officers of the Corporation be, and each of them hereby is, authorized in the name and on behalf of the Corporation, to make all such arrangements, to do and perform all such acts and things, and to execute and deliver all such officer's certificates and other instruments and documents as any such officer deems necessary or appropriate in order fully to effectuate the purposes of these Resolutions; and

BE IT RESOLVED FURTHER, that all acts previously taken by each of the officers of the Corporation in order to effectuate the purposes of these Resolutions are hereby confirmed and ratified.

# [STATE OF CALIFORNIA LOGO]

[SEAL OF THE SECRETARY OF STATE]

# SECRETARY OF STATE

I, BILL JONES, Secretary of State of the State of California, hereby certify:

That the attached transcript of 37 page(s) was prepared by and in this office from the record on file, of which it purports to be a copy, and that it is full, true and correct.

IN WITNESS WHEREOF, I execute this certificate and affix the Great Seal of the State of California this day of MAY 30 2002

/s/ Bill Jones

Secretary of State

[SEAL OF THE STATE OF CALIFORNIA]

# 2070444

## FILED [BQM] in the office of the Secretary of State of the State of California

FEB 25 1998

# /s/ Bill Jones BILL JONES, Secretary of State

#### ARTICLES OF INCORPORATION

#### OF AMERICAN STATES WATER COMPANY

#### ARTICLE I

#### The name of this Corporation is American States Water Company.

#### ARTICLE II

The purpose of this Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of California other than the banking business, the trust business or the practice of a profession permitted to be incorporated by the California Corporations Code.

#### ARTICLE III

The name and address of this Corporation's initial agent for service of process is McClellan Harris III, 630 East Foothill Boulevard, San Dimas, California 91773.

#### ARTICLE IV

The Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,233,200; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per shares and an aggregate stated value of \$15,000,000; 83,200 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$2,080,000; and 30,000,000 shares are to be Common Shares, no par value with a stated value of \$2.50 per share and an aggregate stated value of \$75,000,000.

A statement of the preferences, privileges and restrictions granted to or imposed upon the respective classes or series of shares and/or upon the holders thereof is as follows:

(1) Subject to the provisions of this Article IV, New Preferred Shares of any particular series shall be entitled to such voting rights, if any, as may be specified for shares of such series in the certificate of determination of preferences of such series filed as provided below; all Preferred Shares shall be entitled to voting rights on the basis of one vote per share; and all Common Shares shall be entitled to voting rights on the basis of one-tenth of one vote per share.

(2) New Preferred Shares may be issued from time to time in one or more series. Each such series shall be so designated as to distinguish it from other series of New Preferred Shares and from series of Preferred Shares. Such designation may include an appropriate reference to the dividend rate and/or any other characteristics of such series. The Board of Director is hereby authorized, within the limits of, but to the extent authorized by applicable law and within the limitations and restrictions, if any, stated in this Article IV, to fix or alter, from time to time, the dividend rights, dividend rate, conversion rights, voting rights, right and terms of redemption (including sinking fund provisions), redemption price or prices, and liquidation preferences, or any of them, of any wholly unissued series of New Preferred Shares, and to fix the number of shares constituting any such unissued series, by a resolution or resolutions adopted by the Board of Directors in exercise of the authority hereby granted.

(3) The Preferred Shares may be issued from time to time in any number of series. One such series shall (i) be and hereby is designated the "4 1/4\$ Series", (ii) consist of 32,000 shares, (iii) be entitled to dividends as provided in Paragraph (4) hereof at the rate of 4 1/4% per annum of the par value thereof, and (iv) be redeemable in the manner and otherwise upon the conditions provided in Paragraph (6) hereof by payment of a redemption price equal to the par value thereof and unpaid dividends accrued thereon to and including the date fixed for such redemption and a premium of \$1.50 per share. Another such series of Preferred Shares shall (i) be and hereby is designated as the "4% Series", (ii) consist of 32,000 shares, (iii) be entitled to dividends as provided in Paragraph (4) hereof at the rate of 4% per annum of the par value thereof, and (iv) be redeemable in the manner and otherwise upon the conditions provided in Paragraph (6) hereof by payment of a redemption price equal to the par value thereof and unpaid dividends accrued thereon to and including the date fixed for such redemption and a premium of \$2 per share. Except as to the foregoing particulars no distinction shall

exist between any of the Preferred Shares or any series thereof, and all Preferred Shares, regardless of series, shall be of equal rank and priority.

(4) The holders of the outstanding shares of the several and respective series of Preferred Shares shall be entitled to receive, out of any funds legally available therefor, dividends at the respective rates for the shares of said series, payable in cash quarterly on the first days of March, June, September and December in each year when and as declared by the Board of Directors of this Corporation. Such dividends shall accrue on each such share from the date of its original issuance and shall accrue from day to day whether or not earned or declared. Such dividends shall be cumulative so that if such dividends in respect of any quarterly dividend period at the respective rates fixed therefor shall not have been paid on, or declared and set apart for, all Preferred Shares of each series at the time outstanding, the deficiency shall be fully paid on or declared or set apart for such shares before any dividend or other distribution shall be paid upon or declared or set apart for the Common Shares. No such dividend shall be declared or paid upon or set apart for any outstanding shares of any one of said series unless at the same time such dividends on all outstanding Preferred Shares of all of said series shall be declared and paid in full or set apart for such payment.

(5) In the event of the liquidation, dissolution or winding up of this Corporation, whether voluntary or involuntary, the holders of the shares of the several and respective series of Preferred Shares shall be entitled to receive out of the assets of this Corporation, whether such assets are capital or surplus of any nature, an amount equal to the par value thereof plus all unpaid dividends accrued thereon to the date that such amount is made available for distribution to the holders thereof, and no more, or ratably from available assets if such assets are insufficient to permit payment to said holders of their full preferential amount aforesaid. Such amount shall be paid upon said shares, or shall be set apart for such payment, before any distribution is made or set apart for any Common Shares in any such liquidation, dissolution or winding up. A consolidation or merger of this Corporation with or into any other corporation or corporations shall not be deemed to be a liquidation, dissolution or winding up within the meaning of this Paragraph (5).

(6) This Corporation, at the option of its Board of Directors, may at any time or from time to time redeem the whole or any part of the outstanding shares of any one or more series of the Preferred Shares by paying in cash therefore the amount payable upon the redemption thereof (such amount being hereinafter referred to as the "redemption price"). In case of the redemption of a part only of the outstanding shares of any series of Preferred Shares, this Corporation shall designate by lot, in such manner as the Board of Directors may determine, the shares to be redeemed. At least thirty (30) days' previous notice by mail, postage prepaid, shall be given to the holders of record of the shares to be redeemed, such notice to be addressed to each such shareholder at his post office address as shown by the records of this Corporation at the opening of business on the day of mailing such notice. If on or before the date fixed for redemption and specified in such notice funds necessary for such redemption shall have been set aside at the place designated in said notice so as to be and continue available therefor, then notwithstanding that the certificates evidencing any shares called for such redemption shall not have been surrendered, the dividends with respect to the shares so called for redemption shall cease to accrue after the said date fixed for redemption and all rights with respect to such shares shall forthwith after said date cease and determine except only the right of the holders thereof to receive payment of the redemption price, without interest, upon surrender of their certificates representing the redeemed shares. In case less than all the shares represented by any such surrendered certificate shall have been redeemed, a new certificate shall be issued for the unredeemed shares. Subject to the provisions hereof, the Board of Directors shall have authority to prescribe from time to time the manner in which the Preferred Shares shall be redeemed.

(7) If at any time four (4) quarterly dividends (whether or not consecutive) which have accrued on the outstanding Preferred Shares pursuant to Paragraph (4) hereof shall be in arrears, then at the annual meeting of shareholders next following the fourth such quarterly dividend default, or, if such next annual meeting is not to be held within sixty (60) days following such default, at a special meeting of shareholders called for the purpose on the written request of the holders of not less than ten percent (10%) of the then outstanding Preferred Shares, the holders of said outstanding Preferred Shares shall be entitled, voting separately as a class (regardless of series), to elect the smallest number of directors of this Corporation which shall constitute a majority of the authorized number of such directors and the holders of the New Preferred Shares and the holders of the Common Shares, in accordance with their respective voting rights, shall be entitled to elect the remaining number of such authorized directors; which voting rights by said respective classes of shares shall continue until, but only until, all dividends which shall have accrued for any period under said Paragraph (4) upon the

outstanding Preferred Shares shall have been paid or set apart for payment. At all meetings of shareholders held for the purpose of electing directors during such time as the holders of the Preferred Shares have the right, voting separately as a class, to elect directors pursuant to this Paragraph (7), the presence in person or by proxy of the holders of a majority of the outstanding Preferred Shares (regardless of series), shall be required to constitute a quorum of such class for the election of directors, and the presence in person or by proxy of the holders of a majority of the other outstanding shares entitled to vote at such meeting, as a separate class or classes in accordance with their respective voting rights, shall be required to constitute a quorum of such class for the election of directors; provided, however, that the absence of a quorum of the holders of outstanding shares of either such class or classes shall not prevent the election at any such meeting, or adjournment thereof, of directors by the other such class or classes if the necessary quorum of such other class or classes is present in person or by proxy at such meeting or adjournment; and provided further that in the event that at such meeting or adjournment such a quorum of the holders of classes other than the Preferred Shares is present and such quorum of the holders of the Preferred Shares is not present, then an election of the directors elected at such meeting or adjournment by the holders of classes other than the Preferred Shares shall not be effective and any directors so elected by such holders of classes other than the Preferred Shares shall not assume office until the holders of the Preferred Shares, with such a quorum present, shall elect the directors they are entitled to elect. In the event of an election of directors by holders of Preferred Shares pursuant to this Paragraph (7), the term of office as directors of all persons who are directors of this Corporation at the time of the accrual under this Paragraph (7) of the right of the holders of the Preferred Shares to elect directors shall terminate when the holders of the outstanding Preferred Shares shall have so elected directors. In case any vacancy shall occur among the directors elected as aforesaid by the holders of the Preferred Shares, or among the directors elected as aforesaid by the holders of classes other than the Preferred Shares, during any period for which a majority of the directors shall have been so elected by the holders of the Preferred Shares, such vacancy shall be filled by the vote of a majority of the remaining directors who were so elected by the holders of the Preferred Shares or by the holders of classes other than the Preferred Shares, as the case may be.

(8) Without the approval of the holders of at least two-thirds of the outstanding Preferred Shares, given in person or by proxy, either by written consent or by vote as provided by law, this Corporation shall not

- alter or amend the preferences, voting powers or other special rights or the qualifications, limitations and restrictions imposed in favor of any of the Preferred Shares so as adversely to affect any of the Preferred Shares then outstanding; or
- (ii) authorize or issue any shares of any class, or any securities convertible into shares of any class, ranking prior to the Preferred Shares as to dividends or assets; or
- (iii) reclassify any shares of any class ranking junior to or on a parity with the Preferred Shares into shares of any other class ranking prior to the Preferred Shares; or
- (iv) issue any shares of any class ranking on a parity with the Preferred Shares, unless in either case (a) the aggregate of the par or stated value of the Common Shares to be outstanding immediately after such issue, plus the surplus of this Corporation, all determined in accordance with accepted accounting practice, shall be at least equal to the par or stated value of all shares which rank prior to the Common Shares and which are to be outstanding immediately after such issue; (b) the net earnings of this Corporation, computed in accordance with accepted accounting practice (but after provision for all taxes based upon or measured by income, and after annual interest charges adjusted by provision for amortization of bond discount and expense or of premium on indebtedness, and also after deduction of depreciation as reported in the accounts of this Corporation as filed with the Public Utilities Commission of the State of California or other public authority of said State having jurisdiction to establish or approve the system of accounts of this Corporation), for a period of 12 consecutive calendar months out of the 15 calendar months immediately preceding the date of such issue shall have been at least equal to twice the aggregate of the annual dividend requirements on all shares of this Corporation which rank prior to the Common Shares and which are to be outstanding immediately after such issue, and (c) the net earnings of this Corporation, computed as above (but before interest charges as aforesaid and after deduction for depreciation and provision for all taxes as provided above), for said 12 months period, shall have been at least equal to one and one-half

times the aggregate of all such interest charges and the annual dividend requirements on all shares of this Corporation which rank prior to the Common Shares and which are to be outstanding immediately after such issue.

(9) Without the approval of the holders of a majority of the outstanding Preferred Shares, given in person or by proxy, by written consent or by vote as provided by law, this Corporation shall not (i) issue, assume, or guarantee any unsecured notes or obligations unless immediately thereafter the total principal amount of the unsecured indebtedness of this Corporation shall be less than 10% of the aggregate of the total principal amount of outstanding bonds or other securities representing secured indebtedness issued, assumed or guaranteed by this Corporation, plus its stated capital and surplus; provided, however, that the foregoing provisions of this Paragraph (9) shall not apply to any such notes or obligations which (a) represent liabilities incurred in the ordinary course of business or for construction or acquisition of capital assets or represent tax liability or liability incurred or accrued on account of customers' deposits, or (b) are issued to extend, renew, redeem or refund outstanding indebtedness of this Corporation in principal amount not less than the principal amount of such notes or obligations, or are issued to redeem or refund then outstanding Preferred Shares which have an aggregate par or stated value at least equal to the aggregate principal amount of such notes or obligations, (ii) sell, convey, lease or otherwise dispose of all or substantially all of its assets, property or business, or consolidate or merge with or into any other corporation.

(10) The approval of the holders of outstanding Preferred Shares which in any case may be required by the foregoing Paragraphs (8) and (9) for the taking of any action referred to in any of said paragraphs shall be in addition to any such approval of shareholders of this Corporation as may at the time be required by the laws of the State of California with respect to such action and no such action shall be taken without compliance with such laws of said State as are in effect at the time of taking of any such action.

(11) For the purposes of Paragraphs (8) (iv) and (9) (i) hereof, the certificate or opinion of any independent certified or public accountant of recognized standing (who may be the accountant regularly retained by this Corporation), selected in good faith by the Board of Directors, shall be conclusive with respect to all questions of fact therein required to be determined.

(12) Subject to the dividend preferences provided for herein for all shares of each other class at the time outstanding and to the restrictions set forth above and in this Paragraph (12), the Common Shares shall be entitled to receive dividends when and as declared by the Board of Directors out of any funds of this Corporation legally available therefor. After payment of the full preferential amounts hereinabove provided for all shares of each other class outstanding at the time of any liquidation, dissolution or winding up of this Corporation, whether voluntary or involuntary, all then remaining assets of this Corporation available for distribution to its shareholders shall be distributed ratably upon the Common Shares. No dividend shall be declared on the Common Shares which, after giving effect to such declaration, would reduce the Common Stock Equity of this Corporation as of the end of the calendar month last preceding that in which such dividend was declared to an amount less than 25% of the Total Capitalization of this Corporation as of the end of said last preceding month, except that any such dividend may be declared (a) which would reduce such Common Stock Equity to less than 25% but not less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during the 12 months period terminating at the end of such last preceding calendar month shall not exceed 75% of the net income of this Corporation applicable to its Common Shares for such period, or (b) which would reduce such Common Stock Equity to less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during said 12 months period shall not exceed 50% of the net income of this Corporation applicable to its Common Shares for such period; provided, however, that the foregoing restrictions of this sentence shall not apply to, nor in any way restrict, (a) the payment of any dividend on the Common Shares which is payable in shares of stock of this Corporation, or (b) any reclassification, subdivision, split-up or combination of the Common Shares, or (c) any transfer between the capital and surplus accounts of this Corporation in connection with any such reclassification, subdivisions, split-up or combination or payment of dividend in shares of stock of this Corporation. Common Stock Equity as herein used shall mean the aggregate of (i) par value or stated capital of all outstanding Common Shares, and (ii) the surplus including capital surplus, paid-in surplus and earned surplus as shown by the books of this Corporation after giving effect to the declaration of the proposed dividend, and (iii) premium on Common Shares, less the remaining balance of the amount of organization expenses, as shown on said books. Total Capitalization as herein used shall mean the aggregate of (i) Common Stock Equity, (ii) premium on and the par value or stated capital of all outstanding shares of this

Corporation of any and all classes having preferences over the Common Shares as to dividends or assets, and (iii) the principal amount of all outstanding debt maturing more than 12 months after the close of said 12 months period, all as shown by the books of this Corporation; less the remaining balance of organization expenses, as shown on said books. Net Income as herein used shall be determined in accordance with accepted accounting practice (but after provision for all taxes based upon or measured by income, and after annual interest charges adjusted by provision for amortization of bond discount and expense or of premium on indebtedness, and also after deduction of depreciation for said 12 months period as reported in the accounts of this Corporation as filed with the Public Utilities Commission of the State of California or other public authority of said state having jurisdiction to establish or approve the system of accounts of this Corporation). Net Income applicable to Common Shares as herein used shall mean net income after deduction therefrom of all dividends payable for the period involved on all outstanding shares of any and all classes of this Corporation having preference over the Common Shares as to dividends or assets.

(13) No Preferred Shares of this Corporation which have been reacquired in any manner by this Corporation after the original issue thereof shall ever again be reissued and all such shares so reacquired shall upon such reacquisition cease to be a part of the authorized shares of this Corporation.

(14) Unless such action has been approved by the affirmative vote of at least a majority of the Continuing Directors (as defined below), without the approval of Common Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares representing in the aggregate at least 66 2/3% of the combined voting power of this Corporation's outstanding Common Shares and the New Preferred Shares, voting together as a single class, this Corporation shall not

- subject to subparagraph (iii) below, sell, convey, lease or otherwise dispose of all or substantially all of its assets, property or business;
- (ii) approve the sale, conveyance, lease or other disposition by any subsidiary of this Corporation of all or substantially all of such subsidiary's assets, property or business;
- (iii) sell, transfer, convey or otherwise dispose of more than a majority of the outstanding capital stock of any subsidiary of the Corporation, if such subsidiary holds assets accounting for 50% or more of the Corporation's consolidated assets, other than to an entity the majority of the voting power of the capital stock or other equity interest of which is owned and controlled by this Corporation;
- (iv) consolidate or merge with or into any other corporation or other business entity, except if, immediately after such consolidation or merger, the shareholders of this Corporation immediately prior to such consolidation or merger will own more than 60% of the voting power of the outstanding capital stock or other equity interest of or in the surviving entity; or
- (v) approve the consolidation or merger of any subsidiary of this Corporation, if such subsidiary holds assets accounting for 50% or more of the Corporation's consolidated assets, with or into any other corporation or other business entity.

For purposes of this paragraph (4) of Article IV, the term "Continuing Directors" shall mean any member of the Board of Directors of the Corporation shall mean any member of the Board of Directors of the Corporation (while such person is a member of the Board) who (i) is not an Acquiring Person, or an Affiliate or Associate of an Acquiring Person, or a representative of an Acquiring Person or of any such Affiliate or Associate, and (ii) either (A) was a member of the Board of Directors prior to the time any person became an Acquiring Person, or (B) became a member of the Board of Directors subsequent to the time any person became an Acquiring Person, if such person's nomination for election, or re-election, to the Board was recommended, or approved, by a majority of the Continuing Directors then in office. For purposes of the foregoing definition, (i) "Affiliate" and "Associate" shall have the respective meanings ascribed to such terms in Rule 12b-2 of the General Rules and Regulations under the Exchange Act, as in effect as of the date hereof; (ii) "Acquiring Person" shall mean any person or entity which, alone or together with all Affiliates and Associates of such person or entity, shall be the beneficial owner of 20% or more of the

5

Corporation's voting stock, but shall not include (1) an Exempt Person or (2) any person or entity who or which acquires 20% or more of the Corporation's voting stock in connection with a transaction or series of transactions approved prior to such transaction or transactions by the Board of Directors of the Corporation; provided that no person or entity shall become an Acquiring Person solely as a result of a reduction in the number of shares of the Corporation's voting stock outstanding, unless and until such person or entity shall thereafter become the beneficial owner of additional shares constituting 1% or more of the general voting power of the Corporation. "Exempt Person" shall mean the Corporation, any majority-owned subsidiary of the Corporation, and any employee benefit plan or employee stock plan of the Corporation, or any trust or other entity organized, established or holding Common Shares by, for or pursuant to, the terms of any such plan.

(15) Another series of Preferred Shares shall have the following terms and provisions:

(i) Designation. The designation of said series shall be "Preferred Shares, 5% Series."

(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares. 5% Series, shall be 19,200 shares.

(iii) Dividend Rate. The dividend rate of said Preferred Shares, 5% Series, shall be. pet share. 5% per annum of the share par value.

(iv) Optional Redemption. The redemption prices of the shares of said series, when redeemed by this Corporation at the option of its Board of Directors, shall be an amount per share equal to the par value thereof and unpaid dividends accrued thereon to and including the date fixed for redemption, plus a premium of 5.25 per share.

(v) Sinking Fund for Mandatory Purchases or Redemptions.

(a) So long as any of the Preferred Shares, 5% Series, shall be outstanding. this Corporation. as a sinking fund for the purchase or redemption thereof (hereinafter called the "Sinking Fund"), shall set aside in cash out or any moneys legally available therefor. after full payment or provision jar payment of dividends on all outstanding Preferred Shares of all series and all other shares of this Corporation ranking prior to or on a parity with the Preferred Shares for all prior periods through the end of the last preceding guarterly dividend period for such Preferred Shares and such other shares, on September 5 of each year (hereinafter called the "Sinking Fund payment date"), a sum equal to two percent (2%) of the aggregate par value of the total number of Preferred Shares, 5% Series, theretofore issued. If on any Sinking Fund payment date the Funds of this Corporation legally available therefor shall be insufficient to discharge in full the Sinking Fund requirement then accrued, funds to the extent legally available for such purpose shall be set aside for the Sinking Fund, Such Sinking Fund requirements shall be cumulative so that if for any year or years such requirements shall not be fully discharged as they accrue, funds legally available therefor. after such payment or provision for dividends, for each fiscal year thereafter shall be applied thereto until such requirements are fully discharged.

(b) This Corporation at its option shall be entitled to use as a credit against its Sinking Fund requirement for any year. in an amount equal to the par value thereof. Preferred Shares. 5% Series. which this Corporation shall have theretofore acquired by purchase or redemption. otherwise than through the operation of the Sinking Fund. and for which credit shall not therefore have been taken against any Sinking Fund requirement.

(c) On or before the 60th day next following each Sinking Fund payment date, the cash in the Sinking Fund shall be used to acquire Preferred Shares. 591 Series, by purchase, at a price or prices not exceeding the par value thereof, or by redemption at the par value thereof in the manner provided in Paragraph 6 of Article IV to the Articles of Incorporation of this Corporation, in each case plus an additional amount equal to accrued dividends thereon to the date of such purchase or redemption, which additional amount shall be paid from general fund. If this Corporation legally available therefor and not from the Sinking Fund, or by both such purchase and such redemption. Upon retirement of all Preferred Shares, 5% Series. any cash remaining in the Sinking Fund in excess of that required to complete payment for any shares purchased or agreed to be purchased, or to redeem shares called for redemption through the operation of the Sinking Fund, shall become a part of the general funds of this Corporation.

## ARTICLE V

The liability of the directors of this Corporation for monetary damages shall be eliminated to the fullest extent permissible under California law.

This Corporation is authorized to provide indemnification of agents (as defined in Section 317 of the California Corporations Code) through bylaw provisions, agreements with agents, vote of shareholders or disinterested directors, or otherwise, in excess of the indemnification otherwise permitted by Section 317 of the California Corporations Code, subject only to the applicable limits set forth in Section 204 of the California Corporations Code.

## ARTICLE VI

Notwithstanding any contrary provision of these Articles of Incorporation, any amendment or repeal of paragraph (14) of Article IV. this Article VI or any amendment to these Articles of Incorporation providing for a classified board shall require the affirmative vote of shares representing not less than 66 2/3% of the combined voting power of the outstanding Common Shares. Preferred Shares and, unless otherwise provided in the certificate of determination, the New Preferred Shares, voting together as a single class.

Notwithstanding any contrary provision of these Articles of Incorporation, and except as otherwise expressly provided in the California Corporations Code, none of the following provisions of the Bylaws of the Company may be amended or repealed, except by a majority of the Board or by the shareholders upon the affirmative vote of shares representing at least 66 2/3% of the combined voting power of the outstanding Common Shares. Preferred Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares, voting together as a single class (a) Section 2 of Article II, in Section 15 of Article II, and (c) Section 2 of Article III.

> /s/ Brian Thomas Daly Brian Thomas Daly Incorporator

7

Mar 04 1998

/s/ Bill Jones BILL JONES, Secretary of State

2070444 CERTIFICATE OF AMENDMENT OF ARTICLES OF INCORPORATION OF AMERICAN STATES WATER COMPANY (a California corporation) As of March4, 1998

The undersigned certifies (i) that he is the incorporator of American States Water Company (the "Corporation"). (ii) that no directors of the corporation were named in the Articles of Incorporation or have been elected, and (iii) that no shares of the Corporation have been issued, and hereby takes the following action as of the above date:

The Articles of Incorporation of the Corporation are hereby amended and restated in full in the form attached hereto as Exhibit A.

The undersigned also declares under penalty of perjury under the laws of the State of California that he has read the the foregoing provisions of this certificate and knows the contents thereof and that the same is true of his own knowledge.

Dated as of: March 4, 1998 /s/ Brian Thomas Daly

Exhibit A

Amended and Restated Articles of Incorporation of American States Water Company

# AMENDED AND RESTATED ARTICLES OF INCORPORATION OF

# AMERICAN STATES WATER COMPANY

# ARTICLE I

# The name of this Corporation is American States Water Company.

#### ARTICLE II

The purpose of this Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of California other than the banking business the trust business or the practice of a profession permitted to be incorporated by the California Corporations Code.

#### ARTICLE III

The name and address of this Corporation's initial agent for service of process is McClellan Harris III, 630 East Foothill Boulevard, San Dimas, California 91773.

#### ARTICLE IV

The Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,233,2000; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per shares and an aggregate stated value of \$15,000,000; 83,200 shares are to 30,000,000 shares are to be Common Shares, no par value with a stated value of \$2.50 per shares and an aggregate stated value of \$75,000,000.

A statement of the preferences, privileges and restrictions granted to or imposed upon the respective classes or series of shares and/or upon the holder thereof is as follows:

(1) Subject to the provisions of this Article IV, New Preferred Shares of any particular series shall be entitled to such voting rights, if any, as may be specified for shares of such series in the certificate of determination of preferences of such series filed as provided below; all Preferred Shares shall be entitled to voting rights on the basis of one-tenth of one vote per share.

(2) New Preferred Shares may be issued from time to time in one or more series. Each such series shall be so designated as to distinguish it from other series of New Preferred Shares and from series of Preferred Shares and from series of Preferred Shares. Such designation may include an appropriate reference to the dividend rate and/or any other characteristics of such series. The Board of Director is hereby authorized, within the limits of, but to the extent authorized by applicable law and within the limitations and restrictions, if any, stated in this Articles IV, to fix or alter, from time to time, the dividend rights, dividend rate, conversion rights, voting rights, right and terms of redemption (including sinking fund provisions), redemption price or prices, and liquidation preferences, or any of them, of any wholly unissued series, by a resolution or resolutions adopted by the Board of Directors in exercise of the authority hereby granted.

(3) The Preferred Shares may be issued from time to time in any number of series. One such series shall (i) be and hereby is designated the "4 1/4% Series", (ii) consist of 32,000 shares, (iii) be entitled to dividends as provided in Paragraph (4) hereof at the rate of 4 1/4% per annum of the par value thereof, and (iv) be redeemable in the manner and otherwise upon the conditions provided in Paragraph (6) hereof by payment of a redemption price equal to the par value thereof and unpaid dividends accrued thereon to and including the date fixed for such redemption and premium of \$1.500 per share. Another such series of Preferred Shares shall (i) be and hereby is designated as the "4% Series", (ii) consist of 32,000 shares, (iii) be entitled to dividends as provided in Paragraph (4) hereof at the rate of 4% per annum of the par value thereof, and (iv) be redeemable in the manner and otherwise upon the conditions provided in Paragraph (6) hereof by payment of a redemption price equal to the par value thereof and unpaid dividends accrued thereon to and including the date fixed for such redemption and a premium of \$2 per share. Except as to the foregoing particulars no distinction shall

exist between any of the Preferred Shares or any series thereof, and all Preferred Shares, regardless of series, shall be of equal rank and priority.

(4) The holders of the outstanding shares of the several and respective series of Preferred Shares shall be entitled to receive, out of any funds legally available therefore, dividends at the respective rates for the shares of said series, payable in cash quarterly on the first day of March, June, September and December in each year when and as declared by the Board of Directors of this Corporation. Such dividends shall accrue on each such share from the date of its original issuance and shall accrue from day to day whether or not earned or declared. Such dividends shall be cumulative so that if such dividends in respect of any quarterly dividend period at the respective rates fixed therefore shall not have been paid on, or declared and set apart for, all Preferred Shares of each series at the time outstanding, the deficiency shall be fully paid on or declared or set apart for such shares before any dividend or other distribution shall be paid upon or declared or set apart for the Common Shares. No such dividend shall be declared or paid upon or set apart for any outstanding shares of any one of said series unless at the same time such dividends on all outstanding shares of any one of said series shall be declared and paid in full or set apart for such payments.

(5) In the event of the liquidation, dissolution or winding up of this Corporation, whether voluntary or involuntary, the holders of the shares of the several and respective series of preferred Shares shall be entitled to receive out of the assets of this Corporation, whether such assets are capital or surplus of any nature, an amount equal to the par value thereof plus all unpaid dividends accrued thereon to the date that such amount is made available for distribution to the holders thereof, and no more, or ratably from available assets if such assets are insufficient to permit payment to said holders of their full preferential amount aforesaid. Such amount shall be paid upon said shares, or shall be set apart for such payment, before any distribution is made or set apart for any Common Shares in any such liquidation, dissolution or winding up. A consolidation or merger of this Corporation with or into any other dorporation or winding up within the meaning of this Paragraph (5).

(6) This Corporation, at the option of its Board of Directors, may at any time or from time to time redeem the whole or any part of the outstanding shares of any one or more series of the Preferred Shares by paying in cash therefore the amount payable upon the redemption thereof (such amount being hereinafter referred to as the "redemption price"). In case of the redemption of a part only of the outstanding shares of any series of Preferred Shares, this Corporation shall designate by lot, in such manner as the Board of Directors may determine, the shares to be redeemed. At least thirty (30) days previous notice by mail, postage prepaid, shall be given to the holders of record of the shares to be redeemed, such notice to be addressed to each such shareholder at his post office address as shown by the records of this Corporation at the opening of business on the day of mailing such notice. If on or before the date fixed for redemption and specified in such notice funds necessary for such redemption therefore, then notwithstanding that the certificates evidencing any shares called for such redemption shall not have been surrendered, the dividends with respect to the shares so called for redemption shall cease to accrue after the said date fixed for redemption and all rights with respect to such shares shall forthwith after said date cease and determine except only the right of the holders thereof to receive payment of the redemption price, without interest, upon surrender of their certificates representing the redeemed shares. In case less than all the shares represented by any such surrendered certificate shall have been redeemed, a new certificate shall be issued for the unredeemed shares. Subject to the provisions hereof, the Board of Directors shall have authority to prescribe from time to time the manner in which the Preferred Shares shall be redeemed.

(7) If at any time four (4) quarterly dividends (whether or not consecutive) which have accrued on the outstanding Preferred Shares pursuant to Paragraph (4) hereof shall be in arrears, then at the annual meeting of shareholders next following the fourth such quarterly dividend default, or, if such next annual meeting is not to be held within sixty (60) days following such default, at a special meeting of shareholders called for the purpose on the written request of the holders of not less than ten percent (10%) of the then outstanding Preferred Shares, the holders of said outstanding Preferred Shares shall be entitled, voting separately as a class (regardless of series), to elect the smallest number of directors of this Corporation which shall constitute a majority of the authorized number of such directors and the holders of the New Preferred Shares and the holders of the Common Shares, in accordance with their respective voting rights shall have accrued for any period under said Paragraph (4) upon the

outstanding Preferred Shares shall have been paid or set apart for payment. At all meetings of shareholders held for the purpose of electing directors during such time as the holders of the Preferred Shares have the right, voting separately as a class, to elect directors pursuant to this Paragraph (7), the presence in person or by proxy of the holders of a majority of the outstanding Preferred Shares (regardless of series), shall be required to constitute a quorum of such class for the election of directors, and the presence in person or by proxy of the holders of a majority of the other outstanding shares entitled to vote at such meeting, as a separate class or classes in accordance with their respective voting rights, shall be required to constitute a quorum of such class for the election of directors; provided, however, that the absence of a quorum of the holders of outstanding shares of either such class or classes shall not prevent the election at any such meeting, or adjournment thereof, of directors by the other such class or classes if the necessary quorum of such other class or classes is present in person or by proxy at such meeting or adjournment; and provided further that in the event that at such meeting or adjournment such a quorum of the holders of classes other than the Preferred Shares is present and such guorum of the holders of the Preferred Shares is not present, then an election of the directors elected at such meeting or adjournment by the holders of classes other than the Preferred Shares shall not assume office until the holders of the Preferred Shares, with such a quorum present shall elect the directors they are entitled to elect. In the event of an election of directors by holders of Preferred Shares pursuant to this Paragraph (7), the term of office as directors of all persons who are directors of this Corporation at the time of the accrual under this Paragraph (7) of the right of the holders of the Preferred Shares to elect directors shall terminate when the holders of the outstanding Preferred Shares shall have so elected directors. In case any vacancy shall occur among the directors elected as aforesaid by the holders of the Preferred Shares, or among the directors elected as aforesaid by the holders of classes other than the Preferred Shares, during any period for the which a majority of the directors shall have been so elected by the holders of the Preferred Shares, such vacancy shall be filled by the vote of a majority of the remaining directors who were so elected by the holders of the Preferred Shares or by the holders of classes other than the Preferred Shares, as the case may be.

(8) Without the approval of the holders of at least two-thirds of the outstanding Preferred Shares, given in person or by proxy, either by written consent or by vote as provided by law, this Corporation shall not

- alter or amend the preferences, voting powers or other special rights or the qualifications, limitations and restrictions imposed in favor of any of the Preferred Shares so as adversely to affect any of the Preferred Shares then outstanding; or
- (ii) authorize or issue any shares of any class, or any securities convertible into shares of any class, ranking prior to the Preferred Shares as to dividends or assets; or
- (iii) reclassify any shares of any class ranking junior to or on a parity with the Preferred Shares into shares of any other class ranking prior to the Preferred Shares; or
- (iv) issue any shares of any class ranking on a parity with the Preferred shares, unless in either case (a) the aggregate of the par or stated value of the Common Shares to be outstanding immediately after such issue, plus the surplus of this Corporation, all determined in accordance with accepted accounting practice, shall be at least equal to the par or stated value of all shares which rank prior too the Common Shares and which are to be outstanding immediately after such issue; (b) the net earnings of this Corporation, computed in accordance with accepted accounting practice (but after provision for all taxes based upon or measured by income, and after annual interest charges adjusted by provision for amortization of bond discount and expense or of premium on indebtedness, and also after deduction of depreciation as reported in the accounts of this Corporation as filed with the Public Utilities Commission of the State of California or other public authority of said State having jurisdiction to establish or approve the system of accounts of this Corporation), for a period of 12 consecutive calendar months out of the 15 calendar months immediately preceding the date of such issue shall have been at least equal to twice the aggregate of the annual dividend requirements on all shares of this Corporation which rank prior to the Common Shares and which are to be outstanding immediately after such issue, and (c) the net earnings of this Corporation, computed as above (but before interest charges as aforesaid and after deduction for depreciation and provision for all taxes as provided above), for said 12 months period, shall have been at least equal to one and one-half

times the aggregate of all such interest charges and the annual dividend requirements on all shares of this Corporation which rank prior to the Common Shares and which are to be outstanding immediately after such issue.

(9) Without the approval of the holders of a majority of the outstanding Preferred Shares, given in person or by proxy, by written consent or by vote as provided by law, this Corporation shall not (i) issue, assume, or guarantee any unsecured notes or obligations unless immediately thereafter the total principal amount of the unsecured indebtedness of this Corporation shall be less than 10% of the aggregate of the total principal amount of outstanding bonds or other securities representing secured indebtedness issued, assumed or guaranteed by this Corporation, plus its stated capital and surplus; provided, however, that the foregoing provisions of this Paragraph (9) shall not apply to any such notes or obligations which (a) represent liabilities incurred in the ordinary course of business or for construction or acquisition of capital assets or represent tax liability or liability incurred or accrued on account of customers' deposits, or (b) are issued to extend, renew, redeem or refund outstanding indebtedness of this Corporation in principal amount not less than the principal amount of such notes or obligations, or are issued to redeem or refund then outstanding Preferred Shares which have an aggregate par or stated value at least equal to the aggregate principal amount of such notes or obligations, (ii) sell, convey, lease or otherwise dispose of all substantially all of its assets, property or business, or consolidate or merge with or into any other corporation.

(10) The approval of the holders of outstanding Preferred Shares which in any case may be required by the foregoing Paragraphs (8) and (9) for the taking of any action referred to in any of said paragraphs shall be in addition to any such approval of shareholders of this Corporation as may at the time be required by the laws of the State of California with respect to such action and no such action shall be taken without compliance with such laws of said State as are in effect at time of taking of any such action.

(11) For the purposes of Paragraphs (8)(iv) and (9)(i) hereof, the certificate or opinion of any independent certified or public accountant of recognized standing (who may be the accountant regularly retained by this Corporation), selected in good faith by the Board of Directors, shall be conclusive with respect to all questions of fact therein required to be determined.

(12) Subject to the dividend preferences provided for herein for all shares of each other class at the time outstanding and to the restrictions set forth above and in this Paragraph (12), the Common shares shall be entitled to receive dividends when and as declared by the Board of Directors out of any funds of this Corporation legally available therefore. After payment of the full preferential amounts hereinabove provided for all shares of each other class outstanding at the time of any liquidation, dissolution or winding up of this Corporation, whether voluntary or involuntary, all then remaining assets of this Corporation available for distribution to its shareholders shall be distributed ratably upon the Common Shares. No dividend shall be declared on the Common shares which, after giving effect to such declaration, would reduce the Common Stock Equity of this Corporation as of the end of the calendar month last preceding that in which such dividend was declared to an amount less than 25% of the Total Capitalization of this Corporation as of the end of said last preceding month, except that any such dividend may be declared (a) which would reduce such Common Stock Equity to less than 25% but not less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during the 12 months period terminating at the end of such last preceding calendar month shall not exceed 75% of the net income of this Corporation applicable to its Common Shares for such period, or (b) which would reduce such Common Stock Equity to less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during said 12 months period shall not exceed 50% of the net income of this Corporation applicable to its Common Shares for such period; provided, however, that the foregoing restrictions of this sentence shall not apply to, nor in any way restrict, (a) the payment of any dividend on the Common Shares which is payable in shares of stock of this Corporation, or (b) any reclassification, subdivision, split-up or combination of the Common Shares, or (c) any transfer between the capital and surplus accounts of this Corporation in connection with any such reclassification, subdivisions, split-up or combination or payment of dividend in shares of stock of this Corporation. Common Stock Equity as herein used shall mean the aggregate of (i) par value or stated capital of all outstanding Common Shares, and (ii) the surplus including capital surplus, paid-in surplus and earned surplus as shown by the books of this Corporation after giving effect to the declaration of the proposed dividend, and (iii) premium on Common Shares, less the remaining balance of the amount of organization expenses, as shown on said books. Total Capitalization as herein used shall mean the aggregate of (i) Common Stock Equity, (ii) premium on and the par value or stated capital of all outstanding shares of this

Corporation of any and all classes having preferences over the Common Shares as to dividends or assets, and (iii) the principal amount of all outstanding debt maturing more than 12 months after the close of said 12 months period, all as shown by the books of this Corporation; less the remaining balance of organization expenses, as shown on said books. Net Income as herein used shall be determined in accordance with accepted accounting practice (but after provision for all taxes based upon or measured by income, and after annual interest charges adjusted by provision for amortization of bond discount and expense or of premium on indebtedness, and also after deduction of depreciation for said 12 months period as reported in the accounts of this Corporation as filed with the Public Utilities Commission of the State of California or other public authority of said state having jurisdiction to establish or approve the system of accounts of this Corporation). Net Income applicable to Common Shares as herein used shall mean net income after deduction therefrom of all dividends payable for the period involved on all outstanding shares of any and all classes of this Corporation having preference over the Common Shares as to dividends or assets.

(13) No Preferred Shares of this Corporation which have been reacquired in any manner by this Corporation after the original issue thereof shall ever again be reissued and all such shares so reacquired shall upon such reacquisition cease to be a part of the authorized shares of this Corporation.

(14) Unless such action has been approved by the affirmative vote of at least a majority of the Continuing Directors (as defined below), without the approval of Common Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares representing in the aggregate at least 66 2/3% of the combined voting power of this Corporation's outstanding Common Shares and the New Preferred Shares, voting together as a single class, this Corporation shall not

- subject to subparagraph (iii) below, sell, convey, lease or otherwise dispose of all or substantially all of its assets, property or business;
- (ii) approve the sale, conveyance, lease or other disposition by any subsidiary of this Corporation of all or substantially all of such subsidiary's assets, property or business;
- (iii) sell, transfer, convey or otherwise dispose of more than a majority of the outstanding capital stock of any subsidiary of the Corporation, if such subsidiary holds assets accounting for 50% or more of the Corporation's consolidated assets, other than to an entity the majority of the voting power of the capital stock or other equity interest of which is owned and controlled by this Corporation;
- (iv) consolidate or merge with or into any other corporation or other business entity, except if, immediately after such consolidation or merger, the shareholders of this Corporation immediately prior to such consolidation or merger will own more than 60% of the voting power of the outstanding capital stock or other equity interest of or in the surviving entity; or
- (v) approve the consolidation or merger of any subsidiary of this Corporation, if such subsidiary holds assets accounting for 50% or more of the Corporation's consolidated assets, with or into any other corporation or other business entity.

For purposes of this paragraph (4) of Article IV, the term "Continuing Directors" shall mean any member of the Board of Directors of the Corporation (while such person is a member of the Board) who (i) is not an Acquiring Person, or an Affiliate or Associate of an Acquiring Person, or a representative of an Acquiring Person or of any such Affiliate or Associate, and (ii) either (A) was a member of the Board of Directors prior to the time any person became an Acquiring Person, or (B) became a member of the Board of Directors subsequent to the time any person became an Acquiring Person, if such person's nomination for election, or re-election, to the Board was recommended, or approved, by a majority of the Continuing Directors then in office. For purposes of the foregoing definition, (i) "Affiliate" and "Associate" shall have the respective meanings ascribed to such terms in Rule 12b-2 of the General Rules and Regulations under the Exchange Act, as in effect as of the date hereof; (ii) "Acquiring Person" shall mean any person or entity which, alone or together with all Affiliates and Associates of such person or entity, shall be the beneficial owner of 20% or more of the

Corporation's voting stock, but shall not include (1) an Exempt Person or (2) any person or entity who or which acquires 20% or more of the Corporation's voting stock in connection with a transaction or series of transactions approved prior to such transaction or transactions by the Board of Directors of the Corporation; provided that no person or entity shall become an Acquiring Person solely as a result of a reduction in the number of shares of the Corporation's voting stock outstanding, unless and until such person or entity shall thereafter become the beneficial owner of additional shares constituting 1% or more of the general voting power of the Corporation, "Exempt Person" shall mean the Corporation, any majority-owned subsidiary of the Corporation, or any trust or other entity organized, established or holding Common Shares by, for or pursuant to, the terms of any such plan.

(15) Another series of Preferred Shares shall have the following terms and provisions:

(i) Designation. The designation of said series shall be "Preferred Shares, 5% Series."

(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares. 5% Series, shall be 19,200 shares.

(iii) Dividend Rate. The dividend rate of said Preferred Shares, 5% Series, shall be. pet share. 5% per annum of the share par value.

(iv) Optional Redemption. The redemption prices of the shares of said series, when redeemed by this Corporation at the option of its Board of Directors, shall be an amount per share equal to the par value thereof and unpaid dividends accrued thereon to and including the date fixed for redemption, plus a premium of 5.25 per share.

(v) Sinking Fund for Mandatory Purchases or Redemptions.

(a) So long as any of the Preferred Shares, 5% Series, shall be outstanding. this Corporation. as a sinking fund for the purchase or redemption thereof (hereinafter called the "Sinking Fund"), shall set aside in cash out or any moneys legally available therefor. after full payment or provision jar payment of dividends on all outstanding Preferred Shares of all series and all other shares of this Corporation ranking prior to or on a parity with the Preferred Shares for all prior periods through the end of the last preceding quarterly dividend period for such Preferred Shares and such other shares, on September 5 of each year (hereinafter called the "Sinking Fund payment date"), a sum equal to two percent (2%) of the aggregate par value of the total number of Preferred Shares, 5% Series, theretofore issued. If on any Sinking Fund payment date the Funds of this Corporation legally available therefor shall be insufficient to discharge in full the Sinking Fund requirement then accrued, funds to the extent legally available for such purpose shall be set aside for the Sinking Fund, Such Sinking Fund requirements shall be cumulative so that if for any year or years such requirements shall not be fully discharged as they accrue, funds legally available therefor. after such payment or provision for dividends, for each fiscal year thereafter shall be applied thereto until such requirements are fully discharged.

(b) This Corporation at its option shall be entitled to use as a credit against its Sinking Fund requirement for any year. in an amount equal to the par value thereof. Preferred Shares. 5% Series. which this Corporation shall have theretofore acquired by purchase or redemption. otherwise than through the operation of the Sinking Fund. and for which credit shall not therefore have been taken against any Sinking Fund requirement.

(c) On or before the 60th day next following each Sinking Ftind payment date, the cash in the Sinking Fund shall be used to acquire Preferred Shares. 591 Series, by purchase, at a price or prices not exceeding the par value thereof, or by redemption at the par value thereof in the manner provided in Paragraph 6 of Article IV to the Articles of Incorporation of this Corporation, in each case plus an additional amount equal to accrued dividends thereon to the date of such purchase or redemption, which additional amount shall be paid from general fund. If this Corporation legally available therefor and not from the Sinking Fund, or by both such purchase and such redemption. Upon retirement of all Preferred Shares, 5% Series. any cash remaining in the Sinking Fund in excess of that required to complete payment for any shares purchased or agreed to be purchased, or to redeem shares called for redemption through the operation of the Sinking Fund, shall become a part of the general funds of this Corporation.

#### ARTICLE V

The liability of the directors of this Corporation for monetary damages shall be eliminated to the fullest extent permissible under California law.

This Corporation is authorized to provide indemnification of agents (as defined in Section 317 of the California Corporations Code) through bylaw provisions, agreements with agents, vote of shareholders or disinterested directors, or otherwise, in excess of the indemnification otherwise permitted by Section 317 of the California Corporations Code, subject only to the applicable limits set forth in Section 204 of the California Corporations Code.

# ARTICLE VI

Notwithstanding any contrary provision of these Articles of Incorporation, any amendment or repeal of paragraph (14) of Article IV. this Article VI or any amendment to these Articles of Incorporation providing for a classified board shall require the affirmative vote of shares representing not less than 66 2/3% of the combined voting power of the outstanding Common Shares. Preferred Shares and, unless otherwise provided in the certificate of determination, the New Preferred Shares, voting together as a single class.

Notwithstanding any contrary provision of these Articles of Incorporation, and except as otherwise expressly provided in the California Corporations Code, none of the following provisions of the Bylaws of the Company may be amended or repealed, except by a majority of the Board or by the shareholders upon the affirmative vote of shares representing at least 66 2/3% of the combined voting power of the outstanding Common Shares. Preferred Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares, voting together as a single class (a) Section 2 of Article II, tin Section 15 of Article II, and (c) Section 2 of Article III.

## ARTICLE VII

In the event that the authorized number of directors shall be fixed with at least six (6) but less than nine t9 during any period of time that the Common Shares are listed in the New York Stock Exchange. the Board of Directors shall be divided into two classes, designated Class I and Class II. Each class shall consist of one-half of the directors or as close an approximation as possible. The initial term of office of the directors of Class I shall commence on the date that the Common Shares arc listed on the New York Stock Exchange and shall expire at the annual meeting to be held during fiscal pear 1999 and the initial term of office of the directors of Class hi shall commence on the date that the Common Shares are listed on the New York Stock Exchange and shall expire at the annual meeting to be held during fiscal year 2000. At each subsequent annual meeting, each of the successors to the directors of the class whose term shall have expired at such annual meeting shall he elected for a term running until the second annual meeting next succeeding his or her election and until his or her successor shall have been duly elected and qualified, unless the Common Shares are no longer listed on the New York Stock Exchange.

In the event that the authorized number of directors shall be fixed at nine (9) or more during any period of time that the Common Shares are listed on the New York Stock Exchange. the Board of Directors shall be divided into three classes, designated Class I, Class II and Class III. Each class shall consist of one-third of the directors or as close an approximation as possible. At each subsequent annual meeting, each of the successors to the directors of the class whose term shall have expired at such annual meeting shall be elected for a term running until the third annual meeting next succeeding his or her election until his or her successor shall have been duly elected and qualified, unless the Common Shares are no longer listed on the New York Stock Exchange,

Notwithstanding the rule that the classes shall be as nearly equal in number of directors as possible. in the event of any change in the authorized number of directors, each director then continuing to serve as such, shall nevertheless continue as a director of the class of which he or she is a member until the expiration of his or her current term, or his or her prior death, resignation or removal.

At each annual election, the directors chosen to succeed those whose terms then expire shall be of the same class as the directors they succeed, unless, by reason of any intervening changes in the authorized number

of directors. the Board of Directors shall designate title or more directorships whose term then expires as directorships of another class in order more nearly to achieve equality of number of directors among the classes.

The effective date of the amendment adding this Article VII shall be that date the Common Shares are first listed on the New York Stock Exchange.

AUG 27 1998

/s/ Bill Jones BILL JONES, Secretary of State

2070444

CERTIFICATE OF AMENDMENT

of

AMENDED AND RESTATED ARTICLES OF INCORPORATION

and

# CERTIFICATE OF DETERMINATION

of

# JUNIOR PARTICIPATING PREFERRED STOCK

of

# AMERICAN STATES WATER COMPANY (A CALIFORNIA CORPORATION)

Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly acting President and Secretary, respectively, of American States Water Company (the "Corporation").

2. The Restated Articles of Incorporation of the Corporation shall be amended by striking in its entirety the first full paragraph of Article IV of the restated Articles of Incorporation which now reads:

This Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,233,200; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000; 83,200 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$2,080,000; and 30,000,000 shares are to be Common Shares, no par value with a stated value of \$2.50 per share and an aggregate stated value of \$75,000,000.

and substituting therefore the following paragraph to read in full as follows:

This Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,231,600; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000; 81,600 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$2,040,000; and 30,000,000 shares are to be Common Shares, no par value with a stated value of \$2.50 per share and an aggregate stated value of \$75,000,000.

3. The Restated Articles of Incorporation of the Corporation shall be further amended by striking in its entirety subparagraph (ii) of Paragraph 15 of Article IV of the Restated Articles of Incorporation which now reads:

(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares, 5% Series, shall be 19,200 shares.

and substituting therefor the following paragraph to read in full as follows:

(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares, 5% Series, shall be 17,600 shares.

4. The foregoing amendment set forth in paragraph 3 of this Certificate, as well as the portion of the amendment in paragraph 2 that relates to the "Preferred Shares" are each amendments that may be adopted by the Board of Directors alone (and which were so adopted) because the amendments are required by Section 510 of the California General Corporation Law to reflect the reacquisition of a portion of the Corporation's Preferred Shares, 5% Series, \$25 par value, in accordance with the sinking fund provisions thereof. Such reacquired Preferred Shares cannot be reissued.

5. The foregoing amendments have been duly approved by the Board of Directors as required by Section 905(b) of the California General Corporation Law.

6. The undersigned officers of the Corporation further certify that the following resolution has been duly adopted by the Board of Directors of the Corporation with regards to the Certificate of Determination of Junior Participating Preferred Stock of the Corporation:

RESOLVED, that pursuant to the authority granted to the Board of Directors of the Corporation by the Articles of Incorporation, a series of shares of the New Preferred Shares of the Corporation is hereby established and the number of shares constituting such series and the designation thereof, and the rights, preferences, privileges and restrictions of the shares of such series, are fixed and established as follows:

#### I. Designation and Amount

The shares of such series shall be designated as "Junior Participating Preferred Stock" (the "Junior Preferred Stock") and the number of shares constituting the Junior Preferred Stock shall be 8,958. Such number of shares may be increased or decreased by resolution of the Board of Directors; provided, that no decrease shall reduce the number of shares of Junior Preferred Stock to a number less than the number of shares then outstanding plus the number of shares reserved for issuance upon the exercise of outstanding options, rights or warrants or upon the conversion of any outstanding securities issued by the Corporation convertible into Junior Preferred Stock.

#### II. Dividends and Distributions

(A) Subject to the rights of the holders of any shares of any series of Preferred Shares or New Preferred Shares (or any similar stock) ranking prior and superior to the Junior Preferred Stock with respect to dividends, the holders of shares of Junior Preferred Stock, in preference to the holders of Common Shares of the Corporation, shall be entitled to receive, when, as and if declared by the Board of Directors out of funds legally available for the purpose, quarterly dividends payable in cash on the first day of March, June, September and December in each year (each such date being referred to herein as a "Quarterly Dividend Payment Date"), commencing on the first Quarterly Dividend Payment Date after the first issuance of a share or fraction of a share of Junior Preferred Stock, in an amount per share (rounded to the nearest cent) equal to the greater of (a) 1.00 or (b) subject to the provision for adjustment hereinafter set forth, 1,000 times the aggregate per share amount of all cash dividends, and 1,000 times the aggregate per share amount (payable in kind) of all non-cash dividends or other distributions, other than a dividend pavable in shares of Common Shares or a subdivision of the outstanding shares of Common Shares (by reclassification or otherwise), declared on the Common Shares since the immediately preceding Quarterly Dividend Payment Date or, with respect to the first Quarterly Dividend Payment Date, since the first issuance of any share or fraction of a share of Junior Preferred Stock. In the event the Corporation shall at any time declare or pay any dividend on the Common Shares payable in shares of Common Shares, or effect a subdivision or combination or consolidation of the outstanding shares of Common Shares (by reclassification or otherwise than by payment of a dividend in shares

of Common Shares) into a greater or lesser number of shares of Common Shares, then in each such case the amount to which holders of shares of Junior Preferred Stock were entitled immediately prior to such event under clause (b) of the preceding sentence shall be adjusted by multiplying such amount by a fraction, the numerator of which is the number of shares of Common Shares outstanding immediately after such event and the denominator of which is the number of shares of Common Shares that were outstanding immediately prior to such event.

(B) The Corporation shall declare a dividend or distribution on the Junior Preferred Stock as provided in paragraph (A) of this Section immediately after it declares a dividend or distribution on the Common Shares (other than a dividend payable in shares of Common Shares); provided that, in the event no dividend or distribution shall have been declared on the Common Shares during the period between any Quarterly Dividend Payment Date and the next subsequent Quarterly Dividend Payment Date, a dividend of \$1.00 per share on the Junior Preferred Stock shall nevertheless be payable on such subsequent Quarterly Dividend Payment Date.

(C) Dividends shall begin to accrue and be cumulative on outstanding shares of Junior Preferred Stock from the Quarterly Dividend Payment Date next preceding the date of issue of such shares, unless the date of issue of such shares is prior to the record date for the first Quarterly Dividend Payment Date, in which case dividends on such shares shall begin to accrue from the date of issue of such shares, or unless the date of issue is a Quarterly Dividend Payment Date or is a date after the record date for the determination of holders of shares of Junior Preferred Stock entitled to receive a quarterly dividend and before such Quarterly Dividend Payment Date, in either of which events such dividends shall begin to accrue and be cumulative from such Quarterly Dividend Payment Date. Accrued but unpaid dividends shall not bear interest. Dividends paid on the shares of Junior Preferred Stock in an amount less than the total amount of such dividends at the time accrued and payable on such shares shall be allocated pro rata on a share-by-share basis among all such shares at the time outstanding. The Board of Directors may fix a record date for the determination of holders of shares of Junior Preferred Stock entitled to receive payment of a dividend or distribution declared thereon, which record date shall be not more than 60 days prior to the date fixed for the payment thereof.

#### III. Voting Rights

The holders of shares of Junior Preferred Stock shall have the following voting rights:

4

(A) Subject to the provision for adjustment hereinafter set forth, each share of Junior Preferred Stock shall entitle the holder thereof to 100 votes on all matters submitted to a vote of the shareholders of the Corporation.

(B) Except as otherwise provided herein, or in any other resolutions of the Board creating a series of New Preferred Shares or any similar stock, or by law, the holders of shares of Junior Preferred Stock and the holders of shares of Common Shares, Preferred Shares and any other capital stock of the Corporation having general voting rights shall vote together as one class on all matters submitted to a vote of shareholders of the Corporation.

(C) Except as set forth herein, in the Corporation's Articles of Incorporation or as otherwise provided by law, holders of Junior Preferred Stock shall have no voting rights.

#### IV.Certain Restrictions

(A) Whenever quarterly dividends or other dividends or distributions payable on the Junior Preferred Stock as provided in Section II are in arrears, thereafter and until all accrued and unpaid dividends and distributions, whether or not declared, on shares of Junior Preferred Stock outstanding shall have been paid in full, the Corporation shall not:

 (i) declare or pay dividends, or make any other distributions, on any shares of stock ranking junior (either as to dividends or upon liquidation, dissolution or winding up) to the Junior Preferred Stock;

(ii) declare or pay dividends, or make any other distributions, on any shares of stock ranking on a parity (either as to dividends or upon liquidation, dissolution or winding up) with the Junior Preferred Stock, except dividends paid ratably on the Junior Preferred Stock and all such parity stock on which dividends are payable or in arrears in proportion to the total amounts to which the holders of all such shares are then entitled;

(iii) redeem or purchase or otherwise acquire for consideration shares of any stock ranking junior (either as to dividends or upon liquidation, dissolution or winding up) to the Junior Preferred Stock, provided that the Corporation may at any time redeem, purchase or otherwise acquire shares of any such junior stock in exchange for shares of any stock of the Corporation ranking junior (either as to dividends or upon dissolution, liquidation or winding up) to the Junior Preferred Stock; or

(iv) redeem or purchase or otherwise acquire for consideration any shares of Junior Preferred Stock, or any shares of stock ranking on a parity with the Junior Preferred Stock, except in accordance with a purchase offer made in writing or by publication (as determined by the Board of Directors) to all holders of such shares upon such terms as the Board of Directors, after consideration of the respective annual dividend rates and other relative rights and preferences of the respective series and classes, shall determine in good faith will result in fair and equitable treatment among the respective series or classes.

(B) The Corporation shall not permit any subsidiary of the Corporation to purchase or otherwise acquire for consideration any shares of stock of the Corporation unless the Corporation could, under paragraph (A) of this Section IV purchase or otherwise acquire such shares at such time and in such manner.

#### V. Reacquired Shares

Any shares of Junior Preferred Stock purchased or otherwise acquired by the Corporation in any manner whatsoever shall be retired and cancelled promptly after the acquisition thereof. All such shares shall upon their cancellation become authorized but unissued shares of New Preferred Shares and may be reissued as part of a new series of New Preferred Shares subject to the conditions and restrictions on issuance set forth herein, in the Articles of Incorporation, in any other Certificate of Determination creating a series of New Preferred Shares or any similar stock or as otherwise required by law.

#### VI. Liquidation, Dissolution or Winding Up

Upon any liquidation, dissolution or winding up of the Corporation, no distribution shall be made (1) to the holders of shares of stock ranking junior (either as to dividends or upon liquidation, dissolution or winding up) to the Junior Preferred Stock unless, prior thereto, the holders of shares of Junior Preferred Stock shall have received \$1,000 per share, plus an amount equal to accrued and unpaid dividends and distributions thereon, whether or not declared, to the date of such payment, provided that the holders of shares of Junior Preferred Stock shall be entitled to receive an aggregate amount per share, subject to the provision for adjustment hereinafter set forth, equal to 1,000 times the aggregate amount to be distributed per share to holders of shares of Common Shares, or (2) to the holders of shares of stock ranking on a parity (either as to dividends or upon liquidation, dissolution or winding up) with the Junior Preferred Stock, except distributions made ratably on the Junior Preferred Stock and all such parity stock in proportion to the total amounts to which the holders of all such shares are entitled upon such liquidation, dissolution or winding up. In the event the Corporation shall at any time declare or pay any dividend on the Common Shares payable in shares of Common Shares, or effect a subdivision or combination or consolidation of the outstanding shares of Common Shares (by reclassification or otherwise than by payment of a dividend in shares of Common Shares) into a greater or lesser number of shares of Common Shares, then in each such case the aggregate amount to which holders of shares of Junior Preferred Stock were entitled immediately prior to such event under the proviso in clause (1) of the preceding sentence shall be adjusted by multiplying such amount by a fraction the numerator of which is the number of shares of Common Shares outstanding immediately after such event and the denominator of which is the number of shares of Common Shares that were outstanding immediately prior to such event.

#### VII. Consolidation, Merger, etc.

In case the Corporation shall enter into any consolidation, merger combination or other transaction in which the shares of Common Shares are exchanged for or changed into other stock or securities, cash and/or any other property, then in any

such case each share of Junior Preferred Stock shall at the same time be similarly exchanged or changed into an amount per share, subject to the provision for adjustment hereinafter set forth, equal to 1,000 times the aggregate amount of stock, securities, cash and/or any other property (payable in kind), as the case may be, into which or for which each share of Common Shares is changed or exchanged. In the event the Corporation shall at any time declare or pay any dividend on the Common Shares payable in shares of Common Shares, or effect a subdivision or combination or consolidation of the outstanding shares of Common Shares (by reclassification or otherwise than by payment of a dividend in shares of Common Shares) into a greater or lesser number of shares of Common Shares, then in each such case the amount set forth in the preceding sentence with respect to the exchange or change of shares of Junior Preferred Stock shall be adjusted by multiplying such amount by a fraction, the numerator of which is the number of shares of Common Shares outstanding immediately after such event and the denominator of which is the number of shares of Common Shares that were outstanding immediately prior to such event.

#### VIII. Redemption

The shares of Junior Preferred Stock shall not be redeemable.

#### IX. Rank

The Junior Preferred Stock shall rank, with respect to the payment of dividends and the distribution of assets, junior to all series of any other class of the Corporation's Preferred Shares and New Preferred Shares.

#### X. Amendment

The Articles of Incorporation of the Corporation shall not be amended in any manner which would alter or change the powers, preferences or special rights of the Junior Preferred Stock so as to affect them adversely without the affirmative vote of the holders of at least two-thirds of the outstanding shares of Junior Preferred Stock, voting together as a single class.

#### \* \* \* \*

7. The undersigned officers further certify that the number of shares of New Preferred Shares the Corporation is authorized to issue is 150,000 shares, and that the number of shares constituting the series designated Junior Participating Preferred Stock, none of which has been issued, is 8,958 shares.

/s/ Floyd E. Wicks -----\_\_\_

Floyd E. Wicks President and Chief Executive Officer

\_\_\_\_

/s/ McClellan Harris III

McClellan Harris III Vice President - Finance, Chief Financial Officer, Treasurer and Corporate Secretary

Each of the undersigned declares under penalty of perjury that the matters set forth in the foregoing Certificate of Amendment and Determination are true and correct. Executed at San Dimas, California this 19th Day of August, 1998.

8

/s/ Floyd E. Wicks

- -----Floyd E. Wicks President and Chief Executive Officer

/s/ McClellan Harris III -----McClellan Harris III Vice President - Finance, Chief Financial Officer, Treasurer and Corporate Secretary

#### AUG 13 1999

## /s/ Bill Jones BILL JONES, Secretary of State

2070444 CERTIFICATE OF AMENDMENT

OF AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

AMERICAN STATES WATER COMPANY (A CALIFORNIA CORPORATION)

Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly elected and acting President and Secretary, respectively of American States Water Company (the "Corporation").

2. The Amended and Restated Articles of Incorporation of the Corporation shall be amended by striking in its entirety the first full paragraph of Article IV of the Amended and Restated Articles of Incorporation which now reads:

"This Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,231,600; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000; 81,600 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$2,040,000; and 30,000,000 shares are to be Common Shares with no par value of \$2.50 per share and an aggregate par value of \$75,000,000."

and substituting therefore the following paragraph to read in full as follows:

"This Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,230,000; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000; 80,000 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$2,000,000; and 30,000,000 shares are to be Common Shares with no par value of \$2.50 per share and an aggregate par value of \$75,000,000."

3. The Restated Articles of Incorporation of the Corporation shall be further amended by striking in its entirety paragraph (ii) of Paragraph 15 of Article IV of the Amended and Restated Articles of Incorporation which now reads:

"(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares, 5% Series, shall be 17,600."

and substituting therefore the following paragraph to read in full as follows:

"(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares, 5% Series, shall be 16,000."

4. The foregoing amendments set forth in paragraphs 2 and 3 of this certificate are each amendments that may be adopted by the Board of Directors alone (and which were so adopted) because the amendments are required by Section 510 of the California General Corporation Law to reflect the reacquisition of a portion of the Corporation's Preferred Shares, 5% Series, \$25 par value, in accordance with the sinking fund provisions thereof. Such reacquired Preferred Shares cannot be reissued.

5. The foregoing amendments have been duly approved by the Board of Directors as required by Section 905(b) of the California General Corporation Law.

We further declare, under penalty of perjury under the laws of the State of California, that the matters set forth in this Certificate of Amendment are true and correct.

IN WITNESS WHEREOF, the undersigned have executed this Certificate in San Dimas, California on this 11th day of August, 1999.

/s/ Floyd E. Wicks FLOYD E. WICKS, President

#### FILED

in the office of the Secretary of State of the State of California

NOV 28, 2001 [initials illegible]

\_\_\_\_\_

/s/ Bill Jones

BILL JONES, Secretary of State

C2070444 CERTIFICATE OF AMENDMENT

OF

AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

#### AMERICAN STATES WATER COMPANY (A CALIFORNIA CORPORATION)

Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly elected and acting President and Secretary, respectively of American States Water Company (the "Corporation").

2. The Amended and Restated Articles of Incorporation of the Corporation shall be amended by striking in its entirety the first full paragraph of Article IV of the Amended and Restated Articles of Incorporation which now reads:

"This Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,230,000; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000; 80,000 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$2,000, [illegible]; and 30,000,000 shares are to be Common Shares [illegible] no par value and a stated value of \$2.50 per share and an aggregate par value of \$75,000,000."

and substituting therefore the following paragraph to read in full as follows:

"This Corporation is authorized to issue three classes of stock to be designated, respectively, "New Preferred Shares", "Preferred Shares", and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,230,000; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000; 76,800 shares are to be Preferred Shares with a par value of \$25 per share and an aggregate par value of \$1,920,000; and 30,000,000 shares are to be Common Shares with no par value of \$2.50 per share and an aggregate par value of \$75,000,000."

3. The Restated Articles of Incorporation of the Corporation shall be further amended by striking in its entirety paragraph (ii) of Paragraph 15 of Article IV of the Amended and Restated Articles of Incorporation which now reads:

"(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares, 5% Series, shall be 16,000."

and substituting therefore the following paragraph to read in full as follows:

"(ii) Number of Shares. The authorized number of shares constituting said Preferred Shares, 5% Series, shall be 12,800."

4. The foregoing amendments set forth in paragraphs 2 and 3 of this certificate are each amendments that may be adopted by the Board of Directors alone (and which were so adopted) because the amendments are required by Section 510 of the California General Corporation Law to reflect the reacquisition of a portion of the Corporation's Preferred Shares, 5% Series, \$25 par value, in accordance with the sinking fund provisions thereof. Such reacquired Preferred Shares cannot be reissued.

5. The foregoing amendments have been duly approved by the Board of Directors as required by Section 905(b) of the California General Corporation Law.

We further declare, under penalty of perjury under the laws of the State of California, that the matters set forth in this Certificate of Amendment are true and correct.

IN WITNESS WHEREOF, the undersigned have executed this Certificate in San Dimas, California on this 20th day of November, 2001.

/s/ Floyd E. Wicks FLOYD E. WICKS President and Chief Executive Officer

#### FILED in the Office of the Secretary of State of the State of California

#### MAY 03 2002

/s/ Bill Jones

BILL JONES, Secretary of State

2070444

CERTIFICATE OF AMENDMENT

ΤO

AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

#### AMERICAN STATES WATER COMPANY

Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly elected President and Chief Executive Officer, and the duly elected Chief Financial Officer, Vice President--Finance, Treasurer and Corporate Secretary, respectively, of American States Water Company, a California corporation (the "Company").

2. Article IV of the Company's Amended and Restated Articles of Incorporation is amended to add a new subsection (5) thereto to read in its entirety as follows:

(5) On May 15, 2002, every two shares of Common Shares of this Corporation shall be split into three Common Shares of this Corporation (the "Stock Split"). No certificates evidencing fractional shares will be issued in connection with this Stock Split, but instead, a certificate or certificates evidencing the aggregate of all fractional shares that would be issued (rounded, if necessary, to the next higher whole share) shall be issued to ChaseMellon Shareholder Services, Inc., or its nominee, as transfer agent, for the accounts of all holders of Common Shares of this Corporation otherwise entitled to have a fraction of a share delivered to them in connection with the Stock Split; and the transfer agent shall sell those shares as soon as practicable after receiving them at the then market price and remit the net proceeds to the shareholders entitled thereto.

4. The Company has only shares of Common Shares outstanding.

5. The foregoing amendments of the Amended and Restated Articles of Incorporation have been duly approved by the Board of Directors of the Company alone in accordance with Section 902(c) of the California General Corporation Law.

We further declare under penalty of perjury under the laws of the State of California that the matters set forth in this certificate are true and correct of our own knowledge.

DATED May 3, 2002

/s/ Floyd E. Wicks ------Floyd E. Wicks

#### FILED [illegible initials] in the Office of the Secretary of State of the State of California

#### APR 30 2002

/s/ Bill Jones BILL JONES, Secretary of State

2070444

CERTIFICATE OF AMENDMENT

TO

AMENDED AND RESTATED ARTICLES OF INCORPORATION

OF

#### AMERICAN STATES WATER COMPANY

Floyd E. Wicks and McClellan Harris III certify that:

1. They are the duly elected President and Chief Executive Officer, and the duly elected Chief Financial Officer, Vice President--Finance, Treasurer and Corporate Secretary, respectively, of American States Water Company, a California corporation (the "Company").

2. Article IV of the Company's Amended and Restated Articles of Incorporation is amended in its entirety to read as follows:

This Corporation is authorized to issue two classes of stock to be designated, respectively. "New Preferred Shares" and "Common Shares". The total number of shares which this Corporation is authorized to issue is 30,150,000; 150,000 shares are to be New Preferred Shares with no par value and a stated value of \$100 per share and an aggregate stated value of \$15,000,000, and 30,000,000 shares are to be Common Shares with no par value with a stated value of \$2.50 per share and an aggregate stated value of \$75,000,000.

A statement of the preferences, privileges and restrictions granted to or imposed upon the respective classes or series of shares and/or upon the holders thereof is as follows:

(1) Subject to the provisions of this Article IV, New Preferred Shares of any particular series shall be entitled to such voting rights, if any, as may be specified for shares of such series in the certificate of determination of preferences of such series filed as provided below. All Common Shares shall be entitled to voting rights on the basis of one-tenth of one vote per share.

(2) New Preferred Shares may be issued from time to time in one or more series. Each such series shall be so designated as to distinguish it from other series of New Preferred Shares. Such designation may include an appropriate reference to the dividend rate and/or any other characteristics of such series. The Board of Directors is hereby authorized, within the limits of, but to the extent authorized by applicable law and within the limitations and restrictions, if any, stated in this Article IV, to fix or alter, from time to time, the dividend rights, dividend rate, conversion rights, voting rights, right and terms of redemption (including sinking fund provisions), redemption price or prices, and liquidation preferences, or any of them, of any wholly unissued series of New Preferred Shares, and to fix the number of shares constituting any such unissued series, by a resolution or resolutions adopted by the Board of Directors in exercise of the authority hereby granted.

(3) Subject to the dividend preferences provided for all shares of each other class at the time outstanding and to the restrictions set forth in this Paragraph (3), the Common Shares shall be entitled to receive dividends when and as declared by the Board of Directors out of any funds of this Corporation legally available therefor. After payment of the full preferential amounts hereinabove provided for all shares of each other class outstanding at the time of any liquidation, dissolution or winding up of this Corporation, whether voluntary or involuntary, all then remaining assets of this Corporation available for distribution to its shareholders shall be distributed ratably upon the Common Shares. No dividend shall be declared on the Common Shares which, after giving effect to such declaration, would reduce the Common Stock Equity of this Corporation as of the end of the calendar month last preceding that in which such dividend was declared to an amount less than 25% of the Total Capitalization of this Corporation as of said last preceding month, except that any such dividend may be declared (a) which would reduce such Common Stock Equity to less than 25% but not less than 200%. of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during the 12 months period terminating at the end of such last preceding calendar month shall not exceed 75% of the net income of this Corporation applicable to its Common Shares for such period, or (b) which would reduce such Common Stock Equity to less than 20% of such Total Capitalization if the amount of such dividend plus all dividends on the Common Shares declared during said 12 months period shall not exceed 50% of the net income of this Corporation applicable to its Common Shares for such period; provided, however, that the foregoing restrictions of this sentence shall not apply to, nor in any way restrict, (a) the payment of any dividend on the Common Shares which is payable in shares of stock of this Corporation, or (b) any reclassification, subdivision, split-up or combination of the Common Shares, or (c) any transfer between the capital and surplus accounts of this Corporation in connection with any such reclassification, subdivisions, split-up or combination or payment of dividend in shares of stock of this Corporation. Common Stock Equity as herein used shall mean the aggregate of (i) par value or stated capital of all outstanding Common Shares, and (ii) the surplus (including capital surplus, paid-in surplus and earned surplus) as shown by the books of this Corporation after giving effect to the declaration of the proposed dividend, and (iii) premium on Common Shares, less the remaining balance of the amount of organization expenses, as shown on said books. Total Capitalization as herein used shall mean the aggregate of (i) Common Stock Equity, (ii) premium on and the par value or stated capital of all outstanding shares of this Corporation of any and all classes having preferences over the Common Shares as to dividends or assets, and (iii) the principal amount of all outstanding debt maturing more than 12 months after the close of said 12 months period, all as shown by the books of this Corporation;

less the remaining balance of organization expenses, as shown on said books. Net Income as herein used shall be determined in accordance with accepted accounting practice (but after provision for all taxes based upon or measured by income, and after annual interest charges adjusted by provision for amortization of bond discount and expense or of premium on indebtedness, and also after deduction of depreciation for said 12 months period as reported in the accounts of this Corporation as filed with the Public Utilities Commission of the State of California or other public authority of said state having jurisdiction to establish or approve the system of accounts of this Corporation). Net Income applicable to Common Shares as herein used shall mean net income after deduction therefrom of all dividends payable for the period involved on all outstanding shares of any and all classes of this Corporation having preference over the Common Shares as to dividends or assets.

(4) Unless such action has been approved by the affirmative vote of at least a majority of the Continuing Directors (as defined below), without the approval of Common Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares representing in the aggregate at least 66 2/3% of the combined voting power of this Corporation's outstanding Common Shares and the New Preferred Shares, voting together as a single class, this Corporation shall not

(i) subject to subparagraph (iii) below, sell, convey, lease or otherwise dispose of all or substantially all of its assets, property or business;

(ii) approve the sale, conveyance, lease or other disposition by any subsidiary of this Corporation of all or substantially all of such subsidiary's assets, property or business;

(iii) sell, transfer, convey or otherwise dispose of more than a majority of the outstanding capital stock of any subsidiary of the Corporation, if such subsidiary holds assets accounting for 50% or more of the Corporation's consolidated assets, other than to an entity the majority of the voting power of the capital stock or other equity interest of which is owned and controlled by this Corporation;

(iv) consolidate or merge with or into any other corporation or other business entity, except if, immediately after such consolidation or merger, the shareholders of this Corporation immediately prior to such consolidation or merger will own more than 60% of the voting power of the outstanding capital stock or other equity interest of or in the surviving entity; or

(v) approve the consolidation or merger of any subsidiary of this Corporation, if such subsidiary holds assets accounting for 50% or more of

the Corporation's consolidated assets, with or into any other corporation or other business entity.

For purposes of this paragraph (4) of Article IV, the term "Continuing Directors" shall mean any member of the Board of Directors of the Corporation (while such person is a member of the Board) who (i) is not an Acquiring Person, or an Affiliate or Associate of an Acquiring Person, or a representative of an Acquiring Person or of any such Affiliate or Associate, and (ii) either (A) was a member of the Board of Directors prior to the time any person became an Acquiring Person, or (B) became a member of the Board of Directors subsequent to the time any person became an Acquiring Person, if such person's nomination for election, or re-election, to the Board was recommended, or approved, by a majority of the Continuing Directors then in office. For purposes of the foregoing definition, (i) "Affiliate" and "Associate" shall have the respective meanings ascribed to such terms in Rule 12b-2 of the General Rules and Regulations under the Exchange Act, as in effect as of the date hereof; (ii) "Acquiring Person" shall mean any person or entity which, alone or together with all Affiliates and Associates of such person or entity, shall be the beneficial owner of 20% or more of the Corporation's voting stock, but shall not include (1) an Exempt Person or (2) any person or entity who or which acquires 20% or more of the Corporation's voting stock in connection with a transaction or series of transactions approved prior to such transaction or transactions by the Board of Directors of the Corporation; provided that no person or entity shall become an Acquiring Person solely as a result of a reduction in the number of shares of the Corporation's voting stock outstanding, unless and until such person or entity shall thereafter become the beneficial owner of additional shares constituting 1% or more of the general voting power of the Corporation. "Exempt Person" shall mean the Corporation, any majority-owned subsidiary of the Corporation, and any employee benefit plan or employee stock plan of the Corporation, or any trust or other entity organized, established or holding Common Shares by, for or pursuant to, the terms of any such plan.

3. Article VI of the Company's Amended and Restated Articles of Incorporation is amended in its entirety to read as follows:

Notwithstanding any contrary provision of these Articles of Incorporation, any amendment or repeal of paragraph (4) of Article IV, this Article VI or any amendment to these Articles of Incorporation providing for a classified board shall require the affirmative vote of shares representing not less than 66 2/3% of the combined voting power of the outstanding Common Shares and, unless otherwise provided in the certificate of determination, the New Preferred Shares, voting together as a single class.

Notwithstanding any contrary provision of these Articles of Incorporation, and except as otherwise expressly provided in the California Corporations Code, none of the following provisions of the Bylaws of the Company may be amended or repealed, except by a majority of the Board or by the shareholders upon the affirmative vote of shares

representing at least 66 2/3% of the combined voting power of the outstanding Common Shares and, unless otherwise provided in the certificate of determination for any series of New Preferred Shares, the New Preferred Shares, voting together as a single class: (a) Section 2 of Article II, (b) Section 15 of Article II, and (c) Section 2 of Article III.

4. On the effective date of the filing of this Amendment to the Amended and Restated Articles of Incorporation, the Company's Preferred Stock will be eliminated in accordance with Section 510(b)(2)(A) of the California General Corporation Law. The Company has only shares of Common Shares outstanding, and there will be no effect on the outstanding Common Shares.

5. The foregoing amendments of the Amended and Restated Articles of Incorporation have been duly approved by the Board of Directors of the Company alone in accordance with Section 510(b)(2)(A) of the California General Corporation Law.

We further declare under penalty of perjury under the laws of the State of California that the matters set forth in this certificate are true and correct of our own knowledge.

5

DATED: April 29, 2002

/s/ Floyd E. Wicks ------Floyd E. Wicks

[SEAL OF THE SECRETARY OF STATE]

#### BYLAWS

#### for the regulation, except as otherwise provided by statute or its Articles of Incorporation,

сf

#### AMERICAN STATES WATER COMPANY

#### (a California corporation)

#### ARTICLE I. Offices.

Section 1. PRINCIPAL EXECUTIVE OFFICE. The corporation's principal executive office shall be fixed and located at such place as the Board of Directors (herein called the "Board") shall determine. The Board is granted full power and authority to change said principal executive office from one location to another.

Section 2. OTHER OFFICES. Branch or subordinate offices may be established at any time by the Board at any place or places.

#### ARTICLE II. Shareholders.

Section 1. PLACE OF MEETINGS. Meetings of shareholders shall be held either at the principal executive office of the corporation or at any other place within or without the State of California which may be designated either by the Board or by the written consent of all persons entitled to vote thereat given either before or after the meeting and filed with the Secretary.

Section 2. SPECIAL MEETINGS. Special meetings of the shareholders may be called at any time by the Board, the Chairman of the Board, the President or by the holders of shares entitled to cast not less than ten percent of the votes at such meeting. Upon request in writing to the Chairman of the Board, the President, any Vice President or the Secretary by any person (other than the Board) entitled to call a special meeting of shareholders, the officer forthwith shall cause notice to be given to the shareholders entitled to vote that a meeting will be held at a time requested by the person or persons calling the meeting, not less than thirty-five nor more than sixty days after the receipt of the request. Such request shall be made in accordance with applicable law and these Bylaws. If the notice is not given within twenty days after receipt of the request, the persons entitled to call the meeting may give the notice.

Section 3. ANNUAL MEETINGS. The annual meetings of shareholders shall be held on such date and at such time as may be fixed by the Board. At such meetings, directors shall be elected and any other proper business may be transacted in accordance with applicable law and these Bylaws.

Section 4. NOTICE OF ANNUAL OR SPECIAL MEETINGS. Written notice of each annual or special meeting of shareholders shall be given not less than ten nor more than sixty days before the date of the meeting to each shareholder entitled to vote thereat. Such notice shall state the place, date and hour of the meeting and (i) in the case of a special meeting, the general nature of the business to be transacted, and no other business may be transacted, or (ii) in the case of the annual meeting, those matters which the Board, at the time of the mailing of the notice, intends to present for action by the shareholders, but, subject to the provisions of applicable law and these Bylaws, any proper matter may be presented at the meeting for such action. The notice of any meeting at which directors are to be elected shall include the names of nominees intended at the time of the notice to be presented by management for election.

Notice of a shareholders' meeting shall be given either personally or by mail or by other means of written communication, addressed to the shareholder at the address of such shareholder appearing on the books of the corporation or given by the shareholder to the corporation for the purpose of notice, or, if no such address appears or is given, at the place where the principal executive office of the corporation is located or by publication at least once in a newspaper of general circulation in the county in which the principal executive office is located. Notice by mail shall be deemed to have been given at the time a written notice is deposited in the United States

mails, postage prepaid. Any other written notice shall be deemed to have been given at the time it is personally delivered to the recipient or is delivered to a common carrier for transmission, or actually transmitted by the person giving the notice by electronic means, to the recipient.

Section 5. QUORUM. A majority of the shares entitled to vote, represented in person or by proxy, shall constitute a quorum at any meeting of shareholders. If a quorum is present, the affirmative vote of a majority of the shares represented and voting at the meeting (which shares voting affirmatively also constitute at least a majority of the required quorum) shall be the act of the shareholders, unless the vote of a greater number or voting by classes is required by law or by the Articles, except as provided in the following sentence. The shareholders present at a duly called or held meeting at which a quorum is present may continue to do business until adjournment, notwithstanding the withdrawal of enough shareholders to leave less than a quorum, if any action taken (other than adjournment) is approved by at least a majority of the shares required to constitute a quorum.

Section 6. ADJOURNED MEETINGS AND NOTICE THEREOF. Any shareholders' meeting, whether or not a quorum is present, may be adjourned from time to time by the vote of shareholders entitled to exercise a majority of the voting power represented either in person or by proxy, but in the absence of a quorum (except as provided in Section 5 of this Article) no other business may be transacted at such meeting.

It shall not be necessary to give any notice of the time and place of the adjourned meeting or of the business to be transacted thereat, other than by announcement at the meeting at which such adjournment is taken: provided, however, when any shareholders' meeting is adjourned for more than forty-five days or, if after adjournment a new record date is fixed for the adjourned meeting, notice of the adjourned meeting shall be given as in the case of an original meeting.

Section 7. VOTING. The shareholders entitled to notice of any meeting or to vote at such meeting shall be only persons in whose name shares stand on the stock records of the corporation on the record date determined in accordance with Section 8 of this Article.

Subject to the following sentence and to the provisions of Section 708 of the California General Corporation Law, every shareholder entitled to vote at any election of directors may cumulate such shareholder's votes and give one candidate a number of votes equal to the number of directors to be elected multiplied by the number of votes to which the shareholder's shares are entitled, or distribute the shareholder's votes on the same principle among as many candidates as the shareholder thinks fit. No shareholder shall be entitled to cumulate votes for any candidate or candidates pursuant to the preceding sentence unless such candidate or candidates' names have been placed in nomination prior to the voting and the shareholder has given notice at the meeting prior to the voting of the shareholder's intention to cumulate the shareholder's votes. If any one shareholder has given such notice, all shareholders may cumulate their votes for candidates in nomination.

Elections need not be by ballot; provided, however, that all elections for directors must be by ballot upon demand made by a shareholder at the meeting and before the voting begins.

In any election of directors, the candidates receiving the highest number of votes of the shares entitled to be voted for them up to the number of directors to be elected by such shares are elected.

Voting shall in all cases be subject to the provisions of Chapter 7 of the California General Corporation Law, and to the following provisions:

(a) Subject to clause (g), shares held by an administrator, executor, guardian, conservator or custodian may be voted by such holder either in person or by proxy, without a transfer of such shares into the holder's name; and shares standing in the name of a trustee may be voted by the trustee, either in person or by proxy, but no trustee shall be entitled to vote shares held by such trustee without a transfer of such shares into the trustee's name.

(b) Shares standing in the name of a receiver may be voted by such receiver, and shares held by or under the control of a receiver may be voted by such receiver without the transfer thereof into the receiver's name if authority to do so is contained in the order of the court by which such receiver was appointed.

(c) Subject to the provisions of Section 705 of the California General Corporation Law and except where otherwise agreed in writing between the parties, a shareholder whose shares are pledged shall be entitled to vote such shares until the shares have been transferred into the name of the pledgee, and thereafter the pledgee shall be entitled to vote the shares so transferred. (d) Shares standing in the name of a minor may be voted and the corporation may treat all rights incident thereto as exercisable by the minor, in person or by proxy, whether or not the corporation has notice, actual or constructive, of the nonage, unless a guardian of the minor's property has been appointed and written notice of such appointment given to the corporation.

(e) Shares outstanding in the name of another corporation, domestic or foreign, may be voted by such officer, agent or proxyholder as the bylaws of such other corporation may prescribe or, in the absence of such provision, as the board of directors of such other corporation may determine or, in the absence of such determination, by the chairman of the board, president or any vice president of such other corporation, or by any other person authorized to do so by the chairman of the board, president or any vice president of such other corporation. Shares which are purported to be voted or any proxy purported to be executed in the name of a corporation (whether or not any title of the person signing is indicated) shall be presumed to be voted or the proxy executed in accordance with the provisions of this clause, unless the contrary is shown.

(f) Shares of the corporation owned by any subsidiary shall not be entitled to vote on any matter.

(g) Shares held by the corporation in a fiduciary capacity, and shares of the issuing corporation held in a fiduciary capacity by any subsidiary, shall not be entitled to vote on any matter, except to the extent that the settlor or beneficial owner possesses and exercises a right to vote or to give the corporation binding instructions as to how to vote such shares.

(h) If shares stand of record in the names of two or more persons, whether fiduciaries, members of a partnership, joint tenants, tenants in common, husband and wife as community property, tenants by the entirety, voting trustees, persons entitled to vote under a shareholder voting agreement or otherwise, or if two or more persons (including proxyholders) have the same fiduciary relationship respecting the same shares, unless the Secretary of the corporation is given written notice to the contrary and is furnished with a copy of the instrument or order appointing them or creating the relationship wherein it is so provided, their acts with respect to voting shall have the following effect:

- (i) If only one votes, such act binds all;
- (ii) If more than one vote, the act of the majority so voting binds all;
- (iii) If more than one vote, but the vote is evenly split on any particular matter each faction may vote the securities in question proportionately.

If the instrument is so filed or the registration of the shares shows that any such tenancy is held in unequal interests, a majority or even split for the purpose of this Section shall be a majority or even split in interest.

Section 8. RECORD DATE. The Board may fix, in advance, a record date for the determination of the shareholders entitled to notice of any meeting or to vote or entitled to receive payment of any dividend or other distribution, or any allotment of rights, or to exercise rights in respect of any other lawful action. The record date so fixed shall be not more than sixty days nor less than ten days prior to the date of the meeting nor more than sixty days prior to any other action. When a record date is so fixed, only shareholders of record on that date are entitled to notice of and to vote at the meeting or to receive the dividend, distribution, or allotment of rights, or to exercise of the rights, as the case may be, notwithstanding any transfer of shares on the books of the corporation after the record date. A determination of shareholders shall apply to any adjournment of the meeting unless the Board fixes a new record date to the adjourned meeting. The Board shall fix a new record date if the meeting is adjourned for more than forty-five days.

If no record date is fixed by the Board, the record date for determining shareholders entitled to notice of or to vote at a meeting of shareholders shall be at the close of business on the business day next preceding the day on which notice is given or, if notice is waived, at the close of business on the business day next preceding the day on which the meeting is held. The record date for determining shareholders for any purpose other than set forth in this Section 8 or Section 10 of this Article shall be at the close of business of the day on which the Board adopts the resolution relating thereto, or the sixtieth day prior to the date of such other action, whichever is later.

Section 9. CONSENT OF ABSENTEES. The transactions of any meeting of shareholders, however called and noticed, and wherever held, are as valid as though had at a meeting duly held after regular call

and notice, if guorum is present either in person or by proxy, and if, either before or after the meeting, each of the persons entitled to vote, not present in person or by proxy, signs a written waiver of notice or a consent to the holding of the meeting or an approval of the minutes thereof. All such waivers, consents or approvals shall be filed with the corporate records or made a part of the minutes of the meeting. Attendance of a person at a meeting shall constitute a waiver of notice of and presence at such meeting, except when the person objects, at the beginning of the meeting, to the transactions of any business because the meeting is not lawfully called or convened and except that attendance at a meeting is not waiver of any right to object to the consideration of matters required by the California General Corporation Law to be included in the notice but not so included, if such objection is expressly made at the meeting. Neither the business to be transacted at nor the purpose of any regular or special meeting of shareholders need to be specified in any written waiver of notice, consent to the holding of the meeting or approval of the minutes thereof, except as provided in Section 601(f) of the California General Corporation Law.

Section 10. ACTION WITHOUT MEETING. Subject to Section 603 of the California General Corporation Law, any action which, under any provision of the California General Corporation Law, may be taken at any annual or special meeting of shareholders, may be taken without a meeting and without prior notice if a consent in writing, setting forth the action so taken, shall be signed by the holders of outstanding shares having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted. Unless a record date for voting purposes be fixed as provided in Section 8 of this Article, the record date for determining shareholders entitled to give consent pursuant to this Section 10, when no prior action by the Board has been taken, shall be the day on which the first written consent is given.

Section 11. PROXIES. Every person entitled to vote shares has the right to do so either in person or by one or more persons authorized by a written proxy executed by such shareholder and filed with the Secretary. Any proxy duly executed is not revoked and continues in full force and effect until revoked by the person executing it prior to the vote pursuant thereto by a writing delivered to the corporation stating that the proxy is revoked or by a subsequent proxy executed by the person executing the prior proxy and presented to the meeting, or by attendance at the meeting and voting in person by the person executing the proxy: provided, however, that no proxy shall be valid after the expiration of eleven months from the date of its execution unless otherwise provided in the proxy.

Section 12. INSPECTORS OF ELECTION. In advance of any meeting of shareholders, the Board may appoint inspectors of election to act at such meeting and any adjournment thereof. If inspectors of election be not so appointed, or if any persons so appointed fail to appear or refuse to act, the chairman of any such meeting may, and on the request of any shareholder or shareholder's proxy shall, make such appointment at the meeting. The number of inspectors shall be either one or three. If appointed at a meeting on the request of one or more shareholders or proxies, the majority of shares present shall determine whether one or three inspectors are to be appointed.

The duties of such inspectors shall be as prescribed by Section 707(b) of the California General Corporation Law and shall include: determining the number of shares outstanding and the voting power of each; determining the shares represented at the meeting; determining the existence of a quorum; determining the authenticity, validity and effect of proxies; receiving votes, ballots or consents; hearing and determining all challenges and questions in any way arising in connection with the right to vote; counting and tabulating all votes or consents; determining when the polls shall close; determining the result; and doing such acts as may be proper to conduct the election or vote with fairness to all shareholders. If there are three inspectors of election, the decision, act or certificate of a majority is effective in all respects as the decision, act or certificate of all.

Section 13. CONDUCT OF MEETING. The Chairman of the Board shall preside as chairman at all meetings of the shareholders. The chairman shall conduct each such meeting in a businesslike and fair manner, but shall not be obligated to follow any technical, formal or parliamentary rules or principles of procedure. The chairman's rulings on procedural matters shall be conclusive and binding on all shareholders, unless at the time of a ruling a request for a vote is made to the shareholders holding shares entitled to vote and which are represented in person or by proxy at the meeting, in which case the decision of a majority of such shares shall be conclusive and binding on all shareholders. Without limiting the generality of the foregoing, the chairman shall have all of the powers usually vested in the chairman of a meeting of shareholders.

Section 14. QUALIFICATIONS OF DIRECTORS. Only persons who are nominated in accordance with the procedures set forth in these Bylaws shall be qualified to serve as directors. Nominations of persons for election to the Board may be made at a meeting of shareholders (a) by or at the direction of the Board or (b) by any shareholder of the corporation who is a shareholder of record at the time of giving of notice provided for in this Bylaw, who shall be entitled to vote for the election of directors at the meeting and who complies with the notice procedures set forth in this Bylaw.

Nominations by shareholders shall be made pursuant to timely notice in writing to the Secretary. To be timely as to an annual meeting, a shareholder's notice must be received at the principal executive officers of the corporation not less than 75 days nor more than 90 days prior to the first anniversary of the preceding year's annual meeting; provided, however, that if the date of the annual meeting is changed by more than 30 days from such anniversary date, notice by the shareholder to be timely must be so received not later than the close of business on the 10th day following the earlier of the day on which notice of the date of the meeting was mailed to shareholders or public disclosure of such date was made. To be timely as to a special meeting at which directors are to be elected, a shareholder's notice must be received not later than the close of business on the 10th day following the earlier of the day on which notice of the date of the meeting was mailed to shareholders or public disclosure of such date was made. Such shareholder's notice shall set forth (a) as to each person whom the shareholder proposes to nominate for election or reelection as a director all information relating to such person that is required to be disclosed in solicitations of proxies for election of directors, or is otherwise required, in each case pursuant to Regulation 14A under the Securities Exchange Act of 1934, as amended (including such person's written consent to being named in the proxy statement as a nominee and to serving as a director if elected); (b) as to the shareholder giving the notice (i) the name and address, as they appear on the corporation's books, of such shareholder and (ii) the class and number of shares of the corporation which are beneficially owned by such shareholder and also which are owned of record by such shareholder; and (c) as to the beneficial owner, if any, on whose behalf the nomination is made, (i) the name and address of such person and (ii) the class and number of shares of the corporation which are beneficially owned by such person. At the request of the Board, any person nominated by the Board for election as a director shall furnish to the Secretary that information required to be set forth in the shareholder's notice of nomination which pertains to the nominee

No person shall be qualified to serve as a director of the corporation unless nominated in accordance with the procedures set forth in this Bylaw. The Chairman of the meeting shall, if the facts warrant, determine and declare to the meeting that a nomination was not made in accordance with the procedures prescribed by these Bylaws, and if the Chairman should so determine, that the defective nomination shall be disregarded. Notwithstanding the foregoing provisions of this Bylaw, a shareholder shall also comply with all applicable requirements of the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder with respect to the matters set forth in this Bylaw.

Section 15. PROPER BUSINESS FOR SHAREHOLDER MEETINGS. At a meeting of the shareholders, only such business shall be proper as shall be brought before the meeting (a) pursuant to the corporation's notice of meeting, (b) by or at the direction of the Board or (c) by any shareholder of the corporation who is a shareholder of record at the time of giving of the notice provided for in this Bylaw, who shall be entitled to vote at such meeting and who complies with the notice procedures set forth in this Bylaw.

For business to be properly brought before a meeting by a shareholder pursuant to clause (c) of the first paragraph of this Bylaw, the shareholder must have given timely notice thereof in writing to the Secretary. To be timely as to an annual meeting of shareholders, a shareholder's notice must be received at the principal executive offices of the corporation not less than 75 days nor more than 90 days prior to the first anniversary of the preceding year's annual meeting: provided, however, that if the date of the meeting is changed by more than 30 days from such anniversary date, notice by the shareholder to be timely must be received no later than the close of business on the 10th day following the earlier of the day on which notice of the date of the meeting was mailed to shareholders or public disclosure of such date was made. To be timely as to a special meeting of shareholders, a shareholder's notice must be received not later than the call of the meeting by the Board, the Chairman of the Board or the President, or the date of receipt of a valid request by a person (other than the Board) that the special meeting be called. Such shareholder's notice shall set forth as to each matter the shareholder proposes to bring before the meeting (a) a brief description of such matter and the reasons for proposing such matters(s) at the meeting, (b) the name and address, as they appear on the corporation's books, of the shareholder proposing such business, and the name and address of the beneficial owner, if any, on whose behalf the proposal is made, (c) the class and number of shares of the corporation which are owned beneficially and of record by such shareholder of record and by the beneficial owner, if any, on whose behalf the proposal is made and (d) any material interest of such shareholder of record and the beneficial owner, if any, on whose behalf the proposal is made in such proposal.

Notwithstanding anything in these Bylaws to the contrary, no business shall be proper at a meeting unless brought before it in accordance with the procedures set forth in this Bylaw. The Chairman of the meeting shall, if the facts warrant, determine and declare to the meeting that business was not properly brought before the meeting and in accordance with the procedures prescribed by these Bylaws, and if the Chairman should so determine, that any such business not properly brought before the meeting shall not be transacted. Notwithstanding the foregoing provisions of this Bylaw, a shareholder shall also comply with all applicable requirements of the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder with respect to the matters set forth in this Bylaw.

#### ARTICLE III. Directors.

Section 1. POWERS. Subject to limitations of the Articles, of these Bylaws and of the California General Corporation Law relating to action required to be approved by the shareholders or by the outstanding shares, the business and affairs of the corporation shall be managed and all corporate powers shall be exercised by or under the direction of the Board. Without prejudice to such general powers, but subject to the same limitations, it is hereby expressly declared that the Board shall have the following powers in addition to the other powers enumerated in these Bylaws:

(a) To select and remove all the other officers, agents and employees of the corporation, prescribe the powers and duties for them as may not be inconsistent with law, the Articles or these Bylaws, fix their compensation and require from them security for faithful service.

(b) To conduct, manage and control the affairs and business of the corporation and to make such rules and regulations therefor not inconsistent with law, the Articles or these Bylaws, as they may deem best.

(c) To adopt, make and use a corporate seal, and to prescribe the forms of certificates of stock, and to alter the form of such seal and of such certificates from time to time, as they may deem best.

(d) To authorize the issuance of shares of stock of the corporation from time to time, upon such terms and for such consideration as may be lawful.

(e) To borrow money and incur indebtedness for the purposes of the corporation, and to cause to be executed and delivered therefor, in the corporate name, promissory notes, bonds, debentures, deeds of trust, mortgages, pledges, hypothecations or other evidences of debt and securities therefor.

Section 2. NUMBER OF DIRECTORS. The authorized number of directors shall be not less than five nor more than nine until changed by amendment of the Articles or by a Bylaw duly adopted by the shareholders amending this Section 2. The exact number of directors shall be fixed, within the limits specified, by the Board from time to time in a resolution adopted by a majority of the directors. The exact number of directors shall be seven until changed as provided in this Section 2.

Section 3. ELECTION AND TERM OF OFFICE. Except as otherwise provided in the Articles, the directors shall be elected at each annual meeting of the shareholders, but if any such annual meeting is not held or the directors are not elected thereat, the directors may be elected at any special meeting of shareholders held for that purpose. Each director shall hold office until the next annual meeting and until a successor has been elected and qualified.

Section 4. VACANCIES. Any director may resign effective upon giving written notice to the Chairman of the Board, the President, the Secretary or the Board, unless the notice specifies a later time for the effectiveness of such resignation. If the resignation is effective at a future time, a successor may be elected to take office when the resignation becomes effective.

Vacancies in the Board, except those existing as a result of a removal of a director, may be filled by a majority of the remaining directors, though less than a quorum, or by a sole remaining director, and each director so elected shall hold office until the next annual meeting and until such director's successor has been elected and qualified.

A vacancy or vacancies in the Board shall be deemed to exist in case of the death, resignation or removal of any director, or if the authorized number of directors be increased, or if the shareholders fail, at any annual or special meeting of shareholders at which any director or directors are elected, to elect the full authorized number of directors to be voted for at that meeting.

The Board may declare vacant the office of a director who has been declared of unsound mind by an order of court or convicted of a felony.

The shareholders, subject to applicable law and these Bylaws, may elect a director or directors at any time to fill any vacancy or vacancies not filled by the directors. Any such election by written consent, other than to fill a vacancy created by removal, requires the consent of a majority of the outstanding shares entitled to vote. Any such election by written consent to fill a vacancy created by removal requires unanimous consent.

No reduction of the authorized number of directors shall have the effect of removing any director prior to the expiration of the director's term of office.

Section 5. PLACE OF MEETING. Regular or special meetings of the Board shall be held at any place within or without the State of California which has been designated from time to time by the Board. In the absence of such designation, regular meetings shall be held at the principal executive office of the corporation.

Section 6. REGULAR MEETINGS. Immediately following each annual meeting of shareholders, the Board shall hold a regular meeting for the purpose of organization, election of officers and the transaction of other business.

Other regular meetings of the Board shall be held without call on such dates and at such times as may be fixed by the Board. Call and notice of all regular meetings of the Board are hereby dispensed with.

Section 7. SPECIAL MEETINGS. Special meetings of the Board for any purpose or purposes may be called at any time by the Chairman of the Board, the President, any Vice President, the Secretary or by any two directors.

Special meetings of the Board shall be held upon four days' written notice or forty-eight hours' notice given personally or by telephone, telegraph, telex, or other similar means of communication. Any such notice shall be addressed or delivered to each director at such director's address as it is shown upon the records of the corporation or as may have been given to the corporation by the director for purposes of notice or, if such address is not shown on such records or is not readily ascertainable, at the place in which the meetings of the directors are regularly held.

Notice by mail shall be deemed to have been given at the time a written notice is deposited in the United States mails, postage prepaid. Any other written notice shall be deemed to have been given at the time it is personally delivered to the recipient or is delivered to common carrier for transmission, or actually transmitted by the person giving the notice by electronic means, to the recipient. Oral notice shall be deemed to have been given at the time it is communicated, in person or by telephone or wireless, to the recipient or to a person at the office of the recipient who the person giving the notice has reason to believe will promptly communicate it to the recipient.

Section 8. QUORUM. A majority of the authorized number of directors constitutes a quorum of the Board for the transaction of business, except to adjourn as provided in Section 11 of this Article. Every act or decision done or made by a majority of the directors present at a meeting duly held at which a quorum is present shall be regarded as the act of the Board, unless a greater number be required by law or by the Articles. A meeting at which a quorum is initially present may continue to transact business notwithstanding the withdrawal of directors, if any action taken is approved by at least a majority of the required quorum for such meeting.

Section 9. PARTICIPATION IN MEETINGS BY CONFERENCE TELEPHONE. Members of the Board may participate in a meeting through use of conference telephone or similar communications equipment, so long as all members participating in such meeting can hear one another.

Section 10. WAIVER OF NOTICE. Notice of a meeting need not be given to any director who signs a waiver of notice or consent to holding the meeting or an approval of the minutes thereof, whether before or after the meeting, or who attends the meeting without protesting, prior thereto or at its commencement, the lack of notice to such director. All such waivers, consents and approvals shall be filed with the corporate records or made a part of the minutes of the meetings.

Section 11. ADJOURNMENT. A majority of the directors present, whether or not a quorum is present, may adjourn any directors' meeting to another time and place. Notice of the time and place of holding an adjourned meeting need not be given to absent directors if the time and place be fixed at the meeting adjourned, except as provided in the next sentence. If the meeting is adjourned for more than twenty-four hours, notice of any adjournment to another time or place shall be given prior to the time of the adjourned meeting to the directors who were not present at the time of the adjournment. Section 12. FEES AND COMPENSATION. Directors and members of committees may receive such compensation, if any, for their services, and such reimbursement for expenses, as may be fixed or determined by the Board.

Section 13. ACTION WITHOUT MEETING. Any action required or permitted to be taken by the Board may be taken without a meeting if all members of the Board shall individually or collectively consent in writing to such action. Such consent or consents shall have the same effect as a unanimous vote of the Board and shall be filed with the minutes of the proceedings of the Board.

Section 14. RIGHTS OF INSPECTION. Every director shall have the absolute right at any reasonable time to inspect and copy all books, records and documents of every kind and to inspect the physical properties of the corporation and also of its subsidiary corporations, domestic or foreign. Such inspection by a director may be made in person or by agent or attorney and includes the right to copy and obtain extracts.

Section 15. COMMITTEES. The Board may appoint one or more committees, each consisting of two or more directors, and delegate to such committees any of the authority of the Board except with respect to:

- (a) The approval of any action for which the California General Corporation Law also requires shareholders' approval or approval of the outstanding shares;
- (b) The filling of vacancies on the Board or on any committee;
- (c) The fixing of compensation of the directors for service on the Board or on any committee;
- (d) The amendment or repeal of bylaws or the adoption of new bylaws;
- The amendment or repeal of any resolution of the Board which by its express terms is not so amendable or repealable;
- (f) A distribution to the shareholders of the corporation except at a rate or in a periodic amount or within a price range determined by the Board; or
- (g) The appointment of other committees of the Board or the members thereof.

Any such committee must be designated, and the members or alternate members thereof appointed, by resolution adopted by a majority of the authorized number of directors and any such committee may be designated an Executive Committee or by such other name as the Board shall specify. Alternative members of a committee may replace any absent member at any meeting of the committee. The Board shall have the power to prescribe the manner in which proceedings of any such committee shall be conducted. In the absence of any such prescription, such committee shall have the power to prescribe the manner in which its proceedings shall be conducted. Unless the Board or such committee shall otherwise provide, the regular and special meetings and other actions of any such committee shall be governed by the provisions of this Article applicable to meetings and actions of the Board. Minutes shall be kept of each meeting of each committee.

#### ARTICLE IV. Officers.

Section 1. OFFICERS. The officers of the corporation shall be a President, a Secretary and a Chief Financial Officer. The corporation may also have, at the discretion of the Board, a Chairman of the Board, an Executive Vice President, a Senior Vice President, one or more Vice Presidents, a Treasurer, one or more Assistant Secretaries, one or more Assistant Treasurers, and such other officers as may be elected or appointed in accordance with the provisions of Section 3 of this Article.

Section 2. ELECTION. The officers of the corporation, except such officers as may be elected or appointed in accordance with the provisions of Section 3 or Section 5 of this Article, shall be chosen annually by, and shall serve at the pleasure of, the Board, and shall hold their respective offices until their resignation, removal, or other disqualification from service, or until their respective successors shall be elected.

Section 3. SUBORDINATE OFFICERS. The Board may elect, and may empower the Chairman of the Board, if there be such an officer, or the President, to appoint such other officers as the business of the corporation may require, each of whom shall hold office for such period, have such authority and perform such duties as are provided in these Bylaws or as the Board may from time to time determine.

Section 4. REMOVAL AND RESIGNATION. Any officer may be removed, either with or without cause, by the Board at any time or, except in the case of an officer chosen by the Board, by an officer upon whom such power of removal may be conferred by the Board. Any such removal shall be without prejudice to the rights, if any, of the officer under any contract of employment of the officer.

Any officer may resign at any time by giving written notice to the corporation, but without prejudice to the rights, if any, of the corporation under any contract to which the officer is a party. Any such resignation shall take effect at the date of the receipt of such notice or at any later time specified therein and, unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 5. VACANCIES. A vacancy in any office because of death, resignation, removal, disqualification or any other cause shall be filled in the manner prescribed in these Bylaws for regular election or appointment to such office.

Section 6. CHAIRMAN OF THE BOARD. The Chairman of the Board, if there shall be such an officer, shall be the Chief Executive Officer of the corporation unless, in its sole discretion, the Board should elect the President to be such. The Chief Executive Officer is the general manager and chief executive officer of the corporation and has, subject to the control of the Board, general supervision, direction and control of the business and officers of the corporation. The Chairman of the Board, if there shall be such an officer, shall, if present, preside at all meetings of the shareholders and the Board and exercise and perform such other powers and duties as may be from time to time assigned by the Board.

Section 7. PRESIDENT. Subject to such powers, if any, as may be given to the Chairman of the Board, if there be such an officer, the President shall have the general powers and duties of management usually vested in the office of the president of a corporation and such other powers and duties as may be prescribed by the Board or the Chief Executive Officer, if other than the President. In the absence of the Chairman of the Board, or if there be none, the President shall preside at all meetings of the shareholders and the Board. In the absence or disability of the Chief Executive Officer, if other than the President, the President shall perform all the duties of the Chief Executive Officer and, when so acting, shall have all of the powers of, and be subject to all the restrictions upon, the Chief Executive Officer.

Section 8. VICE PRESIDENTS. The Executive Vice President and Senior Vice President, if any, and other Vice Presidents shall have (subject to the authority of the Board) such powers and perform such duties its from time to time determined by the Chief Executive Officer. in the absence or disability of the President, the Vice Presidents, in the following order, shall perform all the duties of the President and, when so acting, shall have all the powers of, and be subject to all the restrictions upon, the President: the Executive Vice President, if any, the Senior Vice President, if any, and the Vice Presidents in the order of their rank as fixed by the Board, or if not ranked, the Vice President designated by the Board. The Vice President shall have such other powers and perform such other duties as from time to time may be prescribed for them, respectively, by the Board.

Section 9. SECRETARY. The Secretary shall keep or cause to be kept, at the principal executive office and such other place as the Board may order, a book of minutes of all meetings of shareholders, the Board and its committees, with the time and place of holding, whether regular or special, how authorized, the notice thereof given, the names of those present at Board and committee meetings, the number of shares present or represented at shareholders' meetings, and the proceedings thereof. The Secretary shall keep, or cause to be kept, a copy of the Bylaws of the corporation at the principal executive office or business office in accordance with Section 213 of the California Central Corporation Law.

The Secretary shall keep, or cause to be kept, at the principal executive office or at the office of the corporation's transfer agent or registrar, if one be appointed, a share register, or a duplicate share register, showing the names of the shareholders and their addresses, the number of classes of shares held by each, the number and date of certificates issued for the same, and the number and date of cancellation of every certificate surrendered for cancellation.

The Secretary shall give, or cause to be given, notice of all meetings of the shareholders and of the Board and any committees thereof required by these Bylaws or by law to be given, shall keep the seal of the corporation in safe custody, and shall have such other powers and perform such other duties as may be prescribed by the Board. Section 10. CHIEF FINANCIAL OFFICER. The Chief Financial Officer shall keep and maintain, or cause to be kept and maintained, adequate and correct accounts of the properties and business transactions of the corporation, and shall send or cause to be sent to the shareholders of the corporation such financial statements and reports as are by law or these Bylaws required to be sent to them. The books of account shall at all times be open to inspection by any director.

The Chief Financial Officer shall deposit all monies and other valuables in the name and to the credit of the corporation with such depositaries as may be designated by the Board. The Chief Financial Officer shall disburse the funds of the corporation as may be ordered by the Board, shall render to the President and the directors, whenever they request it, an account of all transactions as Chief Financial Officer and of the financial condition of the corporation, and shall have such other powers and perform such other duties as may be prescribed by the Board.

ARTICLE V. Other Provisions.

Section 1. INSPECTION OF CORPORATE RECORDS.

(a) A shareholder or shareholders holding at least five percent in the aggregate of the outstanding voting shares of the corporation or who hold at least percent of such voting shares and have filed a Schedule 14B with the United States Securities and Exchange Commission relating to the election of directors of the corporation shall have the absolute right to do either or both of the following:

> (i) Inspect and copy the record of shareholders' names and addresses and shareholders during usual business hours upon five business days' prior written demand upon the corporation; or

(ii) Obtain from the transfer agent, if any, for the corporation, upon five business days' prior written demand and upon the tender of its usual charges for such a list (the amount of which charges shall be stated to the shareholder by the transfer agent upon request), a list of the shareholders' names and addresses who are entitled to vote for the election of directors and their shareholdings, as of the most recent complied or as of the date specified by the shareholder subsequent to the date of demand.

(b) The record of shareholders shall also be open to inspection and copying by any shareholder or holder of a voting trust certificate at any time during usual business hours upon written demand on the corporation, for a purpose reasonably related to such holder's interest as a shareholder or holder of a voting trust certificate.

(c) The accounting books and records and minutes of proceedings of the shareholders and the Board and committees of the Board shall be open to inspection upon written demand on the corporation of any shareholder or holder of a voting trust certificate at any reasonable time, during usual business hours, for a purpose reasonably related to such holder's interests as a shareholder or as a holder of such voting trust certificate.

(d) Any inspection and copying under this Article may be made in person or by agent or attorney.

Section 2. INSPECTION OF BYLAWS. The corporation shall keep in its principal executive office in the State of California, or if its principal executive office is not in such State at its principal business office in such state, the original or copy of these Bylaws as amended to date, which shall be open to inspection by shareholders at all reasonable times during office hours. If the principal executive office of the corporation is located outside the State of California and the corporation has no principal business office in such state, it shall upon the written request of any shareholder furnish to such shareholder a copy of these Bylaws as amended to date.

Section 3. ENDORSEMENT OF DOCUMENTS, CONTRACTS. Subject to the provisions of applicable law, any note, mortgage, evidence of indebtedness, contract, share certificate, conveyance or other instrument in writing and any assignment or endorsements thereof executed or entered into between the corporation and any other person, when signed by the Chairman of the Board, the President or any Vice President and the Secretary, any Assistant Secretary, the Chief Financial Officer, the Treasurer or any Assistant Treasurer of the corporation, shall be valid and binding on the corporation in the absence of actual knowledge on the part of the other person that the signing officers had no authority to execute the same. Any such instruments may be signed by any other person or persons and in such manner as from time to time shall be determined by the Board, and, unless so authorized by the Board, no officer, agent or employee shall have any power or authority to bind the corporation by any contract or engagement or to pledge its credit or to render it liable for any purpose or amount.

Section 4. CERTIFICATES OF STOCK. Every holder of shares of the corporation shall be entitled to have a certificate signed in the name of the corporation by the Chairman of the Board, the President or a Vice President and by the Chief Financial Officer, the Treasurer or an Assistant Treasurer or the Secretary or an Assistant Secretary, certifying the number of shares and the class or series of shares owned by the shareholder. Any or all of the signatures on the certificate may be facsimile. If any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate shall have ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the corporation with the same effect as if such person were an officer, transfer agent or registrar at the date of issue.

Certificates for shares may be issued prior to full payment under such restrictions and for such purposes as the Board may provide; provided, however, that on any certificate issued to represent any partly paid shares, the total amount of the consideration to be paid therefor and the amount paid thereon shall be stated.

Except as provided in this Section, no new certificate for shares shall be issued in lieu of an old one unless the latter is surrendered and cancelled at the same time. The Board may, however, if any certificate for shares is alleged to have been lost, stolen or destroyed, authorize the issuance of a new certificate in lieu thereof, and the corporation may require that the corporation be given a bond or other adequate security sufficient to indemnify it against any claim that may be made against it (including expense or liability) on account of the alleged loss, theft or destruction of such certificate or the issuance of such new certificate.

Section 5. REPRESENTATION OF SHARES OF OTHER CORPORATIONS. The Chief Executive Officer, the President or any other officer or officers authorized by the Board or the Chief Executive Officer are each authorized to vote, represent and exercise on behalf of the corporation all rights incident to any and all shares of any other corporation or corporations standing in the name of the corporation. The authority herein granted may be exercised either by any such officer in person or by any other person authorized so to do by proxy or power of attorney duly executed by said officer.

Section 6. STOCK PURCHASE PLANS. The corporation may adopt and carry out a stock purchase plan or agreement or stock option plan or agreement providing for the issue and sale for such consideration as may be fixed of its unissued shares, or of issued shares acquired or to be acquired, to one or more of the employees or directors of the corporation or of a subsidiary or to a trustee on their behalf and for the payment for such shares in installments or at one time, and may provide for aiding any such persons in paying for such shares by compensation for services rendered, promissory notes or otherwise.

Any such stock purchase plan or agreement or stock option plan or agreement may include, among other features, the fixing of eligibility for participation therein, the class and price of shares to be issued or sold under the plan or agreement, the number of shares which may be subscribed for, the method of payment therefor, the reservation of title until full payment therefor, the effect of the termination of employment, an option or obligation on the part of the corporation, to repurchase the shares upon termination of employment, restrictions upon transfer of the shares, the time limits of and termination of the plan, and any other matters, not in violation of applicable law, as may be included in the plan as approved or authorized by the Board or any committee of the Board.

Section 7. CONSTRUCTION AND DEFINITIONS. Unless the context otherwise requires, the general provisions, rules of construction and definitions contained in the General Provisions of the California Corporations Code and in the California General Corporation Law shall govern the construction of these Bylaws.

#### ARTICLE VI. Indemnification.

#### Section 1. INDEMNIFICATION OF DIRECTORS AND OFFICERS.

(a) Each person who was or is a party or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (hereinafter a "proceeding"), by reason of the fact that he or she is or was a director or officer of the corporation, or of any predecessor corporation, or is or was a director or officer who is or was serving at the request of the corporation as a director, officer, employee or other agent of another corporation, a partnership, joint venture, trust or other enterprise (including service with respect to corporation-sponsored employee benefit plans), whether the basis at

such proceeding is alleged action or inaction in an official capacity as a director or officer or in any other capacity while serving as a director or officer, shall, subject to the tents of any agreement between the corporation and such person, be indemnified and held harmless by the corporation to the fullest extent permissible under California law and the corporation's Articles, against all expense, liability and loss (including attorneys' fees, judgments, fines, ERISA excise taxes or penalties and amounts paid in settlement) actually and reasonably incurred or suffered by such person in connection therewith; provided, however, that amounts paid in settlement of a proceeding shall be payable only if the settlement is approved in writing by the corporation. Such indemnification shall continue as to a person who has ceased to be a director or officer for acts performed while a director or officer and shall inure to the benefit of his or her heirs, executors and administrators. Notwithstanding the foregoing, the corporation shall indemnify any such person in connection with a proceeding (or part thereof) initiated by such person only if such proceeding (or part thereof) was authorized by the Board of the corporation. The right to indemnification conferred in this Article shall include the right to be paid by the corporation the expenses incurred in defending any proceeding in advance of final disposition to the fullest extent permitted by law; provided, however, that the payment under this Article of such expenses in advance of the final disposition of a proceedings shall be conditioned upon the delivery to the corporation of a written request for such advance and of an undertaking by or on behalf of the director or officer to repay all amounts so advanced if it shall be ultimately determined that such director or officer is not entitled to be indemnified.

(b) Notwithstanding the foregoing or any other provisions under this Article, the corporation shall not be liable under this Article to indemnify a director or officer against expenses, liabilities or losses incurred or suffered in connection with, or make any advances with respect to, any proceeding against a director or officer; (i) as to which the corporation is prohibited by applicable law from paying as an indemnity; (ii) with respect to expenses of defense or investigation, if such expenses were or are incurred without the corporation's consent (which consent may not be unreasonably withheld); (iii) for which payment is actually made to the director or officer under a valid and collectible insurance policy maintained by the corporation, except in respect of any excess beyond the amount of payment under such insurance; (iv) for which payment is actually made to the director or officer under an indemnity by the corporation otherwise than pursuant to this Bylaw Article, except in respect of any excess beyond the amount of payment under such indemnity; (v) based upon or attributable to the director or officer gaining in fact any personal profit or advantage to which he or she was not legally entitled; (vi) for an accounting of profits made from the purchase or sale by the director or officer of securities of the corporation pursuant to the provisions of Section 16(b) of the Securities Exchange Act of 1934 and amendments thereto or similar provisions of any federal, state or local statutory law; or (vii) based upon acts or omissions involving intentional misconduct or a knowing and culpable violation of law.

Section 2. INDEMNIFICATION OF EMPLOYEES AND AGENTS. A person who was, or is a party or is threatened to be made a party to or is involved in any proceeding by reason of the fact that he or she is or was an employee or agent of the corporation or is or was an employee or agent of the corporation who is or was serving at the request of the corporation as an employee or agent of another enterprise, including service with respect to corporation-sponsored employee benefits plans, whether the basis of such action is alleged action or inaction in an official capacity or in any other capacity while serving as an employee or agent, may, upon appropriate action by the corporation and subject to the terms of any agreement between the corporation and such person, be indemnified and held harmless by the corporation's Articles, against all expense, liability and loss (including attorneys' fees, judgments, fines, ERISA excise taxes or penalties and amounts paid or to be paid in settlement) actually and reasonably incurred or suffered by such person in connection therewith.

Section 3. RIGHT OF DIRECTORS AND OFFICERS TO BRING SUIT. If a claim under Section 1 of this Article is not paid by the corporation or on its behalf within 90 days after a written claim has been received by the corporation, the claimant may at any time thereafter bring suit against the corporation to recover the unpaid amount of the claim, and, if successful in whole or in part, the claimant also shall be entitled to be paid the expense of prosecuting such claim.

Section 4. SUCCESSFUL DEFENSE. Notwithstanding any other provision of this Article, to the extent that a director or officer has been successful on the merits or otherwise (including the dismissal of a proceeding without prejudice or the settlement with the written consent of the corporation of a proceeding without admission of liability) in defense of any proceeding referred to in Section 1 or in defense of any claim, issue or matter therein, such director or officer shall be indemnified against expenses (including attorneys' fees) actually and reasonably incurred in connection therewith.

Section 5. INDEMNITY AGREEMENTS. The corporation may enter into agreements with any director, officer, employee or agent of the corporation providing for indemnification to the fullest extent permissible under applicable law and the corporation's Articles.

Section 6. SUBROGATION. In the event of payment by the corporation of a claim under Section 1 of this Article, the corporation shall be subrogated to the extent of such payment to all of the rights of recovery of the indemnified person, who shall execute all papers required and shall do everything that may be necessary or appropriate to secure such rights, including the execution of such documents necessary or appropriate to enable the corporation effectively to bring suit to enforce such rights.

Section 7 NON-EXCLUSIVITY RIGHTS. The right to indemnification provided by this Article shall not be exclusive of any other right which any person may have or hereafter acquire under any statute, bylaw, agreement, vote of shareholders or disinterested directors or otherwise.

Section 8. INSURANCE. The corporation may maintain insurance, at its expense, to protect itself and any director, officer, employee or agent of the corporation or another corporation, a partnership, joint venture, trust or other enterprise against any expense, liability or loss, whether or not the corporation would have the power to indemnify such person against such expense, liability or loss under California law.

Section 9. EXPENSES AS A WITNESS. To the extent that any director, officer or employee of the corporation is by reason of such position a witness in any action, suit or proceeding, he or she will be indemnified against all costs and expenses actually and reasonably incurred by him or her or on his or her behalf in connection therewith.

Section 10. NONAPPLICABILITY TO FIDUCIARIES OF EMPLOYEE BENEFIT PLANS. This Article does not apply to any proceeding against any trustee, investment manager or other fiduciary of an employee benefit plan in such person's capacity as such, even though such person may also be an agent of the corporation. The corporation shall have power to indemnify such trustee, investment manager or other fiduciary to the extent permitted by subdivision (f) of Section 207 of the California General Corporation Law.

Section 11. SEPARABILITY. Each and every paragraph, sentence, term and provision of this Article is separate and distinct so that if any paragraph, sentence, term or provision shall be held to be invalid or unenforceable for any reason, such invalidity or unenforceability shall not affect the validity or enforceability of any other paragraph, sentence, term or provision hereof. To the extent required, any paragraph, sentence, term or provision of this Article may be modified by a court of competent jurisdiction to preserve its validity and to provide the claimant with, subject to the limitations set forth in this Article and any agreement between the corporation and the claimant, the broadest possible indemnification permitted under applicable law.

Section 12. EFFECT OF REPEAL OR MODIFICATION. Any repeal or modification of this Article shall not adversely affect any right of indemnification of a director, officer, employee or agent of the corporation existing at the time of such repeal or modification with respect to any action or omission occurring prior to such repeal or modification.

#### ARTICLE VII. Emergency Provisions.

Section 1. GENERAL. The provisions of this Article shall be operative only during a national emergency declared by the President of the United States or the person performing the Presidents functions, or in the event of a nuclear, atomic or other attack on the United States or a disaster making it impossible or impracticable for the corporation to conduct its business without recourse to the provisions of this Article. Said provisions in such event shall override all other Bylaws of the corporation in conflict with any provisions of this Article, and shall remain operative so long as it remains impossible or impracticable to continue the business of the corporation otherwise, but thereafter shall be inoperative; provided that all actions taken in good faith pursuant to such provisions shall thereafter remain in full force and effect unless and until revoked by action taken pursuant to the provisions of the Bylaws other than those contained in this Article.

Section 2. UNAVAILABLE DIRECTORS. All directors of the corporation who are not available to perform their duties as directors by reason of physical or mental incapacity or for any other reason or who are unwilling to perform their duties or whose whereabouts are unknown shall automatically cease to be directors, with like effect as if such persons had resigned as directors, so long as such unavailability continues.

Section 3. AUTHORIZED NUMBER OF DIRECTORS. The authorized number of directors shall be the number of directors remaining after eliminating those who have ceased to be directors pursuant to Section 2, or the minimum number required by law, whichever number is greater.

Section 4. QUORUM. The number of directors necessary to constitute a quorum shall be one-third of the authorized number of directors as specified in the foregoing Section, or other minimum number as, pursuant to the law or lawful decree then in force, it is possible for the Bylaws of a corporation to specify.

Section 5. CREATION OF EMERGENCY COMMITTEE. In the event the number of directors remaining after eliminating those who have ceased to be directors pursuant to Section 2 is less than the minimum number of authorized directors required by law, then until the appointment of additional directors to make up such required minimum, all the powers and authorities which the Board could by law delegate, including all powers and authorities which the Board could delegate to a committee, shall be automatically vested in an emergency committee, and the emergency committee shall thereafter manage the affairs of the corporation pursuant to such powers and authorities and shall have all other powers and authorities as may by law or lawful decree be conferred on any person or body of persons during a period of emergency.

Section 6. CONSTITUTION OF EMERGENCY COMMITTEE. The emergency committee shall consist of all the directors remaining after eliminating those who have ceased to be directors pursuant to Section 2, provided that such remaining directors are not less than three in number. In the event such remaining directors are less than three in number the emergency committee shall consist of three persons, who shall be the remaining director or directors and either one or two officers or employees of the corporation as the remaining director or directors may in writing designate. If there is no remaining director, the emergency committee shall consist of the three most senior officers of the corporation who are available to serve, and if and to the extent that officers are not available, the most senior employees of the corporation. Seniority shall be determined in accordance with any designation of seniority in the minutes of the proceedings of the Board, and in the absence of such designation, shall be determined by rate of remuneration. In the event that there are no remaining directors and no officers or employees of the corporation available, the emergency committee shall consist of three persons designated in writing by the shareholder owning the largest number of shares of record as of the date of the last record date.

Section 7. POWERS OF EMERGENCY COMMITTEE. The emergency committee, once appointed, shall govern its own procedures and shall have power to increase the number of members thereof beyond the original number, and in the event of a vacancy or vacancies therein, arising at any time, the remaining member or members of the emergency committee shall have the power to fill such vacancy or vacancies. In the event at any time after its appointment all members of the emergency committee shall die or resign or become unavailable to act for any reason whatsoever, a new emergency committee shall be appointed in accordance with the foregoing provisions of this Article.

Section 8. DIRECTORS BECOMING AVAILABLE. Any person who has ceased to be a director pursuant to the provisions of Section 2 and who thereafter becomes available to serve as a director shall automatically become a member of the emergency committee.

Section 9. ELECTION OF BOARD OF DIRECTORS. The emergency committee shall, as soon after its appointment as is practicable, take all requisite action to secure the election of a board of directors, and upon such election, all the powers and authorities of the emergency committee shall cease.

Section 10. TERMINATION OF EMERGENCY COMMITTEE. In the event, after the appointment of an emergency committee, a sufficient number of persons who ceased to be directors pursuant to Section 2 become available to serve as directors, so that if they had not ceased to be directors as aforesaid, there would be enough directors to constitute the minimum number of directors required by law, then all such persons shall automatically be deemed to be reappointed as directors and the powers and authorities of the emergency committee shall be at an end.

#### ARTICLE VIII. Amendments.

Subject to the Articles of Incorporation, these Bylaws may be amended or repealed either by approval of the outstanding shares (as defined in Section 152 of the California General Corporation Law) or by the approval of the Board; provided, however, that after the issuance of shares, a bylaw specifying or changing a fixed number of directors or the maximum or minimum number or changing from a fixed to a variable number of directors or vice versa may only be adopted by approval of the outstanding shares and a bylaw reducing the fixed number or the minimum number of directors to a number less than five shall be subject to the provisions of Section 212(a) of the California General Corporation Law.

CERTIFICATE OF SECRETARY

#### of AMERICAN STATES WATER COMPANY (a California corporation)

I hereby certify that I am the duly elected and acting Secretary of said corporation and that the foregoing Bylaws, comprising 14 pages, constitute the Bylaws of said corporation as duly adopted by unanimous written consent of the Board of Directors on March 5, 1998.

### [SEAL STATE OF CALIFORNIA]

SECRETARY OF STATE

# CERTIFICATE OF STATUS DOMESTIC CORPORATION

I, BILL JONES, Secretary of State of the State of California, hereby certify:

That on the 25th day of February, 1998, AMERICAN STATES WATER COMPANY became incorporated under the laws of the State of California by filing its Articles of Incorporation in this office; and

That no record exists in this office of a certificate of dissolution of said corporation nor of a court order declaring dissolution thereof, nor of a merger or consolidation which terminated its existence; and

That said corporation's corporate powers, rights and privileges are not suspended on the records of this office; and

That according to the records of this office, the said corporation is authorized to exercise all its corporate powers, rights and privileges and is in good legal standing in the State of California: and

That no information is available in this activity or practices of this corporation. office on the financial condition, business

IN WITNESS WHEREOF, I execute this certificate and affix the Great Seal of the State of California this day of July 1, 2002.

[SEAL STATE OF CALIFORNIA]

/s/ Bill Jones

BILL JONES Secretary of State

STATE	OF	ALIFORNIA
=====	====	

FRANCHISE TAX BOARD P.O. BOX 942867 SACREMENTO, CA 94257-0540

> In Reply Refer To: 357JE Date : May 31, 2002

CL@S	Inform	natio	n Servi	ces	
1425	River	Park	Drive,	Suite	110
Sacra	amento,	CA	95815		

Corporation Name : AMERICAN STATES WATER COMPANY Corporation Number : 2070444

|X| 1. The above corporation is in good standing with this agency.

|\_| 2. Information on record with this agency indicates the above corporation is not qualified to transact business in California.

| | 3. The above corporation was incorporated or qualified on

1\_| 4. The above corporation has an unpaid liability of \$ ended
\_\_\_\_\_\_ for income year

I\_I 5. Our records do not show that the above corporation filed franchise tax returns for the income years ended

|\_| 6. The above corporation was effective

|\_| 7. The above corporation's current address on record with this agency
is:

-----

| | 8. We have no current information on the above corporation.

#### Comments

\_\_\_\_\_

\_\_\_\_\_

/s/ Jani Errkine

REPRESENTATIVE

\_\_\_\_\_

#### TELEPHONE ASSISTANCE

Our regular toll-free telephone service is available from 7:00 am, until 8:00 p.m. Monday through Friday from the first working day in January through April 15. The best times to call are between 7:00 and 10:00 in the morning and between 6:00 and 8:00 in the evening. Service is also available from 8:00 am, through 5:00 p.m. on the two Saturdays prior to April 15. After April 15, servIce is

available Monday through Friday. between 8:00 am, and 5:00 p.m.

 From within the united States, call......
 1-800-852-5711

 From outside the United States, call (not toll-free)......
 1-916-845-6500

 For hearing impaired with TDD. call .....
 1-800-822-6268

FIB 4263A (REV 5-95)

#### Consent of Independent Accountants

We hereby consent to the incorporation by reference in the Registration Statement on Form S-8 (No. 333-39482), on Form S-8POS (No. 333-47647) and on Form S-3 (Nos. 333-68299, 333-88979) of American States Water Company of our report dated April 7, 2003 relating to the financial statements, which appears in this Form 10-K. We also consent to the incorporation by reference in the Registration Statement on Form S-3 (No. 333-73378) of Southern California Water Company of our report dated April 7, 2003 relating to the financial statements, which appears in this Form 10-K.

PricewaterhouseCoopers LLP Los Angeles, California April 7, 2003